Prepared for Nevada Environmental Response Trust

Project Number 21-41400C

Prepared by Ramboll Environ Emeryville, California

Date November 15, 2017

## INTERIM REPORT IDENTIFICATION OF COPCs AND DECISION UNITS FOR OU-1 SOILS, REVISION 1 NEVADA ENVIRONMENTAL RESPONSE TRUST SITE HENDERSON, NEVADA



#### Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1

#### Nevada Environmental Response Trust Site (Former Tronox LLC Site) Henderson, Nevada

#### Nevada Environmental Response Trust (NERT) Representative Certification

I certify that this document and all attachments submitted to the Division were prepared at the request of, or under the direction or supervision of NERT. Based on my own involvement and/or my inquiry of the person or persons who manage the system(s) or those directly responsible for gathering the information or preparing the document, or the immediate supervisor of such person(s), the information submitted and provided herein is, to the best of my knowledge and belief, true, accurate, and complete in all material respects.

Office of the Nevada Environmental Response Trust	: 1: 1. all bet sold
mut	A d.V. Sha ing i soldy
Ph.	as (usi ert
Signature: Ju Kilm by	_, not individually, but solely in his
representative capacity as President of the Nevada	Environmental Response Trust Trustee
$()$ $\cdot$ $0$	

**Name:** Jay A. Steinberg, not individually, but solely in his representative capacity as President of the Nevada Environmental Response Trust Trustee

Title: Solely as President and not individually

**Company:** Le Petomane XXVII, Inc., not individually, but solely in its representative capacity as the Nevada Environmental Response Trust Trustee



#### Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1

#### Nevada Environmental Response Trust Site (Former Tronox LLC Site) Henderson, Nevada

#### Responsible Certified Environmental Manager (CEM) for this project

I hereby certify that I am responsible for the services described in this document and for the preparation of this document. The services described in this document have been provided in a manner consistent with the current standards of the profession and, to the best of my knowledge, comply with all applicable federal, state and local statutes, regulations and ordinances.

John M. Pekala, PG Senior Manager

11/15/2017

Date

Certified Environmental Manager Ramboll Environ US Corporation CEM Certificate Number: 2347 CEM Expiration Date: September 20, 2018

The following individuals provided input to this document:

John M. Pekala, PG Allan J. DeLorme, PE Elizabeth Miesner, MS Lynne Haroun, MPH Kimberly Kuwabara, MS Shuo Yu, PhD Amy Manion, PG Craig J. Knox Yuan Zhuang, PhD Jonathan Hunt, PhD Peyton Ward, MEM Erin H. Finestone, MPH Ruben So



DateNovember 15, 2017Prepared byRamboll EnvironDescriptionInterim Report<br/>Identification of COPCs and Decision Units for<br/>OU-1 Soils, Revision 1

Project No. 21-41400C

Ramboll Environ 2200 Powell Street Suite 700 Emeryville, CA 94608 USA T +1 510 655 7400 F +1 510 655 9517 www.ramboll-environ.com



## **CONTENTS**

1.	Introduction	1
1.1	Major Revisions	1
1.2	Report Organization	3
2.	Soil Investigations and Interim Removal Actions	4
2.1	Historical Soil Investigations and Interim Removal Actions	4
2.2	Remedial Investigation – Soil	7
2.3	Soil BHRA Data Set	10
3.	Data Usability Evaluation and Data Analysis	11
3.1	Data Usability Evaluation	11
3.2	Data Analysis	16
4.	Preliminary Identification of Soil COPCs	25
4.1	Step 1 – Concentration/Toxicity Screen	25
4.2	Step 2 – Background Evaluation	27
4.3	Step 3 – Chemical-Specific Evaluations	28
4.4	Study Area Preliminary COPCs	29
5.	Preliminary Identification of Decision Units	30
5.1	Spatial Data Analysis	30
5.2	Proposed Decision Units	32
5.3	Data Usability Evaluation for Individual Decision Units	33
5.4	Identification of Soil COPCs for Individual Decision Units	38
6.	Conclusions and Path Forward	43
7.	References	44

#### **LIST OF TABLES**

Table 1	Data Usability Evaluation
Table 2	Soil Sampling Results for Asbestos (Long Amphibole and Chrysotile Fibers)
Table 3	Evaluation of Sample Quantitation Limits
Table 4	Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics, and Radionuclides: Soil (0-10 ft bgs)
Table 5	Summary Statistics for Organic Compounds: Soil (0-10 ft bgs)
Table 6	Exploratory Data Analysis: Comments for Chlorine Oxyanions, Metals, Other Inorganics and Radionuclides (0-10 ft bgs Soils)
Table 7	Exploratory Data Analysis: Comments for Organic Compounds (0-10 ft bgs)
Table 8	Concentration/Toxicity Screen
Table 9	Results of the Background Evaluation for Metals Carried Forward from the Concentration/Toxicity Screen
Table 10	Results of the Background Evaluation for Radionuclides Carried Forward from the Concentration/Toxicity Screen
Table 11	Preliminary COPCs Identified for Study Area Soils (0-10 ft bgs)
Table 12	Cancer Risks for Radionuclides in Background Soils
Table 13	Evaluation of Sample Quantitation Limits for Individual Decision Units
Table 14	Summary Statistics and Concentration/Toxicity Screen for Individual Decision Units
Table 15	Results of the Background Evaluation for Metals Carried Forward from the Concentration/Toxicity Screen for Individual Decision Units
Table 16	Results of the Background Evaluation for Radionuclides Carried Forward from the Concentration/Toxicity Screen for Individual Decision Units
Table 17	Preliminary COPCs Identified for Soils (0-10 ft bgs) in Individual Decision Units

#### LIST OF FIGURES

- Figure 1 Operations Area Boundary and Tronox Leasehold Area
- Figure 2 Operations Area Features
- Figure 3 Study Area, Excavation Control Areas (ECAs), and Remediation Zone A (RZ-A)
- Figure 4 Soil Sampling Locations Included in the BHRA
- Figure 5 COPC Identification Flowchart
- Figure 6 Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Chlorate and Perchlorate

Figure 7	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Arsenic
Figure 8	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Chromium VI
Figure 9	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Cobalt
Figure 10	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Manganese
Figure 11	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Thallium
Figure 12	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Zirconium
Figure 13	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Uranium-238, Thorium-232 and Uranium-235
Figure 14	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Dioxin TEQs
Figure 15	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Benzo(a)pyrene Equivalents (BaPEqs)
Figure 16	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Naphthalene
Figure 17	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): 4,4'-DDE and 4,4'-DDT
Figure 18	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): beta-BHC, Dieldrin and Toxaphene
Figure 19	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Hexachlorobenzene
Figure 20	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Bis(2-Ethylhexyl)phthalate
Figure 21	Spatial Intensity Plot for Study Area Soils (0-10 ft bgs): Asbestos (Long Amphibole and Long Chrysotile Fibers)
Figure 22a	Spatial Concentration/Risk Plot for Dioxin TEQs (Soil Samples 0-10 ft bgs)
Figure 22b	Spatial Concentration/Risk Plot for Dioxin TEQs (Soil Samples 0-2 ft bgs)
Figure 22c	Spatial Concentration/Risk Plot for Dioxin TEQs (Soil Samples 2-10 ft bgs)
Figure 23a	Spatial Concentration/Risk Plot for Arsenic (Soil Samples 0-10 ft bgs)
Figure 23b	Spatial Concentration/Risk Plot for Arsenic (Soil Samples 0-2 ft bgs)
Figure 23c	Spatial Concentration/Risk Plot for Arsenic (Soil Samples 2-10 ft bgs)
Figure 24a	Spatial Concentration/Risk Plot for Cancer Risk (Soil Samples 0-10 ft bgs)
Figure 24b	Spatial Concentration/Risk Plot for Cancer Risk (Soil Samples 0-2 ft bgs)
Figure 24c	Spatial Concentration/Risk Plot for Cancer Risk (Soil Samples 2-10 ft bgs)
Figure 25a	Spatial Concentration/Risk Plot for Noncancer Hazard Indices (Soil Samples 0- 10 ft bgs)
Figure 25b	Spatial Concentration/Risk Plot for Noncancer Hazard Indices (Soil Samples 0- 2 ft bgs)
Figure 25c	Spatial Concentration/Risk Plot for Noncancer Hazard Indices (Soil Samples 2- 10 ft bgs)
Figure 26a	Spatial Concentration/Risk Plot for Radionuclide Cancer Risk (Soil Samples 0- 10 ft bgs)

- Figure 26b Spatial Concentration/Risk Plot for Radionuclide Cancer Risk (Soil Samples 0-2 ft bgs)
- Figure 26c Spatial Concentration/Risk Plot for Radionuclide Cancer Risk (Soil Samples 2-10 ft bgs)
- Figure 27 Spatial Concentration/Risk Plot for Long Amphibole Fibers (Soil Samples)
- Figure 28 Spatial Concentration/Risk Plot for Long Chrysotile Fibers (Soil Samples)
- Figure 29 Proposed Decision Units

#### **APPENDICES**

#### Appendix A

Response to the NDEP's December 22, 2016 Comments on the August 26, 2016 Interim Report, Identification of COPCs and Exposure Units for Soils

#### Appendix B

Data Validation Summary Reports (CD) and Tables

- Table B-1Summary of Soil Samples Changed in the BHRA Data Set
- Table B-2Summary of Soil Data Excluded During Data Processing
- Table B-3 Summary of Parcel Rejected Soil Data
- Table B-4Summary of Qualified Soil Field Duplicates
- Table B-5
   Revisions of Censored Data for Blank Contamination
- Table B-6Summary of J Qualified Soil Data

#### Appendix C

Soil BHRA Data Set

C-1	Soil BHRA Data Set - Chemicals and Radionuclides (CD)
C-2	Soil BHRA Data Set – Asbestos
Figure C-1	Soil Sampling Locations Included in the BHRA

#### Appendix D

Soil Data Summary Statistics

- Table D-1Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics,<br/>and Radionuclides in Soil (0-10 ft bgs)
- Table D-2Summary Statistics for Organic Compounds in Soil (0-10 ft bgs)

#### Appendix E

RZ-A and BRC/TIMET Soil Background Data Sets (CD)

Figure E-1 RZ-A Soil Background Sample Locations

#### Appendix F

Background Evaluation for Metals and Radionuclides

- Table F-1Summary Statistics for Metals in RZ-A Background Soils and Study<br/>Area Soils (0-10 feet bgs)
- Table F-2Background Comparisons for Metals in Study Area Soils (0-10 feet<br/>bgs)

Table F-3	Summary Statistics for Radionuclides in RZ-A Background Soils and
	Study Area Soils (0-10 feet bgs)
Table F-4	Background Comparisons for Radionuclides in Study Area Soils
	(0-10 feet bgs)
Table F-5A	Equivalence Test for Secular Equilibrium of Uranium Decay Series
Table F-5B	Equivalence Test for Secular Equilibrium of Thorium Decay Series
Table F-6	Correlation Matrices for the Uranium Decay Series and the Thorium
	Decay Series
Figures F1-1 th	rough F1-32 – Background vs. Study Area Boxplots (Metals)
Figures F1-33 t	hrough F1-40 – Background vs. Study Area Boxplots (Radionuclides)
Figures F2-1 th	rough F2-32 – Normal and Lognormal Q-Q Plots (Metals)
Figures F2-33 t	hrough F2-40 – Normal and Lognormal Q-Q Plots (Radionuclides)

### Appendix G

Spatial Quartile Plots

#### Appendix H

NDEP Flowchart for Radionuclide Data Usability

#### Appendix I

ProUCL Output Files

#### Appendix J

Decision Unit-Specific Background Evaluation for Metals and Radionuclides

Table J-1	Summary Statistics for Metals in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs)
Table J-2	Background Comparisons for Metals in Decision Unit Soils (0-10 feet bgs)
Table J-3	Summary Statistics for Radionuclides in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs)
Table J-4	Background Comparisons for Radionuclides in Decision Unit Soils (0- 10 feet bgs)
Table J-5A	Equivalence Test for Secular Equilibrium of Uranium Decay Series (U-238 Chain)
Table J-5B	Equivalence Test for Secular Equilibrium of Thorium Decay Series (Th-232 Chain)
Table J-6	Correlation Matrices for the Uranium Decay Series and the Thorium Decay Series
Figures J1-1 th	rough J1-7 – Background vs. Decision Unit Boxplots (Metals)
Figures J1-8 th	rough J1-15 – Background vs. Decision Unit Boxplots (Radionuclides)
Figures J2-1 th	rough J2-7 – Normal and Lognormal Q-Q Plots (Metals)
Figures J2-8 th	rough J2-15 – Normal and Lognormal Q-Q Plots (Radionuclides)

## ACRONYMS AND ABBREVIATIONS

ACM	asbestos-containing material
AECOM	AECOM, Inc.
AP	ammonium perchlorate
BaP	benzo(a)pyrene
BaPEq	benzo(a)pyrene equivalent
BCL	Basic Comparison Level
bgs	below ground surface
BHC	hexachlorocyclohexane
BHRA	baseline human health risk assessment
BMI	Black Mountain Industrial
box plot	box-and-whisker plot
BRC	Basic Remediation Company
BTEX	benzene, toluene, ethyl benzene, and total xylenes
CAS	Chemical Abstract Service
COPC	chemical of potential concern
CSM	conceptual site model
DDE	dichlorodiphenyldichloroethylene
DDT	dichlorodiphenyltrichloroethane
DQI	data quality indicator
DU	decision unit
DUE	data usability evaluation
DVSR	Data Validation Summary Report
ECA	Excavation Control Area
EDA	exploratory data analysis
ENSR	ENSR Corporation
ENVIRON	ENVIRON International Corporation
EPC	exposure point concentration
EU	exposure unit
GRAS	Generally Recognized as Safe
GWETS	groundwater extraction and treatment system
ні	hazard index

HRA	health risk assessment
HQ	hazard quotient
IQR	interquartile range
IWF	Interceptor Well Field
LOU	Letter of Understanding
MTBE	methyl tertiary butyl ether
NDEP	Nevada Division of Environmental Protection
Neptune	Neptune and Company
NERT	Nevada Environmental Response Trust
Northgate	Northgate Environmental Management, Inc.
OCH	organochlorine herbicide
OCP	organochlorine pesticide
Operations Area	the area comprising the Site, excluding Parcels C, D, E, F, G, and H $$
OPP	organophosphate pesticide
OSSM	Olin Chlor Alkali/Stauffer/Syngenta/Montrose
OU-1	Operable Unit 1, also referred to as the Nevada Environmental Response Trust Site or the Site
p-value	calculated probability
РАН	polycyclic aromatic hydrocarbon
PCB	polychlorinated biphenyl
Phase A investigation	Phase A Source Area Investigation
Phase B investigation	Phase B Source Area Investigation
PQL	practical qualification limit
Q-Q	quartile to quartile plot
QAPP	Quality Assurance Project Plan
QC	quality control
Ra	radium
Ramboll Environ	Ramboll Environ US Corporation
RAW	Removal Action Work Plan for Phase B Soil Remediation
RI/FS	Remedial Investigation and Feasibility Study
RPD	relative percent difference
RSL	regional screening level
RZ	remediation zone

selective ion monitoring
Nevada Environmental Response Trust Site
Site Management Plan
sample quantitation limit
site-related chemical
the area that is the subject of this interim report and that will be evaluated in the BHRA
semivolatile organic compound
toxicity equivalency factor
toxicity equivalent
thorium
Titanium Metals Corporation
total petroleum hydrocarbons
Tronox, LLC
Nevada Environmental Response Trust
uranium
upper confidence limit
Upper Muddy Creek Formation
United States Environmental Protection Agency
volatile organic compound
World Health Organization
cubic yard
feet, foot
milligram per kilogram
microgram per kilogram

## 1. INTRODUCTION

This interim report has been prepared by Ramboll Environ US Corporation (Ramboll Environ) on behalf of the Nevada Environmental Response Trust (NERT or the Trust) and presents a preliminary list of soil chemicals of potential concern (COPCs) and identifies decision units (DUs) for evaluation in the baseline human health risk assessment (BHRA) that will be prepared for a 143-acre area referred to in this report as the "Study Area." The Study Area is located within the NERT Site in Henderson, Nevada ("Site").

The NERT Site comprises 346 acres and has been designated as Operable Unit 1 (OU-1) for the Remedial Investigation and Feasibility Study (RI/FS) currently in progress (Figure 1). Within the NERT Site, the Operations Area<sup>1</sup> is a 259-acre area used by the Trust and their tenant, Tronox, for site operations. Tronox currently leases approximately 114 acres within the Operations Area, on which it operates a chemical manufacturing business; three subtenants to Tronox (Angelo & Newton, Industrial Supply, and Pronto Constructors) also have facilities within the Operations Area (Figure 2). The Operations Area excludes six parcels (Parcels C, D, E, F, G, and H, comprising 87 acres) that are not currently used by the Trust or Tronox.

The BHRA Study Area is an approximately 143-acre non-contiguous area within the Operations Area (Figure 3). The BHRA Study Area excludes 116 acres of the Operations Area consisting of remediation zone A (RZ-A), a contiguous 28-acre area in the southern portion of the Operations Area (see Section 2.1.2) and 38 excavation control areas (ECAs) (approximately 88 acres).

Besides the risk for the Study Area being evaluated under this BHRA, the risks for all other property within OU-1 are being evaluated and/or managed. Specifically, post-remediation HRAs for Parcels C, D, F, G and H are currently underway. A health risk assessment (HRA) has been previously completed for RZ-A (see Section 2.1.2). The 38 ECAs are subject to the Site Management Plan (SMP) (Ramboll Environ 2017), which describes measures to mitigate risks to human health and the environment related to potential exposures to residual COPCs during periods of typical site operations. Considering the risk mitigation measures in place, the Nevada Division of Environmental Protection (NDEP) concurred during a meeting in July 2015 that a BHRA would not be required for the ECAs (Ramboll Environ 2015b). Because soil in Parcel E has not been investigated, the Trust is removing Parcel E from the remedial investigation and BHRA. The Trust is currently reviewing information on the historical use of Parcel E and will develop an approach for investigating Parcel E, if necessary, for discussion with NDEP.

#### 1.1 Major Revisions

Key steps of the BHRA include 1) selection of soil COPCs for which cancer risks and noncancer hazards will be evaluated quantitatively; and 2) identification of DUs for evaluation. In May 2015, the first interim report, *Preliminary Selection of Facility Area COPCs* (2015 interim COPC report) (Ramboll Environ 2015a), was submitted to NDEP; that report focused primarily on the identification of COPCs. In July 2015,

<sup>&</sup>lt;sup>1</sup> The Operations Area is equivalent to the area referred to as the "Facility Area" in previous reports (with the exception of Parcel E, previously considered as part of the Facility Area for risk assessment purposes). These reports include, e.g., the Remedial Investigation and Feasibility Study Work Plan (ENVIRON International Corporation [ENVIRON] 2014a) and the associated risk assessment work plan and report (ENVIRON 2014b, Ramboll Environ 2015a).

representatives of NDEP, NDEP consultants, the Trust, and Ramboll Environ met to discuss the 2015 interim COPC report, and more generally, to identify a path forward for the soil BHRA (Ramboll Environ 2015b). In August 2016, the second interim report, *Identification of COPCs and Exposure Units for Soils*, (2016 interim COPC/EU report) (Ramboll Environ 2016a), was submitted to NDEP. The 2016 interim COPC/EU report updated the list of soil COPCs for the Study Area<sup>2</sup> consistent with the approach agreed upon during the July 2015 meeting. In addition, soil analytical data as part of the Phase 1 Remedial Investigation (RI) were added to the risk assessment data set, and the data usability evaluation (DUE) and a preliminary identification of EUs were included. In December 2016, NDEP provided comments on the 2016 interim COPC/EU report (NDEP 2016).

This report has been prepared to address NDEP comments on the 2016 interim COPC/EU report. NDEP comments, along with the Trust's responses, are included in Appendix A. In addition, this report updates the soil BHRA data set and has been further revised for consistency with recent updates to NDEP guidance. The primary revisions made to this report, as compared with the previously submitted 2016 interim COPC/EU report, are summarized below:

• Changes in the DU identification: In the 2016 interim COPC/EU report, the proposed six EUs were identified based mainly on the current/anticipated future land use and the distribution of COPCs in soil as shown on spatial intensity plots for each chemical by concentration. Because it is difficult to systematically review the spatial plots of all the COPCs in order to evaluate the spatial patterns in concentrations that could be used to identify DUs, the current report reduces the dimensions of the spatial data analysis by focusing on risk-relevant spatial patterns. Spatial risk plots were prepared according to the two health endpoints - cancer risk and noncancer hazard. In addition, dioxin and arsenic concentrations were compared to their site-specific remediation level and background concentrations, respectively.

As agreed with NDEP (and documented in NERT Meeting Minutes dated December 19, 2016, see Appendix A), spatial plots of risk results were initially prepared for the Study Area as a whole. These plots were then used to identify discrete DUs to be evaluated in the BHRA. Based on this evaluation, the Study Area was divided into three DUs.<sup>3</sup> The three DUs include DU-1, the northern portion of the Study Area (north of the L'hoist North America facility), DU-2, the southern portion of the Study Area (south of the L'hoist North America facility), and DU-3, the Central Retention Basin. The identification of the three DUs is discussed in detail in Section 5. DU-specific DUEs and COPC identifications have also been conducted.

• Changes in the soil BHRA data set: Minor changes to the Site, ECA, and parcel boundaries based on current information resulted in a few soil samples that were no longer located within the Study Area, and these samples have been removed from the BHRA data set. A few soil samples that were either identified as within or outside an ECA have been updated; those identified as outside an ECA are included in the BHRA data set. In addition, several soil samples (i.e., asbestos samples) included in

<sup>&</sup>lt;sup>2</sup> The 2015 interim COPC report identified COPCs for areas referred to as "outside ECA soils" and "inside ECA soils." Consistent with the decision to exclude ECAs from the BHRA, the 2016 interim COPC/EU report identified COPCs for "outside ECA soils," or equivalently, the Study Area.

<sup>&</sup>lt;sup>3</sup> The term "Decision Unit" is used in place of the term "Exposure Unit" in this report.

the 2016 BHRA data set were confirmed to be associated with soil removed during the interim soil removal actions conducted in 2010 and 2011, and a few soil samples have been identified as being beneath Pond Mn-2 and inaccessible for soil contact; these samples have been excluded from the BHRA data set. The depths of soil samples after the removal actions have been reconfirmed to make sure that only the samples collected in the top 10 feet (ft) below ground surface (bgs) are included in the BHRA data set. Soil samples that have been changed in the current BHRA data set, compared to the 2016 BHRA data set, are identified in Section 3.1.1.

- **Background evaluation:** In the 2016 interim COPC/EU report, the RZ-A background data set was used to evaluate background conditions for metals and radionuclides. As requested by NDEP (Neptune and Company [Neptune] 2017), in the DU-specific analysis, the regional Basic Remediation Company (BRC)/Titanium Metals Corporation (TIMET) data set (BRC/TIMET 2007) was used for the northern portion of the Study Area and the RZ-A background data set was used for the southern portion of the Study Area. Use of the RZ-A background data set was retained for analysis of the entire Study Area.
- **Expansion of DUE:** The DUE has been expanded per NDEP comments by including detailed discussion and data tables for data quality indicators (DQIs).
- **Update of Soil Screening Levels.** The Basic Comparison Levels (BCLs) used for the concentration/toxicity screen have been updated to be consistent with the most recent NDEP revision to the BCLs (NDEP 2017).

The COPCs and DUs established in this report will be used for the evaluation of soil exposure pathways in the BHRA. COPCs for evaluation of the vapor intrusion pathway are not identified in this report. It is anticipated that for soil gas, all analytes detected in one or more samples will be identified as COPCs.

#### 1.2 Report Organization

The following elements are included in the remainder of this interim report:

- Section 2 summarizes soil investigations and interim soil removal actions conducted at the Operations Area from 2006 through 2014.
- Section 3 identifies the sources of soil analytical data available for the BHRA and presents the DUE, including the data analysis step of the DUE.
- Section 4 describes the COPC selection process and presents a preliminary list of soil COPCs for the Study Area.
- Section 5 presents spatial data analysis for the preliminary soil COPCs, identifies the proposed DUs, and provides DU-specific DUEs and COPC identifications.
- Section 6 lists the references cited in this report.

Supporting tables, figures, and appendices follow the text of the report.

# 2. SOIL INVESTIGATIONS AND INTERIM REMOVAL ACTIONS

The following sections summarize soil investigations and interim soil removal actions conducted within the Operations Area (including the Study Area) since the 2005 conceptual site model (CSM) (ENSR Corporation [ENSR] 2005). These investigations include "historical" investigations and removal actions, i.e., those conducted before the formation of the Trust as well as the ongoing RI for the Operations Area. The soil samples collected during these investigations provide the soil analytical data for the BHRA.

#### 2.1 Historical Soil Investigations and Interim Removal Actions

The following sections describe the historical soil investigations and interim soil removal actions conducted prior to the RI.

#### 2.1.1 Historical Soil Investigations

In 2005, a CSM report (2005 CSM report) was prepared for the Site that integrated information from the soil and groundwater investigations conducted prior to 2005 to document information on site-specific sources, release mechanisms, transport pathways, exposure routes, and potential receptors (ENSR 2005). Historical site investigations conducted since completion of the 2005 CSM report include the Phase A and Phase B Source Area Investigations, which were designed to further characterize soil, groundwater, and soil gas across the Site (ENSR 2006, 2007a, 2008a, AECOM, Inc. [AECOM] 2008).

The objectives of the Phase A Source Area Investigation (Phase A investigation) were to refine the 2005 CSM, further characterize site conditions, and provide data for future risk assessments. To identify and characterize the distribution of site-related chemicals (SRCs) in soils, the investigation focused on soil conditions associated with the 192 SRCs identified in the 2005 CSM report and their suspected source areas. A total of 127 soil samples were collected in November and December 2006 from 27 suspected source area locations. The sample locations were selected based on results of past site investigations, as presented in the CSM report (ENSR 2005), information on chemical use at the Site, and the 70 Letter of Understanding (LOU) study areas identified by NDEP in 1994 (NDEP 1994). In addition to the 192 SRCs previously identified, 44 additional constituents were analyzed for and reported by the laboratory.

During the Phase A investigation, soil samples were collected at depths of 0.5 to 1 ft and at 10 ft intervals thereafter, until groundwater was encountered (ENSR 2006). The samples were analyzed for metals; volatile organic compounds (VOCs), including fuel oxygenates; semivolatile organic compounds (SVOCs); polychlorinated biphenyls (PCBs); dioxins and furans; total petroleum hydrocarbons (TPHs as gasoline, diesel, and oil range organics); organochlorine herbicides (OCHs); organochlorine pesticides (OCPs); and organophosphate pesticides (OPPs). In addition, analyses were conducted for radionuclides, asbestos (surface soil samples only), and wet chemistry constituents. Not all samples were analyzed for all analytes, and at some locations, samples were collected at more frequent depth intervals. Samples were also collected from the manganese tailings stockpile (Figure 2) for analysis of metals and radionuclides. Two near surface

soil samples (1.5 to 3 ft bgs) were collected and analyzed for physical and geotechnical parameters.

Considering the results of the Phase A investigation, the objective of the Phase B Source Area Investigation (Phase B investigation) was to further characterize and evaluate the LOUs in the Operations Area and evaluate their potential impact on soils. Samples were collected at initial soil depths of 0.5 and 10 ft bgs, at the capillary fringe, and at the midpoint between the capillary fringe and 10 ft bgs, without exceeding 20 ft between each vertical sample. Judgmental samples were collected at 0.5 and 10 ft bgs in locations where certain surface features were noted, e.g., minor stains or above-ground pipelines. Soil samples were analyzed for the following analytical groups and analytes: metals (including hexavalent chromium), VOCs, SVOCs, organic acids, PCB Aroclors and congeners, dioxins/furans, OCPs, OPPs, TPHs, chlorate, perchlorate, cyanide, formaldehyde, and radionuclides. In addition, surface samples were collected from 0 to 2 inches and analyzed for asbestos fibers. Samples for wet chemistry and geotechnical parameters were also collected (AECOM 2008).

Supplemental sampling of soils for the Phase B investigation was conducted in December 2009 in accordance with two Tronox memoranda entitled *Scope for Additional Sampling of Area I* (Northgate Environmental Management, Inc. [Northgate] 2009a) and *Scope for Additional Sampling of Area II* (Northgate 2009b). A total of 129 soil samples were collected at or near Phase B investigation locations where reported concentrations of constituents exceeded NDEP BCLs. The purpose of the sampling was to fill remaining data gaps in the pre-excavation data and to provide additional information for excavation planning.

#### 2.1.2 Interim Soil Removal Actions

Interim soil removal actions were conducted in 2010 and 2011 in response to a 2009 order issued by NDEP to Tronox to remove impacted soil from the Site to minimize potential health risks associated with the continued presence of contaminated soil (NDEP 2009a). The Phase A and B investigations identified a number of SRCs within the upper 10 ft of soil with reported concentrations greater than NDEP worker BCLs or modified risk-based goals (as agreed upon by NDEP). These constituents included metals; SVOCs (including hexachlorobenzene); PCBs; OCPs; dioxins/furans; asbestos; and perchlorate. Based on the investigation findings, a detailed scope of work for soil removal was developed, as presented in the *Removal Action Work Plan for Phase B Soil Remediation of Remediation Zones RZ-B through RZ-E* (the RAW) (Northgate 2010a).

For purposes of soil excavation activities, the Operations Area was divided into five RZs based roughly on geographic groupings of elevated detections and CSM considerations (Northgate 2010b), as follows:

- RZ-A: area in the southern portion of the Site;
- RZ-B: area around the Unit Buildings;
- RZ-C: ammonium perchlorate (AP) production area, Koch Materials area, pond and diesel storage tank area, and manganese tailings stockpile;
- RZ-D: former Trade Effluent ponds and AP pad/drum recycling area (including the former hazardous waste landfill); and
- RZ-E: Beta Ditch.

For RZ-A, the results of a HRA for soils (Northgate 2010c) indicated that exposures to residual chemicals in the upper 10 ft of soil were below NDEP's point of departure for cancer risks and noncancer effects. Specifically, the estimated cancer risks were less than one-in-a-million  $(1 \times 10^{-6})$  for indoor commercial workers, outdoor commercial/industrial workers, and construction workers, and the HIs were less than one (1). The upper-bound risks of death from lung cancer or mesothelioma for asbestos exposures to outdoor commercial/industrial workers were less than or equal to  $1 \times 10^{-6}$  for long chrysotile and long amphibole fibers. The best estimate and upper-bound risk estimates for construction workers were less than or equal to  $1 \times 10^{-6}$  for long chrysotile fibers and ranged from zero to  $6 \times 10^{-5}$  for long amphibole fibers. Based on the HRA results, RZ-A was not included in the interim soil removal action.

For RZ-B through RZ-E, Voronoi/Thiessen polygons were generated to define areas of impacted or contaminated soils (Northgate 2010a). The RAW defined contaminated soil as areas with concentrations exceeding worker BCLs or other NDEP-approved concentrations. The remediation strategy consisted of 1) excavation of soils within designated polygons, 2) sampling of discolored soil, 3) removal of discolored soil where above BCLs or otherwise deemed appropriate to remove, and 4) designation of ECAs for inaccessible areas, including certain areas with contaminants above BCLs and/or discolored soil left in place.

To further define the areas for excavation (i.e., the polygons), pre-confirmation sampling was conducted from April to November 2010 in accordance with a pre-confirmation sampling work plan (Northgate 2010b). Borings were advanced during the pre-confirmation sampling program, as follows: 1) 84 borings at existing locations (adjacent to Phase A and B investigation sampling locations) and 2) 91 borings at new locations. Data from "existing locations" were used to establish polygon depths, while data from "new locations" were used to define the horizontal and vertical extent of excavation of near-surface soils (i.e., 0 to 10 ft bgs soils).

Discolored soil was encountered in various locations during removal activities. Considering 1) the location of the discolored soil, 2) available analytical results from adjacent or nearby areas, 3) the anticipated extent of discolored soil, and 4) the excavation activities currently in progress, some areas of discolored soil were removed. Other areas of discolored soil were sampled and evaluated to determine if the soil should be removed or could be left in place in accordance with the *Work Plan for Evaluation of Discolored Soil and Confirmation Soil Sampling in Visually-Impacted Areas* (ENVIRON 2011). Where removal of discolored soil was conducted, confirmation soil samples were collected to verify that remaining soil concentrations were below BCLs. Typically, if the analytical results indicated that concentrations were above BCLs, additional soil was removed and additional confirmation soil sampling was performed.

The interim soil removal action was conducted between August 2010 and November 2011, in which accessible soils with concentrations greater than BCLs were removed down to a maximum depth of 10 ft bgs. An estimated 567,770 cubic yards (cy) of contaminated soil (not including asbestos-containing material [ACM]) were removed during this period. An estimated 11,026 cy of asbestos-containing soil and 1,419 linear feet of ACM piping were also removed and disposed of as part of the soil removal action. Excavated areas were partially backfilled and graded with clean fill from on- and off-site sources. The final grading plan included construction of two retention basins, the Central Retention Basin and the Northern Retention Basin (shown on Figure 2). Areas of inaccessible soils (with concentrations greater than BCLs) and incompletely characterized soils (due to access issues) were designated as ECAs. Removal activities and post-removal conditions at the Site are described in details in the *Revised Interim Soil Removal Action Completion Report* (ENVIRON 2012). The 38 current ECAs (which are subject to change depending on site investigation and/or interim removal actions), are listed in Appendix A of the most recent SMP (Ramboll Environ 2017).

An interim removal action was also conducted at the manganese tailings stockpile, as presented in the *Manganese Tailings Removal Technical Memorandum* (Northgate 2012). The manganese tailings stockpile was located in RZ-C, north of the Manganese Leach Plant and south of the Mn-1 Pond (in the current location of the Mn-2 Pond, as shown on Figure 2). The area is approximately 8.6 acres and was used from 1975 through 2004 for the disposal of manganese tailings from the leach plant process. Manganese tailings from all former storage locations at the Site were consolidated to this location and covered with soil sometime prior to 1985. Since 2004, manganese tailings from Tronox operations (current tailings production) have been shipped to an off-site landfill.

A total of 284,232 tons of tailings and minor debris were removed from the manganese tailings stockpile as part of the interim removal action. In accordance with a request from NDEP, a confirmation sampling program was implemented subsequent to tailings removal. Based on the results of the confirmation sampling program, additional shallow soil excavation was conducted concurrent with Phase B soil remediation in accordance with the RAW and Appendix A of the *Revised Excavation Plan for Phase B Soil Remediation of RZ-C, Addendum to the Remedial Action Work Plan* (Northgate 2010d). The post-confirmation sampling excavation was conducted to address soil that contained concentrations of arsenic, cobalt, manganese and/or asbestos that exceeded NDEP worker BCLs or modified risk-based goals (Northgate 2010a, 2010d).

One ECA has been remediated since the ECAs were established in 2012. ECA #E3 (Facilities at East End of the Beta Ditch) was remediated via soil excavation in October 2013, concurrent with TIMET's excavation and grading of their property. Confirmation soil sample results were collected. The activities are detailed in the *Excavation of Beta Ditch at NERT-TIMET Property Line* report (ENVIRON 2014c).

#### 2.2 Remedial Investigation – Soil

The following sections describe the soil investigations being performed as part of the RI.

#### 2.2.1 Phase 1 RI Data Gap Investigation

In accordance with the 2011 Interim Consent Agreement between the Trust and NDEP, the Trust is in the process of conducting a RI/FS. Per the RI/FS Work Plan (ENVIRON 2014a), Ramboll Environ conducted field work for a Phase 1 RI data gap investigation between October 2014 and May 2015. The Phase 1 data gap investigation was deemed necessary for areas that required further characterization to determine the nature and extent of COPCs in soil and groundwater. For purposes of soil characterization, four main areas were initially identified for collection of additional physical and chemical data from both shallow and deep soils. These areas included the AP-5 pond area, the debris pile, soil in the area between the debris pile and AP-5 pond, and the area west of the Mn-1 pond (Figure 2). Based on further review, additional investigation areas were identified and the Phase 1 RI soil data gap investigation focused on eight investigation areas

(Areas 1 through 8) within the Operations Area. Investigation was performed in Areas 2 through 8. Soil samples from Area 1 are scheduled to be collected upon the completion of the AP-5 pond decommissioning, currently anticipated to be completed in 2018 or 2019.

In Areas 2 through 8, approximately 450 soil samples were collected for chemical analysis and 35 soil samples were collected for determination of physical properties from 55 soil borings, eight on-site well pilot borings, and three test pits. The results of the Phase 1 data gap investigation are summarized in the *Technical Memorandum, Remedial Investigation Data Evaluation* (Ramboll Environ 2016b). Additional data gaps to be addressed in the Phase 2 RI Data Gap Investigation (Phase 2 RI Investigation) were identified in the same submittal.

All of the applicable soil data within the Study Area from the Phase 1 RI are included in the current soil BHRA data set.

#### 2.2.2 Phase 2 RI Data Gap Investigation

In accordance with the *Remedial Investigation Data Evaluation Technical Memorandum* (Ramboll Environ 2016b), the Trust implemented a second phase of remedial investigation (Phase 2 RI) from January to November 2017. Field work was conducted both at the NERT Site and within the Off-Site RI Study Area. The primary purposes of the Phase 2 RI were to obtain data necessary to further understand the nature and extent of impacts to soil and groundwater and to obtain data necessary to conduct feasibility study evaluations for the selection of the final remedy.

At the NERT Site, 63 monitoring wells and 57 borings were installed as part of the Phase 2 RI. Areas of investigation included 1) the future downgradient Site boundary, 2) between the Interceptor Well Field (IWF)/barrier wall and the downgradient Site boundary, 3) the central area of the Site, 4) west of the GW-11 pond, and 5) upgradient of the IWF. Approximately 957 soil samples were collected for chemical analysis and approximately 292 samples were collected for physical properties.

All soil borings and monitoring well installations were completed using rotary sonic drilling methods. Monitoring well development at the Site and within the Off-Site NERT RI Study Area and an initial round of groundwater sampling at all new Phase 2 monitoring wells were completed following drilling activities. In addition, slug testing was conducted in all new Phase 2 monitoring wells, as well as in 10 nearby existing Trust owned wells that had not been tested previously, in order to provide more complete hydraulic characterization in the Upper Muddy Creek Formation (UMCf) on-Site and in areas outside the paleochannels off-Site.

The results of Phase 1 and Phase 2 RI, along with the results of the AECOM Downgradient Study Area effort and the Tetra Tech Unit 4/5 Buildings Investigation (see Section 2.2.3 below), will be incorporated into a comprehensive RI report currently slated to be submitted to NDEP in late 2018.

Most soil samples collected during the Phase 2 RI were collected at depths below 10 feet bgs. It is anticipated that approximately 100 soil samples collected from 50 boring locations during this investigation will be added to the soil BHRA data set.

#### 2.2.3 Phase 3 Data Gap Investigation

As discussed in the *RI/FS Work Plan Addendum: Phase 3 Remedial Investigation*, *Revision 1*, submitted to NDEP on October 6, 2017 (currently awaiting NDEP approval), the Trust plans to implement a third phase of remedial investigation (Phase 3 RI) within the Eastside Study Area, located immediately east of the NERT Site and Off-Site Study Area. The investigation is designed to determine the extent of COPC contamination originating from the NERT Site in the Eastside Study Area, obtain data to support future feasibility study evaluations to address COPCs, and to assist in the selection of the final remedy for COPCs in the Eastside Study Area.

Work performed as part of the Phase 3 RI will be outside of OU-1; therefore, soil data obtained during the investigation will not be included in this soil BHRA data set.

#### 2.2.4 Unit 4 and 5 Buildings Investigation

As part of the RI/FS, the *Unit 4 and 5 Buildings Investigation Work Plan* (Tetra Tech 2015) was submitted to NDEP on March 30, 2015 and approved by NDEP on April 13, 2015. The work plan documented the proposed environmental investigation in the area of the Unit 4 and 5 buildings located at the Site. The work plan replaced Section 5.4.1.2 of the RI/FS Work Plan and included demolition of the Unit 4 building floor, construction of an access ramp, and an environmental investigation using conventional drilling techniques on the basement level of the building. The work was divided into three field mobilizations and subsequent reporting, as summarized below.

#### First Field Mobilization

Fieldwork for the first mobilization was conducted in late 2015 and included advancing four boreholes and collecting discrete-depth groundwater samples from the four exterior corners of the Unit 4 cell floor. The objective of the first mobilization was to obtain preliminary lithologic and analytical data that would be used as a baseline to direct and refine the scope of work for the second field mobilization.

Results of the first field mobilization are summarized in the *Technical Memorandum: Unit 4 and 5 Buildings Investigation First Mobilization* (Tetra Tech 2016a), which was submitted to NDEP on May 6, 2016 and approved on June 28, 2016 following submittal of a response to NDEP comments on June 24, 2016 (Tetra Tech 2016b). Soil samples were collected from four boreholes outside of the four corners of the Unit 4 Building. Since these soil borings were located within ECA B1, these soil samples are not included in the soil BHRA data set.

#### Second Field Mobilization

Fieldwork for the second mobilization was conducted from June 2016 to January 2017, and was summarized in the *Technical Memorandum: Unit 4 and 5 Buildings Investigation Second Mobilization* (Tetra Tech 2017a), which was submitted to NDEP on May 4, 2017. NEDP approved this memorandum on June 8, 2017 along with a request for a written response about how their comments on the memorandum will be addressed in the third mobilization or final RI report. A response to NDEP comments was submitted to NDEP on July 12, 2017 (Tetra Tech 2017b) and approved by NDEP on August 15, 2017.

Sixty nine boreholes were advanced throughout the investigation area as part of the second field mobilization, including 47 boreholes along five transects, three boreholes in the vicinity of a sump located along the southwest corner of the Unit 4 basement, and

additional 19 boreholes that were added during the second mobilization. Borings were advanced to depths ranging from 90 to 250 feet. Soil and discrete-depth groundwater samples were collected from the borings at selected intervals and analyzed for perchlorate, hexavalent chromium, total chromium, VOCs, total dissolved solids (groundwater only), nitrate and sulfate, and chlorate.

Most soil samples collected during the second mobilization of Unit 4 and 5 Buildings Investigation were collected at depths below 10 feet bgs or within ECAs. It is anticipated that approximately 85 soil samples collected from 17 boring locations from this investigation will be added to the soil BHRA data set.

#### Third Field Mobilization

Fieldwork associated with the third mobilization is anticipated to be completed in the fourth quarter of 2017 or early January 2018. The third mobilization includes the installation of five to seven monitoring wells and the sampling of new wells and existing wells in the vicinity of the investigation area (Tetra Tech 2015). The advancement of additional boreholes within the investigation area to supplement data collected during the second mobilization was also recommended (Tetra Tech 2017a). Any soil samples collected outside of the ECA and within the Study Area, will be included in the BHRA. Following completion of the third mobilization, the results from all three field mobilizations will be incorporated into the comprehensive RI report.

#### 2.3 Soil BHRA Data Set

As discussed further in Section 3, the current soil BHRA data set includes 1,131 soil samples collected from 0-10 ft bgs at 431 locations within the Study Area. The soil sampling locations are shown on Figure 4. It is anticipated that approximately 185 soil samples collected from 0-10 ft bgs at 67 locations within the Study Area from the Phase 2 RI and Tetra Tech Unit 4 and 5 Buildings Investigation will be added to the updated soil BHRA data set in the forthcoming BHRA report.

## 3. DATA USABILITY EVALUATION AND DATA ANALYSIS

This section presents the DUE. Section 3.1 presents the first component of the DUE, in which the available soil data are reviewed to ensure that the quality of the data is sufficient to support the BHRA; this component of the evaluation focuses on the quality of each individual data point. Section 3.2 presents the data analysis component of the DUE, which focuses on the entire BHRA data set. In particular, through statistical summaries, background evaluation (for soil metals and radionuclides only), spatial plots, and other exploratory analyses, the data are reviewed relative to our current understanding of the Study Area (as represented by the CSM) and for possible data gaps or other investigation issues.

#### 3.1 Data Usability Evaluation

The DUE was conducted in accordance with NDEP's Supplemental Guidance for Assessing Data Usability for Environmental Investigations at the Black Mountain Industrial (BMI) Facility in Henderson, NV (NDEP 2010a), which is based on the United States Environmental Protection Agency's (USEPA's) Guidance for Data Usability in Risk Assessment (Parts A and B) (USEPA 1992a, b). The USEPA DUE framework provides the basis for identifying and evaluating uncertainties in HRAs with regard to site characterization data. USEPA (1992a) states that "data usability is the process of assuring or determining that the quality of data generated meets the intended use," and that when risk assessment is the intended use, USEPA's guidance "provide[s] direction for planning and assessing analytical data collection activities for the HRA." USEPA has established a specific framework to provide risk assessors a consistent basis for making decisions about the minimum quality and quantity of environmental analytical data to support risk assessment decisions (USEPA 1992a, b; NDEP 2010a). The USEPA data usability guidance identifies the following data quality criteria for evaluating the usability of site investigation data in the risk assessment process:

- Criterion I Reports to Risk Assessor;
- Criterion II Documentation;
- Criterion III Data Sources;
- Criterion IV Analytical Methods and Detection Limits;
- Criterion V Data Review; and
- Criterion VI Data Quality Indicators.

In this section, Criteria I through VI are evaluated for the entire Study Area and not for the individual proposed DUs. It is noted that the conclusions reached for Criteria I through V for the Study Area also apply to the individual DUs. Review of the DU-specific data relative to Criterion VI is discussed in Section 5.3.

The soil data set evaluated using the data quality criteria is identified in Section 3.1.1. Sections 3.1.2 through 3.1.7 briefly describe the evaluation criteria and results of the evaluation. The detailed results are presented in tabular form (Table 1) using the worksheet templates provided by NDEP (2010a).

#### 3.1.1 Soil Data Set and Data Processing

The soil BHRA data set comprises the analytical results that remain representative of current conditions within the Study Area. Specifically, the data set includes samples collected at 0-10 ft bgs<sup>4</sup> as part of the following investigations:

- "Remaining" soil samples<sup>5</sup> from the Phase A investigation in 2006;
- "Remaining" soil samples from the Phase B investigation in 2008 and 2009;
- "Remaining" soil samples from the Phase B supplemental investigation in 2009;
- "Remaining" pre-confirmation soil samples in 2010;
- "Remaining" confirmation samples collected in 2011 to inform the interim removal actions; and
- RI investigation samples collected in 2014.

Additional RI soil samples collected from the 0-10 ft depth interval within the Study Area during recent and upcoming soil investigations (i.e., Phase 2 RI Data Gap Investigation, Unit 4 and 5 Buildings Investigation) will be included in the updated DUE section of the forthcoming BHRA report. Because the planned additional data collection efforts are being conducted to support the evaluation of fate and transport of perchlorate and other constituents in soil and groundwater, it is not anticipated that the additional data will impact the proposed DUs or the selection of COPCs presented in this report. Nevertheless, in the forthcoming BHRA, the DUE, COPC selection, and DU identification provided in this report will be revised, using the updated soil BHRA data set, to ensure that the additional soil results are incorporated into the understanding of soil COPCs and their spatial distribution.

For each soil sample collected from the above investigations, sampling locations were verified relative to current parcel, Operations Area, and ECA boundaries. Samples were excluded from the soil BHRA data set if 1) locations were outside the current boundaries of the Study Area; 2) the sampling depths were greater than 10 ft bgs; or 3) location and/or depth information were not available. In addition, samples collected at three locations (TSB-GJ-03, TSB-GJ-04, and TSB-GR-02) previously believed to be in Parcel G, are now considered to be within the Study Area; these samples are also included in the soil BHRA data set.

In addition to a few sample re-assignments resulting from the minor changes to the current parcel, Operations Area, and ECA boundaries, several soil samples (i.e., asbestos samples) included in the 2016 BHRA data set were confirmed to be associated with soil removed during the interim soil removal actions conducted in 2010 and 2011, and a few soil samples have been identified as being beneath the Mn-2 pond and inaccessible for soil contact; these samples have been excluded from the BHRA data set. The depths of soil samples after the removal actions have been reconfirmed to make sure that only the samples collected from the top 10 ft bgs are included in the BHRA data set. The soil

<sup>&</sup>lt;sup>4</sup> In this and all remaining sections of the report, the 0 to 10 ft interval refers to the post-excavation soil horizon (i.e., following any excavation, backfilling, and grading completed at the Site) unless otherwise stated.

<sup>&</sup>lt;sup>5</sup> "Remaining" soil samples refers to soil samples collected in areas that were not excavated during the interim soil removal actions described in Section 2.1.2 and for which analytical results remain representative of current site conditions.

samples which have been changed in the current BHRA data set, compared to the 2016 BHRA data set, are identified in Appendix B, Table B-1.

The "NERT project database," maintained by Ramboll Environ on behalf of the Trust, houses the analytical data collected during historical and ongoing investigations at the NERT Site and throughout the RI Study Area.<sup>6</sup> After identifying the preliminary set of data for the BHRA, an initial task before the DUE was implemented to 1) identify and correct inconsistencies in data field entries and 2) create additional fields to support data management and interpretation for the BHRA data set. The following items were completed:

- Standardize chemical names and Chemical Abstract Service (CAS) registry numbers;
- Standardize reporting units, e.g., milligrams per kilogram (mg/kg) for metals and micrograms per kilogram (μg/kg) for organic compounds;
- Standardize analytical method names;
- Correct errors in data entry (e.g. typos in sample identification codes);
- Identify a unique result for use in the BHRA for sample/analyte pairs for which more than one result was reported. For example, if two results were reported for benzo[a]pyrene (BaP) in the same sample one by USEPA Method 8270 and the second by USEPA Method 8270 Selective Ion Monitoring (SIM) the result to be used in the BHRA was identified as that from the 8270 SIM analysis because of the greater sensitivity (lower reporting limits) of this method;
- Enter BCLs and confirm that BCLs correspond to the chemical form or species reported. For example, the database compared analytical results for phosphorus with the BCL for white phosphorus. There is no evidence to suggest that white phosphorus is present in Site soils. The most abundant form of phosphorus in soil is orthophosphate. Analytical methods were reviewed to confirm that the analyses were not for white phosphorus; and
- Develop database queries and confirm that queries returned the correct output.

The above steps were necessary due to the approximately 10-year period over which the soil data was collected and the differences in sampling, analysis, and data entry across investigations. This can be understood in the context of soil samples collected by different entities, analyzed by different analytical laboratories for overlapping suites of chemicals, and the use of different reporting conventions.

No change was made to a datum without first understanding the issue and the steps necessary to correct the issue. As needed, sampling plans, laboratory reports, Data Validation Summary Reports (DVSRs), and other supporting documents were reviewed. Data points were considered unusable for risk assessment if information could not be located to confirm and/or correct an identified issue. Soil data excluded from the BHRA data set during data processing are summarized in Appendix B, Table B-2.

To ensure calculation consistency, dioxin toxicity equivalents (TEQs) were calculated (or recalculated) using the results for dioxins, furans, and dioxin-like PCBs and the World

<sup>&</sup>lt;sup>6</sup> Historically, the database has been managed by different entities responsible for investigations and data collection at the Site. Ramboll Environ assumed responsibility for the database in early 2011.

Health Organization (WHO) toxicity equivalency factors (TEFs) scheme (van den Berg et al. 2006). BaP equivalents (BaPEqs) were also calculated (or recalculated) for the seven carcinogenic polycyclic aromatic hydrocarbons (PAHs) (i.e., BaP, benzo[a]anthracene, benzo[b]fluoranthene, benzo[k]fluoranthene, chrysene, dibenzo[a,h]anthracene, and indeno[1,2,3-c,d]pyrene) for which USEPA has derived TEFs (USEPA 2017). Nondetects were addressed using the Kaplan-Meier approach from USEPA's TEQ calculator.

TPH data were excluded from the soil BHRA data set, consistent with NDEP guidance (NDEP 2017). TPH was evaluated through the indicator chemicals, including benzene, toluene, ethyl benzene, and total xylenes (BTEX); methyl tert-butyl ether (MTBE); and PAHs. Over 450 samples in the BHRA data set were analyzed for PAHs and over 200 samples were analyzed for BTEX and MTBE (Appendix C). These samples provide an adequate data set for evaluating TPH.

For asbestos, long chrysotile fibers in four samples were reported as "<1" or "<2" in the laboratory reports and DVSRs (see Table 2; DVSRs are included in Appendix B). In the absence of information allowing for interpretation of these data, the results will not be included in the BHRA data set, but will be discussed in the BHRA.

The soil BHRA data set, which is presented in Appendix C (Table C-1 for chemicals and radionuclides, and Table C-2 for asbestos), includes 1,131 soil samples collected from 0-10 ft bgs at 431 locations.

#### 3.1.2 Criterion I – Reports to Risk Assessor

Criterion I requires confirmation that the reports relied upon are complete and appropriate for use in the HRA. The required information specified under this criterion was verified and is available in the documents associated with the Study Area data collection efforts, as listed in Table 1.

#### 3.1.3 Criterion II – Documentation

The objective of the documentation review is to ensure that each analytical result can be associated with a specific sampling location and that the procedures used to collect the samples are appropriate. As part of this DUE step, Ramboll Environ completed a comprehensive review of the soil samples collected and reported in the documents listed under Criterion I and/or in the NERT project database. The steps completed during the review are listed in Table 1. Figure 4 depicts the location of all soil samples included in the BHRA data set; the analytical results for each sample are included in Appendix C.

#### 3.1.4 Criterion III – Data Sources

The objective of the data sources review is to ensure that adequate sample coverage of source areas has been obtained and that the analytical methods are appropriate to identify chemicals and derive associated exposure point concentrations (EPCs) for the BHRA.

The review of sample coverage from the current BHRA data set is described in Table 1, which is based on the distribution of sample locations prior to excavation, as well as samples collected as part of the Phase 1 RI. Other areas with data gaps are being evaluated by recent or ongoing investigations (i.e., Phase 2 RI, Unit 4 and 5 Buildings Investigation). A review of the updated BHRA data set will be conducted in the forthcoming BHRA to ensure that sample coverage is considered adequate for purposes of the BHRA.

The analytical methods used in the Study Area investigations are described in Table 1. The USEPA analytical methods were adequate for characterizing potential contaminants in soils and provide quantitative analytical results that are of adequate quality for deriving EPCs.

#### 3.1.5 Criterion IV – Analytical Methods and Detection Limits

Criterion IV requires that the analytical method appropriately identifies the chemical form or species, and that for each chemical, the sample quantitation limit (SQL) is sufficiently low for risk characterization. The analytical methods used for the historical and RI investigations are listed in Table 1.

For analytes where the detection frequency was less than 100%, the SQLs from the BHRA data set were compared to 0.1 times the BCL  $(0.1 \times BCL)^7$  (NDEP 2017) to confirm that they were sufficiently low for risk characterization. For chemicals where a BCL was not available, representative surrogates were identified and used for the comparison. For dioxin TEQs, the SQLs were compared to the NDEP-approved site-specific action level of 0.0027 mg/kg, derived based on a study that evaluated the bioaccessibility of dioxins in soils collected from the NERT Site (Northgate 2010e). Table 3 presents the results of the SQL evaluation along with the screening levels. Chemicals with SQLs above screening levels are summarized in Table 1.

Overall, the SQLs were sufficiently low for risk characterization. The impacts of the few exceptions with elevated SQLs on the COPC selection and DU identification are further discussed in Section 5.3.2.

#### 3.1.6 Criterion V – Data Review

The data review included evaluation of completeness, instrument calibration, laboratory precision, laboratory accuracy, blanks, adherence to method specification and quality control (QC) limits, and method performance in sample matrix. Details of this review are presented in Table 1. In summary, the tabular summaries of the data qualifications included in the NDEP-approved DVSRs listed in Criterion I were reviewed, and with the exception of the rejected data discussed in the DVSRs, all data are deemed to be usable for risk assessment purposes. These data qualifications are further discussed below as a component of Criterion VI.

#### 3.1.7 Criterion VI – Data Quality Indicators

The project Quality Assurance Project Plan (QAPPs) (ENSR 2008b, AECOM and Northgate 2009, ENVIRON 2014d) identified five DQIs to insure that the overall quality of the data is sufficient to support the risk assessment, as follows: completeness, comparability, representativeness, precision, and accuracy. The DQIs provide quantitative and qualitative measures for evaluating the risk assessment data as they relate to uncertainties in the selection of COPCs, characterization of EPCs, and risk descriptors used in support of the BHRA and the risk management decisions that will be made for the Study Area. Specifically, the DQIs address field and analytical data quality aspects as they affect uncertainties in the data collected for site characterization and risk assessment.

<sup>&</sup>lt;sup>7</sup> The lower of the indoor and outdoor industrial/commercial worker BLC was used for the comparison.

As discussed previously in Section 2.1.2, an interim soil removal action was conducted at the Operations Area in 2010 and 2011 to minimize potential health risks associated with the continued presence of contaminated soil. Areas designated for removal were identified based on the results of the Phase A and B investigations, pre-confirmation sampling, and supplemental sampling conducted to provide additional information for excavation planning. Over 570,000 cy of soil and over 284,000 tons of manganese tailings were removed (ENVIRON 2012, Northgate 2012). Over 1,100 individual samples were marked as "removed" in the project database. As noted in Section 3.1.1, these samples are not included in the BHRA data set.

The DQI evaluation is presented in Table 1. Based on the evaluation, the overall goals for data quality for risk assessment were achieved, and all DVSRs were reviewed and approved by NDEP (with the exception of the DVSRs for soil samples collected during the RI<sup>8</sup>). In summary, with the exception of the rejected data discussed in Table 1 and listed in Appendix B, Table B-3, all data are deemed to be usable for risk assessment purposes.

#### 3.2 Data Analysis

As described in NDEP guidance (NDEP 2010a), the purpose of the data analysis step is to "use simple exploratory data analysis to compare data to the expectations of the CSM, to determine if the data adequately represent the source terms and exposure areas or evaluation areas." Consistent with guidance, the steps of the exploratory data analysis (EDA), as described in the following sections, include 1) preparation of summary statistics for the BHRA data set (Section 3.2.1), 2) evaluation of background conditions for metals and radionuclides (Section 3.2.2), and 3) preparation and review of spatial plots for detected analytes (Section 3.2.3). Section 3.2.4 discusses the results of the EDA in the context of current and former manufacturing operations at the Operations Area and the CSM.

#### 3.2.1 Summary Statistics

This section presents summary statistics for the entire Study Area. Summary statistics for individual DUs are presented in Section 5.4.1.

Summary statistics for analytical data collected from the shallow soils (i.e., sample locations currently located between 0 and 10 feet bgs) are presented as follows:

- Table 2 Soil sampling results for asbestos (long amphibole and chrysotile fibers);
- Table 4 Summary statistics for chlorine oxyanions, metals, other inorganics, and radionuclides; and
- Table 5 Summary statistics for organic compounds including dioxin TEQs and BaPEqs.

Tables 4 and 5 include analytes detected in one or more soil samples; Appendix D presents summary statistics for all analytes (i.e., detected analytes and analytes reported at less than the SQL in all samples). Individual sample locations are shown on Figure 4. In developing the summary statistics, soil samples with primary and field duplicate results were treated as independent samples. The effect of duplicate treatment on the COPC selection and DU identification is further discussed in Section 5.3.4.

<sup>&</sup>lt;sup>8</sup> Phase 1 RI DVSRs are being submitted to NDEP in 2017. DVSRs for the Phase 2 RI, Phase 3 RI, and Unit 4 and 5 Building Investigations are anticipated to be submitted to NDEP in 2018.

For most analytes, the summary statistics are based on the results of between 200 and 700 samples, although for some analytes the analytical data set is much more limited (<20 samples). However, the analytes with limited sample size were never detected (e.g., chlorite, some SVOCs and VOCs) and/or were not SRCs (e.g., lithium, niobium, chlorite, fluoride, some SVOCs and VOCs). Therefore, the limited sample size for these analytes does not have any impact on the COPC selection and DU identification.

Considering both the data review conducted by Ramboll Environ and the data reviews conducted by NDEP prior to the removal action and the NDEP-approved RAW, the existing data set is considered adequate for risk assessment purposes. Planned additional data collection efforts are being conducted to support the evaluation of fate and transport of perchlorate and other constituents in soil and groundwater. Any additional soil data collected from the 0-10 ft depth interval within the Study Area will be included in the BHRA data set. If additional data are added, the COPC selection and DU identification will be updated using the complete BHRA data set in the forthcoming BHRA report.

Table 2 presents the soil data summary results for asbestos (long amphibole and long chrysotile fibers). Results are reported in terms of the number of long fibers (i.e., >10  $\mu$ m long and <0.4  $\mu$ m wide) observed in the sample. As shown in Table 2, one or more long amphibole fibers were observed in four out of 133 post-abatement samples in the Study Area, and one or more long chrysotile fibers were observed in 21 out of 129 post-abatement samples in the Study Area.

#### 3.2.2 Background Evaluation

To support the EDA, a background evaluation was first conducted for the entire Study Area. A separate background evaluation was conducted for each proposed DU to identify DU-specific metal and radionuclide COPCs (see Section 5.4.2).

As identified in the NDEP-approved BHRA Work Plan, analytical results for 0 to 10 ft soil samples within RZ-A were used as the background data set for metals (ENVIRON 2014b).<sup>9</sup> A detailed discussion of this data set is presented in the Revised Technical Leaching Memorandum (Northgate 2010f). In summary, 31 soil samples, including three field duplicates, were collected from 14 borings<sup>10</sup> within RZ-A during the Phase B investigation; 16 of these samples were collected between 0 and 2 ft bgs and 15 samples were collected between 10 and 11.5 ft bgs. Consistent with the background evaluation conducted for Parcels C, D, F, G, and H (Northgate 2014), a single Phase A boring location (SA02) and five Phase B boring locations (RSAU4, RSAU5, SA28, SA146, and SA147) within LOU 62 (former State Industries, Inc. operational area and boron source area) were excluded from the RZ-A background set due to elevated concentrations of boron and other metals (arsenic, chromium, cobalt, iron, molybdenum, nickel, platinum, and sodium).<sup>11</sup>

<sup>&</sup>lt;sup>9</sup> As noted in the BHRA work plan, NDEP investigated the differences observed in metals concentrations among available BMI background data sets and determined that the RZ-A data set was appropriate for statistical background analysis of metals at the Tronox facility (presently the NERT Site) (NDEP 2010b).

<sup>&</sup>lt;sup>10</sup> As shown on Appendix E, Figure E-1, RSAT7, RSAT8, and RSAS6 are located outside the boundaries of RZ-A and the Site. These three off-site samples are retained in the background data set.

<sup>&</sup>lt;sup>11</sup> Although metals concentrations in these samples were elevated relative to background, the results of the RZ-A HRA indicated that exposures to residual chemicals in the upper 10 ft of soil were below risk levels of concern (see Section 2.1.2).

The RZ-A samples identified for the metals background evaluation are also used for the radionuclide background evaluation. The NDEP-approved BHRA work plan (ENVIRON 2014b) states that the McCullough background data set presented in *Background Shallow Soil Summary Report, BMI Complex and Common Area Vicinity* (BRC and TIMET 2007) would be used for the radionuclide background evaluation. However, in comments on the Parcels Soil HRA (NDEP 2015), NDEP clarified that the RZ-A data set (and not the McCullough data) should also be used for the radionuclides. The RZ-A data set used for the background evaluation of both metals and radionuclides and the background sample locations are included in Appendix E.

The background evaluation was performed using normal and lognormal quartile to quartile plots (Q-Q plots), and side-by-side box-and-whisker plots (box plots). These plots are included in Appendix F. Normal and lognormal Q-Q plots provide a visual assessment of how closely the data follow a normal or lognormal distribution. Data points that fall roughly on a straight line may be considered to follow a normal or lognormal distribution. Both background and Study Area data are included on these plots such that the Q-Q plots provide a direct visual comparison of the two distributions. The Shapiro-Wilk test was used to more formally evaluate the consistency of each data set with a normal or lognormal distribution.

Box plots provide a visual comparison between the Study Area and background data. For each data set, the "box" in the box-and-whisker plot encompasses the central 50 percent of the results (i.e., the results from the 25th to 75th percentiles, or equivalently, between quartile 1 [Q1] and quartile 3 [Q3]). Substantial overlap between the boxes for background and Study Area data indicates that the Study Area data may not be significantly different from background. The whiskers demarcate one "step" above the 75th percentile and one step below the 25th percentile. One "step" is defined as 1.5 times the interquartile range (IQR, the difference between the 75th and 25th percentiles). Data points above and below the whiskers are considered potential outliers from the distribution and are shown on the plots as open circles for non-detected values and as crosses for detected values. As used here, "outliers" may indicate potential hotspots for spatial analysis.

The computer statistical software program Guided Interactive Statistical Decision Tools (GiSdT®, Neptune 2007) was used to perform all statistical tests.<sup>12</sup> Specifically, statistical background comparisons were performed using the t-test, Gehan test, Quantile test, and Slippage test. This suite of tests is sometimes referred to as "Gilbert's Toolbox." The t-test is a parametric test (i.e., an underlying condition is that the data or log-transformed data are normally distributed). In contrast, the Gehan test, Quantile test, and Slippage test are nonparametric, and thus do not require that the data are normally or lognormally distributed (USEPA 2002, NDEP 2009b). These tests are described below:

**The two-sample t-test** tests for equality of the means of the Study Area and background concentrations. An underlying assumption of the test is that concentrations are normally distributed for both data sets.

**The Gehan test** is a modification of the Wilcoxon Rank Sum test that evaluates the difference between the sums of the ranks for two populations. This is a nonparametric

<sup>&</sup>lt;sup>12</sup> Neptune provided Ramboll Environ with a copy of the GiSdT® program used for the statistical evaluation.

method for assessing differences in the centers of the distributions and is based solely on the relative order (or ranking) of the observations from the two samples. This test has less power than the two-sample t-test when the data are normally distributed, but the assumptions are not as restrictive. The GiSdT® version of the Gehan test uses the Mantel approach for ranking the data, which is equivalent to using the Gehan ranking system. The Gehan ranking system is used to rank non-detects with the detected concentrations (NDEP 2009b).

**The Quantile test** evaluates "tail effects" that are not specifically considered in the Wilcoxon Rank Sum test. The Quantile test looks for differences in the right tails (upper end of the distribution), rather than evaluating central tendency. The Quantile test was performed using a defined quantile of 0.80, consistent with the approach used in the Parcels HRA (Northgate 2014).

**The Slippage test** looks for a shift to the right in the extreme right tail of the background data set as compared with the extreme right tail of the Study Area data set. This test evaluates whether the number of Study Area samples with concentrations greater than the maximum background concentration is greater than would be expected statistically if the Study Area and background distributions were the same.

NDEP guidance (2008a) recommends including field duplicates in a data set when the variance of the duplicates is similar to the variance of the primary samples. As noted in the guidance, field duplicate samples represent a discrete and unique measurement of soil chemical conditions proximal to the primary sample (unlike split samples). For the preliminary background evaluation presented in this report, soil samples with primary and field duplicate results were treated as independent samples, consistent with Option 2 in NDEP guidance (NDEP 2008a). The effect of duplicate treatment on the COPC selection and DU identification is further discussed in Section 5.3.4.

Consistent with NDEP guidance (NDEP 2009b), non-detect results are set equal to one-half the limit of detection for the parametric tests and equal to the detection limit for the non-parametric tests. Substitution is not required for the non-parametric tests, which use the Gehan ranking scheme to rank non-detects. For the *t*-test, the Gehan ranking scheme cannot be used; in comments on Revision 2 of the Parcels C, D, F, G, and H HRA, NDEP stated that the value of one-half the detection limit for non-detects is preferred (NDEP 2009b).

The results of all statistical tests included in Gilbert's Toolbox are presented in Appendix F. However, given the large amount of data, the Shapiro-Wilk test is very likely to reject a hypothesis of normality or of any other distribution. On the other hand, ttests are very robust to deviations from underlying normality assumptions (a large amount of data points causes the mean to be approximately normal). Therefore, the results of the t-tests are not reasonable to use in the background evaluation of a large amount of data. As requested by NDEP (2016), the determination of background consistency was only based on the results of the non-parametric tests.

#### 3.2.2.1 Metals

The background evaluation for metals in the Study Area is presented in Appendix F, as follows:

- Table F-1 presents summary statistics for each metal, including the total number of samples, number of detections, percent detections, minimum SQL, maximum SQL, minimum detected value, maximum detected value, median, mean, and standard deviation. Consistent with NDEP guidance (NDEP 2008b), the median, mean, and standard deviation are calculated based on detected concentrations only. The results of the Shapiro-Wilk test are also presented;
- Table F-2 includes the calculated probability (p-values) for the four statistical tests and the overall determination as to whether Study Area soil concentrations are greater than background levels (Five results are shown in the table because the ttest was performed twice, once on the raw data set and once on the logtransformed data set);
- Figures F1-1 through F1-32 present boxplots for metals in background soils and Study Area soils (upper 10 ft); and
- Figures F2-1 through F2-32 present normal and lognormal Q-Q plots for metals in background soils and Study Area soils (upper 10 ft).

A significance level of alpha = 0.025 was used to evaluate the statistical significance of the Gilbert's Toolbox results, consistent with NDEP guidance (NDEP 2009b).

#### 3.2.2.2 Radionuclides

The background (RZ-A) data set includes results for the long-lived radionuclides in the uranium [U]-238 decay series (U-238, U-234, thorium [Th]-230, and radium [Ra]-226) and in the Th-232 series (Th-232, Ra-228, and Th-228). The RZ-A background data set also includes data for U-235, but not for the U-235 decay chain. NDEP guidance (2009c) notes that most isotopes of the U-235 decay chain are barely discernible from the minimal detectable concentrations. The background evaluation and tests for secular equilibrium are presented in Appendix F, as follows:

- Table F-3 presents summary statistics for each radionuclide, including the total number of samples, number of detections, percent detections, minimum and maximum detected values, median, mean, and standard deviation. Consistent with NDEP guidance (NDEP 2008b), the median, mean, and standard deviation are calculated based on detected concentrations only. The results of the Shapiro-Wilk test are also presented;
- Table F-4 includes the p-values for the four statistical tests and the overall determination as to whether Study Area soil concentrations are greater than background levels (Five results are shown in the table because the t-test was performed twice, once on the raw data set and once on the log-transformed data set);
- Tables F-5A and F-5B present the results of the equivalence testing for secular equilibrium of the uranium decay series (U-238 chain) and thorium decay series (Th-232 chain), respectively;
- Table F-6 presents the correlation matrices for the uranium decay series and the thorium decay series;
- Figures F1-33 through F1-40 present the boxplots for radionuclides in background soils and Study Area soils (upper 10 ft); and

• Figures F2-33 through F2-40 present normal and lognormal Q-Q plots for radionuclides in background soils and Study Area soils (upper 10 ft).

The significance level used for the background evaluation of metals (alpha = 0.025) was also used for the background evaluation of radionuclides.

#### 3.2.3 Spatial Analysis

Spatial quartile plots (included in Appendix G) were prepared for detected chemicals in the Study Area to illustrate the spatial distribution of the data, identify potential hotspots, and compare the results to the expectations of the CSM. Each spatial quartile plot presents the following information:

- Areas occupied by ECAs;
- A grid overlay, using 200 × 200 ft squares;
- Sample locations within the grid; and
- Chemical concentrations. The concentration shown in each square is the maximum detected concentration for all samples within the square for soils from 0-10 ft bgs, unless results for all samples within the square were reported as less than the detection limits; concentration bins are defined as follows:
  - ✓ Dark green concentrations < detection limits;
  - ✓ Light green concentrations <Q1;
  - $\checkmark$  Yellow concentrations within the IQR;
  - ✓ Orange concentrations >Q3 and <(Q3 +  $1.5 \times IQR$ ); and
  - ✓ Red concentrations > (Q3 +  $1.5 \times IQR$ ).

Spatial quartile plots are presented for 44 analytes, as follows:

- Chloroxyanions chlorate and perchlorate;
- Metals all metal COPCs (identified in Section 4), metals with concentrations greater than background, and specific metals identified as possible SRCs for certain LOUs (with the exception of calcium, iron, and sodium). Additional metals for which plots were prepared include those commonly associated with industrial operations (e.g., cadmium, copper, and lead);
- Radionuclides U-238, Th-232 and U-235 (the parent radionuclides); and
- Organics all organic COPCs (identified in Section 4) and organic SRCs with a detection frequency of five percent or greater (with the exception of common field/laboratory contaminants). In addition, certain organics (e.g., dichlorodiphenyltrichloroethane [DDT]/dichlorodiphenyldichloroethylene [DDE]) were selected for plotting to enable spatial analysis of chemically-related contaminants.

The plots are presented in Appendix G (organized alphabetically by chemical name) and discussed in the following section.

#### 3.2.4 Comparison with CSM

The EDA (including the review of the Appendix G spatial quartile plots) is presented in Table 6 for chlorine oxyanions, metals, other inorganics, and radionuclides, and in

Table 7 for organic compounds. The following presents an overall summary of the soil data in the context of our current understanding of site history and the Study Area CSM.

- Chloroxyanions. Chlorate and perchlorate manufacturing operations have been conducted within the Study Area since approximately 1945 (Ramboll Environ 2016b). These compounds are detected throughout the Study Area (Figures G-10 and G-30). Although the 2010-2011 interim removal action addressed chlorate and perchlorate concentrations greater than the industrial worker BCL in the upper 10 ft of soil (relative to original site grades), localized areas remain in which perchlorate concentrations are above BCLs within 10 ft of the post-removal ground surface, since backfill of excavated areas was not placed to original grade in all locations. The highest concentrations are found in the central portion of the Site, consistent with the former chlorate and perchlorate production activities in this area and the discharge of process streams to the Beta Ditch and surface impoundments.
- Metals. The 2011 NDEP Action Memorandum (NDEP 2011) identified "metals" as • possible site-related contaminants at LOUs within the Study Area. Individual metals specifically identified as potential contaminants at LOUs within the Study Area included arsenic, barium, boron, chromium (total), hexavalent chromium, iron, lead, magnesium, manganese, nickel, platinum, selenium, and zinc. Calcium and sodium were also identified, presumably referring to their salts. Results of the background evaluation of Study Area metals (Appendix F) show that post-removal soil concentrations were greater than background (as compared with the RZ-A background set) for arsenic, barium, chromium (total), cobalt, manganese, sodium, thallium, and tungsten, consistent with the former use of these metals in the Study Area. Boron and hexavalent chromium had low detection frequencies, suggesting that background comparison results may not be applicable. Although historical information indicates that calcium was used in significant quantities, Study Area concentrations were consistent with background levels, possibly due to the relatively high background levels (approximately 30,000 mg/kg), masking possible releases.

Cobalt, present at concentrations greater than background, was not specifically identified as related to former operations by NDEP (2011) or in other reports reviewed. However, the spatial plots for cobalt (Figure G-14) and manganese (Figure G-24) show substantial overlap in the spatial distribution of these metals, suggesting that cobalt may be a by-product of manganese production. A World War II-era report mentioned cobalt-bearing manganese deposits in southern Appalachia, indicating that cobalt can be present in manganese deposits.<sup>13</sup> Stronger evidence of the association of cobalt with manganese was identified by reviewing analysis of manganese ore and tailing samples in the NERT project database. In one sample, manganese and cobalt ore concentrations were 560,000 mg/kg and 871 mg/kg, respectively, and concentrations in tailings were 79,600 mg/kg and 1,840 mg/kg. The higher cobalt concentration in tailings (both the absolute concentration and concentration relative to that of manganese) is consistent with the expectation that the concentration of cobalt would increase in tailings since manganese was preferentially removed from the ore. The highest concentration of cobalt in soil at the Study Area was 284 mg/kg, detected in the area just north of the leach

<sup>&</sup>lt;sup>13</sup> This is not intended to infer that Appalachia was the source of manganese ores at the Site, only that an association between manganese and cobalt has been reported.

plant/manganese tailings pile. For comparison, the maximum cobalt concentration in the RZ-A background data set is 9.1 mg/kg.

Thallium, present at concentrations greater than background, was not identified as a specific contaminant at an LOU. However, the spatial plot for thallium (Figure G-35) shows elevated thallium concentrations in the eastern area of the Site, suggesting a possible historical presence.

Tungsten, present at concentrations greater than background, was not specifically identified as a potential contaminant at LOUs within the Study Area. However, US Vanadium (LOU70) within ECA C18 formerly produced tungsten compounds at the Site. The spatial plot for tungsten (Figure G-39) shows elevated tungsten concentrations in the eastern and the central areas of the Site, suggesting other possible historical presence.

- Other Inorganics. This group of inorganic compounds includes common industrial chemicals that are used as chemical feedstocks and/or expected to be present in process waste streams. With the exception of fluoride and nitrate, all compounds were historically identified as SRCs at the Operations Area. These compounds are generally highly soluble when present as free anions or cations. Many of these compounds are physiological electrolytes and/or occur naturally in foods. Although all of the listed inorganics occur naturally in soil, RZ-A background data sets are not available to conduct a background analysis. At the concentrations detected in soil, these inorganics do not present human health concerns. Generally, these inorganics are of greater concern when detected as contaminants in groundwater than when present at elevated concentrations in soil.
- **Radionuclides.** Radionuclides are not known to be associated with any of the former operations identified for the Operations Area. Although no specific source areas were identified, the Study Area soil investigations included analyses for radionuclides in the U-238 and Th-232 decay series and for U-235.
- Dioxins/Furans. As shown in Table 5, dioxins/furans were detected in 99% of the soil samples collected within the Study Area. This high detection frequency is not unexpected given that 1) analytical detection limits are very low (less than 0.001 mg/kg), and 2) dioxin/furans are formed during various combustion processes (in the presence of a source of hydrocarbons and chlorine) and are by-products of the production of certain chlorinated chemicals, including pesticides. Dioxins are typically detected in shallow surface soils as a result of airborne deposition. They are extremely persistent in soils and over time will accumulate in the presence of a continuing source. During review of the post-removal data set, soil concentrations exceeding the site-specific action level (0.0027 mg/kg) were noted in the area southwest of the GW-11 Pond, an area that had not been identified for excavation in the 2010/2011 interim removal action. Additional samples were collected in this area as part of the Phase 1 RI to further delineate dioxin in soil (see discussion in Section 5.1).
- **PAHs.** PAHs are ubiquitous environmental contaminants and formed during incomplete combustion of organic materials. The detection frequencies of PAHs in the Study Area were generally very low.

- OCPs and Other Organochlorine Compounds. OCPs and chlorinated by-products • of pesticides and other manufacturing processes involving chlorine have been detected at the Site. These chemicals include aldrin, chlordane, DDT (and DDE) (Figures G-16 through G-18), hexachlorocyclohexane (BHC) (Figures G-1 and G-6), dieldrin (Figure G-19), endosulfan, endrin, hexachlorobenzene (Figure G-21), methoxychlor, PCBs, and toxaphene (Figure G-38) (See above for a separate discussion of dioxins/furans). The detection of these compounds is consistent with former site operations, including the manufacture of chlorobenzenes and DDT by Hardesty/AMECCO from 1946 to 1949 (Kleinfelder 1993, Ramboll Environ 2016b), as well as the manufacture of chlorinated compounds at the adjacent Olin Chlor Alkali/Stauffer/ Syngenta/Montrose (OSSM) facility. Stauffer produced lindane at the former Lindane Plant from 1946 through 1958, and Montrose produced organic chemicals, including chlorobenzene, PCBs, chloral, and 4,4'-dichlorobenzil from 1947 through 1983. In addition to possible air emissions (and deposition) from these processes, associated wastes streams were conveyed to the former Beta Ditch between 1971 and 1976 (Ramboll Environ 2016b).
- **OPPs.** Only two OPPs were detected in the Study Area with very low detection frequencies (i.e., dimethoate and stirophos). Although OPPs were historically listed as SRCs, NDEP did not identify these pesticides as a specific contaminant at an LOU.
- SVOCs. Eight SVOCs were detected in the Study Area with very low detection frequencies, including six phthalates, hexachlorobutadiene, and octachlorostyrene. Phthalates were not historically listed as SRCs, and are common field/laboratory contaminants. Hexachlorobutadiene was not historically listed as a SRC, although it can be a byproduct of reactions involving chlorine and hydrocarbons. Octachlorostyrene was historically listed as a SRC, and is highly persistent in soil. It is a by-product of many industrial chemical processes, and forms during incineration and combustion processes involving chlorinated compounds.
- VOCs. Consistent with results observed in investigations at other industrial facilities, a number of VOCs were detected in soils, but at low frequencies (typically less than 3%). The primary exceptions are chloroform (33% detected in soil samples, widely present in groundwater beneath and in the vicinity of the Site [Ramboll Environ 2016b]), and certain VOCs that have been identified by USEPA (1989) as "common laboratory contaminants", including acetone, 2-butanone, methylene chloride, and toluene (detection frequencies ranging from 20 to 40%). All the VOCs in soil were detected at low concentrations, not indicative of a potential source area.

In the BHRA report, the CSM discussion will be integrated with the updated CSM presented in the forthcoming RI report.
# 4. **PRELIMINARY IDENTIFICATION OF SOIL COPCS**

Soil COPCs for quantitative evaluation in the BHRA are identified following a two-tiered process. The first tier, presented in this section, is the identification of COPCs for the entire Study Area based on the following three-step approach:

- 1. Concentration/toxicity screen;
- 2. Background evaluation for metals and radionuclides; and
- 3. Chemical-specific considerations.

The chemicals that "fail" these steps are retained as COPCs for the Study Area and those that "pass" are excluded as COPCs,<sup>14</sup> as described in Sections 4.1 through 4.3 and shown on Figure 5.

In Section 5, using a COPC-specific, risk-based spatial analysis and a series of spatial plots of concentrations, cancer risks, and noncancer HIs, the Study Area was divided into DUs. The second tier of the process is the identification of COPCs specific to each DU, which is presented in Section 5.4.

#### 4.1 Step 1 – Concentration/Toxicity Screen

The concentration/toxicity screen is conducted to identify those chemicals that could contribute significantly to the cancer risk and/or noncancer hazard estimate (i.e., the HI). The screen considers the maximum detected concentration in soils in the Study Area and chemical-specific toxicity, as reflected in the BCL (or other criteria established for the Site); specifically, a chemical is excluded as a COPC if the maximum detected concentration is less than 0.1 times the BCL ( $0.1 \times BCL$ ). Chemicals that pass this screen for the entire Study Area are eliminated as COPCs. Chemicals that fail this screen (i.e., are present at concentrations greater than or equal to  $0.1 \times BCL$ ) are further screened under Step 2 and/or Step 3.

The BHRA data set identified in Section 3 is the starting point for the concentration/ toxicity screen. This data set includes the results for all analytes detected in one or more samples from the 0 to 10 ft depth interval,<sup>15</sup> with the exception of the analytical results excluded based on the DUE, as discussed in Section 3. For most analytes, the BCL used for the concentration/toxicity screen is the minimum of the indoor and outdoor industrial/commercial worker BCL (NDEP 2017). Because BCLs have not been established for all Study Area analytes, surrogate values were identified where possible. Surrogates and other chemical-specific exceptions as well as the results of the screen are presented in Table 8 and discussed in the following sections.

#### 4.1.1 Surrogates

The concentration/toxicity screen can be implemented only for chemicals for which a BCL or appropriate surrogate is available. Surrogates were identified for most, but not all, chemicals for which a BCL was not available, as follows:

<sup>&</sup>lt;sup>14</sup> The three screening steps are consistent with the COPC identification steps outlined in the BHRA work plan (ENVIRON 2014b). However, as agreed upon by NDEP (Ramboll Environ 2015b), the order of the steps has been changed.

<sup>&</sup>lt;sup>15</sup> An underlying assumption is that soils from depths of up to 10 ft could be brought to the surface during excavation or other activities, leading to potential worker exposures.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

Analyte	Surrogate
Acenanhthylene	Acenanhthene
gamma-Chlordane	Chlordane
Cyanide (total)	Free cyanide (CN-)
Chromium (total)	Chromium III
2,4'-DDE	4,4'-DDE
Endosulfan I	Endosulfan
Endosulfan sulfate	Endosulfan
Endrin ketone	Endrin
Ethyl tert-butyl ether	Methyl tert-butyl ether
ortho-Phosphate	Phosphoric acid
Phosphorus (total)	Phosphoric acid

Surrogates were identified for all but eight analytes (calcium, palladium, potassium, silicon, sodium, sulfate, sulfur, and octachlorostyrene). By default, these analytes are carried forward to Step 2 (metals) and Step 3 (all other analytes) of the COPC selection process.

#### 4.1.2 Chemicals with Non-Health Based BCLs

NDEP has established a non-health based upper-limit soil concentration or a "not-toexceed" concentration of 100,000 mg/kg for metals and for inorganic and organic compounds with low toxicity. That is, if the calculated health-based BCL is greater than 100,000 mg/kg, a limit value of 100,000 mg/kg is listed in the BCL table. For all detected analytes with a NDEP-established upper limit BCL, the maximum concentration is less than 100,000 mg/kg. Thus, these compounds were not identified as COPCs based on concentration considerations (Table 8).

For health-based considerations (i.e., COPC identification), it is appropriate to use the NDEP-calculated health-based BCL (and not the concentration-limit value of 100,000 mg/kg). The health-based BCLs were taken from the BCL calculation table (NDEP 2017). Due to the very low toxicity, the calculated health-based BCL for some chemicals is greater than one million parts per million. The chemicals for which health-based BCLs are used in place of non-health based BCLs are identified in Table 8.

#### 4.1.3 Arsenic, Dioxin TEQs, and Lead

As presented in the BHRA work plan (ENVIRON 2014b), site-specific values are used for arsenic and dioxin TEQs:

- For arsenic, the maximum detected concentration is compared to the site-specific remediation goal of 7.2 mg/kg (NDEP 2010c), which is the maximum arsenic concentration reported for the BRC/TIMET background data set (BRC and TIMET 2007); arsenic would be eliminated as a COPC if the maximum concentration is less than this screening value.
- For dioxin TEQs, the maximum detected value is compared to the site-specific action level of 0.0027 mg/kg; this value was derived based on an *in vitro* soil

bioaccessibility study conducted using Site soils (Northgate 2010e); NDEP approved this value based on the information presented in the study (NDEP 2010c).

USEPA has not established toxicity values (i.e., a cancer slope factor or reference dose) for lead (USEPA 2017). Instead, USEPA used a blood-lead model to establish a regional screening level (RSL) of 800 mg/kg, which NDEP has adopted as the BCL for commercial/industrial workers. Because the health endpoint for lead (i.e., a blood lead concentration) is not cancer risk or noncancer HI, the maximum detected concentration is compared directly to the commercial/industrial worker BCL of 800 mg/kg, and not to  $0.1 \times BCL$ .

#### 4.1.4 Asbestos

BCLs have not been established for asbestos (long amphibole and long chrysotile fibers). Exposure and risk assessments for asbestos are highly dependent on sample size. Even for the case where fibers are not identified (i.e., zero fibers), upper-bound cancer risk estimates can be greater than  $1 \times 10^{-6}$ , depending on sample size. For these reasons, amphibole and chrysotile are retained as COPCs.

#### 4.1.5 Results of Concentration/Toxicity Screen

The concentration/toxicity screen is presented in Table 8. For each listed chemical, the maximum detected concentration and the BCL (or other screening value) are presented. The final column indicates whether the chemical "passed" or "failed" the screen. Of the 132 analytes listed in Table 8, 98 chemicals passed, 26 chemicals failed based on the BCL (or other screening criteria) comparison, and eight chemicals did not have a screening level. Chemicals that failed or that did not have a screening level are carried forward to Steps 2 and/or 3.

#### 4.2 Step 2 – Background Evaluation

The background evaluation step is consistent with USEPA (1989) and NDEP (2009d) guidance, which indicate that metals and radionuclides can be eliminated as COPCs if site concentrations are consistent with background levels. Metals and radionuclides that are present at concentrations greater than background and those for which a background data set are not available are then further screened under Step 3.

The metals and radionuclides that either failed the concentration/toxicity screen or for which a BCL was not available for screening are listed in Tables 9 and 10, respectively. The results of the background evaluation presented in Section 3.2.2 and Appendix F are also presented. Of the 12 metals carried forward from Step 1, two metals (calcium and potassium)<sup>16</sup> were present at concentrations consistent with background and are eliminated as COPCs. Five metals (arsenic, cobalt, manganese, sodium, and thallium) were present at concentrations greater than background, and background data were not available for four metals (palladium, silicon, sulfur, and zirconium). In addition, background comparison results may not be applicable for chromium VI due to low detection frequency (<25%) in both Study Area and RZ-A background data sets; however, chromium VI concentrations in the Study Area were greater than background based on the box plot (Figure F1-9) and Q-Q plot (Figure F2-9). Therefore, except calcium and potassium, all the other metals are carried forward to Step 3.

<sup>&</sup>lt;sup>16</sup> NDEP (2017) notes that calcium and potassium are essential nutrients and do not need to be evaluated in a BHRA.

For radionuclides, as presented in the NDEP flowchart (Appendix H), when approximate secular equilibrium is exhibited in an isotope decay chain, in theory radionuclides in the same decay chain should yield similar background comparison results; if any radionuclide is greater than background, all the radionuclides in that decay chain would be carried forward in the risk assessment. When approximate secular equilibrium is not exhibited in an isotope decay chain, those radionuclides that fail the background evaluation would be carried forward in the risk assessment. As indicated in Table 10, both the U-238 and Th-232 decay series were found to be in secular equilibrium in the Study Area, and statistical testing indicated that activities of all the radionuclides were consistent with RZ-A background levels.

However, the validity of the statistical testing is complicated by several issues associated with sample preparation and analytical methods for radionuclides in both the BHRA and RZ-A background data sets. Similar issues have previously been identified by NDEP in the radionuclide analytical data sets for soil samples collected across the BMI Complex (NDEP 2009c). RZ-A background samples were collected and analyzed in 2009, while Study Area samples were collected and analyzed between 2006 and 2014, i.e., both before and after NDEP issued guidance for evaluating radionuclide data (NDEP 2009c). Over this time period, samples were submitted for analysis to different analytical laboratories and analyzed using different preparation and analytical methods. For example, the analytical methods for Ra-228 included beta spectroscopy and gamma spectroscopy, depending on the laboratory, which may be the reason for the lack of correlation with Ra-228 in the Th-232 decay chain (Table F-6). It is also an unexpected finding that for the RZ-A background data set, the Th-232 decay chain was not in secular equilibrium (Table F-5B).

Given the factors above, the results of background analysis for radionuclides must be interpreted with caution, and it is difficult to consider them as a reliable basis for the COPC selection. Therefore, all radionuclides are retained as COPCs for further DU-specific evaluation in Section 5.4.

### 4.3 Step 3 – Chemical-Specific Evaluations

For the final step of COPC identification, chemicals commonly recognized as having low toxicity and for which a BCL was not available (such that a concentration/toxicity screen could not be conducted) were further reviewed. These chemicals include macronutrients or essential micronutrients and/or are listed on the Generally Recognized as Safe (GRAS) list developed by the U.S. Food and Drug Administration.<sup>17</sup> These include:

- Silicon (essential nutrient; present in foods, with a typical dietary intake of over 20 mg/day in adults);
- Sodium (essential macronutrient, required in large quantities; high consumption from foods; GRAS); NDEP (2017) identifies sodium as an element that typically does not need to be included in a risk assessment because of its low toxicity; and
- Sulfur, sulfate (essential macronutrients, required in large quantities; high consumption from foods; GRAS).

<sup>&</sup>lt;sup>17</sup> <u>http://www.ecfr.gov/cgi-bin/text-</u> <u>idx?SID=e956d645a8b4e6b3e34e4e5d1b690209&mc=true&node=pt21.3.184&rgn=div5</u>

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

Silicon, sodium, sulfur, and sulfate were eliminated as COPCs based on their low toxicity.

#### 4.4 Study Area Preliminary COPCs

The preliminary COPCs identified for Study Area soils are listed in Table 11. The 30 COPCs include chlorate and perchlorate, seven metals, eight radionuclides (U-238 and Th-232 decay series, and U-235), 11 organic compounds, and two asbestos fibers. For two COPCs (palladium and octachlorostyrene), BCLs (and associated toxicity values) are not available; in the absence of toxicity values, these COPCs will be evaluated qualitatively in the BHRA. Also, RZ-A background data are not available for palladium and zirconium and therefore a background evaluation cannot be conducted.

Spatial intensity plots were developed for most COPCs, including chlorate, perchlorate, arsenic, chromium VI, cobalt, manganese, thallium, zirconium, U-238, Th-232 and U-235 (the parent radionuclides), dioxin TEQ, BaPEq, naphthalene, beta-BHC, 4,4'-DDE, 4,4'-DDT, dieldrin, hexachlorobenzene, toxaphene, bis(2-ethylhexyl)phthalate, and long amphibole and long chrysotile fibers (Figures 6 through 21).<sup>18</sup> Plots were not prepared for palladium and octachlorostyrene for which BCLs are not available, and their spatial distributions are presented in the spatial quartile plots (Figures G-29 and G-28, respectively).

Each COPC spatial intensity plot presents the following information:

- A grid overlay, using 200×200 ft squares;
- Sample locations within the grid;
- Areas occupied by ECAs;
- COPC concentrations. The concentration shown in each square is the maximum detected concentration from all samples collected from 0-10 ft deep soil within the square, unless results for all samples within the square were reported as less than the detection limits; concentrations are binned relative to BCLs or other screening criteria, as shown on the individual plots. Results for samples reported as less than the detection limit are colored dark green.

In Section 5.1, for the purpose of identifying DUs, the spatial intensity plots were reviewed along with other spatial plots for concentrations, cancer risks, and noncancer HIs to identify possible "hot spots"<sup>19</sup> and other spatial patterns of the data.

<sup>&</sup>lt;sup>18</sup> In addition to the spatial quartile plots discussed in Section 3.2.3 (and included in Appendix G) as part of the Data Analysis, additional spatial intensity plots were developed for most COPCs. The two sets of plots differ in the concentration bins used for plotting. The concentration bins for the Appendix G plots are based on quartiles of the concentration distribution. The concentration bins used for the COPC plots are based on the BCLs or other screening criteria.

<sup>&</sup>lt;sup>19</sup> For the purpose of this report, "hotspot" is used generically and refers to a localized area in which concentrations tend to be at the upper end of the distribution. These areas are identified based on review of the spatial intensity plots; a statistical hotspot analysis has not been conducted.

# 5. PRELIMINARY IDENTIFICATION OF DECISION UNITS

The DUs define the spatial boundaries of the individual subareas within the Study Area for which exposures and risks will be estimated in the BHRA. USEPA risk assessment guidance (USEPA 1989) recommends using a concentration term that represents "a reasonable estimate of the concentration likely to be contacted over time", when assessing exposures and risk. The purpose of the DU identification is to avoid "diluting" or lowering EPCs by averaging concentrations from hot spots (if present) with samples collected from areas with significantly lower concentrations.

Spatial data analysis for the preliminary soil COPCs considered to establish the DUs are described in Section 5.1. The proposed DUs are described in Section 5.2. DU-specific DUEs and COPC identifications are presented in Sections 5.3 and 5.4, respectively.

#### 5.1 Spatial Data Analysis

As presented in Section 3, tables of summary statistics and spatial quartile plots were prepared for 44 detected soil analytes. As presented in Section 4, through a three-step COPC selection process, 30 preliminary soil COPCs were identified for the entire Study Area, and spatial intensity plots were prepared for these COPCs. Since a total of 30 COPCs were identified, it is difficult to systematically review the spatial plots of all the COPCs in order to evaluate the spatial patterns in concentrations that could be used to identify DUs. To reduce the dimensions of the spatial data analysis by focusing DUs on risk-relevant spatial patterns, spatial risk plots (as described below) were prepared according to the two health endpoints of cancer risk and noncancer hazard. In addition, since arsenic and dioxin TEQ are evaluated based on comparison to site-specific screening criteria, the concentrations plots for these two COPC plots were prepared separately. Finally, asbestos spatial plots were prepared based on fiber counts. Spatial risk/concentration plots were created with the data separated into three depth intervals: 0-10 ft bgs, 0-2 ft bgs (surface soil), and 2-10 ft bgs (subsurface soil), except that only the surface soil depth interval was applicable to asbestos spatial plots.

In summary, the following 17 spatial risk/concentration plots were created:

- Dioxin TEQ: The maximum detected concentration within the specific depth interval at each location was compared to the site-specific action level of 0.0027 mg/kg (Figures 22a through 22c);
- Arsenic: The maximum detected concentration within the specific depth interval at each location was compared to the maximum BRC/TIMET background value of 7.2 mg/kg (Figures 23a through 23c);
- Cancer Risk: Cancer risks for each carcinogenic COPC, calculated as a ratio of the maximum detected concentration within the specific depth interval at each location to the commercial/industrial BCL for cancer effects (corresponding to a cancer risk of 10<sup>-6</sup>), were summed for all chemicals (Figures 24a through 24d);
- Noncancer HI: Hazard quotients (HQs) for each COPC, calculated as a ratio of the maximum detected concentration within the specific depth interval at each location to the commercial/industrial BCL for noncancer effects (corresponding to a noncancer HQ of one), were summed for all chemicals (Figures 25a through 25c);

- Radionuclide Cancer Risk: Cancer risks for each radionuclide, calculated as a ratio of the maximum detected concentration within the specific depth interval at each location to the commercial/industrial BCL for cancer effects (corresponding to a cancer risk of 10<sup>-6</sup>), were summed for all radionuclides(Figures 26a through 26c); and
- Long Amphibole/Chrysotile Fiber: Maximum fiber count at each location (Figures 27 and 28).

The spatial patterns of the spatial risk/concentration plots are discussed below:

#### <u>Dioxin TEQs</u>

As indicated in Figure 22a, dioxin TEQs exceeded the site-specific action level of 0.0027 mg/kg at five locations in the northern portion of the Study Area (north of the L'hoist North America facility), four of which were at 0-2 ft bgs (Figure 22b) and one of which was at 2-10 ft bgs (Figure 22c). In a small area southwest of the GW-11 Pond, one soil sample at 2.5 ft bgs at SSAK3-05 (0.011 mg/kg) was initially found to be at a concentration above the site-specific action level, but this area had not been identified for excavation in the 2010/2011 interim removal action. During the Phase 1 RI, soil samples were collected at three additional locations (RISB-50, 51, and 52) in this area to further delineate dioxin in soil, the results at 0.5 ft bgs were all above the site-specific action level. Dioxin TEQs also exceeded the site-specific action level at one location (RISB-14; 0.0047 mg/kg) in the eastern portion of the Site.

#### <u>Arsenic</u>

As indicated in Figure 23a, arsenic concentrations exceeded the maximum BRC/TIMET background value of 7.2 mg/kg at several scattered locations throughout the northern portion of the Study Area, while arsenic concentrations in the southern portion of the Study Area were all below the maximum BRC/TIMET background value. Most of the soil samples with arsenic concentration exceedances were collected at 2-10 ft bgs (Figure 23b). In the northern portion of the Study Area, the maximum arsenic concentration in soils at 0-2 ft bgs was 11.9 mg/kg (DS-C45-2) and the maximum arsenic concentration in soils at 2-10 ft bgs (EE-C25-1).

#### Cancer Risk

As indicated in Figure 24a, all the estimated cancer risks from chemical COPCs were below or within the NDEP acceptable cancer risk range of  $10^{-6}$  to  $10^{-4}$ . Cancer risks at nine scattered locations in the northern portion of the Study Area were above  $10^{-6}$ , one of which was above  $10^{-5}$  (2 x  $10^{-5}$  at SA106). Most of the cancer risks exceeding  $10^{-6}$  occurred at 2-10 ft bgs (Figure 24c). The cancer risk drivers included chromium VI (at four locations), hexachlorobenzene (at four locations), and BaPEq (at one location).

#### Noncancer Hazard Index

As indicated in Figure 25a, the estimated noncancer HIs from chemical COPCs were below the NDEP threshold of greater than one, except for four locations. Three locations with HI exceedances (SSAN6-02, SA65, and RSAM5) were in the central retention basin and south of the AP-5 Pond and former Beta Ditch, and the HI exceedances occurred at 0-2 ft bgs for all the three locations (Figure 25b) and at 2-10 ft bgs for SSAN6-02 (Figure 25c). The major chemical contributor to the noncancer HI was perchlorate.

#### Radionuclide Cancer Risk

As indicated in Figures 26a through 26c, the estimated radionuclide cancer risks were either below or at 2 x  $10^{-4}$  at all depth intervals throughout the Study Area. No spatial patterns were observed. Ra-226 and Th-228 were the major contributors to the radionuclide cancer risks.

To provide a point of comparison from a health risk perspective between radionuclides in Study Area soils and background soils, the total radionuclide cancer risks for the RZ-A background soils and BRC/TIMET regional background soils were also estimated based on the 95% upper confidence limit (UCL) on the mean soil activity, calculated by the ProUCL software (Version 5.1). The results of background radionuclide cancer risks are presented in Table 12, and the ProUCL output files are included in Appendix I. As indicated in Table 12, the radionuclide cancer risks for both RZ-A background and BRC/TIMET regional background were 2 x 10<sup>-4</sup>. Therefore, although the radionuclide cancer risks for Study Area were slightly above the NDEP acceptable risk range of 10<sup>-6</sup> to 10<sup>-4</sup>, they are consistent with background in the area.

#### Asbestos

As indicated in Figure 27, at three Phase 1 RI sample locations (RISB-10, 12 and 14) in the northern portion of the Study Area and one sample location (SSAS8-04) in the southeast corner of the Study Area, counts of long amphibole fibers were greater than the RAW specified level<sup>20</sup> of one (1) or more fibers per sample (Northgate 2010a). As indicated in Figure 28, for all soil samples throughout the Study Area, counts of long chrysotile fibers were less than the level presented in the RAW (five or more fibers per sample, Northgate 2010a).

As indicated in the spatial plots provided in this report (i.e., Figures 7 through 10, Figures 23a through 23c, and spatial quartile plots in Appendix G) and in the analyses conducted by Neptune (2017), concentrations of some metals (e.g., arsenic, cobalt and manganese) and some radionuclides (e.g., U-238) in the southern portion of the Site appear lower than in the northern portion of the Site, suggesting that background conditions in the northern and southern portion of the Site may be different (the reasons are not clear; however, these differences could be related to historical site operations).

### 5.2 Proposed Decision Units

Based on the spatial analysis, it is proposed that the Study Area be separated into three DUs as shown on Figure 29. DU-1, the northern portion of the Study Area (north of the L'hoist North America facility), and DU-2, the southern portion of the Study Area (south of the L'hoist North America facility), were selected as separate DUs based on potential differences in background concentrations (metals and radionuclides) and due to higher estimated cancer risks in the northern portion. Because of its geography, and potentially

<sup>&</sup>lt;sup>20</sup> The RAW does not specifically use the term "trigger level" or identify remediation goals. However, areas identified for asbestos abatement were those in which amphibole counts in soil samples were one (1) or more fibers and chrysotile counts were five (5) or more fibers (Northgate 2010a).

different exposure profile than DU-1 or DU-2, the Central Retention Basin has been selected as its own DU, DU-3. DU-3 is the location of three out of four soil locations that exceeded an HI of one; mainly due to elevated perchlorate concentrations in this area. The three DUs are described below.

- DU-1 has a total area of 112 acres. DU-1 includes various ancillary buildings operated by Tronox, roadways and utilities, the AP-6 and Mn-2 ponds, debris pile, and the Northern Retention Basin along with its associated drainage channel. The groundwater extraction and treatment system (GWETS, e.g., treatment system, barrier wall, IWF extraction wells) is also located in DU-1; however, many of the GWETS operational areas are classified as ECAs and are therefore not included in the BHRA. The land is largely vacant with the exception of a high-voltage power line that crosses the property in a linear north-south direction, an office trailer used for Trust operations, the fluidized bed reactors and associated sludge processing treatment plant building, an office trailer used by the treatment plant operator, and scattered non-operational railroad tracks, monitoring wells, and concrete block walls. The immediate area surrounding each power line pole has been designated as an ECA. While some portions of DU-1 are located within the Tronox leasehold, the majority of DU-1 is located outside the leasehold.
- DU-2 has a total area of 23 acres. DU-2 includes the Tronox administrative office, most of the Tronox manufacturing area, various ancillary operations (e.g., security, shipping/receiving), and a rectangular area west of RZ-A with a dirt access road used by Pronto Constructors, a Tronox tenant (Figure 2). While some portions of DU-2 are located outside the Tronox leasehold, the majority of DU-2 is located within the leasehold. Much of this area has been designated as an ECA (e.g., Unit Buildings, leach plant), and will not be evaluated in the BHRA.
- DU-3 is the Central Retention Basin, south of the AP-5 Pond and former Beta Ditch, with a total area of 8.5 acres. It is located outside the Tronox leasehold.

### 5.3 Data Usability Evaluation for Individual Decision Units

In Section 3, the DUE was conducted to evaluate the usability of soil BHRA data for the entire Study Area. The conclusions reached for Criteria I through V for the Study Area also apply to the individual DUs. In this section, the DU-specific data are reviewed relative to the five DQIs of Criterion VI, and the impacts of data quality on the COPC selection and DU identification are further discussed. The DU identification for each BHRA data point is included in Appendix C.

#### 5.3.1 Completeness

As discussed in Table 1, both field completeness for each individual investigation and laboratory completeness for each specific DVSR meet the completeness goals of 90% established in the QAPPs (ENSR 2008b, AECOM and Northgate 2009, ENVIRON 2014d). The conclusions also apply to the individual DUs. Rejected ("R" qualified) data associated with post-remediation soil samples at 0-10 ft bgs in each DU of the Study Area are summarized in Appendix B, Table B-3, and these data are excluded from the DU-specific BHRA data set. Laboratory completeness was calculated for the soil BHRA data set (Appendix C) for each individual DU as 99.9% for DU-1, 99.8% for DU-2, and 99.9% for DU-3. Given the small percentage of rejected data, these rejected data have little impact on the spatial coverage of the soil BHRA data set. Additionally, none of the rejected data

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

were above 0.1xBCL except for benzidine in DU-1, reported as nondetects at 0.66 mg/kg to 0.70 mg/kg (see Table B-3). However, these rejected values were lower than the maximum SQL for benzidine in the BHRA data set for DU-1 (37 mg/kg, see Appendix C). The impacts of elevated SQLs for benzidine on the COPC selection and DU identification are further discussed in Section 5.3.2. In summary, the rejected data do not significantly impact the COPC selection and DU identification.

#### 5.3.2 Comparability

As discussed in Table 1, different reporting limits for the same analyte in soil may impact the comparability of the data sets. The ranges of the SQLs for each analyte where the detection frequency was less than 100% are presented in Table 3. For most of the analytes, the SQLs are well below 0.1xBCL (or other screening criteria); therefore, different reporting limits for the same analyte would not affect the COPC selection and DU identification. In each DU, there are some soil analytes with SQLs exceeding 0.1xBCL (or other screening criteria) (summarized in Table 13), and their impacts on the COPC selection are discussed below.

In DU-1:

- For dieldrin, hexachlorobenzene, BaPEq, naphthalene, and toxaphene, the SQLs exceeded 0.1xBCL in 26, 135, 18, one, and 38 samples reported as nondetects. These chemicals were identified as COPCs for DU-1 (see Section 5.4), and the SQLs exceeded the BCL in fewer samples (0 to 12 samples) out of more than 200 samples reported as nondetects. Therefore, elevated SQLs for these chemicals would have little impact on the COPC selection and DU identification.
- For aldrin, alpha-BHC, 1-methylnaphthalene, anthracene, and Aroclor-1260, the detection frequencies were low (0.85% to 15%). The SQLs exceeded 0.1xBCL in one to eight samples out of 22 to 355 samples reported as nondetects. However, no SQL exceeded the BCL. Therefore, although these chemicals were not identified as COPCs for DU-1, SQLs above 0.1xBCL in a limited number of samples would have an insignificant impact on the COPC selection and DU identification.
- For 19 analytes that were never detected (including heptachlor, heptachlor epoxide, dibrom, Aroclor-1221, Aroclor-1232, Aroclor-1242, Aroclor-1248, Aroclor-1254, 1,4-dioxane, 2,4-dinitrotoluene, 2,6-dinitrotoluene, 3,3'-dichlorobenzidine, 4-chloroaniline, hexachlorocyclopentadiene, hexachloroethane, nitrobenzene, pentachlorophenol, bis(2-chloroethyl) ether, and 1,2-dibromo-3-chloropropane), the SQLs exceeded 0.1xBCL in one to seven samples out of 18 to 362 samples reported as nondetects. However, no samples or only one sample had SQLs exceeding the BCL. Therefore, although these chemicals were not identified as COPCs for DU-1, SQLs above 0.1xBCL or BCL in a limited number of samples would have an insignificant impact on the COPC selection and DU identification.
- Benzidine was only analyzed in 14 soil samples collected from nine locations in DU-1 during the Phase 1 RI, and all the concentrations were reported as nondetects with the SQLs exceeding both 0.1xBCL and BCL. Excluding benzidine as a COPC for DU-1 may have a moderate impact on the risk evaluation.
- N-nitroso-di-n-propylamine concentrations, analyzed in 18 soil samples collected from nine locations in DU-1 during the Phase 1 RI, were all reported as nondetects

with the SQLs exceeding 0.1xBCL. The SQLs exceeded the BCL in only one sample. Therefore, although this chemical was not identified as a COPC for DU-1, the SQL above the BCL in only one sample would have an insignificant impact on the COPC selection and DU identification.

In DU-2:

- For dioxin TEQs, the SQLs exceeded the site-specific action level of 0.0027 mg/kg in only one sample. Therefore, although dioxin TEQ was not identified as a COPC for DU-2, the SQL above the site-specific action level in only one sample would have an insignificant impact on the COPC selection and DU identification.
- For hexachlorobenzene and BaPEq, the SQLs exceeded 0.1xBCL in 16 and six samples reported as nondetects. These chemicals were identified as COPCs for DU-2 (see Section 5.4), and the SQLs did not exceed the BCL in any sample reported as nondetect. Therefore, SQLs above 0.1xBCL in a limited number of samples for these chemicals do not have any impact on the COPC selection and DU identification.
- For dieldrin and toxaphene that were never detected, the SQLs exceeded 0.1xBCL in only three samples reported as nondetect, with no sample having SQLs exceeding the BCL. Therefore, although these chemicals were not identified as COPCs for DU-2, SQLs above 0.1xBCL in a limited number of samples would have an insignificant impact on the COPC selection and DU identification.

In DU-3:

- For hexachlorobenzene and BaPEq, the SQLs exceeded 0.1xBCL in 17 and six samples reported as nondetect. These chemicals were identified as COPCs for DU-3 (see Section 5.4), and the SQLs exceeded the BCL in four and five samples reported as nondetect. Therefore, SQLs above 0.1xBCL or BCL in a limited number of samples for these chemicals would have an insignificant impact on the COPC selection and DU identification.
- For dieldrin, toxaphene, and 1,2-dibromo-3-chloropropane that were never detected, the SQLs exceeded 0.1xBCL in only one sample reported as nondetect, with no sample having SQLs exceeding the BCL. Therefore, although these chemicals were not identified as COPCs for DU-3, SQLs above 0.1xBCL in only one smaple would have an insignificant impact on the COPC selection and DU identification.

In summary, except for benzidine in DU-1 (which is not a SRC), different reporting limits for the same soil analyte would have little impact on the COPC selection and DU identification.

Also, differences in sample preparation and analytical methods exist between the Study Area data set and the RZ-A background and BRC/TIMET regional background data set for both metals and radionuclides, which may affect the statistical testing results of background evaluation. However, as discussed in Section 5.4.2, the metals identified as soil COPCs (arsenic in DU-1 and DU-3, chromium VI in all the three DUs, and cobalt and manganese in DU-1) were present at concentrations greater than the background, while the metals eliminated as soil COPCs (thallium and zirconium in DU-1) were present at concentrations consistent with background concentrations. In addition, radionuclides were excluded as COPCs based on the calculation of total cancer risks, not the statistical testing results of the background evaluation. Therefore, potential changes of statistical Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

testing results of the background evaluation due to the incomparability issues of analytical methods would not have any impact on COPC selection.

#### 5.3.3 Representativeness

As discussed in Table 1, the spatial representativeness of the BHRA data set for the Study Area was achieved by following both judgmental and random sampling approaches, and it will be further discussed in the forthcoming BHRA after incorporating additional soil data collected during ongoing investigations (i.e., Phase 2 RI, Unit 4 and 5 Buildings Investigation). The concentration representativeness of the BHRA data set for the Study Area was achieved by using standard methods for sampling and analysis for all the investigations. These conclusions reached for the Study Area also apply to the individual DUs.

#### 5.3.4 Precision

As presented in Table B-4, in the soil BHRA data set, a total of 156 pairs of primary and field duplicate results for DU-1, 14 pairs for DU-2, and 41 pairs for DU-3 were qualified due to relative percent difference (RPD) or reporting limit exceedance. None of the soil analytes qualified due to RPD or reporting limit exceedance was identified as a soil COPC, except for the following chemicals (see Section 5.4):

- DU-1: chlorate, perchlorate, chromium VI, manganese, dioxin TEQ, beta-BHC, 4,4-DDE, 4,4-DDT, hexachlorobenzene, BaPEq, naphthalene, and octachlorostyrene
- DU-2: chromium VI, beta-BHC, and hexachlorobenzene
- DU-3: arsenic, chromium VI, hexachlorobenzene, and octachlorostyrene

For the above COPCs, the duplicate results qualified due to RPD or reporting limit exceedance (see Table B-4) were less than the maximum detected concentrations used in the COPC selection (see Table 14), and were also below the BCLs based on cancer or noncancer effects or other screening criteria. Therefore, the precision issues for the duplicate samples do not impact on the COPC selection and DU identification.

Soil samples with qualified primary and field duplicate results were treated as independent samples in the COPC selection and DU identification, although the variance of the duplicate and primary samples was not tested. The impacts are discussed as follows:

- First, only 4,4'-DDE in DU-1 was identified as a soil COPC based on the maximum concentration detected in a sample with a duplicate (Table 14). The detected 4,4'-DDE concentration was 6 mg/kg in SSAM3-01-7BPC and 5.8 mg/kg in SSAM3-01-7FD (Appendix C), both of which were above the 0.1xBCL. Therefore, 4,4'-DDE would be identified as a soil COPC for DU-1 regardless of how the duplicates were treated.
- Second, although the treatment of duplicate samples may affect the results of the background evaluation, as discussed in Section 5.4.2, the metals identified as soil COPCs (arsenic in DU-1 and DU-3, chromium VI in all the three DUs, and cobalt and manganese in DU-1) were present at concentrations greater than the background, while the metals eliminated as soil COPCs (thallium and zirconium in DU-1) were present at concentrations. Given the limited amount of duplicate data for these metals (0-20%), treatment of duplicate

samples would not significantly change the selection of metal COPCs. In addition, radionuclides were excluded as COPCs based on the calculation of total cancer risks, not the statistical testing results of the background evaluation.

• Finally, among the sample locations with elevated dioxin TEQ, arsenic concentration, cancer risk, noncancer HI, and count of asbestos fibers (Figures 22 through 28), very limited duplicate samples were available, and therefore the spatial pattern of concentrations/risks would not be affected by the treatment of duplicates.

In summary, there is no impact of duplicate treatment on the COPC selection and DU identification.

#### 5.3.5 Accuracy

The soil analytical data were evaluated in DVSRs presented in Appendix B, with a subset of the data qualified with a J qualifier (J, J-, or J+) based on method blank, field duplicate, and/or other quantitation issues (4,257 out of 49,868 data points, see Table C-1); that is, the reported value was estimated, with no (J), low (J-), or high (J+) bias. The potential impact of the J qualified data on the COPC selection and DU identification was evaluated:

- J and J+ Qualified Data: A review of the J and J+ qualified data indicated that the • estimated results were either below the 0.1xBCL (or other screening criteria) or below/equal to the maximum detected concentration used in the COPC selection (Appendix B, Table B-6). Only dioxin TEQ, dieldrin, and toxaphene in DU-1 and octachlorostyrene in DU-2 were identified as COPCs based on a maximum detected concentration with a J qualifier (see Table 14). No BCL is available for octachlorostyrene, and this chemical will be discussed qualitatively in the forthcoming BHRA. The J qualified concentrations for the three COPCs in DU-1 were three- to seven-fold above the 0.1xBCL (or other screening criteria), and the actual concentrations are still likely above the 0.1xBCL (or other screening criteria). Further, although the J or J+ qualified data may affect the results of the background evaluation, as discussed in Section 5.4.2, the metals identified as soil COPCs based on the statistical testing results of the background evaluation (arsenic in DU-1 and DU-3, chromium VI in all the three DUs, and cobalt and manganese in DU-1) were present at concentrations greater than the background concentration. Given the limited amount of J or J+ qualified data for these metals (1-18%), these data would not significantly change the selection of metal COPCs. In addition, radionuclides were excluded as COPCs based on the calculation of total cancer risks, not the statistical testing results of the background evaluation. Finally, among the sample locations with elevated dioxin TEQ, arsenic concentration, cancer risk, and noncancer HI (Figures 22 through 26), only one J qualified data point was available (0.019 mg/kg for dioxin TEQ at RISB-50), and therefore the spatial pattern of concentrations/risks would not be affected by the J or J+ qualified data. In summary, the J and J+ qualified data do not have any impact on the COPC selection and DU identification.
- J- Qualified Data: A review of the J- qualified data indicated most of the results estimated with low bias were either significantly below the 0.1xBCL (or other screening criteria) or lower than the maximum detected concentration used in the COPC selection (Appendix B, Table B-6). No soil COPCs were identified based on a maximum detected concentration with a J- qualifier (see Table 14). Therefore,

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

> correction for the low bias would not change the selection of COPCs. Further, although the J- qualified data may affect the results of the background evaluation, as discussed in Section 5.4.2, the metals eliminated as soil COPCs based on the statistical testing results of the background evaluation (thallium and zirconium in DU-1) were present at concentrations consistent with background concentrations without any J- qualified data. In addition, radionuclides were excluded as COPCs based on the calculation of total cancer risks, not the statistical testing results of the background evaluation. Finally, correction of J- qualified data would not result in additional samples with elevated dioxin TEQ, arsenic concentrations, cancer risk, and noncancer HI, and therefore the spatial pattern of concentrations/risks would not be affected by the J- qualified data. In summary, the J- qualified data do not have any impact on the COPC selection and DU identification.

As discussed in Table 1, in accordance with the most recent guidance (NDEP 2012) for evaluating data associated with blank contamination, Ramboll Environ queried the censored (or nondetect) data for blank contamination from the project database, and changed them from nondetected values at the practical qualification limit (PQLs) (U qualified) to detected values at reported concentrations (J qualified) if the PQLs were higher than the reported concentrations. The revisions of censored data for blank contamination are summarized in Appendix B, Table B-5. Both the PQLs and the reported concentrations were lower than 0.1xBCL, so none of the analytes affected by blank contamination was identified as a soil COPC. In addition, the revisions of data associated with blank contamination to estimated detected values may affect the background evaluation for some metals (e.g., antimony, boron, mercury, molybdenum, platinum, selenium, tin, and tungsten); however, these metals all passed the concentration/toxicity screen (Table 8) and would not be identified as soil COPCs. Therefore, the revisions of data associated with blank contamination to estimated detected values do not have any impact on the COPC selection and DU identification.

### 5.4 Identification of Soil COPCs for Individual Decision Units

DU-specific COPCs are identified from the list of the Study Area COPCs shown in Table 11 following the same methodology as described in Section 4. The analytes eliminated as Study Area COPCs are not re-visited for the individual DUs, based on the following considerations:

- Metals and other inorganics: with the exception of calcium and potassium, the metals and other inorganics not identified as COPCs were eliminated based on 1) the concentration/toxicity screen (using the maximum detected concentration for the Study Area) or 2) the chemical-specific considerations discussed in Section 4.3. The results of the screening are the same, whether applied to the Study Area or to an individual DU. Although calcium and potassium were eliminated based on the background evaluation, these metals are of very low toxicity and would have been eliminated as Study Area COPCs even if present at concentrations greater than background (NDEP 2017).
- Organic compounds: the concentration/toxicity screen was the only criterion used to eliminate organic compounds as COPCs. The maximum detected concentration across the Study Area was used for this screen. The results of the screen are the same, whether applied to the Study Area or to an individual DU.

For chemicals identified as COPCs for the Study Area, as shown in Table 11, it is possible that when evaluated for individual DUs, they would pass the concentration/toxicity screen if the maximum detected concentration in a certain DU is lower than the maximum detected concentration in the entire Study Area. Also, for metals and radionulcides, the results of the background evaluation may change (e.g., the concentrations of a metal are greater than background levels when considered within the entire Study Area, but are consistent with background levels in a certain DU). For these reasons, the COPCs identified for the Study Area are further evaluated in this section for each individual DU.

#### 5.4.1 Step 1 – Concentration/Toxicity Screen

The summary statistics and concentration/toxicity screen for the 28 non-asbestos COPCs identified for the Study Area are presented in Table 14. For each listed chemical, the maximum detected concentration and the BCL (or other screening value) are presented. The final column indicates whether the chemical "passed" or "failed" the screen. The chemicals that "failed" the screen are carried forward to Step 2 and/or 3, and those that "passed" are excluded as COPCs. The results of concentration/toxicity screen for each DU are discussed as follows:

- DU-1: Two chemicals passed, 25 chemicals failed based on the BCL (or other screening criteria) comparison, and one chemical (octachlorostyrene) do not have a screening level.
- DU-2: 12 chemicals passed, 14 chemicals failed based on the BCL (or other screening criteria) comparison, and two chemicals (palladium and octachlorostyrene) do not have a screening level.
- DU-3: two chemicals (palladium and zirconium) were not analyzed, 12 chemicals passed, 13 chemicals failed based on the BCL (or other screening criteria) comparison, and one chemical (octachlorostyrene) does not have a screening level.

Chemicals that failed or that did not have a screening level are carried forward to Steps 2 and/or 3.

Table 2 presents the soil data summary results for asbestos (long amphibole and long chrysotile fibers) for each DU. As shown in Table 2, one or more long amphibole fibers were observed in three out of 95 post-abatement samples in DU-1 and one out of 36 post-abatement samples in DU-2. Also, one or more long chrysotile fibers were observed in 11 out of 93 post-abatement samples in DU-1 and 10 out of 34 post-abatement samples in DU-2. No long amphibole or chrysotile fibers were observed in the two post-abatement samples in DU-3. As discussed in Section 4.1.4, exposure and risk assessments for asbestos are highly dependent on sample size, and even for the case where fibers are not identified (i.e., zero fibers), upper-bound cancer risk estimates can be greater than  $1 \times 10^{-6}$ , depending on sample size. For these reasons, amphibole and chrysotile are retained as COPCs for the individual DUs.

#### 5.4.2 Step 2 – Background Evaluation

In this section, a background evaluation was conducted for each proposed DU to identify DU-specific metal and radionuclide COPCs. As indicated in the spatial plots in this report (i.e., Figures 7 through 10, Figures 23a through 23c, and spatial quartile plots in Appendix G) and in the analyses conducted by Neptune (2017), the concentrations of

some metals and some radionuclides (e.g., U-238) in the southern portion of the Site were lower than in the northern portion of the Site, suggesting that background conditions in the northern and southern portion of the Site may be different. As recommended by Neptune (2017), the regional BRC/TIMET data set was used for the northern portion of the Study Area (DU-1 and DU-3) and the RZ-A background data set was used for the southern portion of the Study Area (DU-2).

The RZ-A background data set is described in Section 3.2.2. A detailed discussion of the BRC/TIMET regional background data set is presented in the *Background Shallow Soil Summary Report, BMI Complex and Common Area Vicinity* (BRC and TIMET 2007). The BRC/TIMET regional background data set is a combination of soil background data collected by ENVIRON for the City of Henderson in 2003 and by BRC/TIMET in 2005. In the 2003 ENVIRON study, soil samples were collected from eight borings at 0 to 1 ft bgs and 3 to 4 ft bgs. In the 2005 BRC/TIMET study, soil samples were collected from 33 initial sampling locations on 11 undeveloped properties near and upgradient from the Site at 0 to 0.5, 4 to 6, and 9 to 11 ft bgs. The sampling locations of both the 2003 ENVIRON study and 2005 BRC/TIMET study were presented in Appendix A of the *Background Shallow Soil Summary Report, BMI Complex and Common Areas Vicinity*, prepared by BRC and TIMET (BRC and TIMET 2007), which were off-site locations in relatively close proximity to the Site but were upgradient and sufficiently distant so that impacts from Site operations were not likely. The RZ-A and BRC/TIMET background data set used for the background evaluation are included in Appendix E.

The DU-specific background evaluation was conducted using the same methodology as described in Section 3.2.2, with the results of statistical tests, side-by-side box plots, and Q-Q plots presented in Appendix J. As discussed in Section 3.2.2, the results of the t-tests are not reasonable to use in the background evaluation of a large amount of data. Therefore, the determination of background consistency was only based on the results of non-parametric tests for analytes with large sample size (>100), and based on the results of both parametric and non-parametric tests for analytes with small sample size (<=100).

The metals and radionuclides in each individual DU that either failed the concentration/toxicity screen or for which a BCL was not available for screening are listed in Tables 15 and 16, respectively. The results of the background evaluation are also presented in the tables and summarized as follows:

- DU-1: Of the six metals carried forward from Step 1, three metals (arsenic, thallium, and zirconium) were present at concentrations consistent with background. Two metals (cobalt and manganese) were present at concentrations greater than background. In addition, background comparison results may not be applicable for chromium VI due to low detection frequency (<25%) in both DU-1 and BRC/TIMET regional background data sets (Table J-1). Of the eight radionuclides carried forward from Step 1, one radionuclide (Th-228) failed the statistical testing for background consistency, while concentrations of all the other radionuclides were consistent with background.</li>
- DU-2: Of the three metals carried forward from Step 1, RZ-A background data were not available for two metals (palladium and zirconium). In addition, background comparison results may not be applicable for chromium VI due to low detection frequency (<25%) in both DU-2 and RZ-A background data sets (Table J-1). Of the

eight radionuclides carried forward from Step 1, all concentrations were consistent with background.

DU-3: Of the two metals carried forward from Step 1, arsenic was present at concentrations greater than background. In addition, background comparison results may not be applicable for chromium VI due to low detection frequency (<25%) in the BRC/TIMET regional background data set (Table J-1). Of the eight radionuclides carried forward from Step 1, five radionuclides (U-238, U-234, Th-232, Th-230, and Th-228) failed the statistical testing for background consistency, while concentrations of all the other radionuclides were consistent with background.</li>

Background comparison results may not be applicable for chromium VI due to low detection frequency. However, the chromium VI concentrations in the individual DUs are greater than background concentrations based on the box plot (Figure J1-2) and Q-Q plot (Figure J2-2). Therefore, chromium VI is retained as a COPC for all the three DUs.

For arsenic in DU-1, although statistical tests indicated that the concentrations were consistent with the BRC/TIMET background (Table J-2), the Q-Q plot (Figure J2-1) showed that the values were lower than the background concentration in the low concentration range and the values were higher than the background concentration in the high concentration range (the maximum concentration is 34 mg/kg). To be conservative, arsenic is retained as a COPC for DU-1.

For radionuclides, as presented in the NDEP flowchart (Appendix H), when approximate secular equilibrium is exhibited in an isotope decay chain, in theory radionuclides in the same decay chain should yield similar background comparison results; if any radionuclide is greater than background, all the radionuclides in that decay chain would be carried forward in the risk assessment. When approximate secular equilibrium is not exhibited in an isotope decay chain, those radionuclides that fail the background evaluation would be carried forward in the risk assessment. As indicated in Table 16, secular equilibrium is exhibited in the U-238 decay chain for all three DUs, but only exhibited in the Th-232 decay chain for DU-2. Also, it is unexpected that in the U-238 decay chain in DU-3, which is at secular equilibrium, radionuclides both passed and failed the background comparisons. Similar issues have previously been identified by NDEP in the radionuclide analytical data sets for soil samples collected across the BMI Complex (NDEP 2009c).

Sample preparation and analytical methods were important factors in explaining some of the radionuclide data anomalies. The BRC/TIMET regional background samples were collected and analyzed in 2003 and 2005, the RZ-A background samples were collected and analyzed in 2009, and Study Area samples were collected and analyzed between 2006 and 2014, i.e., both before and after NDEP issued guidance for evaluating radionuclide data (NDEP 2009c). Over this period, samples were submitted for analysis to different analytical laboratories and analyzed using different preparation and analytical methods. For example, the analytical methods for Ra-228 included beta spectroscopy and gamma spectroscopy, depending on the laboratory, which may be the reason for the lack of correlation with Ra-228 in the Th-232 decay chain (Table J-6). It is also an unexpected finding that for the RZ-A background data set, the Th-232 decay chain was not in secular equilibrium (Table J-5B).

Given that the validity of the statistical testing is complicated by several issues identified above, it is difficult to interpret the results of background evaluation for radionuclides

and consider them as a reliable basis for the COPC selection. To provide a point of comparison from a health risk perspective, the total radionuclide cancer risk at each sampling location was compared to the total radionuclide cancer risks for the RZ-A background soils and BRC/TIMET regional background soils. As indicated in Figures 26a through 26c, the estimated total radionuclide cancer risks were either below or at 2 x 10<sup>-4</sup> at all depths intervals throughout the Study Area, which were consistent with the estimated total radionuclide cancer risks for the RZ-A background and BRC/TIMET regional background data sets (Table 12). Radionuclides are not known to be associated with any of the former operations at the Site. Based on the above discussion, radionuclides were not identified as COPCs for any DU. The impact of excluding radionuclides as COPCs on the risk results and conclusions will be further discussed in the forthcoming BHRA.

#### 5.4.3 Step 3 – Chemical-Specific Evaluations

Since chemicals commonly recognized as having low toxicity and for which a BCL was not available (i.e., macronutrients or essential micronutrients) were already eliminated as COPCs for the Study Area in Section 4.3, no COPC is further eliminated in this step for the DU-specific evaluation.

#### 5.4.4 Preliminary DU-Specific COPCs

The preliminary COPCs identified for soils in each individual DU are listed in Table 17, and summarized as follows:

- DU-1: The eighteen COPCs identified for DU-1 include chlorate and perchlorate, four metals (arsenic, chromium VI, cobalt, and manganese), dioxin TEQ, two PAHs (BaPEq and naphthalene), six OCPs (beta-BHC, 4,4'-DDE, 4,4'-DDT, dieldrin, hexachlorobenzene, and toxaphene), octachlorostyrene, and asbestos (long amphibole and long chrysotile fibers).
- DU-2: The ten COPCs identified for DU-2 include three metals (chromium VI, palladium, and zirconium), BaPEq, two OCPs (beta-BHC and hexachlorobenzene), two SVOCs (bis(2-ethylhexyl)phthalate and octachlorostyrene), and asbestos (long amphibole and long chrysotile fibers).
- DU-3: The eight COPCs identified for DU-3 include perchlorate, two metals (arsenic and chromium VI), BaPEq, hexachlorobenzene, octachlorostyrene, and asbestos (long amphibole and long chrysotile fibers).

For two COPCs (palladium and octachlorostyrene), BCLs (and associated toxicity values) are not available; in the absence of toxicity values, these COPCs will be evaluated qualitatively in the forthcoming BHRA. Also, RZ-A background data are not available for palladium and zirconium in DU-2, and therefore a background evaluation cannot be conducted with this data set. The background comparison of these two metals will be further discussed in the forthcoming BHRA in comparison to the regional BRC/TIMET background data set (BRC and TIMET 2007). Lastly, for arsenic identified as a COPC in both DU-1 and DU-3, an evaluation of the total arsenic risk as compared to background risk will be performed in the forthcoming BHRA.

# 6. CONCLUSIONS AND PATH FORWARD

In summary, this revised interim report updated the DUE, the selection of preliminary soil COPCs, and the identification of DUs using the updated BHRA soil data set. This interim report builds upon the 2015 interim COPC report (Ramboll Environ 2015a) and the 2016 interim COPC/EU report (Ramboll Environ 2016a), incorporates the strategy discussed with NDEP and its consultants at the December 19, 2016 meeting (see Meeting Minutes in Appendix A), and addresses comments received from NDEP on December 22, 2016. A total of 30 COPCs have been identified for the entire Study Area, and based on a risk-based spatial analysis of concentrations, cancer risks, and noncancer HIs, three new DUs have been proposed. These include DU-1, the northern portion of the Study Area (north of the L'hoist North America facility), DU-2, the southern portion of the Study Area (south of the L'hoist North America facility), and DU-3, the Central Retention Basin. DU-specific DUEs and COPC identifications have also been conducted.

In 2018, the Soil BHRA Report for OU-1 will be prepared and submitted to NDEP. In the Soil BHRA Report, the soil BHRA data set will be updated by incorporating additional soil data from applicable ongoing investigations (i.e., Phase 2 RI, Unit Building 4/5 Investigation) and the COPC selection and DU identification will be updated using the same methodology as described in this report. Further, the exposure assessment, toxicity assessment, and risk characterization will be conducted, and the conclusions of the soil BHRA for OU-1 will be presented.

# 7. **REFERENCES**

- AECOM, Inc. (AECOM) 2008. Revised Phase B Site Investigation Work Plan for Areas I, II, III and IV, Tronox LLC Facility, Henderson, Nevada. December. NDEP approved response to comments (dated January 16, 2009) January 16, 2009.
- AECOM and Northgate Environmental Management, Inc. (Northgate), 2009. Revised Phase B Quality Assurance Project Plan (QAPP), Tronox LLC Facility, Henderson, Nevada. Revised July 20. NDEP approved July 27, 2009.
- Basic Remediation Company (BRC). 2009. BRC Standard Operating Procedure (SOP) 40. Data Review Validation. Revision 4. May 7. NDEP approved May 11, 2009.
- BRC and Titanium Metals Corporation (TIMET), 2007. Background Shallow Soil Summary Report, BMI Complex and Common Areas Vicinity. March 16. NDEP commented May 31, 2007 requested errata pages; Errata submitted June 28, 2007; NDEP approved July 3, 2007; Addendum submitted July 22, 2007. NDEP approved July 26, 2007.
- ENSR Corporation (ENSR). 2005. Conceptual Site Model, Kerr-McGee Facility, Henderson, Nevada. February. NDEP requested response to comments during the next monthly meeting October 22, 2005.
- ENSR. 2006. Phase A Source Area Investigation Work Plan, Tronox LLC Facility, Henderson, Nevada. September. NDEP conditionally approved October 26, 2006.
- ENSR. 2007a. Phase A Source Area Investigation Results Report, Tronox LLC Facility, Henderson, Nevada. September. NDEP approved November 30, 2007.
- ENSR. 2007b. Appendix E. Data Validation Summary Report. September 7. (*The final version of this DVSR was submitted to NDEP electronically as Appendix G of ENSR 2007a. Based on available correspondence, Appendix E [available in Ramboll Environ document files as a PDF] is the same as Appendix G, for which an electronic copy is not available.*)
- ENSR. 2008a. Phase B Source Area Investigation, Soil Gas Survey Work Plan, Tronox LLC Facility, Henderson, Nevada. March. NDEP approved March 26, 2008.
- ENSR. 2008b. Quality Assurance Project Plan, Revision 3, Tronox LLC Facility, Henderson, Nevada. April. NDEP approved April 16, 2008.
- ENVIRON International Corporation (ENVIRON). 2011. Workplan for Evaluation of Discolored Soil and Confirmation Soil Sampling in Visually-Impacted Areas. April 18. NDEP approved May 12, 2011.
- ENVIRON. 2012. Interim Soil Removal Action Completion Report, Nevada Environmental Response Trust Site, Henderson, Nevada, August 2010 – November 2011. January. Revised September 28. NDEP approved December 17, 2012.
- ENVIRON. 2014a. Remedial Investigation and Feasibility Study Work Plan, Revision 2, Nevada Environmental Response Trust Site, Henderson, Nevada. June 19. NDEP Approved July 2, 2014.
- ENVIRON. 2014b. Baseline Health Risk Assessment Work Plan, Revision 0, Nevada Environmental Response Trust Site, Henderson, Nevada. February 28. NDEP approved May 20, 2014.

- ENVIRON. 2014c. Excavation of Beta Ditch at NERT-TIMET Property Line. Nevada Environmental Response Trust Site, Henderson, Nevada. March 31. NDEP Approved June 3, 2014.
- ENVIRON. 2014d. Quality Assurance Project Plan, Revision 1, Nevada Environmental Response Trust Site, Henderson, Nevada. July 18. NDEP approved August 1, 2014.
- Kleinfelder. 1993. Environmental Conditions Assessment, Kerr-McGee Chemical Corporation, Henderson, Nevada Facility. April.
- Laboratory Data Consultants. 2013. Data Validation Summary Report, Revision 4, February to August 2011 Soil Remediation Completion Sampling. October 4. NDEP approved February 14, 2014.
- Neptune and Company, Inc (Neptune). 2007. Guided Interactive Statistical Decision Tools (GISdT).
- Neptune. 2010. Revised Data Validation Summary Report for Shallow Supplemental Soil Sampling in Areas I and II. July 21. NDEP approved July 28, 2010.
- Neptune. 2017. NERT Surface Soils COPC Identification and Delineation of Exposure Areas. May.
- Nevada Division of Environmental Protection (NDEP), 1994. Phase II Letter of Understanding between NDEP and Kerr-McGee Chemical Corporation (KMCC). August 15.
- NDEP. 2008a. Statistical Analysis Recommendations for Field Duplicates and Field Splits, BMI Plant Sites and Common Areas Projects, Henderson, Nevada. November 14.
- NDEP. 2008b. Guidance on the Development of Summary Statistics Tables, BMI Plant Sites and Common Areas Projects, Henderson, Nevada. December 10.
- NDEP. 2009a. Enforcement Action for Failure to Complete Approved Site Remediation Activities and Show Cause Meeting, Tronox, LLC, (Tronox) Henderson, Nevada, NDEP Facility ID Number 8 000539. December 14.
- NDEP. 2009b. Significance Levels for The Gilbert Toolbox of Background Comparison Tests, BMI Plant Sites and Common Areas Projects, Henderson, Nevada, prepared by Neptune and Company Inc.
- NDEP. 2009c. Guidance for Evaluating Radionuclide Data for the BMI Plant Sites and Common Areas Projects, Henderson, Nevada. February 6.
- NDEP. 2009d. Supplemental Guidance on Data Validation, BMI Plant Sites and Common Areas Projects, Henderson, Nevada. April 13.
- NDEP. 2010a. Supplemental Guidance for Assessing Data Usability for Environmental Investigations at the BMI Complex and Common Areas in Henderson, Nevada. September 1.
- NDEP. 2010b. Letter to Mr. Matt Paque, Tronox LLC, from Ms. Shannon Harbour, regarding NDEP Response to: Background Issues and Determination of Background Dataset for TRX. August 17.

- NDEP. 2010c. Letter to Mr. Matt Paque, Tronox LLC, from Ms. Shannon Harbour, regarding NDEP Response to Revised Removal Action Work Plan for Phase B Soil Remediation Zones RZ-B through RZ-E, Tronox LLC, Henderson, Nevada. August 15.
- NDEP. 2011. Action Memorandum: Removal Actions, Nevada Environmental Response Trust Site, Clark County, Nevada. July 21.
- NDEP. 2012. Revised Guidance on Qualifying Data due to Blank Contamination for the BMI Complex and Common Areas, BMI Plant Site and Common Areas Projects, Henderson, Nevada. January 5.
- NDEP. 2015. Draft comments from the Nevada Division of Environmental Protection on the Post-Remediation Screening Health Risk Assessment Report for Parcels C, D, F, G and H, Revision 3, dated June 19, 2014. Transmitted from NDEP to the Trust via email, dated February 5.
- NDEP. 2016. Nevada Division of Environmental Protection (NDEP) Response to: Identification of COPCs and Exposure Units for Soils, Nevada Environmental Response Trust Site, Henderson, Nevada. Interim Report. December 22.
- NDEP. 2017. User's Guide and Background Technical Document for NDEP Basic Comparison Levels (BCLs) for Human Health for the BMI Complex and Common Areas. December 2008, Revision 14, July 2017.
- Northgate Environmental Management Inc. (Northgate). 2009a. Scope for Additional Sampling – Phase B Investigation, Area 1. NDEP approved November 24, 2009.
- Northgate. 2009b. Scope for Additional Sampling Phase B Investigation, Area II. NDEP approved December 11, 2009.
- Northgate. 2010a. Removal Action Work Plan for Phase B Soil Remediation of Remediation Zones RZ-B through RZ-E, Tronox LLC, Henderson, Nevada. Revised June 22; Errata submitted August 13. NDEP approved Errata August 20, 2010.
- Northgate. 2010b. Final Revised Pre-Confirmation Sampling Work Plan, Remediation Zones RZ A through RZ-E, Phase B Investigation, Tronox Facility, Henderson, Nevada. March 25. NDEP approved March 30, 2010.
- Northgate. 2010c. Revised Human Health Risk Assessment for Remediation Zone A, Tronox LLC, Henderson, Nevada. July 23. NDEP approved August 20, 2010.
- Northgate. 2010d. Revised Excavation Plan for Phase B Soil Remediation of RZ-C, Addendum to the Removal Action Work Plan. September 1. NDEP conditionally approved December 1, 2010.
- Northgate. 2010e. Results of Bioaccessibility Study for Dioxin/Furans in Soil, Tronox LLC, Henderson, Nevada. May 24. NDEP approved May 25, 2010.
- Northgate. 2010f. Revised Technical Memorandum: Calculation of Leaching-Based, Site-Specific Levels (LSSLs) for the Soil-to-Groundwater Pathway Using NDEP Guidance, Tronox LLC, Henderson, Nevada. November 18. NDEP Responded January 13, 2011.
- Northgate. 2010g. Revised Data Validation Summary Report, Phase B Investigation Area I Soil, Tronox LLC, Henderson, Nevada. January 15. NDEP approved January 20, 2010.

- Northgate. 2010h. Data Validation Summary Report, Phase B Investigation Area II Soil, Tronox LLC, Henderson, Nevada. February 12. NDEP approved February 18, 2010.
- Northgate. 2010i. Data Validation Summary Report, Phase B Investigation Area III Soil, Tronox LLC, Henderson, Nevada. February 18. NDEP approved March 17, 2010.
- Northgate. 2010j. Revised Data Validation Summary Report, Phase B Investigation Area IV Soil, Tronox LLC, Henderson, Nevada. March 19. NDEP approved March 29, 2010.
- Northgate. 2011. Data Validation Summary Report, Additional Pre-Confirmation Sampling, Tronox LLC, Henderson, Nevada. April 14. NDEP approved DVSR on February 21, 2013; Northgate submitted response to comments on EDD on April 11, 2013; NDEP approved response to comments on April 25, 2013.
- Northgate. 2012. Manganese Tailings Removal Technical Memorandum, Nevada Environmental Response Trust Site, Henderson, Nevada, dated November 9. NDEP approved February 21, 2013.
- Northgate. 2014. Post-Remediation Screening Health Risk Assessment Report for Parcels C, D, F, G and H, Revision 3, Nevada Environmental Response Trust Site, Henderson, Nevada, dated June 19. Revision 4 is currently in preparation.
- Ramboll Environ US Corporation (Ramboll Environ). 2015a. Interim Report, Preliminary Selection of Facility Area COPCs, Nevada Environmental Response Trust Site, Henderson, Nevada, dated May.
- Ramboll Environ. 2015b. July 15, 2015 Meeting Summary and BHRA Path Forward. Letter from Ramboll Environ to the Nevada Environmental Response Trust Site, dated August 29, 2015. Environmental Response Trust Site, Henderson, Nevada.
- Ramboll Environ. 2016a. Interim Report, Identification of COPCs and Exposure Units for Soils, Nevada Environmental Response Trust Site, Henderson, Nevada. Dated August 26, 2016.
- Ramboll Environ. 2016b. Technical Memorandum, Remedial Investigation DataEvaluation, Nevada Environmental Response Trust Site, Henderson, Nevada, datedMay 2. NDEP approved August 23, 2016.
- Ramboll Environ. 2017. Site Management Plan, Revision 3. Nevada Environmental Response Trust Site, Henderson, Nevada. February 13. NDEP approved February 28, 2017.
- Tetra Tech. 2015. Unit 4 and 5 Buildings Investigation Work Plan, Henderson, Nevada. March 30. NDEP approved April 13, 2015.
- Tetra Tech. 2016a. Technical Memorandum: Unit 4 and 5 Buildings Investigation First Mobilization. May 6. NDEP approved June 28, 2016.
- Tetra Tech. 2016b. Response to Comments on NERT Unit 4 and 5 Buildings Investigation, First Mobilization Tech Memo. June 24. NDEP approved June 28, 2016.
- Tetra Tech. 2017a. Technical Memorandum: Unit 4 and 5 Buildings Investigation Second Mobilization. May 4. NDEP approved June 8, 2017.

- Tetra Tech. 2017b. Response to NDEP Comments Dated June 8, 2017, Unit 4 and 5 Buildings Investigation Second Mobilization Report. July 12. NDEP approved August 15, 2017.
- United States Environmental Protection Agency (USEPA). 1989. Risk Assessment Guidance for Superfund: Volume I—Human Health Evaluation Manual (Part A). Interim Final. Office of Emergency and Remedial Response, Washington, D. C. U. S. EPA/540/1-89/002. December.
- USEPA. 1992a. Guidance for Data Usability in Risk Assessment (Part A), Final. Office of Emergency and Remedial Response. April.
- http://www.epa.gov/oswer/riskassessment/datause/parta.htm
- USEPA. 1992b. Guidance for Data Usability in Risk Assessment (Part B), Final. Office of Emergency and Remedial Response. April.
- http://www.epa.gov/oswer/riskassessment/datause/partb.htm
- USEPA. 1999. Contract Laboratory Program, National Functional Guidelines for Organic Data Review. Office of Solid Waste and Emergency Response. October.
- USEPA. 2001. Contract Laboratory Program, National Functional Guidelines for Low Concentration Data Review. Office of Solid Waste and Emergency Response. June.
- USEPA. 2002. Guidance for Comparing Background and Chemical Concentrations in Soil for CERCLA Sites. Office of Emergency and Remedial Response, Washington, DC. EPA 540-R-01-003. September.
- USEPA. 2004. Office of Solid Waste and Emergency Response. Contract Laboratory Program, National Functional Guidelines for Inorganic Data Review. October.
- USEPA. 2005a. Test Methods for Evaluation Solid Waste Physical/Chemical Methods (SW-846), Third Edition, July.
- USEPA. 2005b. Office of Emergency and Remedial Response. Contract Laboratory Program Statement of Work for Chlorinated Dibenzo-p-dioxin and Chlorinated Dibenzofuran: Mult-imedia, Multi-concentration. January.
- USEPA. 2008. Office of Solid Waste and Emergency Response. Contract Laboratory Program, National Functional Guidelines for Organic Data Review. June.
- USEPA. 2009. Office of Solid Waste and Emergency Response. Guidance for Labeling Externally Validated Laboratory Analytical Data for Superfund Use. January.
- USEPA. 2017. Regional Screening Levels User's Guide. June.
- Van den Berg M. et al. 2006. The 2005 World Health Organization Reevaluation of Human and Mammalian Toxic Equivalency Factors for Dioxins and Dioxin-like Compounds. Toxicology. Science 92:223-241.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

**TABLES** 

## Nevada Environmental Response Trust Site

Data Usability Criterion	Evaluation Result
(description of criterion)	
I. Reports to the Risk	Historical Investigations
Assessor List all reports and dates	The work plans and DVSRs <sup>1</sup> for historical investigations completed within the Study Area are reported in the following documents.
and confirm that report(s)	Phase A Investigation (between November 1 and December 8, 2006)
and appropriate for use in	Phase A Work Plan (ENSR 2006, approved by NDEP on October 26, 2006)
the BHRA	<ul> <li>Phase A Source Area Investigation Results Report (ENSR 2007a, approved by NDEP on November 30, 2007)</li> </ul>
	Phase A DVSR (ENSR 2007b, approved by NDEP on December 17, 2007)
	Phase B Investigation (between June 11 and July 11, 2008 and between June 2 and November 5, 2009)
	Phase B Work Plan Areas I-IV (AECOM 2008, approved by NDEP on January 16, 2009)
	(A Phase B investigation results report was not identified.)
	• DVSR, Phase B Investigation Area I Soil (Northgate 2010g, approved by NDEP on January 20, 2010)
	• DVSR, Phase B Investigation Area II Soil (Northgate 2010h, approved by NDEP on February 18, 2010)
	• DVSR, Phase B Investigation Area III Soil (Northgate 2010i, approved by NDEP on March 17, 2010)
	• DVSR, Phase B Investigation Area IV Soil (Northgate 2010j, approved by NDEP on March 29, 2010)
	Phase B Supplemental Investigation (between December 9 and December 22, 2009)
	Phase B Scope for Additional Sampling Area I (Northgate 2009a, approved by NDEP on November 24, 2009)



<sup>&</sup>lt;sup>1</sup> DVSRs are provided in Appendix B.

## Nevada Environmental Response Trust Site

Phase B Scope for Additional Sampling Area II (Northgate 2009b, approved by NDEP on December 11, 2009)
(A Phase B supplemental investigation results report was not identified.)
<ul> <li>DVSR, Phase B Shallow Soil Supplemental Sampling Areas I and II (Neptune 2010, approved by NDEP on July 28, 2010)</li> </ul>
Interim Soil Removal Pre-Confirmation Sampling (between April 6 and November 12, 2010)
• Pre-Confirmation Sampling Work Plan (Northgate 2010b, approved by NDEP on March 30, 2010)
(A pre-confirmation sampling results report was not identified.)
• DVSR, Additional Pre-Confirmation Sampling (Northgate 2011, approved by NDEP on February 21, 2013)
Confirmation and Completion Sampling for the Interim Soil Removal Action (between February 9 and August 31, 2011)
Confirmation Soil Sampling Work Plan (ENVIRON 2011, approved by NDEP on May 12, 2011)
<ul> <li>Interim Soil Removal Action Completion Report (ENVIRON 2012, approved by NDEP on December 17, 2012)</li> </ul>
<ul> <li>DVSR, Revision 4, February to August 2011 Soil Remediation Completion Sampling (Laboratory Data Consultants 2013, approved by NDEP on February 13, 2014)</li> </ul>
Other Removal Actions
• Excavation of Beta Ditch at NERT-TIMET Property Line (ENVIRON 2014c, NDEP approved June 3, 2014)
Samples not included in the above investigations
Samples at three locations (TSB-GJ-03, TSB-GJ-04, and TSB-GR-02), collected as part of the investigations of Parcels C, D, F, G, and H, map within the Study Area. These samples have been removed from the parcels risk assessment data set and are now included in the Study Area data set.
Remedial Investigation (October 2014 to present)



## Nevada Environmental Response Trust Site

	<ul> <li>Remedial Investigation and Feasibility Study Work Plan (ENVIRON 2014a, approved by NDEP on July 2, 2014)</li> </ul>
	Remedial Investigation Data Evaluation Technical Memorandum (Ramboll Environ 2016b, approved by NDEP on August 23, 2016)
	Remedial Investigation DVSR (ongoing; DVSR will be submitted to NDEP for approval)
	Overall, the available reports, and the accompanying laboratory reports and DVSRs, are considered complete for BHRA purposes.
II. Documentation	For this step, Ramboll Environ reviewed the soil samples collected and reported in the documents listed
Confirm that each analytical result is	in chronological order):
associated with a specific sample location and that the appropriate sampling procedure is documented.	• Identification of "removed" samples and re-assignment of sample depths: Following the manganese tailings and interim soil removal actions (described in Section 2.1.2), samples that had been collected within an excavated area were tagged in the NERT project database as "removed." The "remaining" samples were reviewed, as follows: for each sample, the sample location and depth (as reported in the original investigation) were reviewed and the top and bottom depths were re-assigned, as needed, to reflect the post-excavation depth. The re-assignment was necessary because not all excavated areas were back-filled to original grade. For example, a sample with a pre-excavation top depth of 10 ft bgs in an area for which soil was excavated to 8 ft bgs and then backfilled with 5 ft of clean soil, was reassigned a top depth of 7 ft bgs.
	• Confirmation of sample locations: Samples with missing geographic location information (i.e., x, y coordinates and/or depth) were removed from the BHRA data set. (Approximately 23 samples were missing this information.) The geographic location of each "remaining" sample was confirmed relative to the current boundaries of the Operations Area, ECAs, and parcels. Samples located outside the Study Area were removed from the BHRA data set, and samples collected as part of a parcel investigation but actually located in the Study Area were moved into the BHRA data set.
	• Confirmation of sampling procedures: As discussed in the work plans listed under Criterion I, all sample collection and handling procedures were consistent with the NDEP-approved QAPP (ENSR 2008b, AECOM



# Nevada Environmental Response Trust Site

# Henderson, Nevada

	and Northgate 2009, ENVIRON 2014d). Ramboll Environ reviewed the chain-of-custody forms prepared in the field and compared them with the analytical data results provided by the laboratories to ensure completeness of the data set.
	The available information is adequate to relate each analytical result retained in the risk assessment dataset to a geographic location, depth interval, and sampling procedure.
III. Data Sources	Historical Investigations
Confirmation that source areas are adequately sampled and that analytical methods are appropriate to identify COPCs and estimate EPCs.	Samples were collected in accordance with the work plans listed under Criterion I. Both judgmental and random sampling approaches were followed, with judgmental samples collected at LOUs that had been identified as source areas. Following each investigation, results were reviewed in consultation with NDEP and areas for additional sampling were identified.
	As part of the work plans and the QAPPs, the use of standard USEPA analytical methods (listed under Criterion IV) were approved by NDEP. Analyses were conducted by NDEP-certified laboratories for the classes of chemical compounds identified as SRCs, including chlorine oxyanions (chlorate and perchlorate), metals and other inorganics, radionuclides, asbestos, dioxins/furans, organic acids, PAHs, PCBs, OCPs, OPPs, SVOCs, TPHs (diesel, gasoline, and oil/grease), and VOCs.
	Remedial Investigation
	As part of the ongoing RI/FS (ENVIRON 2014a; Ramboll Environ 2016b), soil samples were collected to address data gaps identified during review of available historical soil and groundwater data. Surface and subsurface samples were collected from two areas (referred to as Areas 5 and 6 in the work plan) to address spatial data gaps during Phase 1 RI. Review of the analytical results indicates that these spatial data gaps have been addressed. Other areas with data gaps are being evaluated by undergoing investigations (i.e., Phase 2 RI, Unit 4 and 5 Buildings Investigation).
	The specific analyses conducted (asbestos, chlorine oxyanions, metals, VOCs, SVOCs, OCPs, PAHs, PCBs, dioxins/furans, organic acids, radionuclides, TPH, and general chemistry) were identified based on the review of the historical sampling results; Analyses with standard USEPA analytical methods (listed under Criterion IV) were conducted by NDEP-certified laboratories.

Page 4 of 13



## Nevada Environmental Response Trust Site

	In summary, the review of sampling coverage from the current BHRA data set is based on the distribution of sample locations prior to excavation, as well as samples collected as part of the Phase 1 RI. A review of the updated BHRA data set will be conducted in the forthcoming BHRA to ensure that sample coverage is considered adequate for purposes of the BHRA. The USEPA analytical methods are adequate for characterizing potential contaminants in soils and provide quantitative analytical results that are of adequate quality for deriving EPCs.
IV. Analytical Methods	Standard analytical methods were used for all analyses as listed below.
and Detection Limits	Historical Investigations
Confirm that analytical methods appropriately	USEPA Method 6020 or 6010 (metals)
identify the chemical form	USEPA Method 7199 or 7196A (chromium VI)
or species and that the	USEPA Method 7471A (mercury)
concentration appropriate	USEPA Method 350.1 (ammonia)
for the BHRA.	USEPA Method 540-R-97-028 (asbestos)
	USEPA Method 9056 or 300 (bromide, chloride, nitrate, ortho-phosphate, sulfate)
	USEPA Method 365.1 or 6020 (phosphorus)
	• USEPA Method 9056, 300 or 300.1 (chlorate)
	USEPA Method 9012A (cyanide [total])
	USEPA Method 300.1 (chlorite)
	USEPA Method 300 (fluoride)
	USEPA Method 9056 or 353.2 (nitrite)
	USEPA Method 314.0 (perchlorate)
	USEPA Method 8015 or 9071B (TPHs and fuel alcohols)



## Nevada Environmental Response Trust Site

USEPA Method 8081 (OCPs)
USEPA Method 8082 (PCB Aroclors)
USEPA Method 1668A (PCB congeners)
USEPA Method 8290 or 8290 Screen (dioxins/furans)
DOE EML HASL 300 (Th, U)
DOE EML HASL 300, USEPA Method 901.1, or USEPA Method 903.1 (Ra-226)
DOE EML HASL 300, USEPA Method 901.1, or USEPA Method 904.0 (Ra-228)
USEPA Method 8141A (OPPs)
USEPA Method 8151 (herbicides)
USEPA Method 8260 (VOCs)
USEPA Method 8270 SIM or 8270 (SVOCs)
USEPA Method 8315A (formaldehyde)
HPLC-UV per Alpha (organic acids)
Remedial Investigation
USEPA Method 6020 or 6010 (metals)
USEPA Method 7199 (chromium VI)
USEPA Method 7471 (mercury)
• SM 4500 (ammonia)
USEPA Method 540-R-97-028 (asbestos)
USEPA Method 300 (bromide, chloride, nitrate, nitrite, ortho-phosphate, sulfate)
USEPA Method 300.1 (chlorate)



# Nevada Environmental Response Trust Site

USEPA Method 314.0 (perchlorate)
USEPA Method 8015 (TPHs)
USEPA Method 8081 (OCPs)
USEPA Method 8082 (PCB Aroclors)
USEPA Method 1668A (PCB congeners)
USEPA Method 8290 (dioxins/furans)
DOE EML HASL 300 (Th, U)
• USEPA Method 903.0 (Ra-226)
• USEPA Method 904.0 (Ra-228)
USEPA Method 8141A (OPPs)
USEPA Method 8260 (VOCs)
USEPA Method 8270 or 8270 SIM (SVOCs)
USEPA Method 8270 (organic acids)
The above methods are adequate to characterize the corresponding chemical groups in soil.
The SQLs were evaluated to confirm that they were sufficiently low for risk characterization (i.e., below 0.1xBCL, as established in NDEP 2017). As shown in Table 3, maximum SQLs were less than the stated screening levels, with the following exceptions:
<ul> <li>For 28 analytes (OCPs, OPPs, PAHs, PCB Aroclors, SVOCs, and VOCs), the SQLs exceeded 0.1xBCL in 0.2 to 15% of the samples reported as non-detected.</li> </ul>
• For dioxin TEQs, the SQLs exceeded the site-specific action level of 0.0027 mg/kg in one out of three samples reported as non-detected, while the detection frequency was 99% (463 out of 466 samples).



# Nevada Environmental Response Trust Site

	• For hexachlorobenzene, the SQLs exceeded 0.1xBCL in in 168 out of 289 samples reported as non- detected, while the detection frequency was 57% (385 out of 674 samples).
	• Benzidine and n-nitroso-di-n-propylamine were reported as less than detection limits in all samples; the SQLs exceeded 0.1xBCL in 100% and 71% of the non-detected samples, respectively.
	Overall, the SQLs were sufficiently low for risk characterization. The impacts of the few exceptions with elevated SQLs on the COPC selection and DU identification are further discussed in Section 5.3.2.
V. Data Review	The laboratory results from historical investigations and the RI were subjected to formal data validation
Confirm that the quality of the analytical data is assessed by professionals knowledgeable in field collection procedures and analytical chemistry and that data quality is adequate to estimate EPCs.	consistent with USEPA guidelines (USEPA 1999, 2001, 2004, 2005a,b, 2008, 2009), the BMI Plant Site Specific Supplemental Guidance on Data Validation (NDEP 2009d), and BRC Standard Operating Procedure (SOP) 40 and Data Review/Validation (BRC 2009). The USEPA guidelines, which were prepared for Contract Laboratory Program data, were adapted to reflect the analytical methods and measurement quality objectives established for the individual sampling events and NDEP guidance.
	The NDEP-approved DVSRs listed in Criterion I for soil data included in the BHRA data set are provided in Appendix B, in which the names and qualifications of the reviewers, the specific data validation procedures, and the qualification findings are presented. Each DVSR includes the following tabular summaries of the data qualifications:
	Summary of data qualified due to holding time exceedances
	Summary of data qualified due to detection below quantitation limit
	Summary of data qualified due to laboratory blank contamination
	Summary of data qualified due to field blank contamination
	• Summary of data qualified due to matrix spike (MS)/matrix spike duplicate (MSD) recovery exceedances
	Summary of data qualified due to laboratory control spike (LCS) recovery exceedances
	Summary of data qualified due to field/laboratory duplicate
	Summary of data qualified due to surrogate recovery exceedances



## Nevada Environmental Response Trust Site

	Summary of data qualified due to calibration violations
	Summary of data qualified due to calibration range exceedances
	Summary of data qualified due to internal standard recovery exceedances
	Summary of data qualified due to serial dilutions
	Summary of qualified data results
	Summary of rejected data results
	These data qualifications are further discussed below as a component of Criterion VI.
VI. Data Quality	Completeness
Indicators Document that sampling and analysis DQIs are evaluated using criteria specific to the risk assessment.	The completeness criterion includes evaluation of field completeness and laboratory completeness. Field completeness is defined as the percentage of samples collected versus those intended to be collected as specified in the sampling work plans. Laboratory completeness is defined as the percentage of samples reported by the laboratories versus those requested on the chain-of-custodies. The completeness goal stated in the QAPPs is 90% or greater. First, completeness was reviewed as reported in the DVSR prepared for each individual investigation contributing to the soil BHRA data set. A comparison of samples reported in the NERT project database with the work plans for soil investigations listed under Criterion I indicates an actual field completeness of 99% to 100% for all sampling events. In addition, all chain-of-custodies requests were executed by the laboratories,
	with minor exceptions detailed in the DVSRs. Depending on the specific DVSR, 99.2% to 100% laboratory completeness was archived based on validated data, with 0% to 0.8% of the data qualified as rejected ("R" qualified).
	Rejected ("R" qualified) data associated with post-remediation soil samples at 0-10 ft bgs in the Study Area are summarized in Appendix B, Table B-3. Laboratory completeness was calculated for the soil BHRA data set (Appendix C) for the Study Area as 99.9%.
	In summary, both field and laboratory completeness meet the completeness goals of 90% established in the QAPPs. Rejected data are excluded from the soil BHRA data set, and a discussion of how these rejected data



# Nevada Environmental Response Trust Site

occurrences potentially affect the COPC selection and DU identification are further discussed in Section 5.3.1.
Comparability
Comparability is a qualitative characteristic expressing the confidence with which one data set can be combined with another for purposes of estimating exposure. More specifically, comparability is a qualitative expression of the measure of confidence that two or more data sets may contribute to a common analysis. In general, comparability of data is maximized by using standard methods for sampling and analysis, reporting data, and data validation.
Soil samples identified for the BHRA were collected by different entities and analyzed by different analytical laboratories (and in some cases, different analytical methods were used for the same analyte); overall, the investigations from which data are being used span a period of approximately 10 years. Different reporting limits for the same analyte may also impact the comparability of the data sets. The ranges of the SQLs for each analyte where the detection frequency was less than 100% are presented in Table 3. For most of the analytes, the SQLs are well below 0.1xBCL (or other screening criteria); therefore, different reporting limits for the same analyte would not affect the COPC selection and DU identification. There are a few analytes with SQLs exceeding 0.1xBCL (or other screening criteria), and their impacts on the COPC selection and DU identification are further discussed in Section 5.3.2.
Of particular concern are possible differences in the background and Study Area data sets for both metals and radionuclides as a result of different sample preparation methods, modified (or different) analytical methods, and possible systematic differences among the internal laboratory SOPs. For example, the Q-Q plots for cadmium and iron indicate that Study Area concentrations are generally less than background (see Section 3.2.2). These observations indicate possible differences in the data sets, possibly associated with sample extraction, analytical methods, or other less-identifiable differences across the SOPs used by the different laboratories. For radionuclides, such issues were even more obvious, and may be important factors in explaining some of the radionuclide data anomalies. RZ-A background samples were collected and analyzed in 2009, while Operations Area samples were collected and analyzed between 2006 and 2014, i.e, both before and after NDEP issued guidance for evaluating radionuclide data (NDEP 2009c). Over this time period, samples were submitted for analysis to different analytical laboratories and analyzed using different



# Nevada Environmental Response Trust Site

preparation and analytical methods. The impact of these issues on the COPC selection and DU identification are further discussed in Section 5.3.2.
Representativeness
Representativeness is the degree to which data accurately and precisely represent a characteristic of the population at a sampling point or an environmental condition. There is no standard method or formula for evaluating representativeness, which is a qualitative term. Spatial representativeness is achieved through selection of sampling locations that are appropriate relative to the objective of the specific investigation, and by collection of an adequate number of samples from locations identified in relation to the investigation objectives. Concentration representativeness is achieved by obtaining analytical results of sufficient quality, as specified in the QAPP.
Spatial representativeness was discussed previously under Criterion III. As noted, both judgmental and random sampling approaches were followed, with judgmental samples collected at LOUs that had been identified as source areas. The objectives of the sampling programs were met, considering the phased approach used to delineate contaminated areas. Other areas with data gaps are being evaluated by undergoing investigations (i.e., Phase 2 RI, Unit 4 and 5 Buildings Investigation). The spatial representativeness of the updated BHRA data set will be discussed in the forthcoming BHRA to ensure that the post-excavation data provide a conservative representation of current conditions within the Study Area in the context of the CSM.
As presented in the DVSRs listed under Criterion I, standard methods for sampling and analysis were used for all the investigations, which confirmed that the analytical data are representative of the soil concentrations at the locations sampled.
Precision
Precision is a measure of the degree of agreement between replicate measurements of the same source (field precision) or sample (analytical precision). Field precision is evaluated by calculating the RPD between the primary field sample and its field duplicate. Laboratory precision is quantitated for each laboratory data batch by calculating the RPD using data for the LCS/laboratory control spike duplicate (LCSD) and/or data for the MS/MSD. The field precision goal established in the QAPPs is a RPD of less than or equal to 50%, except for the case in which one (or both) of the primary or duplicate result is less than five times the


reporting limit. For the latter case, the acceptance criteria is the reporting limit (i.e., the absolute value of the difference between the primary and duplicate result is less than or equal to the reporting limit). Laboratory precision goals are defined for specific analytical methods, as indicated in the QAPP (see Table 2 of ENVIRON [2014d]).
Field precision for the Study Area samples was assessed by evaluating the field duplicate results in accordance with the Statistical Analysis Recommendations for Field Duplicates and Field Splits (NDEP 2008a), where the primary sample and field duplicate are independent samples. A total of 211 pairs of primary and field duplicate results were qualified due to RPD or reporting limit exceedance (see Appendix B, Table B-4). For laboratory duplicates, there were 573 data qualified due to RPD or reporting limit exceedance (see DVSRs tables in Appendix B). All data with precision exceedances were qualified as "J/Estimated" or "UJ/Estimated non-detected" and are determined to be usable for purposes of the BHRA, and the effects of these qualified data on the COPC selection and DU identification are further discussed in Section 5.3.4.
Accuracy
Accuracy measures the level of bias that an analytical method or measurement exhibits. Both field accuracy and laboratory accuracy are evaluated under this DQI. Accuracy in the field is assessed through the use of trip and equipment blanks and through adherence to all sample handling, preservation, and holding time requirements. As specified in the QAPPs, the objective for trip and equipment blanks is for no analyte to be present at levels greater than the PQL. Accuracy in the laboratory analytical data is a measure of the overestimation or underestimation of reported concentrations. Several QC parameters are used to evaluate the accuracy of reported analytical results, including:
Holding times;
Field and laboratory blanks;
MS/MSD percent recovery;
Surrogate spike recovery; and
LCS percent recovery.
All qualified results (i.e., U, J, J-, and J+ qualified data) for the non-asbestos analytes are presented in Appendix C, Table C-1, and the reasons for these qualified results are summarized in the DVSRs (see Appendix B). Although laboratory limits were exceeded for certain compounds or analyses, as identified by the laboratory (and confirmed during data validation), there does not appear to be a systematic or widespread impact on the quality of the analytical results. Furthermore, based on a review of the laboratory



#### TABLE 1. Data Usability Evaluation

#### Nevada Environmental Response Trust Site

narratives (provided in the laboratory reports in each DVSR), the laboratory does not believe that the observed exceedances of laboratory criteria are cause for concern. Therefore, the qualified data are determined to be usable and valid for purposes of the BHRA and are included in the BHRA data set. The impacts of qualified data on the COPC selection and DU identification are further discussed in Section 5.3.5.
Data collected before 2012 and associated with field and laboratory blank contamination were originally qualified as nondetects based on the NDEP guidance at that time. As requested by NDEP and in accordance with the most recent guidance (NDEP 2012) for evaluating data associated with blank contamination, Ramboll Environ queried the censored data for blank contamination from the project database, and changed them from nondetected values at PQLs (U qualified) to detected values at reported concentrations (J qualified) if the PQLs were higher than the reported concentrations. The revisions of censored data for blank contamination are summarized in Appendix B, Table B-5, and the impacts on COPC selection and DU identification are further discussed in Section 5.3.5.
In summary, with the exception of the rejected data discussed listed in Appendix B, Table B-3, all data are acceptable through the DQI evaluation and deemed to be usable for risk assessment purposes.



Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
RISB-09-0.5-20141211	DU-1	N	12/11/2014	0.5	0	0	8960000	8960000
RISB-09-5.0-20141211	DU-1	N	12/11/2014	5	0	0	8900000	8900000
RISB-10-0.5-20141215	DU-1	N	12/15/2014	0.5	1	0	8970000	8970000
RISB-10-5.0-20141215	DU-1	N	12/15/2014	5	0	0	8860000	8860000
RISB-11-0.5-20141217	DU-1	N	12/17/2014	0.5	0	0	8900000	8900000
RISB-11-5.0-20141217	DU-1	N	12/17/2014	5	0	0	8970000	8970000
RISB-12-0.5-20141215	DU-1	N	12/15/2014	0.5	1	2	8900000	8900000
RISB-12-5.0-20141216	DU-1	N	12/16/2014	5	0	0	8900000	8900000
RISB-13-0.5-20141217	DU-1	N	12/17/2014	0.5	0	0	8970000	8970000
RISB-13-5.0-20141218	DU-1	N	12/18/2014	5	0	0	8940000	8940000
RISB-14-0.5-20141216	DU-1	N	12/16/2014	0.5	0	0	8970000	8970000
RISB-14-5.0-20141216	DU-1	N	12/16/2014	5	1	0	0 8920000	
RSAH3-0.0	DU-1	N	6/11/2008	0.5	0	1	2998000	2998000
RSAK4-0.0	DU-1	N	6/12/2008	0	0	1	1 2991000	
RSAK6-0.0	DU-1	N	6/17/2008	0	0	0	2976000	2976000
RSAL4-0.0	DU-1	N	6/12/2008	0	0	0	2999026	2999026
RSAL5-0.0	DU-1	N	6/12/2008	0	0	1	2966000	2966000
RSAL6-0.0B	DU-1	Ν	9/16/2009	0	0	0	8860000	8860000
RSAL7-0.0	DU-1	N	6/17/2008	0	0	0	2981000	2981000
RSAL8-0.0	DU-1	N	6/17/2008	0	0	0	2991000	2991000
RSAM2-0.0	DU-1	Ν	6/18/2008	0	0	0	2959000	2959000
RSAM3-0.0	DU-1	Ν	6/18/2008	0.5	0	0	2966000	2966000
RSAM4-0.0	DU-1	Ν	6/18/2008	0	0	0	2969000	2969000
RSAM6-0.0B	DU-1	N	7/17/2009	0	0	0	8860000	8860000
RSAM7-1.00BPC	DU-1	Ν	6/16/2010	0.5	0	0	8224442	8224442
RSAN2-0.0	DU-1	Ν	6/18/2008	1	0	0	2985000	2985000
RSAN3-0.0	DU-1	N	6/18/2008	0	0	0	2983000	2983000
RSAN4-0.0	DU-1	N	6/18/2008	0	0	0	2983000	2983000
RSAN5-0.0B	DU-1	N	7/28/2009	0	0	0	8850000	8850000
RSAN7-0.0B	DU-1	Ν	8/4/2009	0.5	0	0	8850000	8850000
RSA02-0.0	DU-1	N	6/16/2008	0	0	0	2974627	2974627

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
RSAO6-0.33BPC	DU-1	N	4/15/2010	0	0	0	8930000	8930000
RSAO6-0.33BPC-FD	DU-1	FD	4/15/2010	0	0	0	8910000	8910000
RSAP5-0.0B	DU-2	N	8/7/2009	0.5	0	2	8910000	8910000
RSAP7-0.0B	DU-2	N	10/27/2009	0	0	0	8970000	8970000
RSAQ4-0.0B	DU-2	N	8/7/2009	0.5	0	0	8850000	8850000
RSAQ7-0.0B	DU-2	N	10/2/2009	0	0	0	8850000	8850000
RSAS5-0.0B	DU-2	N	9/24/2009	0.5	0	< 1	8870000	8870000
RSAS8-0.33BPC	DU-2	N	4/8/2010	1.5	0	0	8880000	8880000
S2-PG-1-1-0.0	DU-2	N	4/8/2010	0	0	0	8940000	8940000
S2-PG-1-1-0.0-FD	DU-2	FD	4/8/2010	0	0	0	8910000	8910000
SA05-033BPC	DU-2	N	5/13/2010	1	0	0 8310000		8310000
SA100-0.0	DU-1	N	6/18/2008	0.5	0	0 2969000		2969000
SA103-0.0B	DU-2	N	8/7/2009	0	0	2	8870000	8870000
SA11-0.33BPC	DU-1	N	4/12/2010	0	0	0	0 8860000	
SA121-0.33BPC	DU-2	N	4/7/2010	0.5	0	3	3 2980000	
SA122-0.0B	DU-2	Ν	9/11/2009	3.5	0	1	8930000	8930000
SA123-0.0B	DU-1	Ν	7/13/2009	1	0	1	8850000	8880000
SA126-0.0B	DU-2	Ν	8/5/2009	0.5	0	0	8920000	8920000
SA136-0.33BPC	DU-2	Ν	4/8/2010	1	0	0	8930000	8930000
SA138-0.0B	DU-2	Ν	8/7/2009	0	0	0	8970000	8970000
SA144-0.0B	DU-1	Ν	7/29/2009	0.5	0	0	8910000	8910000
SA144009-0.0B	DU-1	FD	7/29/2009	0.5	0	0	8910000	8910000
SA145-0.0B	DU-1	N	7/23/2009	0	0	0	8850000	8850000
SA151-0.0B	DU-1	Ν	7/28/2009	0	0	0	8910000	8910000
SA151-0.33BPC	DU-1	Ν	4/12/2010	0.5	0	0	8870000	8870000
SA151009-0.0B	DU-1	FD	7/28/2009	0	0	0	8910000	8910000
SA152-0.0	DU-1	N	6/16/2008	0	0	0	2826974	2826974
SA157-0.0B	DU-1	Ν	10/2/2009	0	0	0	8860000	8860000
SA158-0.0B	DU-1	Ν	8/7/2009	0	0	0	8910000	8910000
SA166-0.0	DU-1	N	6/18/2008	0	0	0	2969000	2969000
SA170-0.0B	DU-2	N	8/11/2009	1.5	0	0	8860000	8860000

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SA170009-0.0B	DU-2	FD	8/11/2009	1.5	0	0	8920000	8920000
SA176-0.0	DU-1	N	6/18/2008	0	0	0	2991000	2991000
SA18	DU-1	N	12/3/2006	0	0	2	2995000	5990000
SA185-0.0B	DU-1	N	7/31/2009	0.5	0	1	8930000	8930000
SA186-0.0B	DU-1	N	7/31/2009	0	0	0	8920000	8920000
SA189-0.33BPC	DU-1	N	5/3/2010	0.5	0	0	8960000	8960000
SA197-0.0B	DU-3	N	7/13/2009	0	0	0	8920000	8920000
SA200-0.0B	DU-1	N	7/29/2009	0.5	0	0	8850000	8850000
SA21	DU-1	N	12/2/2006	0	0	0	2935000	2935000
SA211-0.0B	DU-2	N	8/7/2009	0.5	0	0	8940000	8940000
SA211009-0.0B	DU-2	FD	8/7/2009	0.5	0	0	8940000	8940000
SA212-0.0B	DU-2	N	8/7/2009	0	0	1 8850000		8850000
SA31-0.0B	DU-2	N	9/15/2009	0	0	0 8910000		8910000
SA47-0.0	DU-1	N	6/19/2008	0.5	0	0 2969000		2969000
SA54-0.0B	DU-1	N	7/29/2009	1	0	0 8870000		8870000
SA55-0.0	DU-1	N	6/18/2008	0	0	0	2978000	2978000
SA6	DU-2	N	12/7/2006	0.5	0	0	2846000	2846000
SA62-0.0B	DU-1	N	7/17/2009	0	0	0	8850000	8850000
SA64-0.0B	DU-3	N	7/13/2009	0	0	0	8960000	8960000
SA67-0.0	DU-1	N	6/18/2008	0	0	0	2978000	2978000
SA69-0.0	DU-1	N	6/18/2008	0	0	0	2991000	2991000
SA7	DU-2	N	12/7/2006	0	0	1	2988000	2990000
SA70-0.0B	DU-1	N	7/13/2009	0.5	0	0	8910000	8910000
SA71-0.0B	DU-1	N	7/17/2009	0	0	0	8860000	8860000
SA73-0.0B	DU-1	N	9/16/2009	0	0	0	8920000	8920000
SA74-0.0	DU-1	N	6/12/2008	0	0	0	2969000	2969000
SA75-0.0	DU-1	N	6/12/2008	0	0	0	2978000	2978000
SA77-0.33BPC	DU-2	N	4/20/2010	1	0	0 0 8930000		8930000
SA77-0.33BPC_FD	DU-2	FD	4/20/2010	1	0	0	8930000	8930000
SA8	DU-2	N	12/7/2006	0.5	0	2	2997000	5990000
SA85-0.0	DU-1	N	6/18/2008	0	0	1	2991000	2991000

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SA87-0.0	DU-1	Ν	6/18/2008	0	0	0	2982000	2982000
SSAK2-01-0.00BPC	DU-1	N	4/15/2010	2.5	0	0 8900000		8900000
SSAK2-01-0.00BPC_FD	DU-1	FD	4/15/2010	2.5	0	0	8900000	8900000
SSAK5-02-0.00BPC	DU-1	N	4/21/2010	0	0	0	8940000	8940000
SSAK5-03-0.00BPC	DU-1	N	5/12/2010	0	0	0	8870000	8870000
SSAL2-03-0.00BPC	DU-1	N	4/28/2010	0	0	3	8940000	8940000
SSAL3-03-0.00BPC	DU-1	N	5/11/2010	0	0	3	7500000	7500000
SSAL4-02-0.00BPC	DU-1	N	4/15/2010	0	0	0	8960000	8960000
SSAL4-02-0.00BPC_FD	DU-1	FD	4/15/2010	0	0	0	8930000	8930000
SSAL4-03-0.00BPC	DU-1	N	4/15/2010	0	0	0	8940000	8940000
SSAL5-01-0.00BPC	DU-1	N	5/13/2010	0	0	0	8000000	8000000
SSAL5-02-0.00BPC	DU-1	N	5/13/2010	0	0	0 8130000		8130000
SSAL7-03-0.00BPC	DU-1	N	4/23/2010	0.5	0	0	0 8870000	
SSAM4-01-0.67BPC	DU-1	N	8/3/2010	1	0	0 8940000		8940000
SSAM4-03-0.00BPC	DU-1	N	8/3/2010	0	0	0	8900000	8900000
SSAM4-04-0.00BPC	DU-1	N	6/28/2010	0.5	0	0	8910000	8910000
SSAM7-01-0.00BPC	DU-1	N	4/22/2010	0	0	0	8870000	8870000
SSAM7-02-0.33BPC	DU-1	N	4/19/2010	0.5	0	0	8840000	8840000
SSAM7-08-0.00BPC	DU-1	N	8/23/2010	0.5	0	< 1	8108643	8108643
SSAN3-01-0.00 BPC	DU-1	N	4/9/2010	0	0	0	8930000	8930000
SSAN3-02-0.00BPC	DU-1	N	4/8/2010	0	0	0	8930000	8930000
SSAN4-01-0.00_1_BPC	DU-1	N	10/28/2010	2	0	< 2	7516394	7516394
SSAN5-01-0.00BPC	DU-1	N	4/12/2010	0	0	0	8850000	8850000
SSAN6-05-0.00BPC	DU-1	N	4/12/2010	0	0	1	8920000	8920000
SSAN7-03-0.00BPC	DU-1	N	5/12/2010	0.5	0	0	7650000	7650000
SSAN7-03-0.00BPC_FD	DU-1	FD	5/12/2010	0.5	0	0	7380000	7380000
SSA05-01-0.00BPC	DU-1	N	4/20/2010	0.5	0	0	8960000	8960000
SSAO5-01-0.00BPC_FD	DU-1	FD	4/20/2010	0.5	0	0	8930000	8930000
SSAO6-04-0.00BPC	DU-1	N	4/12/2010	0	0	0	8930000	8930000
SSA07-01-0.00BPC	DU-1	N	4/12/2010	0	0	0	8930000	8930000
SSAQ3-04-0.00_1_BPC	DU-2	N	9/24/2010	0	0	< 2	7993975	7993975

### TABLE 2. Soil Sampling Results for Asbestos (Long Amphibole and Chrysotile Fibers) Nevada Environmental Response Trust Site Handessen Nevada

Henderson, Nevada

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SSAQ4-01-0.00BPC	DU-2	N	4/7/2010	0	0	3	2990000	2990000
SSAQ4-02-0.00 BPC	DU-2	N	4/9/2010	0	0	0	8910000	8910000
SSAS8-01-0.00BPC	DU-2	N	5/19/2010	1	0	0	7910000	7910000
SSAS8-02-0.00BPC	DU-2	N	8/18/2010	0	0	1	8870000	8870000
SSAS8-03-0.33BPC	DU-2	N	8/18/2010	0	0	0	8870000	8870000
SSAS8-04-0.00BPC	DU-2	N	8/16/2010	0	3	0	8880000	8850000
SSAS8-04-0.33BPC	DU-2	N	8/16/2010	0.33	0	0	8860000	8860000
TSB-GJ-03-0	DU-2	N	11/19/2007	0	0	0	2987831	2987831
TSB-GR-02-0	DU-2	N	11/19/2007	0	0	1	2968961	2968961

#### Notes:

bgs = below ground surface

ft = feet

s/g  $PM_{10}$  = structure (fiber) per gram of particulate matter (< 10 micrometer)

s/sample = structure (fiber) per sample

BHRA = Baseline Health Risk Assessment

DU = Decision unit

DVSR = Data Validation Summary Report

FD = Field Duplicate

N = Normal Sample

inc inc

indicates value reported as "< X" in the DVSR or analytical report. Datum will not be included in the quantitative risk evaluation. indicates that analytical sensitivity is different between long amphibole and long chrysotile as reported in the DVSRs or analytical reports. The higher analytical sensitivity will be conservatively used in the BHRA risk calculations.

									Nondetects		
Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
Chlorine	Chlorate	38,900	mg/kg	254	198	78	0.044	5.8	0	0	
Oxyanions	Perchlorate	908	mg/kg	337	321	95	0.035	0.43	0	0	
Metals	Antimony	519	mg/kg	257	77	30	0.50	2.3	0	0	
	Boron	259,000	mg/kg	260	236	91	1.4	13	0	0	Use health-based BCL instead of non-hea
	Cadmium	1,260	mg/kg	260	139	53	0.0050	0.51	0	0	
	Chromium VI	7.0	mg/kg	215	48	22	0.11	0.49	0	0	
	Mercury	389	mg/kg	262	230	88	0.0067	0.040	0	0	Mercury compounds BCL is used
	Molybdenum	6,490	mg/kg	260	194	75	0.052	2.0	0	0	
	Niobium	130	mg/kg	19	1	5.3	0.76	2.1	0	0	
	Palladium	N/A	mg/kg	25	7	28	0.048	0.060	N/A	N/A	
	Platinum	649	mg/kg	195	143	73	0.010	0.24	0	0	
	Selenium	6,490	mg/kg	260	16	6.2	0.16	4.7	0	0	
	Silver	6,490	mg/kg	260	49	19	0.20	1.5	0	0	
	Sulfur	N/A	mg/kg	25	16	64	211	430	N/A	N/A	
	Thallium	13	mg/kg	260	176	68	0.10	0.28	0	0	
	Tin	779,000	mg/kg	195	184	94	9.4	12	0	0	Use health-based BCL instead of non-hea
	Tungsten	1,040	mg/kg	213	173	81	0.10	5.6	0	0	
Other Inorganics	Ammonia	6,140	mg/kg	199	36	18	0.080	6.5	0	0	
	Bromide	441,000	mg/kg	209	24	11	0.063	28	0	0	Use health-based BCL instead of non-hea
	Chloride	113,000	mg/kg	206	196	95	2.1	85	0	0	Use health-based BCL instead of non-hea
	Chlorite	38,900	mg/kg	7	0	0	0.040	0.040	0	0	
	Cyanide (total)	179	mg/kg	132	2	1.5	0.13	1.2	0	0	Conservatively use BCL for CN-
	Fluoride	51,900	mg/kg	7	1	14	0.25	0.25	0	0	
	Nitrate	2,080,000	mg/kg	210	187	89	0.048	6.1	0	0	Use health-based BCL instead of non-hea
	Nitrate/Nitrite	130,000	mg/kg	18	17	94	1.2	1.2	0	0	Minimum BCL of nitrate and nitrite, use he
	Nitrite	130,000	mg/kg	202	40	20	0.080	22	0	0	Use health-based BCL instead of non-hea
	ortho-Phosphate	30,400,000	mg/kg	48	6	13	1.1	57	0	0	Use phosphoric acid as a surrogate, use I
	Sulfate	N/A	mg/kg	210	208	99	2.1	22	N/A	N/A	
Dioxin/Furans	2,3,7,8-TCDD TEQ*	0.0027	mg/kg	466	463	99	0.000044	0.0037	1		Site-specific action level
Other Organics	Benzenesulfonic acid	649,000	mg/kg	27	0	0	0.50	0.50	0	0	Use health-based BCL instead of non-hea
	4-Chlorobenzenesulfonic acid	117	mg/kg	27	0	0	0.50	0.50	0	0	
	o,o-Dimethyl Phosphorodithioate	130,000	mg/kg	27	0	0	2.5	2.5	0	0	Use health-based BCL instead of non-hea
	Diethylphosphorodithioate	104,000	mg/kg	27	0	0	0.50	0.50	0	0	Use health-based BCL instead of non-hea
	Phthalic acid	1,830,000	mg/kg	52	0	0	0.25	70	0	0	Use health-based BCL instead of non-hea
PAHs	Acenaphthene	118	mg/kg	474	7	1.5	0.00016	1.1	0	0	
	Acenaphthylene	118	mg/kg	474	6	1.3	0.00017	1.8	0	0	Use acenaphthene as a surrogate
	Anthracene	4.3	mg/kg	474	14	3.0	0.00072	1.8	0	1	
	BaPEq*	0.32	mg/kg	474	83	18	0.00090	2.4	15	30	
	Benzo(g,h,i)perylene	25,300	mg/kg	473	53	11	0.0011	1.7	0	0	
	Fluoranthene	33,700	mg/kg	474	68	14	0.0010	3.9	0	0	
	Fluorene	93	mg/kg	474	3	0.60	0.00047	1.9	0	0	

Screening Level Note
Ith based upper-limit
Ith based upper-limit
Ith based upper-limit
lth based upper-limit (consider chloride as non-volatile)
lth based upper-limit
ealth-based BCL instead of non-health based upper-limit
alth based upper-limit
nealth-based BCL instead of non-health based upper-limit
Ith based upper-limit
Ith based upper-limit
Ith based upper-limit
Ith based upper-limit

					No. of				Nondetects			
Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen		
PAHs	1-Methylnaphthalene	81	mg/kg	26	4	15	0.00026	8.3	0	1		
	2-Methylnaphthalene	368	mg/kg	474	10	2.1	0.00031	3.9	0	0		
	Naphthalene	18	mg/kg	536	23	4.3	0.00032	3.3	0	1		
	Phenanthrene	25	mg/kg	474	64	14	0.0011	1.8	0	0		
	Pyrene	44	mg/kg	474	90	19	0.0011	1.3	0	0		
PCBs	Aroclor-1016	33	mg/kg	48	0	0	0.034	0.37	0	0		
	Aroclor-1221	1.1	mg/kg	48	0	0	0.034	0.74	0	1		
	Aroclor-1232	1.1	mg/kg	48	0	0	0.034	0.37	0	1		
	Aroclor-1242	1.1	mg/kg	48	0	0	0.034	0.37	0	1		
	Aroclor-1248	1.1	mg/kg	48	1	2.1	0.034	0.37	0	1		
	Aroclor-1254	1.1	mg/kg	48	0	0	0.034	0.37	0	1		
	Aroclor-1260	1.1	mg/kg	66	2	3.0	0.017	0.37	0	1		
Pesticides - OCPs	Aldrin	0.21	mg/kg	281	2	0.70	0.000088	0.092	0	8		
	alpha-BHC	0.49	mg/kg	281	15	5.3	0.000096	0.092	0	2		
	beta-BHC	1.7	mg/kg	281	161	57	0.00035	0.15	0	0		
	delta-BHC	334	mg/kg	281	7	2.5	0.000083	0.092	0	0		
	gamma-BHC	2.8	mg/kg	281	3	1.1	0.000083	0.11	0	0		
	alpha-Chlordane	7.3	mg/kg	281	0	0	0.00010	0.092	0	0	Use chlordane as a surrogate	
	gamma-Chlordane	7.3	mg/kg	279	1	0.40	0.000086	0.092	0	0	Use chlordane as a surrogate	
	trans/gamma-Chlordane	7.3	mg/kg	1	0	0	0.00055	0.00055	0	0	Use chlordane as a surrogate	
	Chlordane (total)	7.3	mg/kg	247	1	0.40	0.00021	0.45	0	0		
	2,4'-DDD	15	mg/kg	7	0	0	0.00011	0.00011	0	0	Use 4,4'-DDD as a surrogate	
	4,4'-DDD	15	mg/kg	280	10	3.6	0.00016	0.18	0	0		
	2,4'-DDE	9.5	mg/kg	32	9	28	0.000089	0.015	0	0	Use 4,4'-DDE as a surrogate	
	4,4'-DDE	9.5	mg/kg	281	153	54	0.00024	0.18	0	0		
	4,4'-DDT	7.5	mg/kg	281	122	43	0.00043	0.18	0	0		
	Dieldrin	0.16	mg/kg	281	4	1.4	0.000073	0.18	2	30		
	Endosulfan I	5,500	mg/kg	281	2	0.70	0.000083	0.092	0	0	Use endosulfan as a surrogate	
	Endosulfan II	5,500	mg/kg	281	0	0	0.00015	0.18	0	0	Use endosulfan as a surrogate	
	Endosulfan sulfate	5,500	mg/kg	281	2	0.70	0.00012	0.18	0	0	Use endosulfan as a surrogate	
	Endrin	30	mg/kg	281	2	0.70	0.000083	0.18	0	0		
	Endrin aldehyde	30	mg/kg	281	0	0	0.00011	0.18	0	0	Use endrin as a surrogate	
	Endrin ketone	30	mg/kg	281	10	3.6	0.00038	0.18	0	0	Use endrin as a surrogate	
	Heptachlor	0.81	mg/kg	279	0	0	0.00021	0.092	0	2		
	Heptachlor epoxide	0.40	mg/kg	280	0	0	0.00012	0.098	0	7		
	Hexachlorobenzene	0.23	mg/kg	674	385	57	0.00028	10	16	168		
	Methoxychlor	4,580	mg/kg	281	16	5.7	0.00043	0.92	0	0		
	Toxaphene	2.3	ma/ka	281	1	0.40	0.0071	3.6	2	42		
	2.4.5-TP	7.330	ma/ka	1	0	0	0.021	0.021	0	0		
Pesticides - OPPs	Atrazine	11	ma/ka	16	0	0	0.012	0.15	0	0		
	Chlorpvrifos	916	ma/ka	57	0	0	0.0062	0.082	0	0		
	Coumaphos	N/A	mg/ka	57	0	0	0.0027	0.036	N/A	N/A		
L		1	99	-		1	1		1			

Screening Level Note

					No. of				Nondetects		
Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
Pesticides - OPPs	Dasanit	N/A	mg/kg	57	0	0	0.0078	0.10	N/A	N/A	
	Demeton (O + S)	37	mg/kg	16	0	0	0.0072	0.096	0	0	
	Demeton-O	37	mg/kg	57	0	0	0.0051	0.067	0	0	Use demeton as a surrogate
	Demeton-S	37	mg/kg	57	0	0	0.0047	0.062	0	0	Use demeton as a surrogate
	Diazinon	732	mg/kg	57	0	0	0.0070	0.093	0	0	
	Dibrom	1.3	mg/kg	57	0	0	0.022	0.29	0	1	
	Dichlorovos	8.8	mg/kg	57	0	0	0.0071	0.094	0	0	
	Dimethoate	183	mg/kg	57	3	5.3	0.0068	0.090	0	0	
	Disulfoton	52	mg/kg	57	0	0	0.0074	0.098	0	0	
	Ethoprop	N/A	mg/kg	57	0	0	0.0047	0.063	N/A	N/A	
	Ethyl p-nitrophenyl benzenethiophosphate	13	mg/kg	57	0	0	0.0035	0.047	0	0	
	Famphur	N/A	mg/kg	57	0	0	0.0031	0.041	N/A	N/A	
	Fenthion	N/A	mg/kg	57	0	0	0.0084	0.11	N/A	N/A	
	Guthion	2,750	mg/kg	57	0	0	0.0034	0.045	0	0	
	Malathion	18,300	mg/kg	57	0	0	0.0045	0.059	0	0	
	Merphos	1.0	mg/kg	57	0	0	0.0049	0.065	0	0	
	Methyl parathion	229	mg/kg	57	0	0	0.0061	0.081	0	0	
	Mevinphos	N/A	mg/kg	57	0	0	0.0044	0.059	N/A	N/A	
	Parathion	5,500	mg/kg	57	0	0	0.0051	0.067	0	0	
	Phorate	183	mg/kg	57	0	0	0.0055	0.073	0	0	
	Prothiophos	N/A	mg/kg	57	0	0	0.0038	0.050	N/A	N/A	
	Ronnel	27	mg/kg	57	0	0	0.015	0.19	0	0	
	Simazine	21	mg/kg	16	0	0	0.021	0.28	0	0	
	Stirophos	107	mg/kg	57	1	1.8	0.0042	0.055	0	0	
	Sulfotepp	458	mg/kg	57	0	0	0.0060	0.080	0	0	
	Sulprofos	N/A	mg/kg	57	0	0	0.0041	0.054	N/A	N/A	
	Thionazin	N/A	mg/kg	57	0	0	0.0054	0.071	N/A	N/A	
	o-Ethyl o-2,4,5-trichlorophenyl ethyl-phosphonothioate	N/A	mg/kg	57	0	0	0.0060	0.080	N/A	N/A	
SVOCs	Acetophenone	2,520	mg/kg	7	0	0	0.033	0.033	0	0	
	Aniline	450	mg/kg	25	0	0	0.033	4.7	0	0	
	Azobenzene	33	mg/kg	7	0	0	0.033	0.033	0	0	
	Benzenethiol	1,260	mg/kg	7	0	0	0.12	0.12	0	0	
	Benzidine	0.011	mg/kg	14	0	0	0.67	37	14	14	
	Benzoic acid	3,670,000	mg/kg	24	0	0	0.033	19	0	0	Use health-based BCL instead of non-hea
	Benzyl alcohol	91,600	mg/kg	25	0	0	0.033	8.3	0	0	
	bis(2-Chloro-1-methylethyl) ether	1,020	mg/kg	7	0	0	0.033	0.033	0	0	
	bis(2-Chloroethoxy)methane	2,750	mg/kg	25	0	0	0.033	7.4	0	0	
	bis(2-Chloroethyl) ether	1.3	mg/kg	25	0	0	0.033	3.9	1	1	
	bis(2-Ethylhexyl)phthalate	183	mg/kg	469	97	21	0.033	5.0	0	0	

Screening Level Note
Ith based upper-limit

									Nondetects		
Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
SVOCs	bis(4-Chlorophenyl) disulfide	N/A	mg/kg	7	0	0	0.20	0.20	N/A	N/A	
	bis(4-Chlorophenyl) sulfone	733	mg/kg	7	0	0	0.33	0.33	0	0	
	4-Bromophenyl-phenyl ether	N/A	mg/kg	25	0	0	0.033	4.2	N/A	N/A	
	Butylbenzylphthalate	1,350	mg/kg	469	7	1.5	0.0025	4.6	0	0	
	Carbazole	128	mg/kg	7	0	0	0.033	0.033	0	0	
	4-Chloro-3-methylphenol	91,600	mg/kg	25	0	0	0.033	3.9	0	0	
	4-Chloroaniline	18	mg/kg	25	0	0	0.033	7.4	0	1	
	2-Chloronaphthalene	175	mg/kg	25	0	0	0.033	3.7	0	0	
	2-Chlorophenol	6,490	mg/kg	25	0	0	0.033	3.9	0	0	
	4-Chlorophenyl-phenyl ether	N/A	mg/kg	25	0	0	0.033	4.7	N/A	N/A	
	4-Chlorothioanisole	N/A	mg/kg	7	0	0	0.0076	0.0076	N/A	N/A	
	4-Chlorothiophenol	N/A	mg/kg	7	0	0	0.19	0.19	N/A	N/A	
	Dibenzofuran	171	mg/kg	25	0	0	0.033	3.7	0	0	
	3,3'-Dichlorobenzidine	5.7	mg/kg	25	0	0	0.033	8.3	1	1	
	2,2'-/4,4'-Dichlorobenzil	389	mg/kg	7	0	0	0.070	0.70	0	0	Use 4,4-dichlorobenzil as a surrogate
	2,4-Dichlorophenol	3,220	mg/kg	25	0	0	0.033	3.7	0	0	
	Diethylphthalate	733,000	mg/kg	469	5	1.1	0.024	5.3	0	0	Use health-based BCL instead of non-hea
	2,4-Dimethylphenol	18,300	mg/kg	25	0	0	0.033	7.2	0	0	
	Dimethylphthalate	9,160,000	mg/kg	469	54	12	0.00093	3.7	0	0	Use health-based BCL instead of non-hea
	Di-n-butylphthalate	91,600	mg/kg	469	33	7.0	0.027	5.0	0	0	
	2,4-Dinitrophenol	1,830	mg/kg	25	0	0	0.33	18	0	0	
	2,4-Dinitrotoluene	8.3	mg/kg	25	0	0	0.033	4.5	0	1	
	2,6-Dinitrotoluene	2.4	mg/kg	25	0	0	0.033	5.3	1	1	
	Di-n-octylphthalate	9,160	mg/kg	469	2	0.40	0.0012	5.0	0	0	
	1,4-Dioxane	36	mg/kg	449	0	0	0.0052	7.1	0	1	
	Diphenyl disulfide	N/A	mg/kg	7	0	0	0.029	0.029	N/A	N/A	
	Diphenyl sulfide	N/A	mg/kg	7	0	0	0.0035	0.0035	N/A	N/A	
	Diphenyl sulfone	733	mg/kg	7	0	0	0.0067	0.0067	0	0	
	1,2-Diphenylhydrazine	3.2	mg/kg	7	0	0	0.033	0.033	0	0	
	Hexachlorobutadiene	6.1	mg/kg	256	5	2.0	0.00028	0.033	0	0	
	Hexachlorocyclopentadiene	8.2	mg/kg	25	0	0	0.13	7.4	0	1	
	Hexachloroethane	9.3	mg/kg	25	0	0	0.033	7.4	0	1	
	Hydroxymethyl phthalimide	N/A	mg/kg	7	0	0	0.043	0.043	N/A	N/A	
	Isophorone	2,700	mg/kg	25	0	0	0.033	3.7	0	0	
	2-Methylphenol	45,800	mg/kg	25	0	0	0.080	4.5	0	0	
	3&4-Methylphenol	45,800	mg/kg	25	0	0	0.067	7.4	0	0	Minimum BCL of 4-methylphenol and 3-m
	2-Nitroaniline	8,880	mg/kg	25	0	0	0.033	3.7	0	0	
	3-Nitroaniline	3,660	mg/kg	25	0	0	0.033	7.4	0	0	Use 4-nitroaniline as a surrogate (noncane
	4-Nitroaniline	128	mg/kg	25	0	0	0.13	7.4	0	0	
	Nitrobenzene	25	mg/kg	469	0	0	0.0019	3.9	0	1	
	2-Nitrophenol	7,330	mg/kg	25	0	0	0.033	7.4	0	0	Use 4-nitrophenol as a surrogate
	4-Nitrophenol	7,330	mg/kg	25	0	0	0.14	7.8	0	0	

Screening Level Note
palth based upper-limit
ealth based upper-limit
nethylphenol
ncer endpoint)

					No. of				Nondetects		
Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
SVOCs	n-Nitroso-di-n-propylamine	0.37	mg/kg	25	0	0	0.033	3.9	1	18	
	n-Nitrosodiphenylamine	524	mg/kg	25	0	0	0.033	4.5	0	0	
	Octachlorostyrene	N/A	mg/kg	467	78	17	0.0037	130	N/A	N/A	
	Pentachlorobenzene	19	mg/kg	7	0	0	0.033	0.033	0	0	
	Pentachlorophenol	4.5	mg/kg	25	0	0	0.33	19	1	1	
	Phenol	275,000	mg/kg	25	0	0	0.033	5.0	0	0	Use health-based BCL instead of non-hea
	Pyridine	1,300	mg/kg	469	0	0	0.028	14	0	0	
	1,2,4,5-Tetrachlorobenzene	8.0	mg/kg	7	0	0	0.033	0.033	0	0	
	2,4,5-Trichlorophenol	91,600	mg/kg	25	0	0	0.033	7.2	0	0	
	2,4,6-Trichlorophenol	233	mg/kg	25	0	0	0.033	4.2	0	0	
VOCs	Acetone	1,040,000	mg/kg	256	102	40	0.0017	0.041	0	0	Use health-based BCL instead of non-hea
	Acetonitrile	3,750	mg/kg	7	0	0	0.0020	0.0020	0	0	
	t-Amyl methyl ether	70,900	mg/kg	249	0	0	0.00011	0.010	0	0	Use methyl tert-butyl ether as a surrogate
	Benzene	5.8	mg/kg	256	0	0	0.00017	0.010	0	0	
	Bromobenzene	679	mg/kg	256	0	0	0.00023	0.010	0	0	
	Bromochloromethane	692	mg/kg	256	0	0	0.00015	0.010	0	0	
	Bromodichloromethane	1.4	mg/kg	256	2	0.80	0.00011	0.010	0	0	
	Bromoform	104	mg/kg	256	1	0.40	0.00012	0.010	0	0	
	Bromomethane	33	mg/kg	256	0	0	0.00025	0.013	0	0	
	2-Butanone	28,400	mg/kg	256	75	29	0.00073	0.020	0	0	
	tert-Butyl alcohol	21,300	mg/kg	249	1	0.40	0.0052	0.20	0	0	
	n-Butylbenzene	108	mg/kg	256	0	0	0.00028	0.010	0	0	
	sec-Butylbenzene	145	mg/kg	256	0	0	0.00025	0.010	0	0	
	tert-Butylbenzene	183	mg/kg	256	0	0	0.00025	0.010	0	0	
	Carbon disulfide	735	mg/kg	7	0	0	0.00055	0.00055	0	0	
	Carbon tetrachloride	3.2	mg/kg	256	1	0.40	0.00032	0.010	0	0	
	Chlorobenzene	18,300	mg/kg	256	6	2.3	0.00012	0.010	0	0	
	Chloroethane	2,110	mg/kg	256	0	0	0.00035	0.010	0	0	
	Chloroform	1.5	mg/kg	256	85	33	0.00014	0.0091	0	0	
	Chloromethane	510	mg/kg	256	0	0	0.00039	0.010	0	0	
	2-Chlorotoluene	907	mg/kg	256	0	0	0.00026	0.010	0	0	
	4-Chlorotoluene	18,300	mg/kg	256	0	0	0.00039	0.010	0	0	
	Cumene	91,600	mg/kg	256	0	0	0.00018	0.010	0	0	
	p-Cymene	647	ma/ka	256	1	0.40	0.00024	0.010	0	0	
	1.2-Dibromo-3-chloropropane	0.071	ma/ka	256	0	0	0.00030	0.010	0	5	
	Dibromochloromethane	43	ma/ka	256	0	0	0.00029	0.010	0	0	
	1 2-Dibromoethane	0.18	ma/ka	249	0	0	0.00026	0.010	0	0	
	Dibromomethane	21,000,000	ma/ka	256	0	0	0.00032	0.010	0	0	Use health-based BCL instead of non-hea
	1.2-Dichlorobenzene	376	ma/ka	256	5	20	0.00015	0.010	0	0	
	1 3-Dichlorobenzene	373	ma/ka	256	0		0.00013	0.010	0	0	
	1 4-Dichlorobenzene	475	ma/ka	256	4	16	0.00013	0.010	0	0	
1		715	iiig/kg	200		1.0	0.00011	0.010		l v	

Screening Level Note
alth haard waara limit
ann based upper-innit
ealth based upper-limit
e (noncancer endpoint)
ealth based upper-limit

Chemical Group         Analyte Mark         Society beside         No. of Spress         No. of Spress         No. of Spress         No. of Sample Mark         No. of Sample Mark         No. of Sample Mark           VCGS         Debicotifurormation         400         maple         55         0         0.0002         0.00002         0.0002         0.0002										Nondetects		
VOCa         Diethonomikanonmehnen         440         mg/g         256         1         0         0.00026         0.010         0	Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
1-10-bit locations         17         mg/kg         286         1         0.40         0.0011         0.10         0         0	VOCs	Dichlorodifluoromethane	403	mg/kg	256	0	0	0.00026	0.010	0	0	
I - 2-biolocolamine         1.00         ng/kg         286         0         0         0.00034         0.00         0         -           I - Delinocalimen         1.00         ng/kg         7         0         0         0.00044         0.00044         0.0004         0.00         0         0         0           I - Delinocalimen         1.80         ng/kg         256         1         0.40         0.0002         0.010         0         0         -           I - Delinocalimen         1.80         ng/kg         256         1         0.40         0.0002         0.010         0         0         -           I - Delinocalimption         5.0         mg/kg         256         0         0         0.00017         0.010         0         0         0.00         1.010-12.delinopropane as a surragate (           I - Delinocarpoprio         27.50         mg/kg         256         0         0         0.0002         0.010         0         0.00         1.012-1.delinopropane as a surragate (         1.61-3.Delinopropane         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8         3.8		1,1-Dichloroethane	17	mg/kg	256	1	0.40	0.00011	0.010	0	0	
1,1-Dicklosophene         1,20         mgkn         280         mgkn         7         0         0.00034         0.0018         0         0         Minum BCL of trans1-2-Dicklosophene           cis1-2-Dicklosophene         12.80         mgkn         256         1         0.40         0.00034         0.0018         0         0         -           trans1-2-Dicklosophene         18.300         mgkn         256         0         0         0.00028         0.018         0         0         -           1,3-Dicklosophane         18.300         mgkn         256         0         0         0.00028         0.010         0         0         -           2,3-Dicklosophene         17.3         mgkn         256         0         0         0.00028         0.010         0         Use 1.3-dicklosophene as a surgate (ast-3.20cklosophene as as (ast-3.20cklosophene as a surgate (ast-3.20cklosophene as (ast-3.20cklosophene as (ast-3.20cklosophene as (ast-3.20cklosophene as (ast-3.20cklosophene as (ast		1,2-Dichloroethane	2.3	mg/kg	256	0	0	0.00034	0.010	0	0	
1.2.Dickinocohomo         2.300         mg/kg         7         0         0         0.00054         0.00054         0.0         Minnum BC. et tame 1.2.Dickhoroscheme           1.4.Dickhoroscheme         18.300         mg/kg         256         1         0.40         0.00028         0.010         0         0         -           1.5.Dickhoroscheme         18.300         mg/kg         256         0         0         0.00018         0.010         0         0         -           1.5.Dickhoroscheme         18.300         mg/kg         256         0         0         0.00017         0.010         0         0.00         1.4         -2.dickhoroscheme as a surragate         1.5         1.5         0.00017         0.010         0         0.001         0.00         0.001         0.00         0.001         0.0001         0.0001         0.00         0.001         0.00         0.001         0.00         0.001         0.00         0.001         0.0001         0.0001         0.0001         0.00         0.0001         0.0002         0.0010         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001         0.0001 <td< td=""><td></td><td>1,1-Dichloroethene</td><td>1,100</td><td>mg/kg</td><td>256</td><td>4</td><td>1.6</td><td>0.00030</td><td>0.010</td><td>0</td><td>0</td><td></td></td<>		1,1-Dichloroethene	1,100	mg/kg	256	4	1.6	0.00030	0.010	0	0	
ion 1.2.Dichlorouchamo         3.300         mg/kg         256         0         0         0.00020         0.010         0.0         0		1,2-Dichloroethene	2,360	mg/kg	7	0	0	0.00054	0.00054	0	0	Minimum BCL of trans-1,2-Dichloroethene
trans-12-Dickhoreshame         16.300         mgkg         258         0         0         0.00028         0.010         0         0		cis-1,2-Dichloroethene	2,360	mg/kg	256	1	0.40	0.00028	0.010	0	0	
1.2-Dichloropropane       5.0       mg/kg       286       0       0       0.0010       0       0		trans-1,2-Dichloroethene	18,300	mg/kg	256	0	0	0.00020	0.010	0	0	
1.3-bickhoregropane         19.300         mg/kg         268         0         0         0.0017         0.010         0         0         0         0           2.2-Dichloroppone         73         mg/kg         286         0         0         0.00017         0.010         0         0.00         0.00         0.0002         0.010         0         0.00		1,2-Dichloropropane	5.0	mg/kg	256	0	0	0.00028	0.010	0	0	
22-binkhorgongene         73         mg/kg         256         0         0         0.017         0.010         0         Use 1-2-dickhorgongene as a surrogate ( 1-1)Okhorgongene           11-Dickhorgongene         26         mg/kg         256         0         0         0.0003         0.010         0         Use 1-3-dickhorgongene as a surrogate ( 1-3-Dickhorgongene         26         mg/kg         256         0         0         0.0003         0.010         0         Use 1-3-dickhorgongene as a surrogate 1-3-dickhorgongene         26         mg/kg         276         0         0         0.0003         0.010         0         0         Use 1-3-dickhorgongene as a surrogate 1-3-dickhorgongene as a surrogate 1-3-dickhorgongene         26         mg/kg         7         0         0         0.00021         N/A         N/A            2-Dimethylpentare         N/A         mg/kg         7         0         0         0.00022         N/A         N/A            2-Dimethylpentare         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            2-Dimethylpentare         N/A         mg/kg         7         0         0         0.00020         N/A         N/A         -		1,3-Dichloropropane	18,300	mg/kg	256	0	0	0.00018	0.010	0	0	
11-bickhorpopene       27.500       mg/kq       256       0       0.00027       0.010       0       0       Use 1.3-dichkorpopopene as a surrogate ( sis 1.3-Dichkorpopene         tisn-1.3-Dickhorpopene       26       mg/kq       256       0       0       0.00023       0.010       0       Use 1.3-dichkorpopene as a surrogate ( sis 1.3-Dichkorpopene         Dissproyl ether       2.80       mg/kq       249       0       0       0.00021       0.00021       0.0       0       -         2.20methylpentane       N/A       mg/kq       7       0       0       0.00021       0.00021       N/A       N/A       -         2.3-Dimethylpentane       N/A       mg/kq       7       0       0       0.00021       0.00021       N/A       N/A       -         3.3-Dimethylpentane       N/A       mg/kq       7       0       0       0.00019       0.00021       N/A       N/A       -         Ethand       1510.0000       mg/kq       228       0       0       0.00021       0.0002       N/A       N/A       -         Ethyl benzane       233       mg/kq       228       0       0       0.00021       0.00021       0.00021       0.00021       0.00021 <td></td> <td>2,2-Dichloropropane</td> <td>73</td> <td>mg/kg</td> <td>256</td> <td>0</td> <td>0</td> <td>0.00017</td> <td>0.010</td> <td>0</td> <td>0</td> <td>Use 1,2-dichloropropane as a surrogate (r</td>		2,2-Dichloropropane	73	mg/kg	256	0	0	0.00017	0.010	0	0	Use 1,2-dichloropropane as a surrogate (r
is-1.3-bichicoprogene         26         mg/kg         256         0         0         0.00033         0.010         0         Use 1.3-dichloroprogene as a surrogate           trans-1.3-Dichloropropene         2.60         mg/kg         24         0         0         0.00023         0.010         0         Use 1.3-dichloropropene as a surrogate           Dinettyl disullide         N/A         mg/kg         7         0         0         0.00023         0.0010         0         -           2.3-Dimettylpentane         N/A         mg/kg         7         0         0         0.00028         0.00021         N/A         N/A         -           2.3-Dimettylpentane         N/A         mg/kg         7         0         0         0.00022         N/A         N/A         -           3.3-Dimettylpentane         N/A         mg/kg         7         0         0         0.00020         0.00020         N/A         N/A         -           3.3-Dimettylpentane         N/A         mg/kg         28         0         0         0.00020         N/A         N/A         -           Ethyl bersize         15,00000         mg/kg         28         0         0.00021         0.00021         0.00		1,1-Dichloropropene	27,500	mg/kg	256	0	0	0.00027	0.010	0	0	Use 1,3-dichloropropene as a surrogate (r
frame-13-Dichloropropene         26         mg/kg         256         0         0         0.00020         0.010         0         User         User           Disopropyl ether         2,260         mg/kg         249         0         0         0.00021         0.0101         0         0         -           Disopropyl ether         XA         mg/kg         7         0         0         0.00028         N/A         N/A         -           2.2-Dimethylpentane         N/A         mg/kg         7         0         0         0.00028         N/A         N/A         -           2.4-Dimethylpentane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A         -           3.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A         -           Ethylotnith         f5.100.000         mg/kg         1         0         0         0.0010         0         0         Use health-based BCL instead of non-head           Ethylotnith         f5.100.000         mg/kg         248         1         0.40         0.00021         N/A         N/A         -		cis-1,3-Dichloropropene	26	mg/kg	256	0	0	0.00033	0.010	0	0	Use 1,3-dichloropropene as a surrogate
Discopoyle ther         2.20         mg/kg         2.49         0         0         0.00021         0.010         0         0         -           Dimethyl disulfido         N/A         mg/kg         7         0         0         0.00021         N/A         N/A         -           2.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00022         N/A         N/A         -           2.4-Dimethylpentane         N/A         mg/kg         7         0         0         0.00022         N/A         N/A         -           3.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00021         0.00022         N/A         N/A         -           Ethanol         15,100,000         mg/kg         7         0         0         0.00023         0.010         0         0         -		trans-1,3-Dichloropropene	26	mg/kg	256	0	0	0.00020	0.010	0	0	Use 1,3-dichloropropene as a surrogate
Dimethy disulfie         N/A         mg/kg         7         0         0         0.00021         0.00021         N/A         N/A		Diisopropyl ether	2,260	mg/kg	249	0	0	0.00023	0.010	0	0	
2.2-Dimethylpentane         N/A         mg/kg         7         0         0         0.00028         0.00028         N/A         NA		Dimethyl disulfide	N/A	mg/kg	7	0	0	0.00021	0.00021	N/A	N/A	
2.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00022         N/A         N/A		2,2-Dimethylpentane	N/A	mg/kg	7	0	0	0.00028	0.00028	N/A	N/A	
2.4-Dimethylpentane         N/A         mg/kg         7         0         0         0.00019         0.00019         N/A         N/A         -           3.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00020         0.00020         N/A         N/A         -           Ethanol         15.00.000         mg/kg         13         0         0         0.0019         0.00         0         Use health-based BCL instead of non-hee           Ethyl ten-butyl ether         70.900         mg/kg         248         0         0         0.00021         0.010         0         -           Ethyl ten-butyl ether         70.900         mg/kg         6         0         0         53         60         0         Use health-based BCL instead of non-hee           3-Ethylpentane         N/A         mg/kg         7         0         0         0.00021         N/A         N/A         -           -Heptane         220         mg/kg         7         0         0         0.00021         N/A         N/A         -           -Heptane         1.650         mg/kg         7         0         0         0.00028         0.020         0         -		2,3-Dimethylpentane	N/A	mg/kg	7	0	0	0.00022	0.00022	N/A	N/A	
3.3-Dimethylpentane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A		2,4-Dimethylpentane	N/A	mg/kg	7	0	0	0.00019	0.00019	N/A	N/A	
Ethanol         15,100,000         mg/kg         13         0         0         0.19         60         0         0         Use health-based BCL instead of non-head           Ethyl benzene         233         mg/kg         228         0         0         0.00019         0.010         0         0         -           Ethyl tert-buly lether         70.900         mg/kg         249         1         0.400         0.00023         0.010         0         0         Use methyl tert-buly lether as a surrogate           Ethylpentane         N/A         mg/kg         6         0         0         0.00021         N/A         N/A         -           Formaldehyde         80         mg/kg         3         0         0         0.21         0.22         0         0         -           1-Heptane         220         mg/kg         7         0         0         0.00016         0         0         -           2-Hexanone         1,650         mg/kg         256         0         0         0.00026         0.00         0         -           Methyl leth-buly lether         238         mg/kg         256         0         0         0.000017         0.010         0		3,3-Dimethylpentane	N/A	mg/kg	7	0	0	0.00020	0.00020	N/A	N/A	
Ethyl benzene         233         mg/kg         228         0         0         0.0019         0.010         0         0         -           Ethyl terl-bulyl ether         70,900         mg/kg         249         1         0.400         0.00023         0.010         0         0         Use methyl terl-bulyl eth-bulyl eth-bulyl eth-bulyl eth-as a surrogate           Ethylen glycol         1,830,000         mg/kg         7         0         0         0.00021         N/A         N/A         -           3-Ethylpentane         N/A         mg/kg         7         0         0         0.00021         0.00021         N/A         N/A         -           Formaldehyde         80         mg/kg         7         0         0         0.00016         0.00021         0         -         -           2-Hoxanone         1,650         mg/kg         7         0         0         0.00026         0.0020         0         -         -           Iddomethane         1,750,000         mg/kg         256         0         0         0.00026         0.00         0         -         -         -         -         -         -         -         -         -         -         -		Ethanol	15,100,000	mg/kg	13	0	0	0.19	60	0	0	Use health-based BCL instead of non-hea
Ethyl tert-butyl ether         70,900         mg/kg         249         1         0.40         0.0023         0.010         0         Use methyl tert-butyl ether as a surrogate           Ethylene glycol         1,830,000         mg/kg         6         0         0         53         60         0         0         Use methyl tert-butyl ether as a surrogate           3-Ethylpentane         N/A         mg/kg         7         0         0         0.00021         N/A         N/A            Formaldehyde         80         mg/kg         7         0         0         0.21         0.22         0         0            -Heptane         220         mg/kg         7         0         0         0.00028         0.00016         0         0            2-Hexanone         1,650         mg/kg         7         0         0         0.00028         0.0002         0         0            Idomethane         1,510         mg/kg         7         0         0         0.00026         0.0002         0         0            Methyl tert-butyl ether         238         mg/kg         256         0         0         0.00020         0.000		Ethyl benzene	233	mg/kg	228	0	0	0.00019	0.010	0	0	
Ethylene glycol         1.830,000         mg/kg         6         0         0         53         60         0         0         Use health-based BCL instead of non-heal           3-Ethylpentane         N/A         mg/kg         7         0         0         0.00021         N/A         N/A         -           Formaldehyde         80         mg/kg         3         0         0         0.21         0.22         0         0         -           n-Heptane         220         mg/kg         7         0         0         0.00016         0         0         -           2-Hexanone         1.650         mg/kg         256         0         0         0.00028         0.0002         0         0         -           Iodomethane         1.510         mg/kg         7         0         0         0.00028         0.00026         0         0         -         -           Methanol         1.750.000         mg/kg         256         0         0         0.00026         0.00020         0         -         -           Methyleth-butyl ether         2380         mg/kg         256         68         27         0.00037         0.010         0         - </td <td></td> <td>Ethyl tert-butyl ether</td> <td>70,900</td> <td>mg/kg</td> <td>249</td> <td>1</td> <td>0.40</td> <td>0.00023</td> <td>0.010</td> <td>0</td> <td>0</td> <td>Use methyl tert-butyl ether as a surrogate</td>		Ethyl tert-butyl ether	70,900	mg/kg	249	1	0.40	0.00023	0.010	0	0	Use methyl tert-butyl ether as a surrogate
3-Ethylpentane         N/A         mg/kg         7         0         0         0.00021         0.00021         N/A         N/A            Formaldehyde         80         mg/kg         3         0         0         0.21         0.22         0         0            n-Heptane         220         mg/kg         7         0         0         0.00016         0.00016         0         0            2-Hexanone         1.650         mg/kg         7         0         0         0.00028         0.020         0         0            Idomethane         1.510         mg/kg         6         0         0         0.00028         0.00026         0         0            Methyl tert-butyl ether         238         mg/kg         256         0         0         0.00023         0.010         0             Methyl-spentanone         3.360         mg/kg         256         68         27         0.00037         0.010         0             Methylene Chloride         1.550         mg/kg         7         0         0         0.00020         0.00020         N/A		Ethylene glycol	1,830,000	mg/kg	6	0	0	53	60	0	0	Use health-based BCL instead of non-hea
Formaldehyde         80         mg/kg         3         0         0         0.21         0.22         0         0		3-Ethylpentane	N/A	mg/kg	7	0	0	0.00021	0.00021	N/A	N/A	
n-Heptane         220         mg/kg         7         0         0         0.00016         0.00016         0         0            2-Hexanone         1,650         mg/kg         256         0         0         0.00026         0.00026         0         0            Iodomethane         1,510         mg/kg         7         0         0         0.00026         0.00026         0         0            Methanol         1,750,000         mg/kg         6         0         0         53         60         0         0             Methyl tert-buly lether         238         mg/kg         256         0         0         0.00017         0.010         0             4-Methyl-2-pentanone         3,360         mg/kg         256         68         27         0.00037         0.010         0             2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.0017         0.0017         0 <td< td=""><td></td><td>Formaldehyde</td><td>80</td><td>mg/kg</td><td>3</td><td>0</td><td>0</td><td>0.21</td><td>0.22</td><td>0</td><td>0</td><td></td></td<>		Formaldehyde	80	mg/kg	3	0	0	0.21	0.22	0	0	
2-Hexanone         1,650         mg/kg         256         0         0         0.00028         0.020         0         -         -           Iodomethane         1,510         mg/kg         7         0         0         0.00026         0.00026         0         0         -           Methanol         1,750,000         mg/kg         6         0         0         53         60         0         0         -         -           Methanol         1,750,000         mg/kg         256         0         0         0.00017         0.010         0         0         -         -           4-Methyl-thylether         238         mg/kg         256         0         0         0.00017         0.010         0         0         -         -           4-Methyl-thylether         1,550         mg/kg         256         68         27         0.00037         0.010         0         -		n-Heptane	220	mg/kg	7	0	0	0.00016	0.00016	0	0	
Iodomethane         1,510         mg/kg         7         0         0         0.00026         0.00026         0            Methanol         1,750,000         mg/kg         6         0         0         53         60         0         0         Use health-based BCL instead of non-head           Methyl tert-butyl ether         238         mg/kg         256         0         0         0.00017         0.010         0         0            4-Methyl-2-pentanone         3,360         mg/kg         256         0         0         0.00026         0.020         0         0            2-Methylexane         1,550         mg/kg         256         68         27         0.00037         0.010         0         0            2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.0017         0.0017         0         0             2-Nitropropane         0.066         mg/kg         7         0         0         0.00028         N/A         <		2-Hexanone	1,650	mg/kg	256	0	0	0.00028	0.020	0	0	
Methanol         1,750,000         mg/kg         6         0         0         53         60         0         0         Use health-based BCL instead of non-head           Methyl tert-butyl ether         238         mg/kg         256         0         0         0.00017         0.010         0         0            4-Methyl-2-pentanone         3,360         mg/kg         256         0         0         0.00017         0.010         0         0            4-Methyl-2-pentanone         3,360         mg/kg         256         68         27         0.00037         0.010         0         0            2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         0.0020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.00014         0.0014         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.00029         0.010         0             n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.0002		lodomethane	1,510	mg/kg	7	0	0	0.00026	0.00026	0	0	
Methyl tert-butyl ether         238         mg/kg         256         0         0         0.00017         0.010         0         0            4-Methyl-2-pentanone         3,360         mg/kg         256         0         0         0.00063         0.020         0         0            Methylene Chloride         1,550         mg/kg         256         68         27         0.00037         0.010         0         0            2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.0017         0.0101         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.017         0         0             n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00029         0.010         0         0            n-Propylbenzene         264         mg/kg         256         0         0         0.00022         0.010		Methanol	1,750,000	mg/kg	6	0	0	53	60	0	0	Use health-based BCL instead of non-hea
4-Methyl-2-pentanone         3,360         mg/kg         256         0         0         0.00063         0.020         0         0            Methylene Chloride         1,550         mg/kg         256         68         27         0.00037         0.010         0         0            2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.00014         0.0014         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.0017         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00029         0.010         0             n-Propylbenzene         264         mg/kg         256         0         0         0.00032         0.010         0         0            1,1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00012         0.010         0		Methyl tert-butyl ether	238	mg/kg	256	0	0	0.00017	0.010	0	0	
Methylene Chloride         1,550         mg/kg         256         68         27         0.00037         0.010         0         0            2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.0017         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00029         0.010         0             n-Propylbenzene         264         mg/kg         256         0         0         0.00032         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00022         0.010         0         0            1,1,2.2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0		4-Methyl-2-pentanone	3,360	mg/kg	256	0	0	0.00063	0.020	0	0	
2-Methylhexane         N/A         mg/kg         7         0         0         0.00020         N/A         N/A            3-Methylhexane         N/A         mg/kg         7         0         0         0.00014         0.00014         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.0017         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00029         0.0107         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00029         0.010         0            n-Propylbenzene         264         mg/kg         256         0         0         0.00032         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00032         0.010         0         0            1,1,2.2-Tetrachloroethane         10         mg/kg         256         0         0         0.00014         0.010         0         0         -		Methylene Chloride	1,550	mg/kg	256	68	27	0.00037	0.010	0	0	
3-Methylhexane         N/A         mg/kg         7         0         0         0.00014         0.00014         N/A         N/A            2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.0017         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00088         0.00088         N/A         N/A            n-Propylbenzene         264         mg/kg         256         0         0         0.00029         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00027         0.010         0         0            Tetrachloroethene         117         mg/kg         256         1         0.40         0.00027         0.010		2-Methylhexane	N/A	mg/kg	7	0	0	0.00020	0.00020	N/A	N/A	
2-Nitropropane         0.066         mg/kg         7         0         0         0.0017         0.0017         0         0            n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00088         0.00088         N/A         N/A            n-Propylbenzene         264         mg/kg         256         0         0         0.00029         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00022         0.010         0         0            Tetrachloroethane         3.2         mg/kg         256         0         0         0.00027         0.010         0         0            Tetrachloroethene         117         mg/kg         256         1         0.40         0.00027         0.010		3-Methylhexane	N/A	mg/kg	7	0	0	0.00014	0.00014	N/A	N/A	
n-Nonyl aldehyde         N/A         mg/kg         7         0         0         0.00088         0.00088         N/A         N/A            n-Propylbenzene         264         mg/kg         256         0         0         0.00029         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00032         0.010         0         0            1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00022         0.010         0         0            Tetrachloroethane         3.2         mg/kg         256         0         0         0.00027         0.010         0         0            Tetrachloroethane         117         mg/kg         256         1         0.40         0.00027         0.010         0         0		2-Nitropropane	0.066	mg/kg	7	0	0	0.0017	0.0017	0	0	
n-Propylbenzene         264         mg/kg         256         0         0         0.00029         0.010         0         0            Styrene         867         mg/kg         256         1         0.40         0.00032         0.010         0         0            1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00014         0.010         0         0            Tetrachloroethane         3.2         mg/kg         256         1         0.40         0.00027         0.010         0         0		n-Nonyl aldehyde	N/A	mg/kg	7	0	0	0.00088	0.00088	N/A	N/A	
Styrene         867         mg/kg         256         1         0.40         0.00032         0.010         0         0            1,1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00014         0.010         0         0            1,1,2,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00014         0.010         0         0            Tetrachloroethane         117         mg/kg         256         1         0.40         0.00027         0.010         0         0		n-Propylbenzene	264	mg/kg	256	0	0	0.00029	0.010	0	0	
1,1,2-Tetrachloroethane         10         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00022         0.010         0         0            1,1,2,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00014         0.010         0         0            Tetrachloroethene         117         mg/kg         256         1         0.40         0.00027         0.010         0         0		Styrene	867	mg/kg	256	1	0.40	0.00032	0.010	0	0	
1,1,2,2-Tetrachloroethane         3.2         mg/kg         256         0         0         0.00014         0.010         0         0            Tetrachloroethene         117         mg/kg         256         1         0.40         0.00027         0.010         0         0		1,1,1,2-Tetrachloroethane	10	mg/kg	256	0	0	0.00022	0.010	0	0	
Tetrachloroethene         117         mg/kg         256         1         0.40         0.00027         0.010         0		1,1,2,2-Tetrachloroethane	3.2	mg/kg	256	0	0	0.00014	0.010	0	0	
		Tetrachloroethene	117	mg/kg	256	1	0.40	0.00027	0.010	0	0	

Screening Level Note
ne and cis-1.2-Dichloroethene
(noncancer endpoint)
(noncancer endpoint)
ealth based upper-limit
e (noncancer endpoint)
ealth based upper-limit
ealth based upper-limit

#### Henderson, Nevada

									Nondetects		
Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	
VOCs	Toluene	817	mg/kg	256	53	21	0.00013	0.010	0	0	
	1,1,2-Trichloro-1,2,2- trifluoroethane	900	mg/kg	11	0	0	0.00054	0.0055	0	0	
	1,2,3-Trichlorobenzene	151	mg/kg	256	2	0.80	0.00038	0.010	0	0	
	1,2,4-Trichlorobenzene	125	mg/kg	256	6	2.3	0.00037	0.010	0	0	
	1,3,5-Trichlorobenzene	285	mg/kg	7	0	0	0.00068	0.00068	0	0	Use 1,2,4-trichlorobenzene as a surrogate
	1,1,1-Trichloroethane	638	mg/kg	256	1	0.40	0.00015	0.010	0	0	
	1,1,2-Trichloroethane	5.8	mg/kg	256	0	0	0.00028	0.010	0	0	
	Trichloroethene	6.9	mg/kg	256	2	0.80	0.00012	0.010	0	0	
	Trichlorofluoromethane	1,210	mg/kg	256	5	2.0	0.00032	0.010	0	0	
	1,2,3-Trichloropropane	0.12	mg/kg	256	0	0	0.00041	0.010	0	0	
	1,2,4-Trimethylbenzene	218	mg/kg	256	6	2.3	0.00022	0.010	0	0	
	1,3,5-Trimethylbenzene	182	mg/kg	256	1	0.40	0.00021	0.010	0	0	
	2,2,3-Trimethylbutane	N/A	mg/kg	7	0	0	0.00021	0.00021	N/A	N/A	
	Vinyl acetate	2,750	mg/kg	7	0	0	0.00018	0.00018	0	0	
	Vinyl chloride	2.2	mg/kg	256	1	0.40	0.00024	0.010	0	0	
	m,p-Xylene	387	mg/kg	233	7	3.0	0.00053	0.010	0	0	Minimum BCL of m-xylene and p-xylene
	o-Xylene	434	mg/kg	233	3	1.3	0.00031	0.010	0	0	
	Xylenes (total)	259	mg/kg	30	0	0	0.00086	0.013	0	0	

#### Notes:

-- = Not applicable OPP = Organophosphorus pesticide mg/kg = milligram per kilogram PAH = Polycyclic aromatic hydrocarbon BaPEq = Benzo(a)pyrene equivalent PCB = Polychlorinated biphenyl BCL = Basic Comparison Level SQL = Sample Quantitation Limit BHC = Hexachlorocyclohexane SVOC = Semivolatile organic compound DDD = Dichlorodiphenyldichloroethane TCDD = Tetrachlorodibenzo-p-dioxin DDE = Dichlorodiphenyldichloroethylene TEQ = Toxicity equivalent DDT = Dichlorodiphenyltrichloroethane TP = Trichlorophenoxy N/A = No screening level available VOC = Volatile organic compound NDEP = Neveda Department of Environmental Protection \* Methodology for equivalent calculations explained in text OCP = Organochlorine pesticide [1] Screening levels are the lowest level among the indoor worker and outdoor worker BCLs (NDEP 2017), unless noted.

#### Source:

NDEP. 2017. User's Guide and Background Technical Document for NDEP Basic Comparison Levels (BCLs) for Human Health for the BMI Complex and Common Areas. Revision 14, July.

Screening Level Note	
(noncancer endpoint)	

#### TABLE 4. Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics, and Radionuclides: Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

Henderson, Nevada

-

Chaminal			No. of	No. of Detects	% Detects	Nonde	etects	Detects						
Group	Analyte	Unit	Samples		% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Chlorine	Chlorate	mg/kg	254	198	78	0.044	5.8	0.045	20,900	3.0	204	1,610	7.9	SA106
Oxyanions	Perchlorate	mg/kg	337	321	95	0.035	0.43	0.012	2,620	8.4	100	288	2.9	RSAM5
Metals	Aluminum	mg/kg	260	260	100			3,900	12,200	9,020	8,930	1,420	0.16	SA43
	Antimony	mg/kg	257	77	30	0.50	2.3	0.11	2.4	0.32	0.70	0.70	1.0	SA114
	Arsenic	mg/kg	571	571	100			0.58	34	3.3	3.7	2.2	0.58	EE-C25-1
	Barium	mg/kg	260	260	100			65	1,780	177	190	110	0.58	SA123
	Beryllium	mg/kg	195	195	100			0.22	0.71	0.46	0.46	0.074	0.16	SA86
	Boron	mg/kg	260	236	91	1.4	13	2.5	1,510	7.3	24	135	5.6	SA62
	Cadmium	mg/kg	260	139	53	0.0050	0.51	0.040	8.9	0.13	0.26	0.77	2.9	SA103
	Calcium	mg/kg	195	195	100			9,930	62,500	26,700	27,500	9,280	0.34	RSAM2
	Chromium (total)	mg/kg	262	262	100			3.5	102	9.4	12	11	0.93	SA106
	Chromium VI	mg/kg	215	48	22	0.11	0.49	0.11	106	0.79	5.8	18	3.1	SA106
	Cobalt	mg/kg	310	310	100			3.2	284	7.6	13	29	2.3	RSAO8
	Copper	mg/kg	260	260	100			8.0	160	18	20	12	0.61	RISB-12
	Iron	mg/kg	260	260	100			7,050	24,000	15,000	14,900	2,690	0.18	RISB-12
	Lead	mg/kg	321	321	100			3.6	267	8.9	13	19	1.5	SA92
	Lithium	mg/kg	7	7	100			10	14	11	12	1.5	0.13	TSB-GR-02
	Magnesium	mg/kg	309	309	100			5,300	71,000	9,700	10,700	5,070	0.48	DS-C39B-1
	Manganese	mg/kg	410	410	100			133	29,200	390	962	2,220	2.3	CS-C44-1
	Mercury	mg/kg	262	230	88	0.0067	0.040	0.0030	1.9	0.017	0.035	0.13	3.7	M-162D
	Molybdenum	mg/kg	260	194	75	0.052	2.0	0.15	55	0.48	1.1	4.9	4.5	RISB-12
	Nickel	mg/kg	260	260	100			6.6	164	15	16	11	0.66	RSAO8
	Niobium	mg/kg	19	1	5.3	0.76	2.1	9.2	9.2	9.2	9.2			TSB-GR-02
	Palladium	mg/kg	25	7	28	0.048	0.060	0.33	0.52	0.44	0.43	0.072	0.17	TSB-GJ-03
	Phosphorus (total)	mg/kg	187	187	100			456	1,600	853	861	188	0.22	RISB-09
	Platinum	mg/kg	195	143	73	0.010	0.24	0.0050	0.16	0.011	0.015	0.017	1.1	SA64
	Potassium	mg/kg	195	195	100			1,230	6,120	2,160	2,320	649	0.28	SA141
	Selenium	mg/kg	260	16	6.2	0.16	4.7	0.70	1.5	1.0	1.0	0.24	0.24	RSAJ3
	Silicon	mg/kg	25	25	100			41	250	140	132	53	0.40	RISB-10
	Silver	mg/kg	260	49	19	0.20	1.5	0.020	7.6	0.12	0.30	1.1	3.6	SA201
	Sodium	mg/kg	195	195	100			198	11,700	756	1,160	1,290	1.1	SA106
	Strontium	mg/kg	213	213	100			73	805	186	208	98	0.47	SA15
	Sulfur	mg/kg	25	16	64	211	430	498	14,000	798	1,680	3,300	2.0	RISB-14
	Thallium	mg/kg	260	176	68	0.10	0.28	0.054	8.4	0.10	0.19	0.64	3.3	SA180
	Tin	mg/kg	195	184	94	9.4	12	0.40	12	4.2	3.8	1.6	0.43	RSAK8
	Titanium	mg/kg	195	195	100			361	1,270	751	743	179	0.24	SA166
	Tungsten	mg/kg	213	173	81	0.10	5.6	0.080	8.5	0.23	0.42	0.80	1.9	RSAK8
	Uranium (total)	mg/kg	213	213	100			0.55	3.6	1.0	1.2	0.52	0.45	SA86
	Vanadium	mg/kg	195	195	100			22	78	42	42	9.0	0.21	RSAK8

#### TABLE 4. Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics, and Radionuclides: Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

Henderson, Nevada

Chamical	Analyte		No of	No. of Detects	% Detects	Nonde	etects	Detects							
Group		Unit	Samples			Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum	
Metals	Zinc	mg/kg	260	260	100			18	300	33	37	25	0.69	RISB-37	
	Zirconium	mg/kg	25	25	100			15	31	22	22	3.9	0.18	RISB-14	
Other Inorganics	Ammonia	mg/kg	199	36	18	0.080	6.5	0.11	563	1.6	39	120	3.0	RSAM5	
	Bromide	mg/kg	209	24	11	0.063	28	0.20	83	1.1	4.7	17	3.6	SA15	
	Chloride	mg/kg	206	196	95	2.1	85	0.90	6,670	75	402	840	2.1	RSAJ2	
	Cyanide (total)	mg/kg	132	2	1.5	0.13	1.2	0.48	1.3	0.89	0.89	0.58	0.65	RSAJ2	
	Fluoride	mg/kg	7	1	14	0.25	0.25	0.52	0.52	0.52	0.52			TSB-GR-02	
	Nitrate	mg/kg	210	187	89	0.048	6.1	0.27	515	5.6	20	46	2.4	SA15	
	Nitrate/Nitrite	mg/kg	18	17	94	1.2	1.2	2.7	37	11	11	8.1	0.73	RISB-10	
	Nitrite	mg/kg	202	40	20	0.080	22	0.090	77	0.47	3.3	12	3.7	SA64	
	ortho-Phosphate	mg/kg	48	6	13	1.1	57	2.4	2,900	6.5	490	1,180	2.4	SA11	
	Sulfate	mg/kg	210	208	99	2.1	22	6.7	15,300	187	839	2,350	2.8	SA65	
Radionuclides	Radium-226	pCi/g	208	208	100			0.20	2.5	0.92	0.95	0.36	0.38	SA92	
	Radium-228	pCi/g	208	208	100			0	3.3	1.2	1.3	0.49	0.39	SA70	
	Thorium-228	pCi/g	205	205	100			0.48	3.0	1.8	1.8	0.36	0.20	SA65	
	Thorium-230	pCi/g	205	205	100			0.43	4.3	1.1	1.2	0.44	0.37	SA74	
	Thorium-232	pCi/g	205	205	100			0.54	2.5	1.6	1.6	0.33	0.21	SA189	
	Uranium-234	pCi/g	187	187	100			0.27	3.4	1.0	1.1	0.44	0.39	SA128	
	Uranium-235	pCi/g	187	187	100			-0.029	0.25	0.052	0.062	0.043	0.70	RSAK6	
	Uranium-238	pCi/g	205	205	100			0.24	3.3	0.96	1.0	0.38	0.37	SA128	

#### Notes:

-- = No value

bgs = below ground surface

ft = feet

mg/kg = milligram per kilogram

pCi/g = picocurie per gram

#### TABLE 5. Summary Statistics for Organic Compounds: Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

	Analyte		No. of Samples	No. of Detects		Nonde	etects	Detects							
Group		Unit			% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum	
Dioxin/Furans	2,3,7,8-TCDD TEQ*	mg/kg	466	463	99	0.000044	0.0037	0.00000055	0.019	0.000018	0.00030	0.0012	3.8	RISB-50	
PAHs	Acenaphthene	mg/kg	474	7	1.5	0.00016	1.1	0.00062	0.70	0.026	0.12	0.26	2.2	EE-C25-1	
	Acenaphthylene	mg/kg	474	6	1.3	0.00017	1.8	0.00066	0.22	0.0014	0.038	0.089	2.4	EE-C25-1	
	Anthracene	mg/kg	474	14	3.0	0.00072	1.8	0.00055	0.30	0.0035	0.028	0.079	2.8	EE-C25-1	
	BaPEq*	mg/kg	474	83	18	0.00090	2.4	0.0069	0.40	0.016	0.047	0.075	1.6	SSAM5-03	
	Benzo(g,h,i)perylene	mg/kg	473	53	11	0.0011	1.7	0.0014	0.38	0.014	0.036	0.062	1.7	SSAO6-04	
	Fluoranthene	mg/kg	474	68	14	0.0010	3.9	0.0017	0.35	0.010	0.038	0.063	1.7	SA180	
	Fluorene	mg/kg	474	3	0.60	0.00047	1.9	0.00052	1.1	0.0010	0.37	0.63	1.7	EE-C25-1	
	1-Methylnaphthalene	mg/kg	26	4	15	0.00026	8.3	0.00062	5.6	0.0022	1.4	2.8	2.0	EE-C25-1	
	2-Methylnaphthalene	mg/kg	474	10	2.1	0.00031	3.9	0.00077	7.9	0.0092	0.81	2.5	3.1	EE-C25-1	
	Naphthalene	mg/kg	536	23	4.3	0.00032	3.3	0.0010	3.1	0.0014	0.23	0.75	3.2	EE-C25-1	
	Phenanthrene	mg/kg	474	64	14	0.0011	1.8	0.0017	1.5	0.0055	0.057	0.22	3.9	EE-C25-1	
	Pyrene	mg/kg	474	90	19	0.0011	1.3	0.0011	1.3	0.011	0.054	0.15	2.8	EE-C25-1	
PCBs	Aroclor-1248	mg/kg	48	1	2.1	0.034	0.37	0.091	0.091	0.091	0.091			RSAS5	
	Aroclor-1260	mg/kg	66	2	3.0	0.017	0.37	0.034	0.061	0.048	0.048	0.019	0.40	RISB-12	
Pesticides - OCP	s Aldrin	mg/kg	281	2	0.70	0.000088	0.092	0.00049	0.00052	0.00051	0.00051	0.000021	0.042	SSAL2-05	
	alpha-BHC	mg/kg	281	15	5.3	0.000096	0.092	0.00024	0.012	0.00059	0.0019	0.0032	1.7	RSAQ4	
	beta-BHC	mg/kg	281	161	57	0.00035	0.15	0.00072	0.87	0.011	0.044	0.10	2.3	SA67	
	delta-BHC	mg/kg	281	7	2.5	0.000083	0.092	0.00048	0.0015	0.00059	0.00079	0.00038	0.48	SA86	
	gamma-BHC	mg/kg	281	3	1.1	0.000083	0.11	0.00083	0.0019	0.0013	0.0013	0.00054	0.40	RSAQ4	
	gamma-Chlordane	mg/kg	279	1	0.40	0.000086	0.092	0.0014	0.0014	0.0014	0.0014			RSAO7	
	Chlordane (total)	mg/kg	247	1	0.40	0.00021	0.45	0.0030	0.0030	0.0030	0.0030			SA66	
	4,4'-DDD	mg/kg	280	10	3.6	0.00016	0.18	0.0014	0.032	0.0046	0.0078	0.0093	1.2	SSAL3-04	
	2,4'-DDE	mg/kg	32	9	28	0.000089	0.015	0.0018	0.088	0.0060	0.027	0.031	1.1	RISB-52	
	4,4'-DDE	mg/kg	281	153	54	0.00024	0.18	0.00040	6.0	0.015	0.27	0.83	3.1	SSAM3-01	
	4,4'-DDT	mg/kg	281	122	43	0.00043	0.18	0.00066	2.3	0.013	0.10	0.27	2.6	SSAM2-01	
	Dieldrin	mg/kg	281	4	1.4	0.000073	0.18	0.00027	0.059	0.016	0.023	0.028	1.2	SSAM2-01	
	Endosulfan I	mg/kg	281	2	0.70	0.000083	0.092	0.00024	0.0015	0.00087	0.00087	0.00089	1.0	SSAL3-01	
	Endosulfan sulfate	mg/kg	281	2	0.70	0.00012	0.18	0.0042	0.016	0.010	0.010	0.0083	0.83	BDT-4-S-15	
	Endrin	mg/kg	281	2	0.70	0.000083	0.18	0.00070	0.0054	0.0031	0.0031	0.0033	1.1	SA180	
	Endrin ketone	mg/kg	281	10	3.6	0.00038	0.18	0.00061	0.020	0.0012	0.0035	0.0059	1.7	SA86	
	Hexachlorobenzene	mg/kg	674	385	57	0.00028	10	0.00032	4.7	0.053	0.19	0.37	2.0	SSAK3-05	
	Methoxychlor	mg/kg	281	16	5.7	0.00043	0.92	0.00050	0.38	0.0021	0.064	0.12	1.9	SSAM2-01	
	Toxaphene	mg/kg	281	1	0.40	0.0071	3.6	0.62	0.62	0.62	0.62			SSAL3-04	
Pesticides - OPP:	s Dimethoate	mg/kg	57	3	5.3	0.0068	0.090	0.011	0.013	0.012	0.012	0.0010	0.083	SA05	
	Stirophos	mg/kg	57	1	1.8	0.0042	0.055	0.041	0.041	0.041	0.041			SA166	
SVOCs	bis(2-Ethylhexyl)phthalate	mg/kg	469	97	21	0.033	5.0	0.058	61	0.097	0.79	6.2	7.8	SSAP4-01	
	Butylbenzylphthalate	mg/kg	469	7	1.5	0.0025	4.6	0.0033	0.053	0.0056	0.017	0.018	1.1	RSAL2	
	Di-n-butylphthalate	mg/kg	469	33	7.0	0.027	5.0	0.035	7.5	0.076	0.59	1.4	2.4	SSAP4-01	
	Di-n-octylphthalate	mg/kg	469	2	0.40	0.0012	5.0	0.084	0.088	0.086	0.086	0.0028	0.033	SSAO4-01	
	Diethylphthalate	mg/kg	469	5	1.1	0.024	5.3	0.042	0.35	0.062	0.11	0.13	1.2	SA86	
	Dimethylphthalate	mg/kg	469	54	12	0.00093	3.7	0.0015	0.79	0.042	0.12	0.17	1.4	BDT-1-S-10	

#### TABLE 5. Summary Statistics for Organic Compounds: Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

Henderson, Nevada

Chamiaal			No. of	No. of		Nond	etects				Detects			
Group	Analyte	Unit	No. of Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
SVOCs	Hexachlorobutadiene	mg/kg	256	5	2.0	0.00028	0.033	0.00041	0.0045	0.00095	0.0017	0.0017	1.0	SA11
	Octachlorostyrene	mg/kg	467	78	17	0.0037	130	0.0021	2.1	0.067	0.11	0.25	2.2	SSAK3-05
VOCs	Acetone	mg/kg	256	102	40	0.0017	0.041	0.0027	0.15	0.012	0.022	0.027	1.2	SA106
	Bromodichloromethane	mg/kg	256	2	0.80	0.00011	0.010	0.00040	0.00069	0.00055	0.00055	0.00021	0.38	SSAO8-10
	Bromoform	mg/kg	256	1	0.40	0.00012	0.010	0.0017	0.0017	0.0017	0.0017			SA102
	2-Butanone	mg/kg	256	75	29	0.00073	0.020	0.00065	0.027	0.0016	0.0034	0.0056	1.7	SSAO7-06
	tert-Butyl alcohol	mg/kg	249	1	0.40	0.0052	0.20	0.0076	0.0076	0.0076	0.0076			RSAM4
	Carbon tetrachloride	mg/kg	256	1	0.40	0.00032	0.010	0.00063	0.00063	0.00063	0.00063			RSAN3
	Chlorobenzene	mg/kg	256	6	2.3	0.00012	0.010	0.00064	0.0019	0.0011	0.0011	0.00044	0.39	SA15
	Chloroform	mg/kg	256	85	33	0.00014	0.0091	0.00031	0.15	0.0014	0.0064	0.022	3.4	SA11
	p-Cymene	mg/kg	256	1	0.40	0.00024	0.010	0.00055	0.00055	0.00055	0.00055			SSAN8-04
	1,2-Dichlorobenzene	mg/kg	256	5	2.0	0.00015	0.010	0.00026	0.00039	0.00038	0.00036	0.000055	0.15	RSAM4
	1,4-Dichlorobenzene	mg/kg	256	4	1.6	0.00011	0.010	0.00056	0.016	0.0012	0.0047	0.0075	1.6	SA08
	1,1-Dichloroethane	mg/kg	256	1	0.40	0.00011	0.010	0.0030	0.0030	0.0030	0.0030			SA08
	1,1-Dichloroethene	mg/kg	256	4	1.6	0.00030	0.010	0.00055	0.0012	0.00081	0.00084	0.00027	0.32	SSAN8-04
	cis-1,2-Dichloroethene	mg/kg	256	1	0.40	0.00028	0.010	0.0041	0.0041	0.0041	0.0041			RISB-57
	Ethyl tert-butyl ether	mg/kg	249	1	0.40	0.00023	0.010	0.00038	0.00038	0.00038	0.00038			RSAO2
	Methylene Chloride	mg/kg	256	68	27	0.00037	0.010	0.00034	0.0082	0.0013	0.0016	0.0012	0.75	RSAM8
	Styrene	mg/kg	256	1	0.40	0.00032	0.010	0.00028	0.00028	0.00028	0.00028			SA55
	Tetrachloroethene	mg/kg	256	1	0.40	0.00027	0.010	0.00068	0.00068	0.00068	0.00068			SA128
	Toluene	mg/kg	256	53	21	0.00013	0.010	0.00023	0.0022	0.00075	0.00084	0.00040	0.48	RSAM3
	1,2,3-Trichlorobenzene	mg/kg	256	2	0.80	0.00038	0.010	0.00081	0.0013	0.0011	0.0011	0.00035	0.33	SA11
	1,2,4-Trichlorobenzene	mg/kg	256	6	2.3	0.00037	0.010	0.00065	0.0037	0.0012	0.0017	0.0012	0.71	SA11
	1,1,1-Trichloroethane	mg/kg	256	1	0.40	0.00015	0.010	0.00095	0.00095	0.00095	0.00095			SA08
	Trichloroethene	mg/kg	256	2	0.80	0.00012	0.010	0.00042	0.0021	0.0013	0.0013	0.0012	0.94	RISB-57
	Trichlorofluoromethane	mg/kg	256	5	2.0	0.00032	0.010	0.00035	0.0017	0.0016	0.0013	0.00057	0.44	SA136
	1,2,4-Trimethylbenzene	mg/kg	256	6	2.3	0.00022	0.010	0.00042	0.0014	0.00084	0.00085	0.00033	0.39	SSAO8-11
	1,3,5-Trimethylbenzene	mg/kg	256	1	0.40	0.00021	0.010	0.00050	0.00050	0.00050	0.00050			SSAO8-11
	Vinyl chloride	mg/kg	256	1	0.40	0.00024	0.010	0.00028	0.00028	0.00028	0.00028			RSAM7
	m,p-Xylene	mg/kg	233	7	3.0	0.00053	0.010	0.00064	0.0023	0.0012	0.0013	0.00053	0.41	RISB-54
	o-Xylene	mg/kg	233	3	1.3	0.00031	0.010	0.00046	0.00074	0.00057	0.00059	0.00014	0.24	RISB-54

#### Notes:

-- = No value

bgs = below ground surface

ft = feet

- mg/kg = milligram per kilogram
- BaPEq = Benzo(a)pyrene equivalent
- BHC = Hexachlorocyclohexane
- DDD = Dichlorodiphenyldichloroethane
- DDE = Dichlorodiphenyldichloroethylene
- DDT = Dichlorodiphenyltrichloroethane

OCP = Organochlorine pesticide

OPP = Organophosphorus pesticide

PAH = Polycyclic aromatic hydrocarbon

PCB = Polychlorinated biphenyl

SVOC = Semivolatile organic compound

TCDD = Tetrachlorodibenzo-p-dioxin

- TEQ = Toxicity equivalent
- VOC = Volatile organic compound

\* Methodology for equivalent calculations explained in text

						Backgrou	nd Evalua	ation			
Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration <sup>[1]</sup>	2005 CSM SRC? <sup>[2]</sup>	Fail Statistical Testing for Background?	Table	Figure	Spatial Plot	Comme	
Chlorine Oxyanions	Chlorate	254	198	20,900	Yes	NA	NA	NA	Figure 6, G-10	Manufactured at the Site from approximately 1945-1998; ch highest post-removal concentrations are found near the Uni former manufacturing and disposal areas. Polygons with co	
	Perchlorate	337	321	2,620	Yes	NA	NA	NA	Figure 6, G-30	(perchlorate) were targeted for removal in 2010/2011; howe remain in certain areas. (See discussion in Section 5.1).	
Metals	Aluminum	260	260	12,200	Yes	No	F2	F1-1 F2-1	NA	Although historically listed as a SRC, NDEP did not identify Concentrations are consistent with background and <0.1xB	
	Antimony	257	77	2.4	Yes	LDF	F2	F1-2 F2-2	NA	Although historically listed as a SRC, NDEP did not identify Concentrations are <0.1xBCL.	
	Arsenic	571	571	34	Yes	Yes	F2	F1-3 F2-3	Figure 7, G-2	Hardesty/AMECCO (LOU4) manufactured sodium arsenite contaminant for LOU4 and LOU60 (Acid Drain system). Pos Study Area with concentrations >background. No clear spat	
	Barium	260	260	1,780	Yes	Yes	F2	F1-4 F2-4	Figure G-3	NDEP identified barium as a potential contaminant at sever Drain System (#60); former State Industries (#62); Kelley Tr Barium concentrations are greater than background, with el Industries area and at scattered locations throughout the St	
	Beryllium	195	195	0.71	Yes	No	F2	F1-5 F2-5	NA	Although historically listed as a SRC, NDEP did not identify Concentrations are consistent with background and <0.1xB	
	Boron	260	236	1,510	Yes	LDF	F2	F1-6 F2-6	Figure G-8	Kerr-McGee manufactured boron at the Site beginning in ap boron plant. The highest levels are located in the central an are <0.1xBCL.	
	Cadmium	260	139	8.9	Yes	No	F2	F1-7 F2-7	Figure G-9	Although historically listed as a SRC, NDEP did not identify Concentrations are consistent with background and <0.1xB	
	Calcium	195	195	62,500	Yes	No	F2	F1-8 F2-8	NA	Although used extensively or formed as a waste product (e. manganese production), concentrations are consistent with high (~30,000 mg/kg), possibly masking calcium releases.	
	Chromium (total)	262	262	102	Yes	Yes	F2	F1-9 F2-9	Figure G-12	NDEP identified chromium (total) as associated with LOU32 are greater than background, with the highest concentration and the groundwater barrier wall. Concentrations are <0.1xt	
	Chromium VI	215	48	106	Yes	LDF	F2	F1-10 F2-10	Figure 8, G-13	In unimpacted soils, chromium VI concentrations are typical hexavalent chromium (as sodium dichromate) was used ext perchlorate. NDEP identified hexavalent chromium as a pot concentrations of hexavalent chromium are co-located with 3 Ponds.	
	Cobalt	310	310	284	Yes	Yes	F2	F1-11 F2-11	Figure 9, G-14	NDEP identified cobalt as a potential contaminant at LOU70 greater than background; spatially, elevated concentrations by-product of manganese production.	
	Copper	260	260	160	Yes	No	F2	F1-12 F2-12	Figure G-15	Although historically listed as a SRC, NDEP did not identify Concentrations are consistent with background and <0.1xB	

ent <sup>[c]</sup>

alorate and perchlorate are frequently co-located. The it Buildings, Beta Ditch, and ponds, corresponding to incentrations >38,900 mg/kg (chlorate) and 908 mg/kg ever, soils with residual perchlorate concentrations

aluminum as a specific contaminant at an LOU. CL.

antimony as a specific contaminant at an LOU.

solution. NDEP identified arsenic as a potential st-removal, scattered locations remain throughout the tial pattern. (See discussion in Section 5.1).

al LOUs, including the Storm Sewer System (#59); Acid rucking (#63); and Nevada Precast Concrete (#65). evated concentrations detected in the former State rudy Area. But concentrations are <0.1xBCL.

beryllium as a specific contaminant at an LOU. CL.

pproximately 1994, and Tronox continues to operate a d eastern areas of the Site. Study Area concentrations

cadmium as a specific contaminant at an LOU. CL.

g., calcium is a process waste from chlorate and background. Calcium background concentrations are

2 (the groundwater remediation unit). Concentrations is in the general area of the former P-2 and P-3 ponds BCL.

Ily below detection limits (i.e., <0.5 mg/kg). Historically, tensively for production of sodium chlorate and sodium cential contaminant at over 25 LOUs. The highest chromium in the general area of the former P-2 and P-

) (US Vanadium Leasehold). Cobalt concentrations are generally co-locate with manganese, since it may be a

copper as a specific contaminant at an LOU. CL.

						Backgrou	nd Evalua	ation		
Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration <sup>[1]</sup>	2005 CSM SRC? <sup>[2]</sup>	Fail Statistical Testing for Background?	Table	Figure	Spatial Plot	Comme
Metals	Iron	260	260	24,000	Yes	No	F2	F1-13 F2-13	NA	NDEP identified iron as a potential contaminant at multiple I electrolytically treated to reduce chromium/other heavy meta site). Iron is also associated with manganese ores. Post-rem and <0.1xBCL.
	Lead	321	321	267	Yes	No	F2	F1-14 F2-14	Figure G-22	NDEP identified lead as a potential contaminant at several I Acid Drain System (LOU60); and State Industries (LOU62). background and less than the lead BCL.
	Lithium	7	7	14	No	NA	NA	NA	NA	Not historically identified as a SRC. Small sample size. RZ-, are <0.1xBCL.
	Magnesium	309	309	71,000	Yes	No	F2	F1-15 F2-15	Figure G-23	Produced at the Site from approximately 1942 to 1944. NDE associated with numerous LOUs. Post-removal soil concent 0.1 times the health-based BCL.
	Manganese	410	410	29,200	Yes	Yes	F2	F1-16 F2-16	Figure 10, G-24	Produced at the Site since 1951; ongoing production by Tro the eastern portion of the Site, from soil depths of 0-10 ft bg background.
	Mercury	262	230	1.9	Yes	No	F2	F1-17 F2-17	NA	Although historically identified as a SRC, NDEP did not ider Concentrations are consistent with background and <0.1xB
	Molybdenum	260	194	55	Yes	No	F2	F1-18 F2-18	Figure G-25	US Vanadium Corporation (LOU70) formerly leased facilities Study Area concentrations are consistent with background a
	Nickel	260	260	164	Yes	No	F2	F1-19 F2-19	Figure G-27	NDEP identified nickel as a potential contaminant within the consistent with background and <0.1xBCL.
	Niobium	19	1	9.2	No	NA	NA	NA	NA	Not historically identified as a SRC. Small sample size. RZ-, are <0.1xBCL.
	Palladium	25	7	0.52	No	NA	NA	NA	Figure G-29	Not historically identified as a SRC. Small sample size. RZ-
	Phosphorus (total)	187	187	1,600	Yes	NA	NA	NA	NA	Although historically identified as a SRC, NDEP did not ider RZ-A background data are not available. See related discussion for "phosphates."
	Platinum	195	143	0.16	Yes	No	F2	F1-20 F2-20	Figure G-32	NDEP identified platinum as a potential contaminant at the I concentrations are consistent with background and <0.1xBC
	Potassium	195	195	6,120	Yes	No	F2	F1-21 F2-21	NA	Although historically listed as a SRC, NDEP did not identify LOU. Concentrations are consistent with background.
	Selenium	260	16	1.5	Yes	LDF	F2	F1-22 F2-22	NA	NDEP identified selenium as a potential contaminant within Low detection frequency; concentrations are <0.1xBCL.
	Silicon	25	25	250	Yes	NA	NA	NA	NA	Although historically listed as a SRC, NDEP did not identify background data are not available.
	Silver	260	49	7.6	Yes	LDF	F2	F1-23 F2-23	NA	Although historically identified as a SRC, NDEP did not ider detection frequency; concentrations are <0.1xBCL.
	Sodium	195	195	11,700	Yes	Yes	F2	F1-24 F2-24	NA	NDEP identified sodium as a potential contaminant at multip
	Strontium	213	213	805	Yes	No	F2	F1-25 F2-25	NA	Although historically listed as a SRC, NDEP did not identify Concentrations are consistent with background and <0.1xB

ent <sup>[c]</sup>

LOUs. At the IWF, extracted groundwater was als and precipitate as iron oxide sludge (disposed of offnoval concentrations are consistent with background

LOUs, including the Storm Sewer System (LOU59); the Post-removal soil concentrations are consistent with

A background data are not available. Concentrations

EP identified magnesium as a potential contaminant trations are consistent with background and less than

nox. Highest concentrations are present primarily in s. Study Area concentrations are greater than

ntify mercury as a specific contaminant at an LOU. CL.

s at the Site for production of molybdenum compounds. and <0.1xBCL.

Storm Sewer System (LOU 59). Concentrations are

A background data are not available. Concentrations

A background data are not available.

ntify phosphorus as a specific contaminant at an LOU.

Platinum Drying Unit (LOU 15). Study Area

potassium as associated with contamination at an

the Acid Drain System (LOU 60).

silicon as specific contaminant at an LOU. RZ-A

ntify silver as a specific contaminant at an LOU. Low

ble LOUs. Concentrations are greater than background.

strontium as a specific contaminant at an LOU. CL.

						Backgrou	nd Evalua	ation		
Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration <sup>[1]</sup>	2005 CSM SRC? <sup>[2]</sup>	Fail Statistical Testing for Background?	Table	Figure	Spatial Plot	Comme
Metals	Sulfur	25	16	14,000	Yes	NA	NA	NA	NA	See discussion for "Other Inorganics."
	Thallium	260	176	8.4	Yes	Yes	F2	F1-26 F2-26	Figure 11, G-35	Although historically listed as a SRC, NDEP did not identify to Concentrations are greater than background, suggesting a provide the second se
	Tin	195	184	12	Yes	LDF	F2	F1-27 F2-27	NA	Although historically listed as a SRC, NDEP did not identify concentrations are <0.1xBCL.
	Titanium	195	195	1,270	Yes	No	F2	F1-28 F2-28	Figure G-37	NDEP identified titanium as a potential contaminant at the for Precast Concrete (LOU 65). Concentrations are consistent v
	Tungsten	213	173	8.5	Yes	Yes	F2	F1-29 F2-29	Figure G-39	US Vanadium (LOU70) formerly produced tungsten compour background, with elevated tungsten concentrations are foun Concentrations are <0.1xBCL.
	Uranium (total)	213	213	3.6	Yes	No	F2	F1-30 F2-30	Figure G-40	Although historically identified as a SRC, NDEP did not iden Concentrations are consistent with background and <0.1 BC
	Vanadium	195	195	78	Yes	No	F2	F1-31 F2-31	NA	Although historically identified as a SRC, NDEP did not iden Concentrations are consistent with background and <0.1 BC
	Zinc	260	260	300	Yes	No	F2	F1-32 F2-32	Figure G-43	NDEP identified zinc as a potential contaminant within the S consistent with background and concentrations are <0.1xBC
	Zirconium	25	25	31	No	NA	NA	NA	Figure 12, G-44	Not historically listed as a SRC. Small sample size. RZ-A ba
Other	Ammonia	199	36	563	Yes	NA	NA	NA	NA	
morganics	Bromide	209	24	83	Yes	NA	NA	NA	NA	
	Chloride	206	196	6,670	Yes	NA	NA	NA	NA	This group of inorganic compounds includes common indust
	Cyanide (total)	132	2	1.3	Yes	NA	NA	NA	NA	were historically identified as SRCs. These compounds are
	Fluoride	7	1	0.52	No	NA	NA	NA	NA	or cations.
	Nitrate	210	187	515	Yes	NA	NA	NA	NA	Although all of the listed inorganics occur naturally in soil, R
	Nitrate/Nitrite	18	17	37	Yes	NA	NA	NA	NA	concerns. (Many are physiological electrolytes and/or occur
	Nitrite	202	40	77	No	NA	NA	NA	NA	greater concern when detected as contaminants in groundw soil.
	ortho-Phosphate	48	6	2,900	Yes	NA	NA	NA	NA	1
	Sulfate	210	208	15,300	Yes	NA	NA	NA	NA	]

#### nt <sup>[c]</sup>

thallium as a specific contaminant at an LOU. possible historical presence at the Site.

tin as a specific contaminant at an LOU.

ormer J.B Kelley Trucking (LOU 63) and Nevada vith background and <0.1xBCL.

unds at the Site. Concentrations are greater than and in the eastern and the central areas of the Site.

tify uranium as a specific contaminant at an LOU. L.

tify vanadium as a specific contaminant at an LOU. L.

Storm Sewer System (LOU59). Soil concentrations are CL.

ckground data are not available.

trial chemicals that are used as chemical feedstocks ith the exception of fluoride and nitrate, all compounds generally highly soluble when present as free anions

Z-A background data sets are not available to conduct il, these inorganics do not present human health naturally in foods.) Generally, these inorgranics are of vater than when present at elevated concentrations in

						Backgrour	nd Evalua	ation		
Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration <sup>[1]</sup>	2005 CSM SRC? <sup>[2]</sup>	Fail Statistical Testing for Background?	Table	Figure	Spatial Plot	Comme
Radio- nuclides	Uranium-238	205	205	3.3	Yes	No	F4	F1-33 F2-33	Figure 13, G-42	
	Uranium-234	187	187	3.4	Yes	No	F4	F1-34 F2-34	NA	
	Thorium-230	205	205	4.3	Yes	No	F4	F1-35 F2-35	NA	
	Radium-226	208	208	2.5	Yes	No	F4	F1-36 F2-36	NA	Although historically listed as SRCs, radionuclides are not k operations at the Site. Although no potential source areas v
	Thorium-232	205	205	2.5	Yes	No	F4	F1-37 F2-37	Figure 13, G-36	analysis must be interpreted with caution given the issues a methods.
	Thorium-228	205	205	3.0	Yes	No	F4	F1-38 F2-38	NA	
	Radium-228	208	208	3.3	Yes	No	F4	F1-39 F2-39	NA	1
	Uranium-235	187	187	0.25	No	No	F4	F1-40 F2-40	Figure 13, G-41	

LDF = Low detection frequency (<25%) in either site or background datasets. Background comparison results may not be applicable.

#### Notes:

bgs = below ground surface ft = feet

mg/kg = milligram per kilogram

pCi/g = picocurie per gram

BCL = Basic comparison level

NA = Not applicable

NDEP = Nevada Division of Environmental Protection

SRC = Site related chemical, as identified in the Conceptual Site Model (ENSR 2005)

CSM = Conceptual site model

Listed analytes are those detected in one or more samples in the Study Area.

[1] Concentrations are in mg/kg for all groups except radionuclides; radionuclide activities are in pCi/g.

[2] From Table 5 of the ENSR (2005) Conceptual Site Model report.

[3] Based on information from: ENSR 2005; ENVIRON 2011; NDEP 2011; and Ramboll Environ 2016.

#### Sources:

ENSR. 2005. Conceptual Site Model, Kerr-McGee Facility, Henderson, Nevada. February. NDEP requested response to comments during the next monthly meeting October 22. ENVIRON. 2011. Phase I Environmental Site Assessment of Tronox LLC, Clark County, Nevada. January.

NDEP. 2011. Action Memorandum: Removal Actions, Nevada Environmental Response Trust Site, Clark County, Nevada. July.

LOU = Letter of Understanding

Ramboll Environ. 2016. Technical Memorandum, Remedial Investigation Data Evaluation, Nevada Environmental Response Trust Site, Henderson, Nevada, dated May 2.

ent <sup>[c]</sup>

known to be associated with any of the former/current were identified, soil samples were analyzed for ith background levels. The results of background associated with sample preparation and analytical

#### TABLE 7. Exploratory Data Analysis: Comments for Organic Compounds (0-10 ft bgs) Nevada Environmental Response Trust Site Henderson, Nevada

Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration (mg/kg)	2005 CSM SRC? <sup>[1]</sup>	Spatial Plot	Comment <sup>[2]</sup>
Dioxins/ Furans	2,3,7,8-TCDD TEQ*	466	463	0.019	Yes	Figure 14, G- 34	Unintentional by-product of high-temperature processes, e.g., incomplete combustion of chlorine is required). Highly persistent. High detection frequency due, in part, to se with concentrations >0.0027 mg/kg targeted for removal in 2010/2011; however, soils >0.0027 mg/kg remain. (See discussion in Section 5.1).
PAHs	Acenaphthene	474	7	0.70	Yes	NA	
	Acenaphthylene	474	6	0.22	Yes	NA	Expected to co-locate with BaPEqs.
	Anthracene	474	14	0.30	Yes	NA	
	BaPEq*	474	83	0.40	Yes	Figure 15, G- 4	Listed as SRCs, PAHs are ubiquitous environmental contaminants, formed during inc materials. The highest concentrations are located in the center of the Study Area.
	Benzo(g,h,i)perylene	473	53	0.38	Yes	Figure G-5	
	Fluoranthene	474	68	0.35	Yes	Figure G-20	Expected to co-locate with BaPEqs.
	Fluorene	474	3	1.1	Yes	NA	
	1-Methylnaphthalene	26	4	5.6	No	NA	Neglik den store kisterier II. Kiteden store det stier fremensies with the kis
	2-Methylnaphthalene	474	10	7.9	No	NA	in the area of the former diesel above-ground storage tanks.
	Naphthalene	536	23	3.1	Yes	Figure 16, G- 26	
	Phenanthrene	474	64	1.5	Yes	Figure G-31	Expected to co-locate with BaPEqs.
	Pyrene	474	90	1.3	Yes	Figure G-33	
PCBs	Aroclor-1248	48	1	0.091	Yes	NA	As reported in the Environmental Conditions Assessment (Kleinfelder 1993), in 1980 were reported at the Site, but only 12 remained as of the date of the 1993 report. The
	Aroclor-1260	66	2	0.061	Yes	NA	frequency.
Pesticides OCPs	Aldrin	281	2	0.00052	No	NA	Although OCPs were historically listed as SRCs, NDEP did not identify aldrin as a spe detection frequency (<1%). Post-removal concentrations are <0.1xBCL.
	alpha-BHC	281	15	0.012	No	Figure G-1	
	beta-BHC	281	161	0.87	No	Figure 18, G- 6	Not listed historically as a SRC. However, the former Stauffer facility (to the west) pro 1946 through 1958; the alpha, beta, and delta isomers are by-products of lindane pro
	delta-BHC	281	7	0.0015	No	NA	were generally found in the western half of the Site.
	gamma-BHC	281	3	0.0019	No	NA	
	Chlordane (total)	247	1	0.0030	No	NA	Although OCPs were historically listed as SRCs, NDEP did not identify chlordane as a
	gamma-Chlordane	279	1	0.0014	No	NA	Low detection frequency (<1%). Post-removal concentrations are <0.1xBCL.
	2,4'-DDE	32	9	0.088	Yes	Figure G-16	
	4,4'-DDD	280	10	0.032	Yes	NA	Historical information indicates that Hardesty/AMECCO (1946-1949, LOU4) listed DD concentrations of DDT and related compounds at the Site are relatively low, significar
	4,4'-DDE	281	153	6.0	Yes	Figure 17, G- 17	6,900 mg/kg; DDD,16,000 mg/kg; and DDE, 38,000 mg/kg) were reported in soil sam the west (PES Environmental, Inc. 2016). 4,4'-DDT and 4,4'-DDE are mostly co-located
	4,4'-DDT	281	122	2.3	Yes	Figure 17, G- 18	the west/central portion of the Study Area.
	Dieldrin	281	4	0.059	No	Figure 18, G- 19	Although OCPs were historically listed as SRCs, NDEP did not identify dieldrin as a s Very persistent in soils; low detection frequency (<2%).

ion and pesticide production (a source sensitive analytical methods. Polygons bils with residual concentrations
incomplete combustion of organic
nighest concentrations generally found
30 22 PCB-containing transformers he Kleinfelder report also noted that in ations are <0.1xBCL. Low detection
specific contaminant at an LOU. Low
produced gamma-BHC (lindane) from production. The highest concentrations
as a specific contaminant at an LOU.
DDT for production. While detected cantly higher concentrations (DDT, amples at the former Stauffer facility to cated; the highest concentrations are in
a specific contaminant at an LOU.

#### TABLE 7. Exploratory Data Analysis: Comments for Organic Compounds (0-10 ft bgs) Nevada Environmental Response Trust Site Henderson, Nevada

Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration (mg/kg)	2005 CSM SRC? <sup>[1]</sup>	Spatial Plot	Comment <sup>[2]</sup>
Pesticides	Endosulfan I	281	2	0.0015	No	NA	
OCPS	Endosulfan sulfate	281	2	0.016	No	NA	Although OCPs were historically listed as SRCs, NDEP did not identify these pesticides as specific contaminants at an
	Endrin	281	2	0.0054	No	NA	concentrations are <0.1xBCL.
	Endrin ketone	281	10	0.020	No	NA	
	Hexachlorobenzene	674	385	4.7	Yes	Figure 19, G- 21	Historically listed as a SRC. Formed as a by-product during the manufacture of other chemicals involving chlorine, mainly solvents and pesticides.
	Methoxychlor	281	16	0.38	No	NA	Although OCPs were historically listed as SPCs. NDEP did not identify these pasticides as a specific contaminant at an
	Toxaphene	281	1	0.62	No	Figure 18, G- 38	LOU. Low detection frequencies (<6%).
Pesticides	Dimethoate	57	3	0.013	No	NA	Although OPPs were historically listed as SRCs, NDEP did not identify these pesticides as a specific contaminant at an
OPPs	Stirophos	57	1	0.041	No	NA	LOU. Low detection frequency (<6%). Post-removal concentrations are <0.1xBCL.
SVOCs	Bis(2-ethylhexyl)phthalate	469	97	61	No	Figure 20, G- 7 Not historically listed as a SRC. Bis(2-ethylhexyl)phthalate is a common field/laboratory contaminant. 7 hot historically listed as a SRC. Bis(2-ethylhexyl)phthalate is a common field/laboratory contaminant. 61 mg/kg).	
	Butylbenzylphthalate	469	7	0.053	No	NA	
	Di-n-butylphthalate	469	33	7.5	No	NA	
Di-n-oct	Di-n-octylphthalate	469	2	0.088	No	NA	Not historically listed as SRCs. Phthalates are common field/laboratory contaminants. Low detection frequencies (0.5 to 12%) and concentrations are <0.1xBCL
	Diethylphthalate	469	5	0.35	No	NA	
	Dimethylphthalate	469	54	0.79	No	NA	
	Hexachlorobutadiene	256	5	0.0045	No	NA	Not historically listed as a SRC, although can be a byproduct of reactions involving chlorine and hydrocarbons. Low detection frequency (<2%). Post-removal concentrations are <0.1xBCL.
	Octachlorostyrene	467	78	2.1	Yes	Figure G-28	Historically listed as a SRC. By-product of many industrial chemical processes; formed during incineration and combustion processes involving chlorinated compounds. Highly persistent.
VOCs	Acetone	256	102	0.15	Yes	NA	
	Bromodichloromethane	256	2	0.00069	No	NA	
	Bromoform	256	1	0.0017	No	NA	See VOC comments below
	2-Butanone	256	75	0.027	No	NA	
	Carbon tetrachloride	256	1	0.00063	No	NA	
	Chlorobenzene	256	6	0.0019	Yes	NA	
	Chloroform	256	85	0.15	Yes	Figure G-11	Although chloroform is the most prevalent groundwater VOC, there are no written records of its use at the Site. Tronox has communicated that chloroform was used in small quantities at the facility lab. Post-removal soil concentrations are very low and not indicative of a potential source area.
	p-Cymene	256	1	0.00055	No	NA	See VOC comments below.
	1,2-Dichlorobenzene	256	5	0.00039	Yes	NA Mono- and dichlorobenzenes were produced by Hard	Mono- and dichlorobenzenes were produced by Hardesty/Amecco from 1946-1949. Soil concentrations are very low and
	1,4-Dichlorobenzene	256	4	0.016	Yes	NA	are not considered indicative of a potential source area. Low detection frequency.
	1,1-Dichloroethane	256	1	0.0030	No	No         NA           No         NA           Soc VOC comments below	
	1,1-Dichloroethene	256	4	0.0012	No		See VOC comments below
	cis-1,2-Dichloroethene	256	1	0.0041	No	NA	
	Ethyl tert-butyl ether	249	1	0.00038	No	NA	

#### TABLE 7. Exploratory Data Analysis: Comments for Organic Compounds (0-10 ft bgs) Nevada Environmental Response Trust Site

#### Henderson, Nevada

Chemical Group	Analyte	Number of Samples	Number of Detects	Maximum Detected Concentration (mg/kg)	2005 CSM SRC? <sup>[1]</sup>	Spatial Plot	Comment <sup>[2]</sup>
VOCs	Methylene chloride	256	68	0.0082	Yes	NA	
	Styrene	256	1	0.00028	No	NA	
	tert Butyl alcohol	249	1	0.0076	No	NA	
	Tetrachloroethene	256	1	0.00068	Yes	NA	
		256	53	0.0022	Yes	NA	
	1,2,3-Trichlorobenzene	256	2	0.0013	No	NA	Historically, a number of individual VOCs were listed as SRCs and NDEP identified
	1,2,4-Trichlorobenzene	256	6	0.0037	No	NA	acetone, 2-butanone, methylene chloride, and toluene. These specific VOCs were d
	1,1,1-Trichloroethane	256	1	0.00095	No	NA	low concentrations) than the other listed VOCs, but the detected levels are not indic
	Trichloroethene	256	2	0.0021	Yes	NA	Other than the common field//aberatory contaminants, detection frequencies and co
	Trichlorofluoromethane	256	5	0.0017	No	NA	are very low and are not indicative of a potential source.
	1,2,4-Trimethylbenzene	256	6	0.0014	No	NA	
	1,3,5-Trimethylbenzene	256	1	0.00050	No	NA	
	Vinyl chloride	256	1	0.00028	No	NA	
	m,p-Xylene	233	7	0.0023	Yes	NA	
	o-Xylene	233	3	0.00074	Yes	NA	<u> </u>

#### Notes:

bgs = below ground surface OCP = Organochlorine pesticides ft = feet OPP = Organophosphorus pesticides PAH = Polycyclic aromatic hydrocarbons mg/kg = milligram per kilogram PCB = Polychlorinated biphenyl BaPEq = Benzo(a)pyrene equivalent BCL = Basic comparison level SRC = site-related chemical BHC = Hexachlorocyclohexane SVOC = Semivolatile organic compound DDD = Dichlorodiphenyldichloroethane TCDD = Tetrachlorodibenzo-p-dioxin DDE = Dichlorodiphenyldichloroethylene TEQ = Toxicity equivalent DDT = Dichlorodiphenyltrichloroethane VOC = Volatile organic compound LOU = Letter of Understanding \* Methodology for equivalent calculations explained in text NA = not applicable

Listed analytes are those detected in one or more samples in the Study Area.

[1] From Table 5 of the ENSR (2005) Conceptual Site Model report.

[2] Based on information from: ENSR 2005; ENVIRON 2011; NDEP 2011; and Ramboll Environ 2016.

#### Sources:

ENSR. 2005. Conceptual Site Model, Kerr-McGee Facility, Henderson, Nevada. February. NDEP requested response to comments during the next monthly meeting October 22.

ENVIRON. 2011. Phase I Environmental Site Assessment of Tronox LLC, Clark County, Nevada. January.

Kleinfelder. 1993. Environmental Conditions Assessment, Kerr-McGee Chemical Corporation, Henderson, Nevada Facility. April.

NDEP. 2011. Action Memorandum: Removal Actions, Nevada Environmental Response Trust Site, Clark County, Nevada. July.

PES Environmental, Inc., 2016. Conceptual Site Model (revised), Vadose Zone, BHC Cake Pile 3 & Former BHC Haul Route (LOU No. 12), Former Stauffer Chemical Company Facility, Olin Chlor Alkali Products, Henderson, Nevada, February 12. Ramboll Environ. 2016. Technical Memorandum, Remedial Investigation Data Evaluation, Nevada Environmental Response Trust Site, Henderson, Nevada, dated May 2.



						Det	tects	· ·		No. of Samples	Concentration/
Chemical Group	Analyte	Unit	No. of Samples	No. of Detects	% Detects	Maximum	Location of Maximum	Screening Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
Chlorine	Chlorate	mg/kg	254	198	78	20,900	SA106	38,900		2	Fail
Oxyanions	Perchlorate	mg/kg	337	321	95	2,620	RSAM5	908		61	Fail
Metals	Aluminum	mg/kg	260	260	100	1,220	SA43	1,240,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Antimony	mg/kg	257	77	30	2.4	SA114	519		0	Pass
	Arsenic	mg/kg	571	571	100	34	EE-C25-1	7.2	Maximum BRC/TIMET background		Fail
	Barium	mg/kg	260	260	100	1,780	SA123	238,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Beryllium	mg/kg	195	195	100	0.71	SA86	2,540		0	Pass
	Boron	mg/kg	260	236	91	1,510	SA62	259,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Cadmium	mg/kg	260	139	53	8.9	SA103	1,260		0	Pass
	Calcium	mg/kg	195	195	100	62,500	RSAM2	N/A		N/A	N/A
	Chromium (total)	mg/kg	262	262	100	102	SA106	1,950,000	Use chromium III as a surrogate, use health-based BCL instead of non-health based upper-limit	0	Pass
	Chromium VI	mg/kg	215	48	22	106	SA106	7.0		25	Fail
	Cobalt	mg/kg	310	310	100	284	RSAO8	385		9	Fail
	Copper	mg/kg	260	260	100	160	RISB-12	36,700		0	Pass
	Iron	mg/kg	260	260	100	24,000	RISB-12	908,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Lead	mg/kg	321	321	100	267	SA92	800			Pass
	Lithium	mg/kg	7	7	100	14	TSB-GR-02	2,600		0	Pass
	Magnesium	mg/kg	309	309	100	71,000	DS-C39B-1	5,200,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Manganese	mg/kg	410	410	100	29,200	CS-C44-1	28,100		23	Fail
	Mercury	mg/kg	262	230	88	1.9	M-162D	389	Mercury compounds BCL is used	0	Pass
	Molybdenum	mg/kg	260	194	75	55	RISB-12	6,490		0	Pass
	Nickel	mg/kg	260	260	100	164	RSAO8	24,700		0	Pass
	Niobium	mg/kg	19	1	5.3	9.2	TSB-GR-02	130		0	Pass
	Palladium	mg/kg	25	7	28	0.52	TSB-GJ-03	N/A		N/A	N/A
	Phosphorus (total)	mg/kg	187	187	100	1,600	RISB-09	9,630,000	Use phosphoric acid as a surrogate, use health-based BCL instead of non- health based upper-limit, adjust BCL based on molecular weight	0	Pass
	Platinum	mg/kg	195	143	73	0.16	SA64	649		0	Pass
	Potassium	mg/kg	195	195	100	6,120	SA141	N/A		N/A	N/A
	Selenium	mg/kg	260	16	6.2	1.5	RSAJ3	6,490		0	Pass
	Silicon	mg/kg	25	25	100	250	RISB-10	N/A		N/A	N/A
	Silver	mg/kg	260	49	19	7.6	SA201	6,490		0	Pass
	Sodium	mg/kg	195	195	100	11,700	SA106	N/A		N/A	N/A
	Strontium	mg/kg	213	213	100	805	SA15	779,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Sulfur	mg/kg	25	16	64	14,000	RISB-14	N/A		N/A	N/A
	Thallium	mg/kg	260	176	68	8.4	SA180	13		1	Fail
	Tin	mg/kg	195	184	94	12	RSAK8	779,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Titanium	mg/kg	195	195	100	1,270	SA166	5,190,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Tungsten	mg/kg	213	173	81	8.5	RSAK8	1,040		0	Pass
	Uranium (total)	mg/kg	213	213	100	3.6	SA86	3,830		0	Pass

						De	tects	0		No. of Samples	Concentration/
Chemical Group	Analyte	Unit	No. of Samples	No. of Detects	% Detects	Maximum	Location of Maximum	Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
Metals	Vanadium	mg/kg	195	195	100	78	RSAK8	6,420		0	Pass
	Zinc	mg/kg	260	260	100	300	RISB-37	389,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Zirconium	mg/kg	25	25	100	31	RISB-14	104		25	Fail
Other Inorganics	Ammonia	mg/kg	199	36	18	563	RSAM5	6,140		0	Pass
	Bromide	mg/kg	209	24	11	83	SA15	441,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Chloride	mg/kg	206	196	95	6,670	RSAJ2	113000	Use health-based BCL instead of non-health based upper-limit (consider chloride as non-volatile)	0	Pass
	Cyanide (total)	mg/kg	132	2	1.5	1.3	RSAJ2	179	Conservatively use BCL for CN-	0	Pass
	Fluoride	mg/kg	7	1	14	0.52	TSB-GR-02	51,900		0	Pass
	Nitrate	mg/kg	210	187	89	515	SA15	2,080,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Nitrate/Nitrite	mg/kg	18	17	94	37	RISB-10	130,000	Minimum BCL of nitrate and nitrite, use health-based BCL instead of non- health based upper-limit	0	Pass
	Nitrite	mg/kg	202	40	20	77	SA64	130,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	ortho-Phosphate	mg/kg	48	6	13	2,900	SA11	30,400,000	Use phosphoric acid as a surrogate, use health-based BCL instead of non- health based upper-limit	0	Pass
	Sulfate	mg/kg	210	208	99	15,300	SA65	N/A		N/A	N/A
Radionuclides	Radium-226	pCi/g	208	208	100	2.5	SA92	0.023		208	Fail
	Radium-228	pCi/g	208	208	100	3.3	SA70	0.041		207	Fail
	Thorium-228	pCi/g	205	205	100	3.0	SA65	0.025		205	Fail
	Thorium-230	pCi/g	205	205	100	4.3	SA74	8.4		187	Fail
	Thorium-232	pCi/g	205	205	100	2.5	SA189	7.4		202	Fail
	Uranium-234	pCi/g	187	187	100	3.4	SA128	11		69	Fail
	Uranium-235	pCi/g	187	187	100	0.25	RSAK6	0.35		149	Fail
	Uranium-238	pCi/g	205	205	100	3.3	SA128	1.4		205	Fail
Dioxin/Furans	2,3,7,8-TCDD TEQ*	mg/kg	466	463	99	0.019	RISB-50	0.0027	Site-specific action level		Fail
PAHs	Acenaphthene	mg/kg	474	7	1.5	0.70	EE-C25-1	118		0	Pass
	Acenaphthylene	mg/kg	474	6	1.3	0.22	EE-C25-1	118	Use acenaphthene as a surrogate	0	Pass
	Anthracene	mg/kg	474	14	3.0	0.30	EE-C25-1	4.3		0	Pass
	BaPEq*	mg/kg	474	83	18	0.40	SSAM5-03	0.32		25	Fail
	Benzo(g,h,i)perylene	mg/kg	473	53	11	0.38	SSAO6-04	25,300		0	Pass
	Fluoranthene	mg/kg	474	68	14	0.35	SA180	33,700		0	Pass
	Fluorene	mg/kg	474	3	0.60	1.1	EE-C25-1	93		0	Pass
	1-Methylnaphthalene	mg/kg	26	4	15	5.6	EE-C25-1	81		0	Pass
	2-Methylnaphthalene	mg/kg	474	10	2.1	7.9	EE-C25-1	368		0	Pass
	Naphthalene	mg/kg	536	23	4.3	3.1	EE-C25-1	18		2	Fail
	Phenanthrene	mg/kg	474	64	14	1.5	EE-C25-1	25		0	Pass
	Pyrene	mg/kg	474	90	19	1.3	EE-C25-1	44		0	Pass
PCBs	Aroclor-1248	mg/kg	48	1	2.1	0.091	RSAS5	1.1		0	Pass
	Aroclor-1260	mg/kg	66	2	3.0	0.061	RISB-12	1.1		0	Pass
Pesticides - OCPs	Aldrin	mg/kg	281	2	0.70	0.00052	SSAL2-05	0.21		0	Pass
	alpha-BHC	mg/kg	281	15	5.3	0.012	RSAQ4	0.49		0	Pass

						De	tects	0		No. of Samples	Concentration/
Group	Analyte	Unit	No. of Samples	No. of Detects	% Detects	Maximum	Location of Maximum	Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
Pesticides - OCPs	beta-BHC	mg/kg	281	161	57	0.87	SA67	1.7	-	11	Fail
	delta-BHC	mg/kg	281	7	2.5	0.0015	SA86	334		0	Pass
	gamma-BHC	mg/kg	281	3	1.1	0.0019	RSAQ4	2.8	-	0	Pass
	gamma-Chlordane	mg/kg	279	1	0.40	0.0014	RSAO7	7.3	Use chlordane as a surrogate	0	Pass
	Chlordane (total)	mg/kg	247	1	0.40	0.0030	SA66	7.3	-	0	Pass
	4,4'-DDD	mg/kg	280	10	3.6	0.032	SSAL3-04	15		0	Pass
	2,4'-DDE	mg/kg	32	9	28	0.088	RISB-52	9.5	Use 4,4'-DDE as a surrogate	0	Pass
	4,4'-DDE	mg/kg	281	153	54	6.0	SSAM3-01	9.5	-	10	Fail
	4,4'-DDT	mg/kg	281	122	43	2.3	SSAM2-01	7.5		6	Fail
	Dieldrin	mg/kg	281	4	1.4	0.059	SSAM2-01	0.16	-	2	Fail
	Endosulfan I	mg/kg	281	2	0.70	0.0015	SSAL3-01	5,500	Use endosulfan as a surrogate	0	Pass
	Endosulfan sulfate	mg/kg	281	2	0.70	0.016	BDT-4-S-15	5,500	Use endosulfan as a surrogate	0	Pass
	Endrin	mg/kg	281	2	0.70	0.0054	SA180	30		0	Pass
	Endrin ketone	mg/kg	281	10	3.6	0.020	SA86	30	Use endrin as a surrogate	0	Pass
	Hexachlorobenzene	mg/kg	674	385	57	4.7	SSAK3-05	0.23		256	Fail
	Methoxychlor	mg/kg	281	16	5.7	0.38	SSAM2-01	4,580		0	Pass
	Toxaphene	mg/kg	281	1	0.40	0.62	SSAL3-04	2.3		1	Fail
Pesticides - OPPs	Dimethoate	mg/kg	57	3	5.3	0.013	SA05	183		0	Pass
	Stirophos	mg/kg	57	1	1.8	0.041	SA166	107		0	Pass
SVOCs	bis(2-Ethylhexyl)phthalate	mg/kg	469	97	21	61	SSAP4-01	183	-	1	Fail
	Butylbenzylphthalate	mg/kg	469	7	1.5	0.053	RSAL2	1,350	-	0	Pass
	Diethylphthalate	mg/kg	469	5	1.1	0.35	SA86	733,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Dimethylphthalate	mg/kg	469	54	12	0.079	BDT-1-S-10	9,160,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Di-n-butylphthalate	mg/kg	469	33	7.0	7.5	SSAP4-01	91,600		0	Pass
	Di-n-octylphthalate	mg/kg	469	2	0.40	0.088	SSAO4-01	9,160		0	Pass
	Hexachlorobutadiene	mg/kg	256	5	2.0	0.0045	SA11	6.1		0	Pass
	Octachlorostyrene	mg/kg	467	78	17	2.1	SSAK3-05	N/A		N/A	N/A
VOCs	Acetone	mg/kg	256	102	40	0.15	SA106	1,040,000	Use health-based BCL instead of non-health based upper-limit	0	Pass
	Bromodichloromethane	mg/kg	256	2	0.80	0.00069	SSAO8-10	1.4		0	Pass
	Bromoform	mg/kg	256	1	0.40	0.0017	SA102	104		0	Pass
	2-Butanone	mg/kg	256	75	29	0.027	SSAO7-06	28,400		0	Pass
	tert-Butyl alcohol	mg/kg	249	1	0.40	0.0076	RSAM4	21,300		0	Pass
	Carbon tetrachloride	mg/kg	256	1	0.40	0.00063	RSAN3	3.2		0	Pass
	Chlorobenzene	mg/kg	256	6	2.3	0.0019	SA15	18,300		0	Pass
	Chloroform	mg/kg	256	85	33	0.15	SA11	1.5		0	Pass
	p-Cymene	mg/kg	256	1	0.40	0.00055	SSAN8-04	647		0	Pass
	1,2-Dichlorobenzene	mg/kg	256	5	2.0	0.00039	RSAM4	376		0	Pass
	1,4-Dichlorobenzene	mg/kg	256	4	1.6	0.016	SA08	475		0	Pass
	1,1-Dichloroethane	mg/kg	256	1	0.40	0.0030	SA08	17		0	Pass

Ohamiaal			No. of			Detects		Sorooning		No. of Samples	Concentration/
Group	Analyte	Unit	NO. Of Samples	No. of Detects	% Detects	Maximum	Location of Maximum	Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
VOCs	1,1-Dichloroethene	mg/kg	256	4	1.6	0.0012	SSAN8-04	1,100		0	Pass
	cis-1,2-Dichloroethene	mg/kg	256	1	0.40	0.0041	RISB-57	2,360		0	Pass
	Ethyl tert-butyl ether	mg/kg	249	1	0.40	0.00038	RSAO2	70,900	Use methyl tert-butyl ether as a surrogate (noncancer endpoint)	0	Pass
	Methylene Chloride	mg/kg	256	68	27	0.082	RSAM8	1,550		0	Pass
	Styrene	mg/kg	256	1	0.40	0.00028	SA55	867		0	Pass
	Tetrachloroethene	mg/kg	256	1	0.40	0.00068	SA128	117		0	Pass
	Toluene	mg/kg	256	53	21	0.0022	RSAM3	817		0	Pass
	1,2,3-Trichlorobenzene	mg/kg	256	2	0.80	0.0013	SA11	151		0	Pass
	1,2,4-Trichlorobenzene	mg/kg	256	6	2.3	0.0037	SA11	125		0	Pass
	1,1,1-Trichloroethane	mg/kg	256	1	0.40	0.00095	SA08	638		0	Pass
	Trichloroethene	mg/kg	256	2	0.80	0.0021	RISB-57	6.9		0	Pass
	Trichlorofluoromethane	mg/kg	256	5	2.0	0.0017	SA136	1,210		0	Pass
	1,2,4-Trimethylbenzene	mg/kg	256	6	2.3	0.0014	SSAO8-11	218		0	Pass
	1,3,5-Trimethylbenzene	mg/kg	256	1	0.40	0.00050	SSAO8-11	182		0	Pass
	Vinyl chloride	mg/kg	256	1	0.40	0.00028	RSAM7	2.2		0	Pass
	m,p-Xylene	mg/kg	233	7	3.0	0.0023	RISB-54	387	Minimum BCL of m-xylene and p-xylene	0	Pass
	o-Xylene	mg/kg	233	3	1.3	0.00074	RISB-54	434		0	Pass

#### Notes:

-- = Not applicable

mg/kg = milligram per kilogram

pCi/g = picocurie per gram

BaPEq = Benzo(a)pyrene equivalent

BCL = Basic Comparison Level

BHC = Hexachlorocyclohexane

COPC = Chemical of Potential Concern

DDD = Dichlorodiphenyldichloroethane

DDE = Dichlorodiphenyldichloroethylene

DDT = Dichlorodiphenyltrichloroethane

N/A = BCL (other screening value) not available for screen

NDEP = Neveda Department of Environmental Protection

OCP = Organochlorine pesticide

OPP = Organophosphorus pesticide

PAH = Polycyclic aromatic hydrocarbon

PCB = Polychlorinated biphenyl

SVOC = Semivolatile organic compound

TCDD = Tetrachlorodibenzo-p-dioxin

TEQ = Toxicity equivalent

VOC = Volatile organic compound

\* Methodology for equivalent calculations explained in text

[1] Screening levels are the lowest level among the indoor worker and outdoor worker BCLs (NDEP 2017), unless noted.

indicates analyte is carried forward to COPC identification Step 2. For arsenic, lead, and 2,3,7,8-TCDD TEQ, the maximum detected concentration is compared directly to the screening level. For all other analytes, the maximum detected concentration is compared to 0.1 x screening level. If the maximum detected concentration is greater than or equal to the 0.1 x screening level, the analyte "fails" and is carried forward to Step 2. If less than the 0.1 x screening level, the analyte "fails" and is carried forward to Step 2. If less than the 0.1 x screening level, the analyte "passes" and is eliminated as a COPC. By default, analytes for which screening levels are not available are retained for Step 2 (metals) and Step 3 (organics).

#### Source:

NDEP. 2017. User's Guide and Background Technical Document for NDEP Basic Comparison Levels (BCLs) for Human Health for the BMI Complex and Common Areas. Revision 14, July.

# TABLE 9. Results of the Background Evaluation for Metals CarriedForward from the Concentration/Toxicity ScreenNevada Environmental Response Trust SiteHenderson, Nevada

Chemical Name	Study Area Concentrations Greater than Background Levels? <sup>[1]</sup>
Arsenic	Yes
Calcium	No
Chromium VI <sup>[2]</sup>	LDF
Cobalt	Yes
Manganese	Yes
Palladium	NA
Potassium	No
Silicon	NA
Sodium	Yes
Sulfur	NA
Thallium	Yes
Zirconium	NA

#### Notes:

LDF = Low detection frequency (<25%) in either Study Area or background data sets. Background comparison results may not be applicable.

NA = Background data are not available

element is present at concentrations greater than background or background data are not available.

[1] Based on background evaluation presented in Appendix F.

[2] For chromium VI, although background comparison results may not be applicable due to low detection frequency in both Study Area and RZ-A background data sets, concentrations in the Study Area are greater than background based on the box plot (Figure F1-9) and Q-Q plot (Figure F2-9).

#### TABLE 10. Results of the Background Evaluation for Radionuclides Carried from the Concentration/Toxicity Screen Nevada Environmental Response Trust Site Henderson, Nevada

Chain	Secular Equilibrium?	Radionuclide	Study Area Concentrations Greater than Background Levels? <sup>[1],[2]</sup>	Hydrofluoric Acid Digestion?		
		Uranium-238	No			
I Iranium-238	Vec	Uranium-234	No	Yes		
Oranium-236	165	Thorium-230	No	165		
		Radium-226	Study AreaConcentrationsHydrofleGreater thanAciaBackgroundDigestiLevels?PressNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYesNoYes			
		Thorium-232	No			
Thorium-232	Yes	Radium-228	No	Yes		
		Thorium-228	No			
Uranium-235	Not Evaluated	Uranium-235	No	Yes		

#### Notes:

radionuclide is present at concentrations greater than background.

[1] Based on background analysis presented in Appendix F.

[2] The validity of statistical testing for radionuclide background evaluation is confounded by sample preparation and analytical method issues in both Study Area and RZ-A background data sets. For a full discussion of these limitations, see Section 4.2 of the report.

#### TABLE 11. Preliminary COPCs Identified for Study Area Soils (0-10 ft bgs) Nevada Environmental Response Trust Site Henderson, Nevada

Chemical Group	СОРС
Chlorine Oxyanions	Chlorate
	Perchlorate
Metals	Arsenic
	Chromium VI
	Cobalt
	Manganese
	Palladium <sup>[1],[2]</sup>
	Thallium
	Zirconium <sup>[2]</sup>
Radionuclides	Thorium-232 Series <sup>[3]</sup>
	Uranium-238 Series <sup>[3]</sup>
	Uranium-235 <sup>[3]</sup>
Dioxin/Furans	2,3,7,8-TCDD TEQ*
PAHs	BaPEq*
	Naphthalene
Pesticides - OCPs	beta-BHC
	4,4'-DDE
	4,4'-DDT
	Dieldrin
	Hexachlorobenzene
	Toxaphene
SVOCs	Bis(2-Ethylhexyl)phthalate
	Octachlorostyrene <sup>[1]</sup>
Asbestos	Long amphibole fibers
	Long chrysotile fibers

#### Notes:

bgs = below ground surface	DDT = Dichlorodiphenyltrichloroethane
ft = feet	OCP = Organochlorine pesticide
BaPEq = Benzo(a)pyrene equivalent	PAH = Polycyclic aromatic hydrocarbon
BCL = Basic Comparison Level	SVOC = Semivolatile organic compound
BHC = Hexachlorocyclohexane	TCDD = Tetrachlorodibenzo-p-dioxin
COPC = Chemical of Potential Concern	TEQ = Toxicity equivalent
DDE = Dichlorodiphenyldichloroethylene	

[1] Retained as a COPC in the absence of a BCL or other screening level. This COPC will be discussed qualitatively in the uncertainties assessment.

[2] RZ-A background data are not available for this chemical, and therefore a background evaluation cannot be conducted

[3] Although statistical testing results indicate all decay products in the thorium-232 and uranium-238 decay series and uranium-235 are consistent with background, the radionuclides are retained for further evaluation for each individual decision unit.

## TABLE 12. Cancer Risks for Radionuclides in Background SoilsNevada Environmental Response Trust SiteHenderson, Nevada

		Commercial/I	RZ-A Bac	ckground	BRC/TIMET Background			
Chain	Radionuclide	ndustrial BCL (pCi/g)	95% UCL (pCi/g)	Cancer Risk	95% UCL (pCi/g)	Cancer Risk		
Uranium-238	Uranium-238	1.4	1.1	7.8E-07	1.1	8.2E-07		
	Uranium-234	11	1.2	1.0E-07	1.2	1.1E-07		
	Thorium-230	8.4	1.2	1.4E-07	1.3	1.6E-07		
	Radium-226	0.023	1.1	4.6E-05	1.2	5.1E-05		
Thorium-232	Thorium-232	7.4	1.6	2.1E-07	1.7	2.2E-07		
	Radium-228	0.041	1.4	3.5E-05	2.0	4.9E-05		
	Thorium-228	0.025	1.8	7.3E-05	1.7	6.9E-05		
Uranium-235	Uranium-235	0.35	0.065	1.9E-07	0.072	2.1E-07		
	Т	otal Cancer Risk		2E-04		2E-04		

Notes:

-- = Not applicable

pCi/g = picocurie per gram

BCL = Basic Comparison Level

UCL = Upper Confidence Limit

#### TABLE 13. Evaluation of Sample Quantitation Limits for Individual Decision Units

#### Nevada Environmental Response Trust Site

					No. of	No. of						
Decision Unit	Chemical Group	Analyte	Screening Levels <sup>[1]</sup>	Unit	No. of Samples	No. of Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	Screening Level Note
	PAHs	Anthracene	4.3	mg/kg	367	12	3.3	0.00072	1.8	0	1	
		BaPEq*	0.32	mg/kg	367	55	15	0.00090	2.4	10	18	
		1-Methylnaphthalene	81	mg/kg	26	4	15	0.00026	8.3	0	1	
		Naphthalene	18	mg/kg	421	22	5.2	0.00032	3.3	0	1	
	PCBs	Aroclor-1221	1.1	mg/kg	25	0	0	0.034	0.74	0	1	
		Aroclor-1232	1.1	mg/kg	25	0	0	0.034	0.37	0	1	
		Aroclor-1242	1.1	mg/kg	25	0	0	0.034	0.37	0	1	
		Aroclor-1248	1.1	mg/kg	25	0	0	0.034	0.37	0	1	
		Aroclor-1254	1.1	mg/kg	25	0	0	0.034	0.37	0	1	
		Aroclor-1260	1.1	mg/kg	43	1	2.3	0.017	0.37	0	1	
	Pesticides-	Aldrin	0.21	mg/kg	236	2	0.85	0.00024	0.092	0	8	
	OCPs	alpha-BHC	0.49	mg/kg	236	10	4.2	0.00021	0.092	0	2	
		Dieldrin	0.16	mg/kg	236	4	1.7	0.00020	0.18	2	26	
		Heptachlor	0.81	mg/kg	235	0	0	0.00021	0.092	0	2	
		Heptachlor epoxide	0.40	mg/kg	235	0	0	0.00041	0.098	0	7	
DU-1		Hexachlorobenzene	0.23	mg/kg	555	319	57	0.00028	10	12	135	
501		Toxaphene	2.3	mg/kg	236	1	0.42	0.015	3.6	2	38	
	Pesticides- OPPs	Dibrom	1.3	mg/kg	41	0	0	0.022	0.29	0	1	
	SVOCs	Benzidine	0.011	mg/kg	14	0	0	0.67	37	14	14	
		Bis(2-Chloroethyl) ether	1.3	mg/kg	18	0	0	0.070	3.9	1	1	
		4-Chloroaniline	18	mg/kg	18	0	0	0.13	7.4	0	1	
		3,3'-Dichlorobenzidine	5.7	mg/kg	18	0	0	0.15	8.3	1	1	
		2,4-Dinitrotoluene	8.3	mg/kg	18	0	0	0.080	4.5	0	1	
		2,6-Dinitrotoluene	2.4	mg/kg	18	0	0	0.095	5.3	1	1	
		1,4-Dioxane	36	mg/kg	342	0	0	0.0052	7.1	0	1	
		Hexachlorocyclopentadiene	8.2	mg/kg	18	0	0	0.13	7.4	0	1	
		Hexachloroethane	9.3	mg/kg	18	0	0	0.13	7.4	0	1	
		Nitrobenzene	25	mg/kg	362	0	0	0.0019	3.9	0	1	
		n-Nitroso-di-n-propylamine	0.37	mg/kg	18	0	0	0.070	3.9	1	18	
		Pentachlorophenol	4.5	mg/kg	18	0	0	0.34	19	1	1	
	VOCs	1,2-Dibromo-3-chloropropane	0.071	mg/kg	191	0	0	0.00030	0.010	0	4	
	Dioxin/Furans	2,3,7,8-TCDD TEQ*	0.0027	mg/kg	33	31	94	0.000046	0.0037	1		Site-specific action level
	PAHs	BaPEq*	0.32	mg/kg	45	18	40	0.0081	0.040	0	6	
DU-2	Pesticides-	Dieldrin	0.16	mg/kg	34	0	0	0.000073	0.035	0	3	
	OCPs	Hexachlorobenzene	0.23	mg/kg	50	18	36	0.00030	0.035	0	16	
		Toxaphene	2.3	mg/kg	34	0	0	0.0071	0.53	0	3	

#### TABLE 13. Evaluation of Sample Quantitation Limits for Individual Decision Units

Nevada Environmental Response Trust Site

Henderson, Nevada

Desision	Chomical		Screening		No. of	No. of				Nondetects			
Unit	Group	Analyte	Levels <sup>[1]</sup>	Unit	Samples	Detects	% Detects	Minimum SQL	Maximum SQL	No. of Samples Above Screen	No. of Samples Above 10% Screen	es Screening Level Note	
	PAHs	BaPEq*	0.32	mg/kg	62	10	16	0.0082	0.45	5	6		
	Pesticides-	Dieldrin	0.16	mg/kg	11	0	0	0.0013	0.020	0	1		
DU-3	OCPs	Hexachlorobenzene	0.23	mg/kg	69	48	70	0.0018	0.39	4	17		
		Toxaphene	2.3	mg/kg	11	0	0	0.035	0.59	0	1		
	VOCs	1,2-Dibromo-3-chloropropane	0.071	mg/kg	25	0	0	0.0037	0.0091	0	1		

= Not applicable	PAH = Polycyclic aromatic hydrocarbon
mg/kg = milligram per kilogram	PCB = Polychlorinated biphenyl
BaPEq = Benzo(a)pyrene equivalent	SQL = Sample Quantitation Limit
BCL = Basic Comparison Level	SVOC = Semivolatile organic compound
BHC = Hexachlorocyclohexane	TCDD = Tetrachlorodibenzo-p-dioxin
DU = Decision unit	TEQ = Toxicity equivalent
NDEP = Neveda Department of Environmental Protection	VOC = Volatile organic compound
OCP = Organochlorine pesticide	* Methodology for equivalent calculations explained in text
OPP = Organophosphorus pesticide	[1] Screening levels are the lowest level among the indoor worker and outdoor worker BCLs (NDEP 2017), unless noted.

#### Source:

NDEP. 2017. User's Guide and Background Technical Document for NDEP Basic Comparison Levels (BCLs) for Human Health for the BMI Complex and Common Areas. Revision 14, July.

Ramboll Environ

#### TABLE 14. Summary Statistics and Concentration/Toxicity Screen for Individual Decision Units

Nevada Environmental Response Trust Site

Henderson, Nevada

				No. of	Τ		Nondetects			Detects							No. of Samples	Concentration/	
Decision Unit	Chemical Group	Analyte	Unit	No. of Samples	No. of Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum	Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
	Chlorine	Chlorate	mg/kg	194	152	78	0.044	5.4	0.045	20,900	2.5	245	1,830	7.5	SA106	38,900		2	Fail
	Oxyanions	Perchlorate	mg/kg	254	239	94	0.035	0.43	0.012	1,500	8.8	67	206	3.1	SSAM6-02	908		38	Fail
	Metals	Arsenic	mg/kg	457	457	100			0.58	34	3.2	3.6	2.3	0.62	EE-C25-1	7.2	Maximum BRC/TIMET background		Fail
		Chromium VI	mg/kg	154	34	22	0.11	0.49	0.13	106	1.3	7.9	21	2.7	SA106	7.0		22	Fail
		Cobalt	mg/kg	249	249	100			4.6	284	7.8	14	32	2.3	RSAO8	385		9	Fail
		Manganese	mg/kg	347	347	100			133	29,200	407	1,070	2,400	2.2	CS-C44-1	28,100		23	Fail
		Palladium	mg/kg	18	0	0	0.048	0.060								N/A		N/A	Pass
		Thallium	mg/kg	199	133	67	0.21	0.28	0.060	8.4	0.10	0.22	0.73	3.3	SA180	13		1	Fail
		Zirconium	mg/kg	18	18	100			15	31	22	22	4.6	0.21	RISB-14	104		18	Fail
	Radionuclides	Radium-226	pCi/g	147	147	100			0.20	2.5	0.91	0.93	0.36	0.39	SA92	0.023		147	Fail
		Radium-228	pCi/g	147	147	100			0	3.3	1.2	1.2	0.50	0.41	SA70	0.041		146	Fail
		Thorium-228	pCi/g	147	147	100			0.48	2.8	1.8	1.8	0.36	0.20	SA82	0.025		147	Fail
		Thorium-230	pCi/g	147	147	100			0.43	4.3	1.1	1.2	0.38	0.32	SA74	8.4		135	Fail
		Thorium-232	pCi/g	147	147	100			0.66	2.5	1.6	1.6	0.32	0.20	SA189	7.4		145	Fail
D0-1		Uranium-234	pCi/g	129	129	100			0.27	2.8	0.98	1.1	0.37	0.35	SA74	11		42	Fail
		Uranium-235	pCi/g	129	129	100			-0.026	0.25	0.052	0.061	0.041	0.67	RSAK6	0.35		107	Fail
		Uranium-238	pCi/g	147	147	100			0.24	2.4	0.95	1.0	0.31	0.31	SA74	1.4		147	Fail
	Dioxin/Furans	2,3,7,8-TCDD TEQ*	mg/kg	395	394	99.8	0.000044	0.000044	0.00000074	0.019	0.000017	0.00033	0.0012	3.8	RISB-50	0.0027	Site-specific action level		Fail
	PAHs	BaPEq*	mg/kg	367	55	15	0.00090	2.4	0.0069	0.29	0.015	0.039	0.059	1.5	EE-C25-1	0.32		14	Fail
		Naphthalene	mg/kg	421	22	5.2	0.00032	3.3	0.0010	3.1	0.0014	0.25	0.77	3.1	EE-C25-1	18		2	Fail
	Pesticides -	beta-BHC	mg/kg	236	139	59	0.00064	0.15	0.00072	0.87	0.010	0.044	0.10	2.4	SA67	1.7		9	Fail
	OCPs	4,4'-DDE	mg/kg	236	133	56	0.00024	0.18	0.00040	6.0	0.015	0.29	0.88	3.0	SSAM3-01	9.5		10	Fail
		4,4'-DDT	mg/kg	236	104	44	0.00058	0.18	0.00066	2.3	0.013	0.11	0.29	2.7	SSAM2-01	7.5		6	Fail
		Dieldrin	mg/kg	236	4	1.7	0.00020	0.18	0.00027	0.059	0.016	0.023	0.028	1.2	SSAM2-01	0.16		2	Fail
		Hexachlorobenzene	mg/kg	555	319	57	0.00028	10	0.00032	4.7	0.048	0.19	0.40	2.1	SSAK3-05	0.23		208	Fail
		Toxaphene	mg/kg	236	1	0.42	0.015	3.6	0.62	0.62	0.62	0.62			SSAL3-04	2.3		1	Fail
	SVOCs	bis(2-Ethylhexyl)phthalate	mg/kg	362	80	22	0.044	5.0	0.058	0.82	0.097	0.15	0.14	0.94	SSAO4-01	183		0	Pass
		Octachlorostvrene	ma/ka	360	61	17	0.0037	130	0.0021	2.1	0.067	0.11	0.27	2.4	SSAK3-05	N/A		N/A	N/A
	Chlorine	Chlorate	ma/ka	36	29	81	0.22	5.8	0.072	1.510	5.8	100	354	3.4	SA156	38.900		0	Pass
	Oxyanions	Perchlorate	ma/ka	40	40	100			0.032	34	1.7	6.0	8.9	1.5	SA07	908		0	Pass
	Metals	Arsenic	ma/ka	48	48	100			1.2	6.3	2.8	2.9	0.89	0.31	SA156	7.2	Maximum BRC/TIMET background		Pass
		Chromium VI	ma/ka	36	6	17	0.11	0.43	0.11	2.4	0.39	0.83	0.95	1.1	SA05	7.0		2	Fail
		Cobalt	ma/ka	36	36	100			5.4	8.8	7.1	7.1	0.95	0.13	SA05	385		0	Pass
		Manganese	ma/ka	36	36	100			249	1 290	368	395	168	0.43	SA07	28 100		0	Pass
		Palladium	ma/ka	7	7	100			0.33	0.52	0.44	0.43	0.072	0.17	TSB-G-I-03	N/A		N/A	N/A
		Thallium	mg/kg	36	22	61	0.10	0.26	0.074	0.02	0.10	0.40	0.072	0.23	SA103	13		0	Pass
		Zirconium	mg/kg	7	7	100			21	25	22	22	1.5	0.067	TSB-GB-02	104		7	Fail
	Radionuclides	Radium-226	nCi/a	36	36	100			0.46	1.3	0.89	0.86	0.23	0.27	SA06	0.023		36	Fail
		Radium-228	pCi/g	36	36	100			0.40	2.4	1.5	1.5	0.20	0.33	SA31	0.020		36	Fail
		Thorium-228	pCi/g	36	36	100			1.2	2.4	1.0	1.5	0.43	0.00	TSB-G.I-04	0.025		36	Fail
DU-2		Thorium-230	pCi/g	36	36	100			0.74	2.2	1.0	1.0	0.20	0.14	TSB-G I-04	8.4		32	Fail
		Thorium-232	pCi/g	36	36	100			1 1	2.0	1.1	1.1	0.22	0.20	SA03	7.4		36	Fail
		Uranium-234	pCi/g	36	36	100			0.77	1.0	1.4	1.5	0.32	0.24	SA122	11		14	Fail
		Uranium-235	pCi/g	36	36	100			-0.029	0.21	0.047	0.056	0.27	0.24	SA03	0.35		26	Fail
		Uranium-238	pCi/g	36	36	100			0.025	1.7	0.047	1.0	0.040	0.00	SA03	1 /		36	Fail
	Dioxin/Eurans		mg/kg	33	21	0/	0.000046	0.0037	0.00000055	0.0016	0.90	0.00012	0.20	0.24	TSB-CP-02	0.0027	Site-specific action lovel		Pass
	PAHs	2,0,7,0°TODD TEQ BaPEa*	mg/kg	33 45	10	94 40	0.00040	0.0037	0.000000055	0.0010	0.0000052	0.00012	0.00031	1.7	SSAD1-01	0.0027			Fass
		Nanhthalono	mg/kg		0	40	0.0001	0.040	0.0073	0.34	0.012	0.040	0.001	1.7	00AF4-01	10		0	Page
	Pesticides -	hoto-BHC	mg/kg	24	12	20	0.00007	0.037	0.0012	0.20	0.019		0.000			17			Fass
	OCPs		mg/kg	34	13	38	0.00035	0.018	0.0013	0.29	0.018	0.070	0.096	1.4	TSR CLOC	1.7		2	Fail
			mg/kg	34 24	14	41	0.00025	0.035	0.00089	0.91	0.0050	0.080	0.24	3.0	TSD-GJ-04	9.5		0	Pass
		4,4-DUT	mg/kg	34	12	35	0.00043	0.035	0.0020	0.61	0.012	0.069	0.17	2.5	15B-GJ-04	1.5		U	Pass
		Dieidfin	тту/кд	34	U	U	0.000073	0.035								0.16			Pass

Ramboll Environ
#### TABLE 14. Summary Statistics and Concentration/Toxicity Screen for Individual Decision Units

Nevada Environmental Response Trust Site

Henderson, Nevada

	Chemical Group	Analyte		No. of Samples	No. of Detects	% Detects	Nondetects Detects						No. c		Io. of Samples Concentration/				
Decision Unit			Unit				Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum	Levels <sup>[1]</sup>	Screening Level Note	> 0.1 x Screening Level	Toxicity Screen Result
DU-2	Pesticides -	Hexachlorobenzene	mg/kg	50	18	36	0.00030	0.035	0.00048	0.52	0.0090	0.066	0.15	2.3	SSAQ3-01	0.23		5	Fail
	OCPs SVOCs	Toxaphene	mg/kg	34	0	0	0.0071	0.53								2.3			Pass
		bis(2-Ethylhexyl)phthalate	mg/kg	45	5	11	0.033	0.91	0.089	61	0.096	12	27	2.2	SSAP4-01	183		1	Fail
		Octachlorostyrene	mg/kg	45	2	4.4	0.0069	0.43	0.15	0.21	0.18	0.18	0.042	0.24	SSAQ3-01	N/A		N/A	N/A
	Chlorine Oxyanions	Chlorate	mg/kg	24	17	71	0.22	5.6	0.086	107	2.3	10	25	2.5	SA175	38,900		0	Pass
		Perchlorate	mg/kg	43	42	98	0.054	0.054	0.071	2,620	121	385	551	1.4	RSAM5	908		23	Fail
	Metals	Arsenic	mg/kg	66	66	100			1.7	11	4.5	4.9	1.8	0.36	SA15	7.2	Maximum BRC/TIMET background		Fail
		Chromium VI	mg/kg	25	8	32	0.11	0.44	0.12	1.3	0.53	0.54	0.36	0.67	SA64	7.0		1	Fail
		Cobalt	mg/kg	25	25	100			3.2	8.4	6.7	6.4	1.4	0.22	SA104	385		0	Pass
		Manganese	mg/kg	27	27	100			151	1,500	302	347	251	0.72	SA165	28,100		0	Pass
		Palladium	mg/kg	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	N/A		NS	NS
		Thallium	mg/kg	25	21	84	0.22	0.24	0.054	0.18	0.095	0.097	0.028	0.29	SA60	13		0	Pass
		Zirconium	mg/kg	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	NS	104		NS	NS
	Radionuclides	Radium-226	pCi/g	25	25	100			0.45	2.0	1.2	1.2	0.43	0.36	SA64	0.023		25	Fail
		Radium-228	pCi/g	25	25	100			0.45	1.9	1.2	1.2	0.44	0.36	SA16	0.041		25	Fail
		Thorium-228	pCi/g	22	22	100			0.78	3.0	1.6	1.7	0.47	0.28	SA65	0.025		22	Fail
		Thorium-230	pCi/g	22	22	100			0.53	3.3	1.2	1.6	0.77	0.47	SA128	8.4		20	Fail
DU-3		Thorium-232	pCi/g	22	22	100			0.54	2.4	1.6	1.5	0.42	0.28	SA65	7.4		21	Fail
203		Uranium-234	pCi/g	22	22	100			0.37	3.4	1.3	1.4	0.80	0.56	SA128	11		13	Fail
		Uranium-235	pCi/g	22	22	100			0.0047	0.17	0.075	0.076	0.047	0.62	SA128	0.35		16	Fail
		Uranium-238	pCi/g	22	22	100			0.38	3.3	1.1	1.3	0.75	0.57	SA128	1.4		22	Fail
	Dioxin/Furans	2,3,7,8-TCDD TEQ*	mg/kg	38	38	100			0.00000021	0.0021	0.000092	0.00021	0.00041	2.0	SA175	0.0027	Site-specific action level		Pass
	PAHs	BaPEq*	mg/kg	62	10	16	0.0082	0.45	0.0078	0.40	0.032	0.091	0.12	1.3	SSAM5-03	0.32		5	Fail
		Naphthalene	mg/kg	64	1	1.6	0.0055	0.39	0.0011	0.0011	0.0011	0.0011			SA128	18		0	Pass
	Pesticides - OCPs	beta-BHC	mg/kg	11	9	82	0.0019	0.0068	0.0018	0.052	0.014	0.019	0.015	0.81	SA15	1.7		0	Pass
		4,4'-DDE	mg/kg	11	6	55	0.0018	0.0036	0.0046	0.20	0.051	0.084	0.085	1.0	SSAM5-03	9.5		0	Pass
		4,4'-DDT	mg/kg	11	6	55	0.0018	0.0061	0.0042	0.33	0.054	0.11	0.13	1.2	SSAM5-03	7.5		0	Pass
		Dieldrin	mg/kg	11	0	0	0.0013	0.020								0.16			Pass
		Hexachlorobenzene	mg/kg	69	48	70	0.0018	0.39	0.0042	1.1	0.14	0.22	0.25	1.2	SSAM5-03	0.23		43	Fail
	01/00	Toxaphene	mg/kg	11	0	0	0.035	0.59								2.3			Pass
	SVOCs	bis(2-Ethylhexyl)phthalate	mg/kg	62	12	19	0.047	1.1	0.077	0.85	0.16	0.29	0.29	1.0	SSAM5-03	183		0	Pass
		Octachlorostyrene	mg/kg	62	15	24	0.0071	0.39	0.013	0.68	0.064	0.10	0.16	1.6	SA86	N/A		N/A	N/A

#### Notes:

-- = Not applicable N/A = BCL (other screening value) not available for screen NDEP = Neveda Department of Environmental Protection mg/kg = milligram per kilogram pCi/g = picocurie per gram NS = Not sampled BaPEq = Benzo(a)pyrene equivalent OCP = Organochlorine pesticide PAH = Polycyclic aromatic hydrocarbon BCL = Basic Comparison Level BHC = Hexachlorocyclohexane SVOC = Semivolatile organic compound COPC = Chemical of Potential Concern TCDD = Tetrachlorodibenzo-p-dioxin DDE = Dichlorodiphenyldichloroethylene TEQ = Toxicity equivalent DDT = Dichlorodiphenyltrichloroethane \* Methodology for equivalent calculations explained in text DU = Decision Unit [1] Screening levels are the lowest level among the indoor worker and outdoor worker BCLs (NDEP 2017), unless noted.

indicates analyte is carried forward to COPC identification Step 2. For arsenic, lead, and 2,3,7,8-TCDD TEQ, the maximum detected concentration is compared directly to the screening level. For all other analytes, the maximum detected concentration is compared directly to the screening level. For all other analytes, the maximum detected concentration is compared to 0.1 x screening level. If the maximum detected concentration is compared to 0.1 x screening level, the analyte "fails" and is carried forward to Step 2. If less than the 0.1 x screening level, the analyte "passes" and is eliminated as a COPC. By default, analytes for which screening levels are not available are retained for Step 2 (metals) and Step 3 (organics).

#### Source:

NDEP. 2017. User's Guide and Background Technical Document for NDEP Basic Comparison Levels (BCLs) for Human Health for the BMI Complex and Common Areas. Revision 14, July.

# TABLE 15. Results of the Background Evaluation for Metals Carried Forward from the Concentration/Toxicity Screen for Individual Decision Units

Nevada Environmental Response Trust Site Henderson, Nevada

Decision Unit	Chemical Name	Decision Unit Concentrations Greater than Background Levels? <sup>[1]</sup>				
	Arsenic <sup>[2]</sup>	No				
	Chromium VI <sup>[3]</sup>	LDF				
	Cobalt	Yes				
00-1	Manganese	Yes				
	Thallium	No				
	Zirconium	No				
	Chromium VI <sup>[3]</sup>	LDF				
DU-2	Palladium	NA				
	Zirconium	NA				
2 110	Arsenic	Yes				
00-3	Chromium VI <sup>[3]</sup>	LDF				

### Notes:

COPC = Chemical of Potential Concern

DU = Decision unit

LDF = Low detection frequency (<25%) in either Study Area or background data sets. Background comparison results may not be applicable.

NA = Background data are not available

element is present at concentrations greater than background or background data are not available.

[1] Based on background evaluation presented in Appendix J.

[2] Despite statistical tests indicating that concentrations are consistent with BRC/TIMET background,

conservatively retained as a COPC for DU-1 based on the Q-Q plot (Figure J2-1) .

[3] For chromium VI, although background comparison results may not be applicable due to low detection frequency in the DU, BRC/TIMET, and RZ-A background data sets, concentrations in all three DUs are greater than background based on the box plot (Figure J1-2) and Q-Q plot (Figure J2-2).

# TABLE 16. Results of the Background Evaluation for Radionuclides Carried from the Concentration/Toxicity Screen for Individual Decision Units Nevada Environmental Response Trust Site Henderson, Nevada

Decision Unit	Chain	Secular Equilibrium?	Radionuclide	Decision Unit Concentrations Greater than Background Levels? <sup>[1],[2]</sup>	Hydrofluoric Acid Digestion?	
			Uranium-238	No	Yes	
	I Iranium-238	Yes	Uranium-234	No		
	oranium-200	163	Thorium-230	No		
			Radium-226	No		
00-1			Thorium-232	No		
	Thorium-232	No	Radium-228	No	Yes	
			Thorium-228	Yes		
	Uranium-235	Not Evaluated	Uranium-235	No	Yes	
			Uranium-238	No		
	Uranium-238	Vac	Uranium-234	No	Voc	
		165	Thorium-230	No	165	
			Radium-226	No		
D0-2			Thorium-232	No	Yes	
	Thorium-232	Yes	Radium-228	No		
			Thorium-228	No		
	Uranium-235	Not Evaluated	Uranium-235	No	Yes	
			Uranium-238	Yes	Yes	
	I Iranium 228	Vac	Uranium-234	Yes		
	Oranium-236	165	Thorium-230	Yes		
			Radium-226	No		
00-3			Thorium-232	Yes		
	Thorium-232	No	Radium-228	No	Yes	
			Thorium-228	Yes		
	Uranium-235	Not Evaluated	Uranium-235	No	Yes	

# Notes:

COPC = Chemical of Potential Concern

DU = Decision unit

radionuclide is present at concentrations greater than background.

[1] Based on background analysis presented in Appendix J.

[2] The validity of statistical testing for radionuclide background evaluation is confounded by sample preparation and analytical method issues in the DU, BRC/TIMET, and RZ-A background data sets. For a full discussion of these limitations, see Section 5.4.2 of the report. No radionuclides are identified as COPCs in any DU, because the estimated total radionuclide cancer risks at all sample locations throughout the Study Area are consistent with the estimated total radionuclide cancer risks for the RZ-A background and BRC/TIMET regional background and radionuclides are not known to be associated with any of the former operations at the Site.

# TABLE 17. Preliminary COPCs Identified for Soils (0-10 ft bgs) in Individual Decision UnitsNevada Environmental Response Trust SiteHenderson, Nevada

Chemical Group	COPC	DU-1	DU-2	DU-3
Chlorine Oxyanions	Chlorate	Х		
	Perchlorate	Х		Х
Metals	Arsenic	Х		Х
	Chromium VI	Х	Х	Х
	Cobalt	Х		
	Manganese	Х		
	Palladium <sup>[1],[2]</sup>		Х	
	Zirconium <sup>[2]</sup>		Х	
Dioxin/Furans	2,3,7,8-TCDD TEQ*	Х		
PAHs	BaPEq*	Х	Х	Х
	Naphthalene	Х		
Pesticides - OCPs	beta-BHC	Х	Х	
	4,4'-DDE	Х		
	4,4'-DDT	Х		
	Dieldrin	Х		
	Hexachlorobenzene	Х	Х	Х
	Toxaphene	Х		
SVOCs	Bis(2-Ethylhexyl)phthalate		Х	
	Octachlorostyrene [1]	Х	Х	Х
Asbestos	Long amphibole fibers	Х	Х	Х
	Long chrysotile fibers	Х	Х	Х

#### Notes:

bgs = below ground surface	DDT = Dichlorodiphenyltrichloroethane
ft = feet	DU = Decision unit
BaPEq = Benzo(a)pyrene equivalent	OCP = Organochlorine pesticide
BCL = Basic Comparison Level	PAH = Polycyclic aromatic hydrocarbon
BHC = Hexachlorocyclohexane	SVOC = Semivolatile organic compound
COPC = Chemical of Potential Concern	TCDD = Tetrachlorodibenzo-p-dioxin
DDE = Dichlorodiphenyldichloroethylene	TEQ = Toxicity equivalent

[1] Retained as a COPC in the absence of a BCL or other screening level. This COPC will be discussed qualitatively in the uncertainties assessment.

[2] RZ-A background data are not available for this chemical, and therefore a background evaluation cannot be conducted for DU-2.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

# **FIGURES**











Q: \DRAWINGS\NERT









Cmap\201709 Grid\BHRA\_COPC\_intensity201709.



Cmap\201709 Grid\BHRA\_COPC\_intensity201709.









Cmap\201709 Grid\BHRA\_COPC\_intensity201709.














































H:\LeP





Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

> APPENDIX A RESPONSES TO NDEP'S DECEMBER 22, 2016 COMMENTS ON THE AUGUST 26, 2016 INTERIM REPORT, IDENTIFICATION OF COPCS AND EXPOSURE UNITS FOR SOILS

Response to the NDEP's December 22, 2016 Comments on the August 26, 2016 Interim Report, Identification of COPCs and Exposure Units for Soils Nevada Environmental Response Trust Site, Henderson, Nevada

NDEP Comment	Response to Comment
Fatal Flaw	
<ol> <li>Section 5.1, first bullet, page 26. The Exposure Units do not seem to have any relation to a possible chronic exposure area for some individual. Hence it does not seem appropriate to call them exposure units. They seem to relate almost wholly to current land use and infrastructure. Further substantiation is warranted otherwise alternative EUs should be considered (e.g., based upon the Conceptual Site Model and the potential exposure realm for future on-site populations).</li> </ol>	As discussed at the December 19, 2016 meeting with NDEP (see Meeting Minutes in Attachment 1 to Appendix A), the term "Decision Unit" [DU] will be used in place of the term "Exposure Unit" in upcoming reports. It was proposed at the meeting that the Study Area would be considered as a whole, and a COPC-specific statistical and spatial analysis would be conducted to support the selection of DUs. In the revised COPC/DU report, Ramboll Environ conducted a COPC- specific, risk-based spatial analysis, and generated a series of spatial plots of concentrations, cancer risks, and noncancer hazard indices (HIs) (see Section 5.1). Three new DUs have been proposed based on these analyses (see Section 5.2). A WebEx meeting was also held on October 5, 2017 to present the results of these analyses to NDEP, and NDEP did not provide any further comments on the information presented during the meeting.

**Essential Corrections** 

NDEP Comment	Response to Comment
2. Section 2.2 Remedial Investigation – Soil, Page 7, First paragraph, third line from the bottom, page 7 (Remedial Investigation Data Evaluation Technical Memorandum (Ramboll Environ 2016), which is currently under NDEP review). If the memorandum is still under review, please identify the exposure units (EUs) that may be affected by the pending review and how this might affect the selection of COPCs in soil.	Around the same time as the submittal of the interim COPC/EU report on August 26, 2016, NDEP completed its review of the Remedial Investigation Data Evaluation Technical Memorandum (dated July 13, 2016). NDEP did not comment on the specific investigations conducted to support the baseline health risk assessment and no additional data needs were identified for risk assessment purposes. Although additional data has been or will be collected (Phase 2 RI, Phase
	3 RI, Unit 4 and 5 Buildings Investigation), these data are being collected primarily to evaluate fate and transport of perchlorate and other constituents in soil and groundwater. Further, NDEP agreed that comments on the Remedial Investigation Data Evaluation Technical Memorandum can be addressed in the forthcoming remedial investigation pata Evaluation Technical Memorandum will not be prepared. NDEP approved the memorandum on August 23, 2016.
	Because planned additional data collection efforts are being conducted primarily to evaluate fate and transport of perchlorate and other constituents in soil and groundwater, we do not anticipate that the additional soil data will impact the proposed DU(s) or the selection of COPCs. Any additional soil data collected from the 0-10 foot depth interval within the Study Area will be included in the risk assessment data set. The COPC selection steps will be repeated in the BHRA using the updated data set and the DU(s) will be reviewed to ensure that the additional soil results do not impact our current understanding of the spatial distribution of soil contamination.
3. <u>Section 3.1 Data Usability Evaluation, second paragraph, last</u> <u>line, page 8.</u> Please discuss how the review of EU-specific data relative to Criterion VI with NDEP may affect the selection of COPCs. For example, if there are potential issues with data quality indicators, then this could affect the selection of COPCs (and is there a chance that COPCs have been eliminated prematurely?).	Please see response to Specific Comment #1. A review of the DU- specific data relative to Criterion VI is presented in Section 5.3, and the impacts of potential issues with data quality indicators on the COPC selection and DU identification are also further discussed in the same section. The results indicated that no COPCs have been eliminated prematurely.

	NDEP Comment	Response to Comment
4. <u>Section</u> senter review and R there fill da cavea being	on 3.1.4 Criterion III – Data Sources, 2 <sup>nd</sup> paragraph, last ence, page 11. The document states that based on the w that sample coverage from the historical investigations RI are considered adequate for the BHRA. However, e are some areas that are still undergoing investigation to tha gaps. Therefore, this statement is misleading. Please at by indicating areas where sample coverage is still g investigated.	Please see response to Specific Comment #2. This statement has been revised to reflect the status of the data collection activities to date.
5. <u>Section</u> page Sugge	on 3.1.5, Criterion IV, first paragraph, second sentence, <u>12.</u> Table 2 is Soil Sampling Results for Asbestos. est correct to reference Table 3.	The table number has been revised.
6. <u>Section</u> page sets v data o limita can b	on 3.2.1 Summary Statistics, Last paragraph, last line, 13. The text notes that limitations in the available data will be discussed in the BHRA. However, the purpose of a usability section to select COPCs, is to identify the ations. If addition data collection is ongoing, then this be documented in the explanation.	This sentence has been revised as follows: "For most analytes, the summary statistics are based on the results of between 200 and 700 samples, although for some analytes (particularly those not expected to be site related) the analytical data set is much more limited (<20 samples). However, the analytes with limited sample size were never detected (e.g., chlorite, some SVOCs and VOCs) and/or were not SRCs (e.g., lithium, niobium, chlorite, fluoride, some SVOCs and VOCs). Therefore, the limited sample size for these analytes does not impact the COPC selection and DU identification. Considering both the data review conducted by Ramboll Environ and the data reviews conducted by NDEP prior to the removal action and the NDEP-approved RAW, the existing data set is considered adequate for risk assessment purposes. Planned additional data collection efforts are being conducted to support the evaluation of fate and transport of perchlorate and other constituents in soil and groundwater. Any additional soil data collected from the 0-10 ft depth interval within the Study Area will be included in the BHRA data set. If additional data are added, the COPC selection and DU identification

	NDEP Comment	Response to Comment
7.	Section 3.2.4 Comparison to CSM second bullet, page 18. Review of Table D2 indicates boron and hexavalent chromium had low detection frequencies as indicated by "LDF" and suggests background comparison results may not be applicable. Additionally, thallium and tungsten indicate that study area samples are greater than background samples in the last column of the table but are not addressed in this section of the text. Suggest verifying and revising text/tables accordingly.	In the revised report, the background evaluation has been performed using the updated BHRA data set. The tables have been revised accordingly. Also, clarifying text has been added to Section 3.2.4 to discuss the background comparison results of boron, hexavalent chromium, thallium and tungsten.
	Additionally, Table 6 data for "Study Area > Bkg?" are not in agreement with Table D2 for boron, chromium VI, and iron. Please revise accordingly.	
8.	Section 4.1 Step 1 – Concentration/Toxicity Screen, Second paragraph, third line, page 21. The text states that "with the exception of the analytical results excluded based on the DUE (see Section 3)." However, the DUE section does not discuss the excluded data. In addition, Table 1 does not explicitly list the data that are excluded. Please provide.	The data excluded during data processing are discussed in Section 3.1.1. A list of the data that are excluded are provided in Appendix B, Table B-2. The text has been revised to lead the reader to the location of the appropriate table.
9.	Section 4.1.4, Page 23 There are several soil samples with multiple amphibole protocol structures: 1, 2, 3, 4, 7, 9, and 11 fibers. The "intensity plot" for asbestos only shows grids where fibers were detected, so in fact no intensity at all, just 0/1. Please revise such that intensity is indicated in some manner (e.g., color coding).	The intensity plot for asbestos has been revised by using color coding to indicate different ranges of fiber counts (see Figure 21). Most of the asbestos samples with high fiber counts (> 4) are no longer in the current BHRA data set because several samples included in the 2016 BHRA data set were confirmed as having been removed during the interim soil removal actions conducted in 2010 and 2011.

NDEP Comment	Response to Comment
10. <u>Section 4.4 Study Area Preliminary COPCs last bullet, page</u> <u>25.</u> This radionuclide did not pass the background evaluation. The investigator's note to interpret "with caution" the U-235 results yet the DUE found the data usable for risk assessment. Suggest U-235 be continued into the risk assessment where its overall contribution to risk can be further evaluated.	As discussed in the DUE in Section 3, radionuclide data are usable for risk assessment. The results of the background evaluation for radionuclides are discussed in Section 4.2 for the entire Study Area and Section 5.4.2 for individual DUs. Given that the validity of the statistical testing is complicated by several issues associated with sample preparation and analytical methods, it is difficult to interpret the results of the background evaluation for radionuclides and consider them as a reliable basis for the COPC selection. To provide a point of comparison from a health risk perspective, the total radionuclide cancer risk at each sampling location was compared to the total radionuclide cancer risks for the RZ-A background soils and BRC/TIMET regional background soils. As indicated in Figures 26a through 26c, the estimated total radionuclide cancer risks for the RZ-A background (Table 12). Radionuclides are not known to be associated with any of the former operations at the Site. Therefore, radionuclides were not identified as COPCs for any DU.
11. Section 4.1.4, first full paragraph, first sentence, page 26. It is stated that Th-232 is within background yet Figure D1-30 shows total uranium results that clearly have a tail extending well outside the range of background results. This warrants further discussion.	Ramboll Environ has assumed this comment is related to uranium (total) versus U-238 decay series (instead of Th-232 decay series). These two terms refer to different groups of constituents, were reported in different units (mg/kg for uranium [total] and pCi/g for U-238 decay series), and were evaluated against different background data sets. As indicated in Table F-4, for the U-238 decay series, all radionuclide activities were consistent with background levels. Statistical testing results in Table F-2 indicate that the uranium (total) concentration was consistent with the background level, although Figure F1-30 shows a tail extending well outside the range of background results. This issue will not affect the COPC selection, because uranium (total) passed the toxicity screen (Table 8) and would not be identified as a COPC regardless of the results of background evaluation.

NDEP Comment	Response to Comment
12. <u>Section 5.1, Page 26.</u> The size of the six EUs range from about 12 to 45 acres. There is some narrative of which COPCs are elevated in each EU, but no indication that spatial patterns of contamination were used to define these areas. Please expand this discussion accordingly by elaborating on how and why they were defined in this manner.	Please see the response to Specific Comment #1. A COPC-specific, risk-based spatial analysis was conducted and a series of spatial plots of concentrations, cancer risks, and noncancer HIs were reviewed (see Section 5.1). Three new DUs have been proposed based on these analyses (see Section 5.2).
13. <u>Section 5.1, last paragraph, page 26.</u> The plots showing the spatial distribution of COPCs don't assist with the visualization of whether the outliers in the boxplots of many COPCs occur in common areas or independently. Please see Specific Comment #19, fourth paragraph for additional recommendations.	In the spatial quartile plots presented in Appendix G, the concentrations bins are tied to those used for the box plots (i.e., <q1, +="" [1.5="" and="" iqr]="" q1-q3,="" q3="" x=""> Q3 + [1.5 x IQR]). These plots provided the visualization of whether the outliers in the boxplots of many chemicals occur in common areas or independently.</q1,>
<ol> <li>Section 5.2, Page 27. Figure 19 shows that these areas are not all spatially contiguous and as such they are not plausible exposure units. Please see Fatal Flaw #1 and Specific Comment #12.</li> </ol>	Please see the responses to Specific Comments #1 and #12.
15. <u>Section 5.2, footnote 18, page 27.</u> There should be some basis in an exposure model for how big a "localized area" might be. The 1-acre grid could be a localized area, or 10 grids, or? Please provide a more definitive discussion.	Please see the response to Specific Comment #1.

**NDEP Comment** 

	1
Response to Comment	
es for which the detection frequency was less than 100%,	

<ul> <li>16. Page 5 of Table 1. This table lists DOE EML HASL 300 as a standard analytical method on historical investigations for thorium and uranium. Review of Table 3, Evaluation of Sample Quantitation Limits, only summarizes U-235 and U-238. In addition, Table 1 lists PCB Congeners by EPA Method 1558A as a standard analytical method on historical investigations. However, review of Table 3, Evaluation of Sample Quantitation Limits, does not include any samples associated with this method; same for herbicides by EPA Method 8151 and for formaldehyde by EPA Method 8315A. Suggest clarification/revision.</li> <li>In addition, Criterion IV Analytical Methods and Detection Limits (last set of bullets on page 7 of Table 1): For the analytes where the SQL exceeded the 0.1 x BCL, it would be helpful to have some additional information about these analytes and samples. For example, are these samples with elevated SQLs co-located in one EU? If so, how would this affect the COPC selection process (e.g., is a chemical eliminated because it is non-detect due to elevated SQLs?). Please provide additional information here or in the text (or table).</li> </ul>	For analytes for which the detection frequency was less than 100%, the SQLs from the BHRA data set were compared to 0.1 times the BCL as shown in Table 3 (as discussed in Section 3.1.5, second paragraph). As indicated in Table 4, the detection frequency of all radionuclides was 100%, so the evaluation of their SQLs was not needed. Dioxin-like PCB congeners were evaluated through the calculation of dioxin TEQs, and the evaluation of the SQL for dioxin TEQ is shown in Table 3. Other PCB congeners were not evaluated in the report due to the lack of BCLs (or appropriate surrogates) needed for the concentration/toxicity screen. Only one herbicide (2,4,5-TP) was analyzed by EPA Method 8151. It was included in the chemical group of Pesticides – OCPs, and the evaluation of its SQL is shown in Table 3. Formaldehyde by EPA Method 8315A was included in the chemical group of VOCs, and the evaluation of its SQL is shown in Table 3. Additional discussion about the effects of samples with elevated SQLs on the COPC selection and DU identification is provided in Section 5.3.2.
<ol> <li>Page 4 of Table 6. The data for "Study Area &gt; Bkg?" are not in agreement with Table D4 for U-234, is missing a "no" for U- 238, and indicates "TBD" for Th-232 whereas Table D4 indicates "no". Suggest clarification/revision.</li> </ol>	The background evaluation has been performed using the updated BHRA data set. The tables were checked for consistency and revised accordingly. Table 6 has been updated with the "Fail Statistical Testing for Background?" column indicating "no" for U-234, U-238, and Th-232, consistent with Table F-4.
18. <u>Page 2 of Table 8.</u> Zirconium is highlighted in blue color to indicate the chemical "failed" the toxicity screen. However, the maximum detection reported of 31 mg/kg and a screening level of 104 mg/kg appears to contradict the "fail" status. Please revise as appropriate.	In this table, the screening criteria is whether the maximum concentration for zirconium (31 mg/kg) is greater than 0.1 x BCL (0.1 x 104 = 10.4 mg/kg), so this chemical fails the toxicity screen.

19. Figure 5 and all other spatial intensity plots. The spatial intensity plots are not useful for characterizing the spatial pattern of the data, which is an essential component of the development of a narrative that allows for the defensible delineation of exposure units, or combinations of exposure units over such large areas. The proposed exposure units are larger than a typical industrial receptor's exposure area at this site, and the intent seems to be to combine across smaller exposure units (typical default industrial exposure units (note in comments above that these are not really exposure units). This type of combination of exposure units is only reasonable if the spatial pattern of the data demonstrates relative homogeneity and randomness across the exposure unit. However, the spatial plots do not allow for sufficient evaluation of the potential for areas with comparatively large or small concentrations. Also, because the term exposure unit is often reserved for the size of area to which a receptor might be exposed, some consideration should be given to using another term for a larger area (such as exposure area), with definitions provided for the terms used.	As previously noted in our response to Specific Comment # 1, we will adopt the term "decision unit" in place of "exposure unit" in upcoming reports. In this report, Ramboll Environ conducted a COPC-specific, risk-based spatial analysis, and generated a series of spatial plots of concentrations, cancer risks, and noncancer HIs (see Section 5.1). Three new DUs have been proposed based on these analyses (see Section 5.2).
In addition, the background comparisons have been performed for the entire site, and not for each exposure unit. The same statistical principle applies as for defining exposure areas. This is only appropriate if the data are relatively homogenous across the site (essentially independent and identically distributed – essentially one population). This is an assumption that underlies the background statistical tests. To the extent that the assumption is violated, the study area should be broken up into smaller units that separate area of different concentrations. For example, the challenges that have been presented	
regarding radionuclides appear to be because of relatively large concentrations in the north-east quadrant of the data. It looks doubtful that this quadrant would pass background	

NDEP Comment	Response to Comment
comparisons, but we defer judgment on that until the statistical analysis is redone.	
Plots that are useful to support these EU and background comparison concerns should have a relatively continuous representation of the range of concentration (or counts for asbestos fibers). The current use of three ordinal categorical classes for the concentrations (defined as relative to 0.1 the respective BCL) does not allow for the necessary determination of hot spots and general spatial trends in the data. We suggest the generation of new plots (not as replacements necessarily, but in addition – the current plots might be useful to some who like to be able to distinguish concentrations at these discrete levels).	In this report, Ramboll Environ conducted a COPC-specific, risk-based spatial analysis, and generated a series of spatial plots of concentrations, cancer risks, and noncancer HIs (see Section 5.1). Three new DUs have been proposed based on these analyses (see Section 5.2).
Furthermore, it is not clear from the plots how the existence of multiple samples within a grid cell is depicted. More generally, the basic algorithm underlying the current plots is not provided, and could affect appearance – it should be provided.	As stated in Section 4.4, fourth bullet, page 29, the maximum detected concentration is displayed when multiple samples were collected within a grid cell.
20. <u>Appendix A-1 Data Validation Summary Reports.</u> With the exception of the Data Validation Summary Report for Phase A, tables associated with the DVSRs have not been included. As the tables compliment and complete the DVSR, they should be included in the Appendix or, at least, provided on CD.	The DVSR tables are provided in Appendix B.

NDEP Comment	Response to Comment
21. <u>Appendix A-1 Data Validation Summary Reports.</u> Per the guidance in effect at the time of validation, data in all but one DVSR were censored for blank contamination. The final DVSR included in this report for the August 2011 Soil Remediation Completion Sampling was produced after the guidance changed and the associated data were not censored for blank contamination. In order for all datasets to be treated equivalently, and to avoid the loss of data, perhaps censored results (detects qualified as non-detected), should be considered as estimated detects as they would be per current guidance. These data should be easily queried from the project database and the original result should be used.	The censored data for blank contamination were queried from the project database, and considered as estimated detects per current guidance (See Appendix B, Table B-5). The BHRA data set and all the related evaluation (text, tables, and figures) have been updated accordingly.
Minor Correction	
22. <u>Section 5.2, first full paragraph, page 27.</u> If the spatial plots were reviewed to identify possible hot spots then the outlines of the proposed EUs should be shown in the spatial plots. Currently they are not. In addition, it would be fairly easy to take the ratios of sample results to BCLs and create some contour plots. That could be done across COPCs for cancer risk and HI. This would provide a basis for proposing exposure units.	In this report, Ramboll Environ conducted a COPC-specific, risk-based spatial analysis and generated a series of spatial plots of concentrations, cancer risks, and noncancer HIs (see Section 5.1, Figures 22-28). Three new DUs have been proposed based on these analyses (see Section 5.2, Figure 29).
23. <u>Table 1, Data Quality Indicators, Precision.</u> Text indicates precision goals were met; however, several DVSRs note samples qualified for field duplicate precision outliers. Perhaps this should read, "Precision of the individual investigations met the goals of 50% established in the QAPPs <i>or were qualified as estimated.</i> "	The text has been revised, and the data for qualified field duplicates are summarized in Appendix B, Table B-4.

NDEP Comment	Response to Comment
24. <u>Table 1, Data Quality Indicators, Precision.</u> The precision goal of 50% is strictly for field duplicates; laboratory precision goals are defined for specific methods and should be noted here.	The text has been revised as "The field precision goal established in the QAPPs is a RPD of less than or equal to 50%, except for the case in which one (or both) of the primary or duplicate results is less than five times the reporting limit. For the latter case, the acceptance criteria is the reporting limit (i.e., the absolute value of the difference between the primary and duplicate results is less than or equal to the reporting limit). Laboratory precision goals are defined for specific analytical methods, as indicated in the QAPP (see Table 2 of ENVIRON [2014d])."
25. <u>Table 1, Data Quality Indicators, Accuracy.</u> Text indicates surrogate and LCS percent recoveries were met; however, several DVSRs note samples qualified and rejected for these criteria. Matrix spikes are not included here as having met criteria, but should be. Perhaps this should read, "Surrogate, LCS, and matrix spike percent recoveries met the QC acceptance criteria established in the QAPPs <i>or were qualified</i> <i>as estimated or rejected.</i> "	The text has been revised as proposed by NDEP. All qualified results (i.e., U, J, J-, and J+ qualified data) for the non-asbestos analytes are presented in Appendix C, Table C-1, and the reasons for these qualified results are summarized in the DVSRs (see Appendix B).
26. <u>Table 1, Data Quality Indicators, Accuracy.</u> Sample preservation - may want to consider adding details regarding data rejected because the laboratory did not correctly preserve some SPLP samples (rejected data).	The rejected data, including the reasons for rejection are summarized in Appendix B, Table B-3.
27. <u>Table 1, Data Quality Indicators, Accuracy.</u> For sample integrity, at least one soil sample was noted as having been received with standing water (estimated), likely from melting ice in the cooler. And a number of holding times were missed, resulting in estimated and rejected data.	All qualified results (i.e., U, J, J-, and J+ qualified data) for the non- asbestos analytes are presented in Appendix C, Table C-1, and the reasons for these qualified results are summarized in the DVSRs (see Appendix B). The rejected data, including the reasons for rejection are summarized in Appendix B, Table B-3.
28. <u>Table 1, Data Quality Indicators, Accuracy.</u> Depending on what happens regarding specific comment #3, blank qualifications discussed here may need to be changed.	Discussion about the data qualified for blank contamination is presented in Table 1 for the entire Study Area and in Section 5.3.5 for the individual DUs.
29. <u>Appendix A-1, Acronyms and abbreviations.</u> The correlation coefficient is not r2, it is simply r.	This comment has been made on a DVSR prepared by ENSR in 2007. Therefore, Ramboll Environ is not able to revise the DVSR.

NDEP Comment	Response to Comment
30. <u>Spatial plots.</u> Note that the plots have a legend that has lowest concentrations first, then middle, then high (greater than BCLs for the most part), and the NDs. However, NDs are really the lowest concentrations, so it would make more sense to change the order so that it is monotonic). Also the color scheme is awkward for the NDs (grey) – they clearly are not from the same color palette, and given that their concentrations are usually small, it is not clear why this has been done this way.	The order of the concentrations in the legend has been revised such that the NDs are listed first. We chose a grey color for the NDs to more clearly distinguish samples reported as ND from samples with detected concentrations. In addition, we note that for organic compounds in particular, NDs often represent a range of concentrations and concentrations reported at the higher end of the range may be greater than some of the detected concentrations. However, in response to this comment, a color from the same color palette was selected for the NDs to facilitate interpretation of the plots.
31. <u>Arsenic spatial plot.</u> The arsenic plot indicates that background is 7.2 mg/kg. This is not a correct statement. Clarification is needed on what 7.2 mg/kg actually represents. (That is, it represents the maximum background values from the BRC/TIMET background data set – it does not represent NERT site conditions, and it does not represent average background (for example)).	Relevant text, tables, and figures have been revised to indicate that 7.2 mg/kg is not a site-specific background, but represents the maximum background value from the BRC/TIMET background data set.
32. <u>Table D4 Shapiro-Wilk tests.</u> The Shapiro-Wilk test results are not useful. Given the amount of data, these tests are very likely to reject a hypothesis of normality or of any other distribution. In addition, t-tests are very robust to deviations from underlying normality assumptions, in large part because large number of data points causes the mean to be approximately normal. There are large numbers of data points here. The non-parametric tests are reasonable, and presenting t-tests for the actual data and the log data is ok, but the results should now be interpreted for every test that is run, without picking one or the other t-test based on a Shapiro-Wilk test that is likely to reject.	Ramboll Environ has understood that the all tests included in Gilbert's Tool box should be presented. However, as requested, for analytes with large sample size (>100), the results of the background comparison were updated in the related tables and text based on the non-parametric tests only.

NDEP Comment	Response to Comment				
33. <u>Missing information</u> . Although the box plots and quantile plots show the background data, it would also be helpful if the background data summaries were included in the summary statistics tables. A numerical comparison can then be made, as well as the graphical comparisons that are provided.	Per discussion with NDEP at the December 19, 2016 meeting, this comment is disregarded.				
Editorial Changes					
34. <u>Section 5.1.1, Last paragraph, 5th line, page 35.</u> This sentence should be removed. The fact that there is a less than 10-fold difference between min and max does not provide evidence that the data are background. This might be even more the case with such a low coefficient of variation, which could imply a large mean (compared to the standard deviation). Chemical data tend to get noisier at low concentrations, not more stable. In addition, the coefficient of variation is not a sufficient statistic, and, as such should not be used for this type of conclusion.	This sentence has been deleted in the revised report.				

Response to the NDEP's December 22, 2016 Comments on the August 26, 2016 Interim Report, Identification of COPCs and Exposure Units for Soils Nevada Environmental Response Trust Site, Henderson, Nevada

# ATTACHMENT 1

December 19, 2016 NERT Meeting Minutes

# **NERT Meeting Minutes**

# December 19, 2016

Allan DeLorme, Ramboll Environ Liz Miesner, Ramboll Environ Lynne Haroun, Ramboll Environ Steve Clough, NERT Weiquan Dong, NDEP Carlton Parker, NDEP Kirk Stowers, Broadbent & Associates, Inc. Paul Black, Neptune & Company, Inc. Paul Hackenberry, Hackenberry Associates, LLC (via phone) Kurt Fehling, The Fehling Group, LLC

# Meeting Agenda

Discuss preliminary NDEP comments on the following reports:

- 1. Interim Report, Identification of COPCs and Exposure Units for Soils, dated August 26, 2016
- 2. Soil Gas Investigation and Health Risk Assessment for Parcels C, D, F, G, and H, Revision 1, dated September 23, 2016

# **COPC Selection and Decision Units for the NERT Operations Area**

- The term Decision Unit (DU) will be used in place of Exposure Unit
- Intent was to create a preliminary list of COPCs; not definitive; may be revised per the ongoing RI (which is focused on groundwater and not soils); Phase I RI data are in the Deliverable. Depending on timing, Phase II (which will not be available until summer 2017), may or may not be included in the revised COPC.DU report, but will be included in the BHRA report<sup>1</sup>. This data, which is not being collected for risk assessment purposes, is not expected to impact COPC selection or the identification of the DU.
- Specific Comments #1, 12, 14, and 15: Right now, it appears the site will remain industrial and there are no plans for redevelopment. NDEP suggested working backwards by determining what the Trust is willing to do and ensure—such as deed restricting it to commercial/industrial. If something changes in the future, then reassess and address as appropriate. **Decision: May consider the site one decision unit for the baseline health risk assessment.**
- Specific Comments #7, 10, and 19: Background may be used as needed but if used, make sure it is properly justified and that the background populations and decision units are of one population if combined. For arsenic, present a comparison of risks for site concentrations to risks for background concentration.
- Specific Comments #13, and 19: The spatial plots provided may be sufficient, but Neptune prefers continuous concentration plots while Ramboll Environ prefers the grid

<sup>&</sup>lt;sup>1</sup> Post-meeting note: Additional soil samples are being collected as part of the Phase II RI to support background evaluations. While it is not anticipated that this data set will impact the metals identified as preliminary COPCs (most metals were deleted based on the concentration/toxicity screen), it is possible that the background evaluation of radionuclide concentrations will be impacted.

plots provided in the report; Neptune indicated that they may plot the data (which is provided on CD in the report binders) for their own internal review.

- Radionuclides: NERT understands more is to be done with respect to these for the Study Area. Neptune is evaluating the radionuclide background issues for Parcel H<sup>2</sup>.
- Specific Comment #33: Disregard this comment per NDEP.

# Soil Gas HRA for Parcels C, D, F, G, and H

- Trust goal is an NFA; 0-10 foot soils with land use restrictions.
- May have a potential buyer(s) in place for these parcels.
- Cumulative risk for soil-related pathways and soil gas (vapor intrusion and outdoor air) will be presented in either the soil HRA or soil gas HRA.
- Soil gas: NDEP will provide more direction on whether or not a trench model will be evaluated.
- For benzene in groundwater, consider using a vapor intrusion model specific to petroleum hydrocarbons.
- Soil gas: use multiple lines of evidence including soil gas data and groundwater modeling; include evaluation of outdoor air and hypothetical future indoor.
  - Groundwater evaluation will be presented in main report (not in an appendix).
  - J&E modeling will use EPA's current version (using some site-specific parameters as discussed).
  - May look at temporal trends on a sample by sample (well by well) basis.
  - May look at averaging across well groups but this would likely be a more appropriate discussion for the Uncertainty Section.
  - For off-site scenarios, NERT will bolster the existing discussion.
  - Specific Comment #6: NERT will note in the text why samples outside of Parcel H are not used.
  - Specific Comment #7: NERT will no longer try to correlate soil gas and groundwater but will use multiple lines of evidence and delete this discussion.
  - Specific Comment #8: NDEP has removed this comment.
  - Specific Comment #9: The building height of 8 ft was approved in the work plan and therefore, NDEP will accept it here.
  - Specific Comment #18: NERT will evaluate whether or not the soil property indicators are applicable to the shallow soil (~5 ft) depth.
  - Specific Comment #2: will be revised based upon the temporal study.
- The Trust may want to consider removing Parcel F from the risk assessment (or evaluating Parcel F separately) considering the encroaching DNAPL plume and associated uncertainties.
- Presentation of risk results
  - Present risk results separately for chemicals, radionuclides, and asbestos.
  - For chemicals, cumulative risk is presented for soil-related pathways and soil gas pathways (vapor intrusion, outdoor air).

<sup>&</sup>lt;sup>2</sup> Parcel H is outside the Operations Area and is being evaluated separately as part of the risk assessment for Parcels C, D, F, G, and H.

 Also, present the background risk for arsenic (soil-related pathways), the estimated site risk for arsenic, and cumulative risk (all chemicals)±arsenic risk.

# **Other Notes**

• **NDEP will send final comments to the Trust**; responses to the comments will be included as Attachment A to the soil gas HRA and the COPC Selection/Decision Unit report.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

> APPENDIX B DATA VALIDATION SUMMARY REPORTS (CD) AND TABLES

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	BDT-1-N-20-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-1-N-20-10BPC_FD	10	11	FD	End depth exceeds 10 ft
Remove	BDT-1-S-10-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-1-S-15-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-1-S-20-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-1-S-5-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-3-N-20-10BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-4-N-15-14.0BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-4-N-20-14.0BPC	10	11	N	End depth exceeds 10 ft
Remove	BDT-4-S-15-12BPC	10	11	N	End depth exceeds 10 ft
Remove	BERM-J7-01-1BPC	1.5	2.5	Ν	Change of ECA D3 Northern Boundary
Remove	BERM-J7-01-2BPC	2.5	3.5	N	Change of ECA D3 Northern Boundary
Remove	BERM-J7-01-3BPC	3.5	4.5	Ν	Change of ECA D3 Northern Boundary
Remove	CS-E08B-1	0	0	Ν	Already removed during remediation
Remove	CS-E11-2	0	0	Ν	Already removed during remediation
Remove	EE-D02-1	0	0	N	Sample within ECA
Remove	M-161D-10.0-20141203	10	10.5	N	End depth exceeds 10 ft
Remove	M-161D-10.0-20141203-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	M-162D-10.0-20141209	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-09-10.0-20141211	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-10-10.0-20141215	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-11-10.0-20141217	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-11-10.0-20141217-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	RISB-12-10.0-20141216	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-13-10.0-20141218	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-14-10.0-20141216	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-15-10.0-20141103	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-16-10.0-20141029	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-17-10.0-20141028	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-18-10.0-20141028	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-19-10.0-20141027	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-20-10.0-20141030	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-21-10.0-20141031	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-21-10.0-20141031-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	RISB-22-10.0-20141029	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-22-10.0-20141029-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	RISB-23-10.0-20141029	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-24-10.0-20141027	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-26-10.0-20141023	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-27-10.0-20141024	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-27-10.0-20141024-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	RISB-28-10.0-20141027	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-29-10.0-20141027	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-30-10.0-20141118	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-31-10.0-20141119	10	10.5	Ν	End depth exceeds 10 ft

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	RISB-32-10.0-20141120	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-33-10.0-20141119	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-34-10.0-20141119	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-35-10.0-20141119	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-36-10.0-20141118	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-37-10.0-20141118	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-39-0.5-20141121	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-39-0.5-20141202	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-39-5.0-20141121	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-39-5.0-20141121-FD	5	5.5	FD	Change of ECA D3 Northern Boundary
Remove	RISB-39-5.0-20141202	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-39-5.0-20141202-FD	5	5.5	FD	Change of ECA D3 Northern Boundary
Remove	RISB-40-0.5-20141121	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-40-0.5-20141202	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-40-5.0-20141121	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-40-5.0-20141202	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-41-0.5-20141121	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-41-0.5-20141121-FD	0.5	1	FD	Change of ECA D3 Northern Boundary
Remove	RISB-41-0.5-20141202	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-41-0.5-20141202-FD	0.5	1	FD	Change of ECA D3 Northern Boundary
Remove	RISB-41-5.0-20141121	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-41-5.0-20141202	5	5.5	Ν	Change of ECA D3 Northern Boundary
Remove	RISB-48-0.5-20141120	0.5	1	Ν	Change of Nouthern Site Boundary
Remove	RISB-48-0.5-20141202	0.5	1	Ν	Change of Nouthern Site Boundary
Remove	RISB-48-5.0-20141121	5	5.5	Ν	Change of Nouthern Site Boundary
Remove	RISB-48-5.0-20141202	5	5.5	Ν	Change of Nouthern Site Boundary
Remove	RISB-50-10.0-20141029	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-51-10.0-20141030	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-52-10.0-20141030	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-53-10.0-20141107	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-54-10.0-20141117	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-55-10.0-20141114	10	10.5	N	End depth exceeds 10 ft
Remove	RISB-55-10.0-20141114-FD	10	10.5	FD	End depth exceeds 10 ft
Remove	RISB-56-10.0-20141104	10	10.5	Ν	End depth exceeds 10 ft
Remove	RISB-57-10-20141103	10	10.5	Ν	End depth exceeds 10 ft
Remove	RSAI3-14BPC	10	10.5	N	Already removed during remediation
Remove	RSAI4-0.0	0	0.5	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI4-0.5B	0.5	2	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI4-10B	10	11.5	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI5-0.0	0	0.5	N	Change of ECA D3 Northern Boundary
Remove	RSAI5-0.5B	0.5	2	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI5009-10B	10	11.5	FD	Change of ECA D3 Northern Boundary
Remove	RSAI5-10B	10	11.5	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI7-0.0	0	0	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI7-0.5B	0	0.5	Ν	Change of ECA D3 Northern Boundary

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	RSAI7009-1.5BR	1	1.5	FD	Change of ECA D3 Northern Boundary
Remove	RSAI7-1.5BR	1	1.5	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI7-10B	9.5	11	Ν	Change of ECA D3 Northern Boundary
Remove	RSAI7-1BR	0.5	1	Ν	Change of ECA D3 Northern Boundary
Remove	RSAK2-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAK2-10BPC	10	11	Ν	End depth exceeds 10 ft
Remove	RSAK4-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAL3-9BPC	10	11	Ν	End depth exceeds 10 ft
Remove	RSAL4-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAL5-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAL7-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAL8-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAM2-10B	10	11.5	N	End depth exceeds 10 ft
Remove	RSAM4-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAM6-10B	10	11.5	N	End depth exceeds 10 ft
Remove	RSAM7-0.33BPC	0	0	N	Already removed during remediation
Remove	RSAN5-10B	10	11.5	N	End depth exceeds 10 ft
Remove	RSAO2-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSA05-10B	10	11.5	N	End depth exceeds 10 ft
Remove	RSAO7-19B	0	0	Ν	Correct depth at 10-11.5 ft
Remove	RSAO7-29B	8.5	10	Ν	Correct depth at 20 ft
Remove	RSAQ5-0.0B	0.5	1	Ν	Sample within ECA
Remove	RSAQ5-0.5B	1	2.5	N	Sample within ECA
Remove	RSAQ5-1BPC	1.5	2.5	Ν	Sample within ECA
Remove	RSAQ5-2BPC	2.5	3.5	Ν	Sample within ECA
Remove	RSAQ5-3BPC	3.5	4.5	Ν	Sample within ECA
Remove	RSAQ7-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	RSAS8-0.0B	1	1.5	Ν	Already removed during remediation
Remove	SA103009-10B	10	12	FD	End depth exceeds 10 ft
Remove	SA103-10B	10	12	Ν	End depth exceeds 10 ft
Remove	SA121-0.0B	0	0.5	Ν	Already removed during remediation
Remove	SA121-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA136-0.0B	1	1.5	Ν	Already removed during remediation
Remove	SA138009-10B	10	12	FD	End depth exceeds 10 ft
Remove	SA138-10B	10	12	Ν	End depth exceeds 10 ft
Remove	SA139-10B	6.5	8	Ν	Change of ECA C17 Southern Boundary
Remove	SA139-4BPC	0.5	1.5	Ν	Change of ECA C17 Southern Boundary
Remove	SA139-5BPC	1.5	2.5	Ν	Change of ECA C17 Southern Boundary
Remove	SA141-30B	1.5	3	Ν	Correct depth at 16 ft bgs
Remove	SA142-30.5B	3	4.5	N	Correct depth at 10 ft bgs, beneath Pond Mn-2
Remove	SA149009-45B	2	3.5	FD	Correct depth at 17 ft bgs
Remove	SA149-45B	2	3.5	N	Correct depth at 17 ft bgs
Remove	SA151-10B	10	11.5	N	End depth exceeds 10 ft
Remove	SA157-10B	10	11.5	N	End depth exceeds 10 ft
Remove	SA166-10B	10	11.5	N	End depth exceeds 10 ft

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	SA171-15B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA182-4BPC	0	0.5	Ν	Already removed during remediation
Remove	SA182-5BPC	0.5	1.5	Ν	Already removed during remediation
Remove	SA182-6BPC	1.5	2.5	Ν	Already removed during remediation
Remove	SA185-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA186-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA189-0.0	0	0	Ν	Already removed during remediation
Remove	SA189-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA204009-10B	10	11.5	FD	End depth exceeds 10 ft
Remove	SA204-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA21-10	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA23	0.5	0.5	Ν	Change of ECA D3 Northern Boundary
Remove	SA23-0.5	1	2.5	Ν	Change of ECA D3 Northern Boundary
Remove	SA31-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA50009-12B	10	11.5	FD	End depth exceeds 10 ft
Remove	SA50-12B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA52-28B	0	0	Ν	Correct depth at 12.5 ft bgs
Remove	SA55-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA62-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA67-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA69-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA7-10	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA7-10D	10	11.5	FD	End depth exceeds 10 ft
Remove	SA71-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA77-0.0B	0.5	1	Ν	Already removed during remediation
Remove	SA85-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SA87-10B	10	11.5	Ν	End depth exceeds 10 ft
Remove	SSAI2-03-10BPC	8	9	Ν	Change of Western Site Boundary
Remove	SSAI3-02-25BPC	10	11	Ν	End depth exceeds 10 ft
Remove	SSAJ3-06-10BPC	10	11	Ν	End depth exceeds 10 ft
Remove	SSAJ8-02-10BPC	10	11	Ν	End depth exceeds 10 ft
Remove	SSAJ8-03-10BPC	10	11	Ν	End depth exceeds 10 ft
Remove	SSAK6-02-4BPC	4	4.5	Ν	Already removed during remediation
Remove	SSAK6-02-5BPC	4.5	5.5	Ν	Already removed during remediation
Remove	SSAK8-03-10BPC	10	11	Ν	End depth exceeds 10 ft
Remove	SSAM4-01-0.00BPC	0.5	0.5	Ν	Already removed during remediation
Remove	SSAM4-01-0.33BPC	0.5	0.5	Ν	Already removed during remediation
Remove	SSAM5-05-0.00_01_BPC	0	0.17	Ν	Sample within ECA
Remove	SSAM5-05-0_01_BPC	0	0.5	Ν	Sample within ECA
Remove	SSAM5-05-1_01_BPC	1	2	Ν	Sample within ECA
Remove	SSAM5-05-1_01_BPC_FD	1	2	FD	Sample within ECA
Remove	SSAM5-05-10_01_BPC	10	11	N	Sample within ECA
Remove	SSAM5-05-4_01_BPC	4	5	Ν	Sample within ECA
Remove	SSAM5-05-5_01_BPC	5	6	Ν	Sample within ECA
Remove	SSAM6-06-0.00_01_BPC	0	0.17	N	Sample within ECA

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	SSAM6-06-0_01_BPC	0	0.5	N	Sample within ECA
Remove	SSAM6-06-1_01_BPC	1	2	Ν	Sample within ECA
Remove	SSAM6-06-10_01_BPC	10	11	N	Sample within ECA
Remove	SSAM6-06-4_01_BPC	4	5	Ν	Sample within ECA
Remove	SSAM6-06-5_01_BPC	5	6	Ν	Sample within ECA
Remove	SSAM7-02-0.00BPC	0	0	Ν	Already removed during remediation
Remove	SSAM7-02-0.00BPC_FD	0	0	FD	Already removed during remediation
Remove	SSAM7-03-3BPC	2.5	3.5	Ν	Already removed during remediation
Remove	SSAM7-03-4BPC	3.5	4.5	Ν	Already removed during remediation
Remove	SSAM7-03-5BPC	4.5	5.5	Ν	Already removed during remediation
Remove	SSAM7-03-5BPC_FD	4.5	5.5	FD	Already removed during remediation
Remove	SSAN4-01-8BPC	10	11	N	End depth exceeds 10 ft
Remove	SSAN8-03-0.5BPC	2.5	3	Ν	Change of ECA C17 Southern Boundary
Remove	SSAN8-03-0BPC	2	2.5	Ν	Change of ECA C17 Southern Boundary
Remove	SSAN8-05-0.5BPC	1	1.5	N	Change of ECA C17 Southern Boundary
Remove	SSAO8-04-0BPC	1.5	2	N	Beneath Pond Mn-2
Remove	SSAO8-04-0.5BPC	2	2.5	Ν	Beneath Pond Mn-2
Remove	SSAO8-07-0BPC	2.5	3	Ν	Beneath Pond Mn-2
Remove	SSAO8-07-0.5BPC	3	3.5	N	Beneath Pond Mn-2
Remove	SSAO8-13-1_01_BPC	8.5	9.5	N	Beneath Pond Mn-2
Remove	SSAO8-13-2_01_BPC	9.5	10.5	N	Beneath Pond Mn-2
Remove	SSAO8-14-3_01_BPC	0	1	Ν	Beneath Pond Mn-2
Remove	SSAO8-14-3_01_BPC_FD	0	1	FD	Beneath Pond Mn-2
Remove	SSAO8-14-4_01_BPC	1	2	N	Beneath Pond Mn-2
Remove	SSAO8-14-5_01_BPC	2	3	Ν	Beneath Pond Mn-2
Remove	SSAP3-03-10BPC	10	11	N	End depth exceeds 10 ft
Remove	SSAP4-01-10BPC	10	11	N	End depth exceeds 10 ft
Remove	SSAP4-01-10BPC_FD	10	11	FD	End depth exceeds 10 ft
Remove	SSAP8-02-0.00_01_BPC	4	4.5	N	Change of ECA C18 Eastern Boundary
Remove	SSAQ5-06-1BPC	0	1	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-06-2BPC	1	2	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-06-3BPC	2	3	N	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-06-3BPC_FD	2	3	FD	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-07-10BPC	8.5	9.5	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-07-1BPC	0	0.5	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-07-2BPC	0.5	1.5	N	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ5-07-5BPC	3.5	4.5	N	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ6-01-1BPC	0	0.5	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ6-01-2BPC	0.5	1.5	Ν	Change of Site Boundary (Southern Property Line of Lhoist North America)
Remove	SSAQ6-02-0.3_01_BPC	0.5	0.5	N	Sample within ECA
Remove	SSAQ6-02-0.3_01_BPC_FD	0.5	0.5	FD	Sample within ECA
Remove	SSAS8-03-0.00BPC	0	0	Ν	Already removed during remediation
Remove	TSB-GJ-02-0_11/19/2007	0	1.5	Ν	Change of Parcel G Southern Boundary
Remove	TSB-GJ-02-05_11/19/2007	5	6.5	N	Change of Parcel G Southern Boundary
Remove	TSB-GJ-02-0-FD_11/19/2007	0	1.5	FD	Change of Parcel G Southern Boundary

Action	Sample ID	Start Depth (ft bgs)	End Depth (ft bgs)	Sample Type	Reason
Remove	TSB-GJ-04-0_11/19/2007	0	1.5	Ν	Already removed during remediation
Remove	TSB-GJ-05-0_11/19/2007	0	1.5	Ν	Change of RZ-A Western Boundary
Remove	TSB-GJ-05-5_11/19/2007	5	6.5	Ν	Change of RZ-A Western Boundary
Remove	TSB-GJ-07-0_11/19/2007	0	1.5	Ν	Change of Parcel G Southern Boundary
Remove	TSB-GJ-07-5_11/19/2007	5	6.5	Ν	Change of Parcel G Southern Boundary
Add	RISB-54-0.5-20141117	0.5	1	Ν	Correction of ECA C9 Boundary
Add	RISB-54-10.0-20141117	10	10.5	Ν	Correction of ECA C9 Boundary
Add	RISB-54-5.0-20141117	5	5.5	Ν	Correction of ECA C9 Boundary
Add	RSAO7-9B	0	1.5	Ν	Correct depth at 0 ft bgs
Add	RSAO8-11.5B	0	1	Ν	Correct depth at 0 ft bgs
Add	SA141009-14B	0.5	2	FD	Correct depth at 0.5 ft bgs
Add	SA141-14B	0.5	2	Ν	Correct depth at 0.5 ft bgs
Add	SA149-32B	4	5.5	Ν	Correct depth at 4 ft bgs
Add	SSAN7-06-0BPC	0	0.5	Ν	Correct depth at 0 ft bgs
Add	SSAN7-06-0.5BPC	0.5	1	Ν	Correct depth at 0.5 ft bgs
Add	SSAO7-04-0BPC	0	0.5	Ν	Correct depth at 0 ft bgs
Add	SSAO7-04-0.5BPC	0.5	1	Ν	Correct depth at 0.5 ft bgs
Add	SSAO7-07-1_01_BPC	0	1	Ν	Correct depth at 0 ft bgs
Add	SSAO7-07-2_01_BPC	1	2	Ν	Correct depth at 1 ft bgs
Add	SSAO7-07-3_01_BPC	2	3	Ν	Correct depth at 2 ft bgs
Add	SSAO7-07-4_01_BPC	3	4	Ν	Correct depth at 3 ft bgs
Add	SSAO7-07-5_01_BPC	4	4	Ν	Correct depth at 4 ft bgs
Add	SSAO8-05-9.5BPC	9.5	10.5	Ν	Correct depth at 9.5 ft bgs
Add	SSAO8-05-9.5BPC_FD	9.5	10.5	FD	Correct depth at 9.5 ft bgs
Add	SSAO8-08-9.5BPC	9.5	10.5	N	Correct depth at 9.5 ft bgs

<u>Notes:</u> bgs = below ground surface ft = feet ECA = Excavation Control Area FD = Field duplicate ID = Identification N = Normal

# TABLE B-2. Summary of Soil Data Excluded During Data Processing Nevada Environmental Response Trust Site Henderson, Nevada

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit		Detection Flag	Qualifier	Reason for Exclusion
TSB-GJ-03	TSB-GJ-03-0_11/19/2007	Ν	C	Bromide	5	5 mg/kg	Ν	1	UJ	Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded
TSB-GJ-03	TSB-GJ-03-5_11/19/2007	N	5	Bromide	5.3	3 mg/kg	Ν	1	UJ	Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded
TSB-GJ-04	TSB-GJ-04-0_11/19/2007	N	0	Bromide	5.4	l mg/kg		N	UJ	Analyzed twice by the same method and not detected. The data with the higher detection limit was exclude:
TSB-GJ-04	TSB-GJ-04-5_11/19/2007	N	5	Bromide	5.5	mg/kg		N	UJ	Analyzed twice by the same method and not detected. The data with the higher detection limit was exclude:
TSB-GR-02	TSB-GR-02-0_11/19/2007			Bromide	5.1	mg/kg		N	UJ	Analyzed twice by the same method and not detected. The data with the higher detection limit was excluder
TSB-GR-02	TSB-GR-02-0-FD_11/19/2007		5	Bromide	5.3	s mg/kg		N		Analyzed wice by the same method and not detected. The data with the higher detection limit was excluder
TSB-G1-02	TSB-G I-03-0 11/19/2007	N		Chloride	818	8 ma/ka	Y	N (	05	Analyzed twice by the same method and hot detected. The data with the ingref detection minit was excluded
TSB-GJ-03	TSB-GJ-03-5 11/19/2007	N	5	Chloride	78.2	2 ma/ka	Y	/		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
TSB-GJ-04	TSB-GJ-04-0_11/19/2007	N	0	Chloride	89.8	3 ma/ka	Y	/		Analyzed wice by the same method and both detected. The data with the lower value was excluded.
TSB-GJ-04	TSB-GJ-04-5 11/19/2007	N	5	Chloride	74.3	3 ma/ka	Y	(		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
TSB-GR-02	TSB-GR-02-0_11/19/2007	N	C	Chloride	1060	) mg/kg	Y	(		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
TSB-GR-02	TSB-GR-02-0-FD_11/19/2007	FD	C	Chloride	863	3 mg/kg	Y	{		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
TSB-GR-02	TSB-GR-02-5_11/19/2007	N	5	Chloride	1230	) mg/kg	Y	(		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
RSAH3	RSAH3-0.5B	Ν	1	Chromium VI	0.4	l mg/kg	Ν	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAH3	RSAH3009-0.5B	FD	1	Chromium VI	0.41	l mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAJ2	RSAJ2-10B	N	2.5	Chromium VI	0.42	2 mg/kg	N	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAJ3	RSAJ3-10B	N	5	Chromium VI	0.43	3 mg/kg	N	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAK4	RSAK4-0.5B	N	0.5		0.4	l mg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAK4	RSAK4009-0.5B	FD	0.5		0.41	i mg/kg		N /		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAKO		IN N	0.5		0.27	mg/kg	Y	r .1	J	Analyzed twice by the same method and both detected. The data with the lower value was excluded.
RSAK0	RSAR0-10B		9.0		0.43	sing/kg		N /		Analyzed wice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL2	RSAL 2-10B	N	9.5		0.44	1 ma/ka		J		Analyzed whice by the same method and both detected. The data with the lower value was excluded.
RSAL4	RSAL4-0.5B	N	0.5	Chromium VI	0.44	l ma/ka		J		Analyzed wice by the same method and not detected at the same detection imit. One data point was excluded.
RSAL4	RSAL4009-0.5B	FD	0.5	Chromium VI	0.42	2 ma/ka	N	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL5	RSAL5-0.5B	N	0.5	Chromium VI	0.43	3 mg/kg	N	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL6	RSAL6-0.5B	N	C	Chromium VI	0.41	l mg/kg	N	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL6	RSAL6-10B	N	9.5	Chromium VI	0.44	l mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL7	RSAL7-0.5B	N	0.5	Chromium VI	0.41	l mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAL8	RSAL8-0.5B	Ν	0.5	Chromium VI	0.4	l mg/kg	Ν	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM2	RSAM2-0.5B	N	0.5	Chromium VI	0.41	l mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM3	RSAM3-0.5B	N	1	Chromium VI	0.4	l mg/kg	N	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM4	RSAM4-0.5B	N	0.5	Chromium VI	0.43	3 mg/kg		l .		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM5	RSAM5-10B	N	1		0.43	3 mg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAIVIO	RSAMO-0.5B	IN N	0.5		0.41	i mg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM7	RSAM7-0.5B	FD	9.6		0.4	ing/kg 2 ma/ka		N J		Analyzed wice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM7	RSAM7-10B	N	9.5	Chromium VI	0.43	3 ma/ka		J		Analyzed wice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM8	RSAM8-0.5B	N	0.0	Chromium VI	0.39	) ma/ka	N	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAM8	RSAM8-10B	N	g	Chromium VI	0.43	3 mg/kg	N	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN2	RSAN2-0.5B	N	1.5	Chromium VI	0.426	6 mg/kg	N	J		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN3	RSAN3-0.5B	N	C	Chromium VI	0.43	3 mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN3	RSAN3-10B	Ν	9.5	Chromium VI	0.43	3 mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN4	RSAN4-0.5B	N	C	Chromium VI	0.44	l mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN4	RSAN4009-10B	FD	9.5	Chromium VI	0.44	i mg/kg	N	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN4	RSAN4-10B	N	9.5		0.44	i mg/kg	N	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN5	RSANS-U.5B		0.5		0.41	mg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
ROAND			9.5		0.44	+ mg/Kg		N J		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAN7	RSANZ-0.5B	N	9.0	Chromium VI	0.43	lma/ka		N J		Analyzed twice by the same method and not detected at the same detection limit. One data point was evoluded.
RSA02	RSA02-0.5B	N	0.5		0.4	R ma/ka		N .		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.
RSA03	RSA03-0.5B	N	0.0	Chromium VI	0.443	l ma/ka	N	- N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded
RSA03	RSAO3-10B	N	8	Chromium VI	0.43	3 mg/kg		- N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSA05	RSAO5-0.5B	N	0.5	Chromium VI	0.41	mg/ka	N	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAO6	RSAO6-0.5B	N	0	Chromium VI	0.4	I mg/kg	N	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAO6	RSAO6-10B	Ν	8.5	Chromium VI	0.43	3 mg/kg	Ν	١		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAO7	RSAO7-9B	N	C	Chromium VI	0.19	) mg/kg	Ν	۱		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAO8	RSAO8-11.5B	N	C	Chromium VI	0.59	) mg/kg	Y	(		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
RSAO8	RSAO8-21.5B	N	6.5	Chromium VI	0.73	3 mg/kg	Y	(		Analyzed twice by the same method and both detected. The data with the lower value was excluded.
RSAP5	RSAP5-0.5B	N	1	Chromium VI	0.42	2 mg/kg	N	1		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAP7	RSAP7-0.5B	N	0.5	Chromium VI	0.43	3 mg/kg	N.	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAQ4	KSAQ4-0.5B	N	1		0.41	mg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
RSAUG	RSAUD-U.5B	IN N			0.43	simg/kg		N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.
NSAQ0	NOAQO-IUD	IN	/		0.43	pring/Kg		N		TAnaiyzeu twice by the same method and not detected at the same detection limit. One data point was excluded.

#### TABLE B-2. Summary of Soil Data Excluded During Data Processing Nevada Environmental Response Trust Site Henderson, Nevada

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reaso
RSAQ7	RSAQ7-0.5B	Ν	0.5	Chromium VI	0.43	3 mg/kg	Ν		Analyzed twice by the same method and not detected at the same detected
RSAS5	RSAS5-0.5B	Ν	1	Chromium VI	0.41	l mg/kg	N		Analyzed twice by the same method and not detected at the same detec
RSAS8	RSAS8-0.5B	N	1.5	Chromium VI	0.43	3 mg/kg	N		Analyzed twice by the same method and not detected at the same detection
SA100	SA100-0.5B	N	1	Chromium VI	0.41	l mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA102	SA102-1.5BR	N	0	Chromium VI	19.6	6 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA102	SA102-10B	N	8.5		4.12	2 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA102	SA102-1BR	N	0		19.9	a mg/kg	Y		Analyzed twice by the same method and both detected. The data with the
SA103	SA103-0.5B		0.5		0.42	2 mg/kg	IN N		Analyzed twice by the same method and not detected at the same detect
SA104	SA104009-10B	N	9		0.42	2 mg/kg	N V	1	Analyzed twice by the same method and hot detected at the same detected and hot detected. The data with the
SA105	SA105009-10B	FD	5	Chromium VI	0.02	1 mg/kg	I N	3	Analyzed twice by the same method and both detected at the same detected
SA105	SA105-10B	N	5	Chromium VI	0.43	3 ma/ka	N		Analyzed twice by the same method and not detected at the same detected
SA106	SA106-12B	N	8.5	Chromium VI	96.3	3 ma/ka	Y		Analyzed twice by the same method and both detected. The data with th
SA108	SA108-20B	N	1	Chromium VI	1.95	5 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA109	SA109-0.5B	Ν	2.5	Chromium VI	1.18	3 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA114	SA114009-1BR	FD	0	Chromium VI	4.12	2 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA114	SA114-1.5BR	Ν	0	Chromium VI	3.89	9 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA114	SA114-10B	Ν	8	Chromium VI	65.1	l mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA114	SA114-1BR	Ν	0	Chromium VI	4.11	l mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA121	SA121-0.5B	Ν	0.5	Chromium VI	1.56	6 mg/kg	Υ	J	Analyzed twice by the same method and both detected. The data with th
SA121	SA121009-0.5B	FD	0.5	Chromium VI	0.42	2 mg/kg	N	UJ	Analyzed twice by the same method and not detected at the same detec
SA122	SA122-0.5B	Ν	4	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA123	SA123-0.5B	N	1.5	Chromium VI	0.33	3 mg/kg	Y	J	Analyzed twice by the same method and both detected. The data with th
SA126	SA126-0.5B	N	1	Chromium VI	0.42	l mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA128	SA128-10B	N	0	Chromium VI	0.4	1 mg/kg	Y	J	Analyzed twice by the same method and both detected. The data with th
SA131	SA131-10B	N	9.5		0.87	/ mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA134	SA134-10B	N	8.5		0.43	3 mg/kg	N		Analyzed twice by the same method and not detected at the same detection of the same det
SA136	SA136-0.5B	N	1.5		0.43	a mg/kg	N		Analyzed twice by the same method and not detected at the same detection
SA138	SA138-0.5B		0.5		0.42	2 mg/kg	IN N		Analyzed twice by the same method and not detected at the same detection
SA141 SA141	SA141009-14B	N	0.5		0.1	mg/kg	N		Analyzed twice by the same method and not detected at the same detect
SA143	SA143-24B	N	0.0	Chromium VI	0.10	1 mg/kg	Y	.1	Analyzed twice by the same method and hot detected at the same detected and hot detected. The data with the
SA144	SA144-0 5B	N	1	Chromium VI	0.5	1 mg/kg	N	3	Analyzed twice by the same method and both detected at the same detected and both detected at the same detected at
SA145	SA145-0.5B	N	0	Chromium VI	0.3	3 ma/ka	Y	J	Analyzed twice by the same method and hot detected at the same detect
SA145	SA145-10B	N	9.5	Chromium VI	0.43	3 ma/ka	N	-	Analyzed twice by the same method and not detected at the same detected
SA149	SA149-32B	N	4	Chromium VI	0.2	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA150	SA150-10B	N	9.5	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA151	SA151-0.5B	Ν	0.5	Chromium VI	0.41	l mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA152	SA152-0.5B	Ν	0	Chromium VI	0.41	l mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA152	SA152009-0.5B	FD	0	Chromium VI	0.4	1 mg/kg	Ν		Analyzed twice by the same method and not detected at the same detec
SA152	SA152-10B	Ν	9.5	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA154	SA154-10B	N	2.5	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA156	SA156-10B	N	9.5	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA157	SA157-0.5B	N	0.5	Chromium VI	0.43	3 mg/kg	N	UJ	Analyzed twice by the same method and not detected at the same detec
SA157	SA157009-0.5B	FD	0.5		1.99	) mg/kg	Y	J	Analyzed twice by the same method and detected at the same value. Or
SA158	SA158-0.5B	N	0		1.5	i mg/kg	Y N		Analyzed twice by the same method and both detected. The data with the
SA158	SA 158-10B	IN NI	9.5		0.43	sing/kg	IN N		Analyzed twice by the same method and not detected at the same detect
SA165	SA165-10B	N	0.5		0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detect
SA100	SA170-0.5B	N	0.0		0.4	mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA171	SA171-5B	N	0	Chromium VI	0.42	2 mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA175	SA175-10B	N	25	Chromium VI	0.42	1 mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA176	SA176-0.5B	N	0	Chromium VI	0.4	l ma/ka	N		Analyzed twice by the same method and not detected at the same detec
SA176	SA176-10B	N	9.5	Chromium VI	0.42	2 ma/ka	N		Analyzed twice by the same method and not detected at the same detected
SA180	SA180-0.5B	N	0	Chromium VI	0.463	3 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA180	SA180-0.5BD	FD	0	Chromium VI	0.645	5 mg/kg	Y		Analyzed twice by the same method and both detected. The data with th
SA180	SA180-10B	Ν	8.5	Chromium VI	0.436	6 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA182	SA182-10B	N	5.5	Chromium VI	0.43	3 mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA185	SA185-0.5B	Ν	0.5	Chromium VI	2.54	1 mg/kg	Υ		Analyzed twice by the same method and both detected. The data with th
SA186	SA186-0.5B	N	0.5	Chromium VI	0.49	9 mg/kg	Y		Analyzed twice by the same method and not detected at the same detec
SA189	SA189-0.5B	N	0.5	Chromium VI	0.4	l mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA197	SA197-0.5B	N	0	Chromium VI	0.43	3 mg/kg	N		Analyzed twice by the same method and not detected at the same detected
SA197	SA197009-10B	FD	9.5	Chromium VI	0.4	I mg/kg	N		Analyzed twice by the same method and not detected at the same detec
SA197	SA197-10B	N	9.5		0.43	s mg/kg	N	l	Analyzed twice by the same method and not detected at the same detec
SA198	SA198-10B	IN	2		0.43	s mg/kg	IN	I	Analyzed twice by the same method and not detected at the same detec

#### n for Exclusion tion limit. One data point was excluded. tion limit. One data point was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. e lower value was excluded. e lower value was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded. e lower value was excluded. e lower value was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded. tion limit. One data point was excluded. tion limit. One data point was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. ne data point was excluded. e lower value was excluded. ction limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded. tion limit. One data point was excluded. e lower value was excluded. tion limit. One data point was excluded.

# TABLE B-2. Summary of Soil Data Excluded During Data Processing Nevada Environmental Response Trust Site Henderson, Nevada

Sample Location	Sample ID	Sample Type	Start Depth (ft)		Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion	
SA200	SA200-0.5B	Ν	1	Chromium VI		0.7	ˈmg/kg	Y		Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA201	SA201-10B	Ν	5.5	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA207	SA207-20B	N	6	Chromium VI		0.485	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA211	SA211-0.5B	N	1	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA212	SA212-0.5B	N	0.5	Chromium VI		0.41	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA214	SA214-0.5B	N	1	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA31	SA31-0.5B	N	0.5			0.42	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA35	SA35-0.5B	N	7	Chromium VI		0.42	mg/kg	N		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.	
SA39	SA39-10B	N	8.5	Chromium VI		0.40	mg/kg	N		Analyzed twice by the same method and not detected at the same detection imit. One data point was excluded.	
SA42	SA42009-10B	FD	5.5	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA42	SA42-10B	N	5.5	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA43	SA43-10B	Ν	5.5	Chromium VI		0.42	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA44	SA44-10B	Ν	5	Chromium VI		0.41	mg/kg	Ν		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA47	SA47-0.5B	N	1	Chromium VI		0.448	mg/kg	Y		Analyzed twice by the same method and detected at the same value. One data point was excluded.	
SA48	SA48-10B	N	7.5	Chromium VI		0.445	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA49	SA49-10B	N	8.5	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA50	SA50-0.5B					0.41	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA51	SA51-10B	N	7	Chromium VI		0.42	mg/kg	N		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.	
SA54	SA54-0 5B	N	15	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA55	SA55-0.5B	N	0.5	Chromium VI		0.4	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA57	SA57-10B	N	5	Chromium VI		0.436	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA57	SA57-10BD	FD	5	Chromium VI		0.432	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA60	SA60-10B	Ν	9.5	Chromium VI		0.53	mg/kg	Y		Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA62	SA62-0.5B	Ν	0.5	Chromium VI		0.41	mg/kg	Ν		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA64	SA64-0.5B	Ν	C	Chromium VI		1.29	mg/kg	Y		Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA64	SA64-10B	N	9.5	Chromium VI		0.37	mg/kg	Y	J	Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA65	SA65-10B	N	C	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA65	SA65-20B	N	8.5	Chromium VI		0.28	mg/kg	Y	J	Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA67	SA67-0.5B	N	0.5			0.413	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA69 SA70	SA09-0.5B	N	0.5	Chromium VI		0.44	mg/kg	N		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.	
SA71	SA71-0.5B	N	0.5	Chromium VI		0.42	mg/kg	Y		Analyzed twice by the same method and hot detected at the same detection innt. One data point was excluded.	
SA73	SA73-0.5B	N	0.0	Chromium VI		0.43	ma/ka	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA73	SA73-10B	N	9.5	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA74	SA74-0.5B	N	C	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA74	SA74009-0.5B	FD	C	Chromium VI		0.43	mg/kg	Ν		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA74	SA74-10B	Ν	9.5	Chromium VI		0.43	mg/kg	Ν		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA75	SA75-0.5B	N	C	Chromium VI		0.42	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA75	SA75-10B	N	9.5	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA76	SA76-10B	N	4			0.42	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA77 SA82	SA77-0.3D SA82-0 5B	N	1	Chromium VI		0.43	mg/kg	N		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.	
SA85	SA85-0.5B	N	0.5	Chromium VI		0.41	mg/kg	N		Analyzed whice by the same method and not detected at the same detection limit. One data point was excluded.	
SA86	SA86009-10B	FD	5.5	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA86	SA86-10B	N	5.5	Chromium VI		0.44	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA87	SA87-0.5B	N	0.5	Chromium VI		0.427	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA88	SA88-10B	Ν	7	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
SA92	SA92-0.5B	Ν	C	Chromium VI		0.22	mg/kg	Y	J	Analyzed twice by the same method and both detected. The data with the lower value was excluded.	
SA92	SA92-10B	Ν	7.5	Chromium VI		0.43	mg/kg	N		Analyzed twice by the same method and not detected at the same detection limit. One data point was excluded.	
DS-C18-1	DS-C18-1	N	5	2,3,7,8-Tetrac	nlorodibenzo-p-dioxin	0.000014	mg/kg	N		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.	
DS-C18-1	DS-C18-1	N	5	1,2,3,7,8,9-He	xachlorodibenzo-p-dioxin	0.000054	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1,2,3,4,6,7,8-F	leptachlorodibenzo-p-dioxin	0.00004	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	0 1,2,3,4,7,8,9-F 1 2 3 6 7 8-⊔≏	reprachioroubenzoruran	0.000055	ing/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1 2 3 6 7 8-Ho	xachlorodibenzo-n-diovin	0.000035	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1.2.3.4.7 8-He	xachlorodibenzo-p-dioxin	0.00001	ma/ka	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1.2.3.4.5.6.7.8	-Octachlorodibenzofuran	0,00006	ma/ka	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1,2,3,7,8-Pent	achlorodibenzofuran	0.000028	mg/kg	N	1	Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	Ν	5	1,2,3,4,7,8-He	xachlorodibenzofuran	0.000068	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	2,3,4,6,7,8-He	xachlorodibenzofuran	0.000038	mg/kg	Ν		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.	
DS-C18-1	DS-C18-1	Ν	5	2,3,4,7,8-Pent	achlorodibenzofuran	0.000027	′ mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1,2,3,4,5,6,7,8	-Octachlorodibenzo-p-dioxin	0.000073	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
DS-C18-1	DS-C18-1	N	5	1,2,3,4,6,7,8-	leptachlorodibenzofuran	0.000039	mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.	
Sample Location	Sample ID	Sample Start Depth Type (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	fier Reason for Exclusion			
--------------------	---------------------	---------------------------------	--	----------	-------	-------------------	-----------	---	--	--	--
DS-C18-1	DS-C18-1	N 5	2,3,7,8-Tetrachlorodibenzofuran	0.000024	mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-1	DS-C18-1	N 5	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	0.000065	mg/kg	Ν		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.			
DS-C18-1	DS-C18-1	N 5	1,2,3,7,8,9-Hexachlorodibenzofuran	0.00006	mg/kg	Ν		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.000059	mg/kg	Ν		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.			
DS-C18-2	DS-C18-2	N 6.5	2,3,7,8-Tetrachlorodibenzo-p-dioxin	0.000017	mg/kg	N		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.000043	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,4,5,6,7,8-Octachlorodibenzoturan	0.00007	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.000067	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C10-2	DS-C10-2	N 0.3	1,2,3,4,7,0-Rexachiorodibenzofuran	0.00012	mg/kg	IN N		Analyzed twice by the same method and not detected, and the other net detected. The particulation limit was excluded.			
DS-C18-2	DS-C18-2	N 65	1,2,3,0,7,8 9-Hentachlorodibenzofuran	0.000043	mg/kg	N		Analyzed wice by the same method, one detected, and the other not detected. The hondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,4,5,6,7,8-Octachlorodibenzo-n-dioxir	0.000035	ma/ka	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1.2.3.4.6.7.8-Heptachlorodibenzofuran	0.000038	ma/ka	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	2,3,4,7,8-Pentachlorodibenzofuran	0.000044	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	2,3,4,6,7,8-Hexachlorodibenzofuran	0.000048	mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,7,8-Pentachlorodibenzofuran	0.000044	mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,4,7,8-Hexachlorodibenzofuran	0.000085	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	0.00009	mg/kg	Ν		Analyzed twice by the same method and not detected. The data with the higher detection limit was excluded.			
DS-C18-2	DS-C18-2	N 6.5	1,2,3,7,8,9-Hexachlorodibenzofuran	0.000075	mg/kg	Ν		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
DS-C18-2	DS-C18-2	N 6.5	2,3,7,8-Tetrachlorodibenzofuran	0.000026	mg/kg	N		Analyzed twice by the same method, one detected, and the other not detected. The nondetect data was excluded.			
SSAI2-01	SSAI2-01-1BPC	N 1	Hexachlorobenzene	0.0084	mg/kg	N		Analyzed twice by EPA Method 8081, one detected, and the other not detected. The nondetect data was excluded.			
SSAM3-01	SSAM3-01-2BPC		Hexachlorobenzene	0.84	mg/kg	Y		Analyzed twice by EPA Method 8081 and both detected. The data with the lower value was excluded			
SSAM5-03	SSAM5-03-5BPC	N 0.5	Hexachlorobenzene	0.094	mg/kg	Y	J+	Analyzed twice by EPA Method 8081 and both detected. The data with the lower value was exclude:			
SSAM5-03	SSAM5-03-7BPC	N 2.5	Hexachlorobenzene	0.23	mg/kg	Y V		Analyzed wice by EPA Method 8081 and both detected. The data with the lower value was exclude:			
SSAMD-03	SSAND-03-96FC	N 4.5	Hexachlorobenzene	0.19	mg/kg	T N	11	Analyzed wice by EPA Method 8270 and 8270 and 8270 and 8270 was excluded			
SA00	SA0-0.5 SA21-0 5	N OF	Hexachlorobenzene	0.33	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim: The data from EPA Method 8270 was excluder			
SA21	SA21-0.5	N 0.5	Fluoranthene	0.34	ma/ka	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EFA Method 8270 was excluder			
SA21	SA21-0.5	N 0.5	Acenaphthylene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA05	SA5-0.5	N 1	Hexachlorobenzene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA05	SA5-0.5	N 1	Fluoranthene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA05	SA5-0.5	N 1	Acenaphthylene	0.43	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA06	SA6-0.5D	FD 1	Hexachlorobenzene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA06	SA6-0.5	N 1	Hexachlorobenzene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA08	SA8-0.5	N 1	Fluoranthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA06	SA6-0.5	N 1	Fluoranthene	0.35	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:			
SAU6	SA6-0.5D	FD 1	Fluoranthene	0.38	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:			
SAU6	SA6-0.5	IN I	Anthracene	0.35	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:			
SA06	SA6-0 5	N 1	Anthracene	0.35	ma/ka	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 GML The data from EFA Method 8270 was excluder			
SA06	SA6-0.5	N 1	Benzo(g.h.i)pervlene	0.35	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder			
SA06	SA6-0.5D	FD 1	Benzo(g,h,i)perylene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA05	SA5-0.5	N 1	Anthracene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA21	SA21-0.5	N 0.5	Anthracene	0.34	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA21	SA21-0.5	N 0.5	Benzo(g,h,i)perylene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA07	SA7-0.5	N 0.5	Hexachlorobenzene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA07	SA7-0.5	N 0.5	Fluoranthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA07	SA7-0.5	N 0.5	Acenaphthylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA07	SA7-0.5	N 0.5	Anthracene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA07	SA7-0.5	N 0.5	Benzo(g,h,i)perylene	0.35	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:			
SAU5	SA5-0.5	N 1	Benzo(g,n,i)perylene	0.43	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:			
SAU5	SA5-0.5		Pyrene Hoveeblerebenzene	0.43	mg/kg	N N		Analyzed by both EPA Mietnod 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:			
SA03	SA3-0.5D	FD 1.0	Hexachlorobenzene	0.35	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder.			
SA03	SA3-0.5D	FD 1.5	Fluoranthene	0.35	ma/ka	N	<u>U</u>	Analyzed by both ETA Method 8270 and 8270 GMC. The data from ETA Method 8270 was excluder			
SA03	SA3-0.5	N 1.5	Fluoranthene	0.35	ma/ka	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EFA Method 8270 was excluder			
SA03	SA3-0.5D	FD 1.5	Acenaphthylene	0.35	mg/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA03	SA3-0.5	N 1.5	Acenaphthylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA03	SA3-0.5D	FD 1.5	Anthracene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA03	SA3-0.5	N 1.5	Anthracene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA03	SA3-0.5D	FD 1.5	Benzo(g,h,i)perylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA03	SA3-0.5	N 1.5	Benzo(g,h,i)perylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA21	SA21-0.5	N 0.5	Pyrene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
SA06	SA6-0.5	N 1	Acenaphthylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			
5A06	5A6-U.5D	רט 1	Acenaphthylene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded			

Sample Location	Sample ID	Sample Start Depth Type (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	lifier Reason for Exclusion		
SA08	SA8-0.5	N 1	Acenaphthylene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	Pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5D	FD 1	Pyrene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5	N 1	Pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5D	FD 1.5	Pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:		
SA03	SA3-0.5	N 1.5	Pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:		
SA07	SA7-0.5	N 0.5	2-Methylpaphthalone	0.35	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA21	SA21-0.5	N 0.5	2-Methylnaphthalene	0.35	ma/ka	N	U	Analyzed by both EFA Method 8270 and 8270 GMC. The data from EFA Method 8270 was excluder		
SA21	SA21-0.5	N 0.5	Acenaphthene	0.34	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA21	SA21-0.5	N 0.5	Phenanthrene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5D	FD 1	2-Methylnaphthalene	0.38	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	2-Methylnaphthalene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5	N 1	2-Methylnaphthalene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	Benzo(g,h,i)perylene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	Acenaphthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:		
SAU6	SA6-0.5	N 1	Acenaphthene	0.35	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:		
SA06	SA6-0.5	N 1	Phenanthrene	0.35	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder.		
SA06	SA6-0.5D	FD 1	Phenanthrene	0.33	ma/ka	N	U	Analyzed by both EFA Method 8270 and 8270 GMC. The data from EFA Method 8270 was excluder		
SA06	SA6-0.5D	FD 1	Acenaphthene	0.38	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5	N 1	Fluorene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	Fluorene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5D	FD 1	Naphthalene	0.38	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5	N 1	Naphthalene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA06	SA6-0.5D	FD 1	Fluorene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA08	SA8-0.5	N 1	Naphthalene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA21	SA21-0.5	N 0.5	Fluorene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:		
SA21	SA21-0.5	N 0.5	Naphthalene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:		
SA05	SA5-0.5	N I		0.43	mg/kg	N		Analyzed by both EPA Mietnod 8270 and 8270 SIM. The data from EPA Mietnod 8270 was exclude:		
SA05	SA5-0.5	N 1	Phenanthrene	0.43	ma/ka	N	U	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EFA Method 8270 was excluder		
SA05	SA5-0.5	N 1	Fluorene	0.43	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA05	SA5-0.5	N 1	Naphthalene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5	N 1.5	2-Methylnaphthalene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5D	FD 1.5	2-Methylnaphthalene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5D	FD 1.5	Acenaphthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5	N 1.5	Acenaphthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA03	SA3-0.5	N 1.5	Phenanthrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:		
SAU3	SA3-0.5D	FD 1.5	Phenanthrene	0.35	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:		
SA03	SA3-0.5D	FD 1.3	Fluorene	0.35	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder.		
SA03	SA3-0.5D	FD 1.5	Naphthalene	0.35	ma/ka	N	U	Analyzed by both EFA Method 8270 and 8270 GMC. The data from EFA Method 8270 was excluder		
SA03	SA3-0.5	N 1.5	Naphthalene	0.35	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA07	SA7-0.5	N 0.5	Acenaphthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA07	SA7-0.5	N 0.5	Phenanthrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA07	SA7-0.5	N 0.5	Fluorene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA07	SA7-0.5	N 0.5	Naphthalene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded		
SA21	SA21-0.5	N 0.5	Uranium-234	1.28	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA21	SA21-0.5	N 0.5	Thorium-228	1.9	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA21	SA21-0.5	N 0.5	I horium-230	1.19	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A 01-R. The data from DOE EML HASL 300 A-01-R was exclude:		
SA21	SA21-0.5	N 0.5	Uranium-235	0.279	pci/g	N	L.	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-K. The data from DOE EML HASL 300 A-01-K was excluded		
SA15 SA15	SA15-10		Uranium-234	1.41	pci/g	T V	J+  +	Analyzed by both DOE EWE HASE-300 MOD and DOE EWE HASE 300 A-01-R. The data from DOE EWE HASE 300 A-01-R was excluder		
SA18	SA18-0.5	N C	Uranium-234	1.31	pci/g	Y	01	Analyzed by both DOE EME HASE 300 MOD and DOE EME HASE 300 A-01-R. The data from DOE EME HASE 300 A-01-R was excluder		
SA18	SA18-0.5D	FD C	Uranium-234	1.14	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA15	SA15-10	N C	Thorium-228	1.47	pci/g	Y	J+	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA15	SA15-10D	FD C	Thorium-228	1.63	pci/g	Υ	J+	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA18	SA18-0.5	N C	Thorium-228	1.98	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA18	SA18-0.5D	FD C	Thorium-228	1.88	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA18	SA18-0.5	N C	Thorium-230	1.23	pci/g	Υ		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA18	SA18-0.5D	FD C	Thorium-230	1.09	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA15	SA15-10		I norium-230	1.23	pci/g	Y	J+	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded		
SA15	SA 15-10D SA 15-10		I HUHUIII-230	1.26	pci/g	r N	J+	Analyzed by both DOE EMIL FIASE-300 MOD and DOE EMIL HASE 300 A-01 P. The data from DOE EMIL HASE 300 A-01 P. Was excluded		
5715	0/10-10		Graniun-200	0.207	ru"9	· N	5	principated by sour DOE LIVE THOUSOU WOD and DOE LIVE THOUSOUR ATOTAL THE data from DOE EWIL THOUSOUR OF A BUILDER		

Sample Location	Sam	ple ID San Ty	mple /pe	Start Depth (ft)		Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
SA21	SA21-0.5	Ν		0.5	Uranium-238		1.23	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA21	SA21-0.5	N		0.5	Thorium-232		2	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA15	SA15-10D	FD		0	Uranium-235		0.287	pci/g	N	0	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA18	SA18-0.5D	FD		0	Uranium-235		0.293		N		Analyzed by both DOE EMIC HASE-300 MOD and DOE EMIC HASE 300 A-01-R. The data from DOE EMIC HASE 300 A-01-R was excluder
SA15	SA15-10	N		0	Uranium-238		1.77	i pci/g	N	U	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA15	SA15-10D	FD		0	Uranium-238		2.06	pci/g	Y	-	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA18	SA18-0.5D	FD		0	Uranium-238		1.36	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA18	SA18-0.5	N		0	Uranium-238		2.82	pci/g	N		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA15	SA15-10	N		0	Thorium-232		2.02	pci/g	Y	J+	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA18	SA18-0.5	N		0	Thorium 232		1.83	pci/g	Y	1.	Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA15 SA18	SA15-10D SA18-0.5D	FD FD		0	Thorium-232		2.03		T V	J+	
SA11	SA10-0.5D	N N		8	Uranium-234		2.00	pci/g	Y		Analyzed by both DOE EME HASE 300 MOD and DOE EME HASE 300 A-01-R. The data from DOE EME HASE 300 A-01-R was excluded
SA11	SA11-10	N		8	Thorium-230		1.79	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA11	SA11-10	N		8	Uranium-235		0.302	pci/g	N		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA11	SA11-10	N		8	Thorium-228		2.24	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA11	SA11-10	N		8	Uranium-238		2.04	pci/g	N		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
SA11	SA11-10	N		8	Thorium-232		2.19	pci/g	Y		Analyzed by both DOE EML HASL-300 MOD and DOE EML HASL 300 A-01-R. The data from DOE EML HASL 300 A-01-R was excluded
CS-C10B-1	CS-C10B-1	N		0	Hexachloroben	zene	0.17	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-0.5-201	41217 N		0.5	Fluoranthene		0.073	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-11 RISB-14	RISB-11-5.0-201	41217 N 41216 N		5	Fluoranthene		0.076	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder
RISB-10	RISB-10-0.5-201	41210 N		0.5	Fluoranthene		0.07	mg/kg ma/ka	N	<u>U</u>	Analyzed by both ETA Method 8270 and 8270 SM. The data from ETA Method 8270 was excluder
RISB-47	RISB-47-0.5-201	41120 N		0.5	Fluoranthene		0.071	ma/ka	N	<u>U</u>	Analyzed by both ErA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-201	41120 N		5	Fluoranthene		0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-201	41209 N		5	Fluoranthene		0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-201	41211 N		5	Fluoranthene		0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-201	41211 N		0.5	Fluoranthene		0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-201	41218 N		5	Fluoranthene		0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-201	41216 N		0.5	Fluoranthene		0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-13 PISB-10	RISB-13-0.5-201	41217 N 41215 N		0.5	Fluoranthene		0.074	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
CS-C10B-1	CS-C10B-1	41215 N		5	Fluoranthene		0.075	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
M-162D	M-162D-0.5-201	41209 N		0.5	Fluoranthene		0.072	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
RISB-12	RISB-12-2.5-201	41216 N		2.5	Fluoranthene		3.9	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-201	41120 N		5	Anthracene		0.085	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-0.5-201	41120 N		0.5	Anthracene		0.081	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-201	41215 N		0.5	Anthracene		0.08	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-201	41215 N		5	Anthracene		0.085	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
CS-C10B-1	CS-C10B-1	11200 N		0	Anthracene		0.017	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-162D	M-162D-0.5-201	41209 N 41209 N		0.5	Anthracene		0.080	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
RISB-11	RISB-11-0.5-201	41203 N		0.5	Anthracene		0.084	ma/ka	N	<u>U</u>	Analyzed by both ETA Method 8270 and 8270 SM. The data from ETA Method 8270 was excluder
RISB-11	RISB-11-5.0-201	41217 N		5.5	Anthracene		0.087	mg/kg	N	<u> </u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-0.5-201	41217 N	1	0.5	Acenaphthylene	е	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-5.0-201	41217 N		5	Acenaphthylene	e	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-0.5-201	41120 N		0.5	Acenaphthylene	9	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-201	41120 N		5	Acenaphthylene	e	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-201	41120 N		5	Pyrene	- domo	0.085	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-0.5-201	41120 N		0.5	Benzo(g,h,i)per	ylene	0.11	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-47 PISB-13	RISB-47-5.0-201	41120 N 41217 N		5 0.5	Apthracopo	ylene	0.12	mg/kg	N N		Analyzed by both EPA Miethod 8270 and 8270 SIM. The data from EPA Miethod 8270 was exclude:
RISB-13	RISB-13-5 0-201	41217 N		0.5	Anthracene		0.083	mg/kg	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 SMC. The data from EFA Method 8270 was excluder
RISB-14	RISB-14-0.5-201	41216 N		0.5	Anthracene		0.083	mg/kg ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
RISB-14	RISB-14-5.0-201	41216 N		5.0	Anthracene		0.088	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-201	41218 N	1	5	Acenaphthylene	Э	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-201	41216 N		0.5	Acenaphthylene	9	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-201	41217 N		0.5	Acenaphthylene	9	0.074	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-201	41211 N		5	Anthracene		0.083	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-201	41211 N		0.5	Anthracene		0.081	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	KISB-09-5.0-201	41211 N		5	Acenaphthylene	3	0.072	mg/kg	IN N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-201	41211 N 41211 N		0.5	Acenaphthylene	5	0.071	ing/kg ma/ka	N	<u> </u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-11	RISB-11-0 5-201	41217 N		0.5	Pyrene		0.003	ma/ka	N	<u> </u>	Analyzed by both ELA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
				5.5			0.004	····ə···ə	1	-	

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
RISB-11	RISB-11-5.0-20141217	Ν	Ę	5 Benzo(g,h,i)perylene	0.12	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5	5 Pyrene	0.08	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	Ę	5 Pyrene	0.085	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-C10B-1 PISB-10	CS-C10B-1 PISB-10-0 5-20141215	N	0.6	J Benzo(g,n,i)perviene	0.016	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-10	RISB-10-5.0-20141215	N	0.0	5 Benzo(g, h, i)perviene	0.12	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	5	5 Acenaphthylene	0.076	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	ŧ	5 Pyrene	0.086	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	Ν	0.5	5 Pyrene	0.083	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N	0.5	5 Benzo(g,h,i)perylene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	Ę	5 Benzo(g,h,i)perylene	0.12	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-20141217	N	0.5	b Pyrene	0.085	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-13 RISB-14	RISB-13-5.0-20141218 RISB-14-0 5-20141216	N	0.5	5 Benzo(g,n,i)perviene	0.12	mg/kg mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	0.0	5 Benzo(g,h,i)perviene	0.12	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-20141217	N	0.5	5 Benzo(g,h,i)perylene	0.12	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5	5 Acenaphthylene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	Ν	Ę	5 Acenaphthylene	0.075	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-0.5-20141215	N	0.5	5 Fluoranthene	0.072	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-0.5-20141217	N	0.5	5 Benzo(g,h,i)perylene	0.12	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-0.5-20141203	N	0.5	5 Fluoranthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-5.0-20141203	N	5		0.076	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-161D	M-161D-0.5-20141203		0.6		0.067	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
CS-C10B-1	CS-C10B-1	N	0.0	Aranacene	0.001	ma/ka	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 GMC. The data from EFA Method 8270 was excluder
RISB-09	RISB-09-0.5-20141211	N	0.5	5 Pyrene	0.081	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	N	Ę	5 Benzo(g,h,i)perylene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N	0.5	5 Benzo(g,h,i)perylene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	Ν	0.5	5 Acenaphthylene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	Ę	5 Acenaphthylene	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-0.5-20141215	N	0.5	5 Anthracene	0.082	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-2.5-20141216	N	2.5	5 Anthracene	4.5	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-161D DISB-12	M-161D-5.0-20141203	N	24	Acenaphthylene	0.076	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-12	RISB-12-2.5-20141210	N	2.0	5 Acenaphthylene	0.072	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder
M-161D	M-161D-0.5-20141203	N	0.5	5 Acenaphthylene	0.072	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-161D	M-161D-5.0-20141203	N	Ę	5 Pyrene	0.087	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-0.5-20141203	N	0.5	5 Pyrene	0.081	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-C10B-1	CS-C10B-1	N	(	0 2-Methylnaphthalene	0.019	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-C10B-1	CS-C10B-1	N	(	0 Pyrene	0.056	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-0.5-20141215	N	0.5		0.082	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-12 PISB-12	RISB-12-2.5-20141216		2.3	5 Benzo(a hi)pen/lene	4.5	mg/kg	N N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-12	RISB-12-0.5-20141215	N	2.0	5 Benzo(g,n,n)perviene	0.1	mg/kg mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SMC. The data from EPA Method 8270 was excluder
M-161D	M-161D-0.5-20141203	N	0.5	5 Benzo(g,h,i)pervlene	0.11	ma/ka	N	<u>U</u>	Analyzed by both Er A method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-161D	M-161D-5.0-20141203	N	Ę	5 Benzo(g,h,i)perylene	0.12	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-0.5-20141120	N	0.5	5 Pyrene	0.081	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-5.0-20141217	Ν	Ę	5 Pyrene	0.087	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-20141218	N	Ę	5 Pyrene	0.088	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N	0.5	5 Pyrene	0.083	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	Ę		0.088	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
NI-162D	N-162D-5.0-20141209		0.6		0.072	mg/kg	N N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-47	RISB-47-0 5-20141120	N	0.0	5 Acenaphthene	0.07	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SMC. The data from EPA Method 8270 was excluder
CS-C10B-1	CS-C10B-1	N	0.0	0 Acenaphthene	0.000	ma/ka	N	<u>U</u>	Analyzed by both Er A method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-10	RISB-10-0.5-20141215	N	0.5	5 Acenaphthene	0.067	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	ŧ	Acenaphthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	Ν	0.5	5 Acenaphthene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-5.0-20141217	Ν		5 Acenaphthene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-20141120	N	5	5 Acenaphthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-5.0-20141120	N		5 Phenanthrene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	KISB-47-0.5-20141120	N	0.5	Phenanthrene	0.068	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-C10B-1 RISB-10	RISB-10-0 5-201/1215	N	04	5 Phenanthrene	0.017	mg/Kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-10	RISB-10-5 0-20141215	N	0.8	5 Phenanthrene	0.007	ma/ka	N	<u>.</u>	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
		1			0.071	···	1	-	

Sample Location	Sample ID	Sample Start Depth Type (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
RISB-13	RISB-13-0.5-20141217	N 0.5	5 Acenaphthene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-20141218	N 5	Acenaphthene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N 0.5	Acenaphthene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-14 M-162D	RISB-14-5.0-20141216 M-162D-0 5-20141209	N 5	Acenaphthene	0.073	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
M-162D	M-162D-5.0-20141209	N 5	5 Phenanthrene	0.003	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-5.0-20141217	N 5	5 Phenanthrene	0.073	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-0.5-20141217	N 0.5	5 Phenanthrene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N 0.5	5 Phenanthrene	0.07	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-20141217	N 0.5	5 Phenanthrene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N 5	5 Phenanthrene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-20141218	N 5		0.073	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-09	RISB-09-0.5-20141211	N 05		0.069	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N 0.5	5 Phenanthrene	0.068	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	N 5	Phenanthrene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-5.0-20141203	N 5	5 Acenaphthene	0.073	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-0.5-20141203	N 0.5	5 Acenaphthene	0.068	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-2.5-20141216	N 2.5	Acenaphthene	3.7	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-0.5-20141215	N 0.5	5 Acenaphthene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-0.5-20141203	N 0.5	Phenanthrene	0.068	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-12	RISB-12-2.5-20141216	N 2.5		3.7	mg/kg	N		Analyzed by both EPA Method 8270 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
M-161D	M-161D-5 0-20141213	N U.C	2 Phenanthrene	0.009	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder
CS-C10B-1	CS-C10B-1	N C	) Fluorene	0.018	mg/kg mg/kg	N	<u>U</u>	Analyzed by both Er A method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-10	RISB-10-5.0-20141215	N 5	5 Fluorene	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N 0.5	Fluorene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N 5	5 Fluorene	0.076	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N 0.5	5 Fluorene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-20141217	N 0.5	5 Fluorene	0.074	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-47	RISB-47-0.5-20141120	N 0.5		0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-47	RISB-47-5.0-20141120	N 5		0.075	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11 DISB-11	RISB-11-0.5-20141217 RISB-11-5 0-20141217	N 0.3		0.073	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-09	RISB-09-5.0-20141211	N 5	Fluorene	0.072	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EFA Method 8270 was excluder
RISB-09	RISB-09-0.5-20141211	N 0.5	5 Fluorene	0.071	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-20141218	N 5	Fluorene	0.077	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N 0.5	5 Fluorene	0.073	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N 5	Fluorene	0.077	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-11	RISB-11-0.5-20141217	N 0.5	5 Naphthalene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-C10B-1	CS-C10B-1	N (	) Naphthalene	0.031	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-47 DISB-47	RISB-47-0.5-20141120 RISB-47-5 0-20141120	N 0.3		0.068	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-11	RISB-11-5 0-20141217	N 5	Naphthalene	0.071	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SMC. The data from EPA Method 8270 was excluder
RISB-10	RISB-10-5.0-20141215	N 5	5 Naphthalene	0.071	mg/kg mg/kg	N	<u>U</u>	Analyzed by both Er A method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-10	RISB-10-0.5-20141215	N 0.5	Naphthalene	0.067	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-5.0-20141218	N 5	Naphthalene	0.073	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N 5	5 Naphthalene	0.073	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-13	RISB-13-0.5-20141217	N 0.5	5 Naphthalene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N 0.5	5 Naphthalene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N 5	Naphthalene	0.072	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
M-162D	M-162D-0.5-20141209	N 0.5		0.069	mg/kg	N		Analyzed by both EPA Method 8270 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-12	RISB-12-2 5-20141205	N 0.0		0.071	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 Sim. The data from EPA Method 8270 was excluder
RISB-12	RISB-12-0.5-20141215	N 0.5	Fluorene	0.072	mg/kg mg/kg	N	<u>U</u>	Analyzed by both EFA Method 8270 and 8270 SIM. The data from EFA Method 8270 was excluder
M-161D	M-161D-5.0-20141203	N 5	5 Fluorene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-5.0-20141203	N 5	Naphthalene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-0.5-20141215	N 0.5	Naphthalene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-12	RISB-12-2.5-20141216	N 2.5	5 Naphthalene	3.7	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-161D	M-161D-0.5-20141203	N 0.5	Naphthalene	0.068	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	N 5	Naphthalene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
KISB-09	KISB-09-0.5-20141211	IN 0.5		0.068	mg/kg	N N		Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was excluded
SATT SA09	SATI-0.5 SA9-10D	ED F	Nanhthalene	0.0054	mg/kg	N	03	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA21	SA21-0.5	N 0.5	5 Naphthalene	0.0052	mg/ka	N	U U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded
1		1 1 010						· · · · · · · · · · · · · · · · · · ·

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	er Reason for Exclusion				
SA05	SA5-0.5	Ν	1	Naphthalene	0.0065	5 mg/kg	Ν	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
SA11	SA11-0.5D	FD	(	) Naphthalene	0.0057	′ mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA11	SA11-10	N	3	Naphthalene	0.0054	l mg/kg	N	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA16	SA16-10	N	8	3 Naphthalene	0.0056	8 mg/kg	N	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA16	SA16-0.5	N	(	Naphthalene	0.0053	3 mg/kg	N	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA08	SA8-0.5	N	1	Naphthalene	0.00079	) mg/kg	Y	J	Analyzed by both EPA Method 8260 and 82/0 SIM. The data from EPA Method 8260 was excluded				
SA18	SA18-10		9.5	Naphthalene	0.0054	mg/kg	N	0	Analyzed by both EPA Method 8260 and 82/0. The data from EPA Method 8260 was excluded				
SA10	SA16-0.5D				0.0053	a mg/kg	N	<u>U</u>	Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded				
SA18	SA0-0.3D SA18-0 5	N	(	Naphthalene	0.0055	5 mg/kg	N		Analyzed by both EPA Method 8260 and 8270 Smit. The data from EPA Method 8260 was excluder.				
SA15	SA15-10D	FD		Naphthalene	0.0055	5 ma/ka	N	<u>U</u>	Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluder				
SA15	SA15-20	N		Naphthalene	0.0055	5 ma/ka	N	<u> </u>	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluder				
SA06	SA6-0.5	N		Naphthalene	0.0053	3 ma/ka	N	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
SA15	SA15-10	N	(	Naphthalene	0.0059	) mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA12	SA12-10	N	8	Naphthalene	0.0054	l mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA12	SA12-0.5	N	(	Naphthalene	0.0056	6 mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded				
SA03	SA3-0.5	Ν	1.5	5 Naphthalene	0.0053	8 mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
SA03	SA3-0.5D	FD	1.5	Naphthalene	0.0053	8 mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
CS-C10B-1	CS-C10B-1	N	(	Naphthalene	0.00063	8 mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
M-161D	M-161D-5.0-20141203	N	5	Naphthalene	0.001	mg/kg	N	U	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
KISB-12	RISB-12-2.5-20141216	N	2.5		0.0011	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was excluded				
M-161D	M-161D-0.5-20141203	N	0.5	Naphthalene	0.00095	mg/kg	N	0	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was exclude:				
M-162D	M-162D-5.0-20141209	N		Naphthalene	0.0011	mg/kg	N	0	Analyzed by both EPA Method 8260 and 8270 SIM. The data from EPA Method 8260 was exclude:				
M-162D SA07	M-162D-0.5-20141209	N	0.8		0.00099	a mg/kg	N	<u>U</u>	Analyzed by both EPA Method 9260 and 9270 SIM. The data from EPA Method 9260 was excluder				
SAU	SA9-10	N	0.0	Naphthalene	0.0000	Sma/ka	N		Analyzed by both EFA Method 8260 and 8270 Stim. The data from EFA Method 8260 was excluder				
SSAK8-02	SSAK8-02-1BPC	N	0.5	Hexachlorobenzene	0.12	2 ma/ka	Y	<u>J</u>	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluder				
SSA07-02	SSA07-02-1BPC	N	0.5	Hexachlorobenzene	0.03	3 ma/ka	N	•	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SA92	SA92-10B	N	7.5	Hexachlorobenzene	0.0051	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAL8	RSAL8-0.5B	N	0.5	Hexachlorobenzene	0.38	B mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SA182	SA182-10B	N	5.5	Hexachlorobenzene	0.0073	3 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM5	RSAM5-10B	Ν	1	I Hexachlorobenzene	0.011	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SA129	SA129-3BPC	N	6	Hexachlorobenzene	0.41	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SSAM3-01	SSAM3-01-2BPC	N	(	Hexachlorobenzene	0.83	8 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SSAL3-03	SSAL3-03-1BPC	N	1	Hexachlorobenzene	0.032	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RISB-47	RISB-47-5.0-20141120	N	5	Naphthalene	0.001	mg/kg	N	<u>U</u>	Analyzed by both EPA Method 8260 and 82/0 SIM. The data from EPA Method 8260 was exclude:				
RISB-47	RISB-47-0.5-20141120	N	0.5		0.00094	mg/kg	N	U	Analyzed by both EPA Method 8250 and 8270 SIM. The data from EPA Method 8260 was exclude:				
RSAM4	RSAM4-0.5B	N	0.5		0.058	a mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAL6	RSAU 6-10B	N	9 6		0.023	8 ma/ka	N		Analyzed by both EFA Method 8270 and 8001. The data from EFA Method 8270 was excluder				
RSAL6	RSAI 6-0.5B	N	0.0	Hexachlorobenzene	0.0067	/ ma/ka	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluder				
RSAN3	RSAN3-10B	N	9.5	Hexachlorobenzene	0.0073	3 ma/ka	N	•	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAN3	RSAN3-0.5B	N	(	) Hexachlorobenzene	0.0018	3 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SSAO6-04	SSAO6-04-5BPC	N	5	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SSAN7-03	SSAN7-03-1BPC	N	1.5	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAN4	RSAN4-10B	Ν	9.5	Hexachlorobenzene	0.0073	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAN4	RSAN4-0.5B	Ν	(	) Hexachlorobenzene	0.0033	3 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAL5	RSAL5-0.5B	N	0.5	Hexachlorobenzene	0.013	8 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAJ3	RSAJ3-10B	N	5	Hexachlorobenzene	0.3	8 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM7	RSAM7-10B	N	9.5	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAK6	RSAK6-10B	N	9.5	Hexachlorobenzene	0.0073	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAK8	RSAK8-10B	N	1		0.2	2 mg/kg	Ŷ		Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 was excluded				
SA180	SA180-10B	N	8.5		0.0072	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
ROALI SSAK7-03	KSAL7-0.3D	N	0.8		0.006	a mg/kg	N	1	Analyzed by both EPA Method 8270 and 9001. The data from EPA Method 8270 was excluded				
SA86	SA86-10B	N	5.6	Hexachlorobenzene	0.000	ama/ka	Y	<u>.</u>	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM8	RSAM8-10B	N	0.0	Hexachlorobenzene	0.023	3 ma/ka	Ý	-	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SSAN6-02	SSAN6-02-3BPC	N	3.5	Hexachlorobenzene	0.16	Sima/ka	Ý	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM8	RSAM8-0.5B	N	(	Hexachlorobenzene	0.04	l mg/kg	Y	-	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SA86	SA86009-10B	FD	5.5	Hexachlorobenzene	0.097	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM2	RSAM2-0.5B	N	0.5	Hexachlorobenzene	0.0097	′ mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAM7	RSAM7009-10B	FD	9.5	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAL4	RSAL4-0.5B	N	0.5	Hexachlorobenzene	0.087	′ mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
SA180	SA180-0.5B	Ν	(	Hexachlorobenzene	0.02	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				
RSAO5	RSAO5-0.5B	N	0.5	Hexachlorobenzene	0.076	3 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded				

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical R		Unit	Detection Flag	Qualifier	Reaso
RSAO6	RSAO6-0.5B	Ν	0	Hexachlorobenzene	0.0069	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA57	SA57-10BD	FD	5	Hexachlorobenzene	0.036	δ mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA131	SA131-10B	Ν	9.5	Hexachlorobenzene	0.12	2 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAN4	RSAN4009-10B	FD	9.5	Hexachlorobenzene	0.0073	3 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK8-01	SSAK8-01-1BPC	N	1	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA48	SA48-10B	N	7.5	Hexachlorobenzene	0.0073	3 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAL2-05	SSAL2-05-3_01_BPC	N	3	Hexachlorobenzene	0.2	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAL2-05	SSAL2-05-4_01_BPC	N N	4	Hexachlorobenzene	0.03	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAL2-05	SSAL2-05-2_01_BPC	N	2	Hexachlorobenzene	0.028	/ mg/kg		1	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSA07	RSA07-0 5B	N	0.5	Hexachlorobenzene	0.072	nig/kg 2 mg/kg	N	5	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA138	SA138-0.5B	N	0.0	Hexachlorobenzene	0.0072	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAP5	RSAP5-0.5B	N	0.0	Hexachlorobenzene	0.0071	ma/ka	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAP7	RSAP7-0.5B	N	0.5	Hexachlorobenzene	0.0072	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA156	SA156-10B	N	9.5	Hexachlorobenzene	0.007	′ mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAQ4	RSAQ4-0.5B	N	1	Hexachlorobenzene	0.025	5 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAQ6	RSAQ6-10B	Ν	7	Hexachlorobenzene	0.0072	2 mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA121	SA121009-0.5B	FD	0.5	Hexachlorobenzene	0.0072	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA121	SA121-0.5B	Ν	0.5	Hexachlorobenzene	0.0062	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA103	SA103-0.5B	Ν	0.5	Hexachlorobenzene	0.045	5 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAS5	RSAS5-0.5B	Ν	1	Hexachlorobenzene	0.04	l mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA214	SA214-0.5B	Ν	1	Hexachlorobenzene	0.0071	mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA136	SA136-0.5B	Ν	1.5	Hexachlorobenzene	0.022	2 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAN7-01	SSAN7-01-1BPC	Ν	1	Hexachlorobenzene	0.061	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAI2-01	SSAI2-01-1BPC	Ν	1	Hexachlorobenzene	0.029	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAI2-01	SSAI2-01-1BPC_FD	FD	1	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAL2	RSAL2-10B	N	9.5	Hexachlorobenzene	0.007	′ mg/kg	Ŷ	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA180	SA180-0.5BD	FD	0	Hexachlorobenzene	0.019	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAM7	RSAM7-0.5B	N	0	Hexachlorobenzene	0.0068	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA67	SA67-0.5B	N	0.5	Hexachlorobenzene	0.01	mg/kg	Y V		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAKO	RSAR0-0.5B		0	Hexachlorobenzene	0.0055	s mg/kg	ř	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAL4	RSAL4009-0.5B		0.5	Hexachlorobenzene	0.060	o mg/kg	T N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA82	SA82-0 5B	N	1	Hexachlorobenzene	0.0072	/mg/kg			Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSA02	RSA02-0.5B	N	0.5	Hexachlorobenzene	0.017	/mg/kg	Y	1	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA176	SA176-0.5B	N	0.0	Hexachlorobenzene	0.0069	) mg/kg	N	0	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAM6	RSAM6-0.5B	N	0.5	Hexachlorobenzene	0.021	ma/ka	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA176	SA176-10B	N	9.5	Hexachlorobenzene	0.007	′mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA166	SA166-0.5B	Ν	0.5	Hexachlorobenzene	0.016	b mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA128	SA128-10B	Ν	0	Hexachlorobenzene	0.0071	mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAN2	RSAN2-0.5B	Ν	1.5	Hexachlorobenzene	0.007	′ mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA207	SA207-20B	Ν	6	Hexachlorobenzene	0.008	8 mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA85	SA85-0.5B	Ν	0.5	Hexachlorobenzene	0.094	l mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA35	SA35-0.5B	Ν	0	Hexachlorobenzene	0.007	′ mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA35	SA35-10B	N	7	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAM3	RSAM3-0.5B	N	1	Hexachlorobenzene	0.0069	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA165	SA165-10B	N	8.5	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA57	SA57-10B	N	5	Hexachlorobenzene	0.022	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA47	SA47-0.5B	N N	1	Hexachlorobenzene	0.057	mg/kg	ř		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA92	SA92-0.5B	N N	0		0.062	l mg/kg	T V		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAN5	RSAN5-0.5B	N	2.5	Hexachlorobenzene	0.014	amg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-15	BDT-1-S-15-2BPC	N	0.0	Hexachlorobenzene	0.0000	8 ma/ka	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-15	BDT-1-S-15-2BPC FD	FD	2	Hexachlorobenzene	0.03	3 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-15	BDT-1-N-15-10BPC	N	8.5	Hexachlorobenzene	0.22	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-15	BDT-1-S-15-4BPC	N	4	Hexachlorobenzene	0.03	3 ma/ka	N	-	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-15	BDT-1-S-15-6BPC	N	6	Hexachlorobenzene	0.03	B mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK7-06	SSAK7-06-1BPC	N	1	Hexachlorobenzene	0.032	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-20	BDT-1-S-20-4BPC	N	4	Hexachlorobenzene	0.029	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-20	BDT-1-S-20-6BPC	N	6	Hexachlorobenzene	0.028	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-20	BDT-1-S-20-2BPC	Ν	2	Hexachlorobenzene	0.029	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAO4-05	SSAO4-05-1BPC	Ν	0	Hexachlorobenzene	0.046	δ mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK6-03	SSAK6-03-3BPC	N	0	Hexachlorobenzene	0.32	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAK4	RSAK4009-0.5B	FD	0.5	Hexachlorobenzene	0.24	mg/kg	Y	<u>_</u>	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAO3	RSAO3-0.5B	N	0	Hexachlorobenzene	0.13	8 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth

#### on for Exclusion nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec hod 8270 was excluded nod 8270 was excluded hod 8270 was excludec hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec nod 8270 was excluded nod 8270 was excluded

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Depth Chemical F		Unit	Detection Flag	Qualifier	Reaso
RSAK4	RSAK4-0.5B	N	0.5	Hexachlorobenzene	0.25	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-10	BDT-1-S-10-2BPC	Ν	2	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-10	BDT-1-S-10-4BPC	Ν	4	Hexachlorobenzene	0.031	mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-10	BDT-1-S-10-6BPC	Ν	6	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK5-03	SSAK5-03-1BPC	Ν	0.5	Hexachlorobenzene	0.087	′ mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-15	BDT-1-S-15-8BPC	Ν	8	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-5	BDT-1-S-5-8BPC	Ν	8	Hexachlorobenzene	0.03	8 mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-15	BDT-1-N-15-8BPC	Ν	6.5	Hexachlorobenzene	0.47	′ mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-10	BDT-1-N-10-8BPC_FD	FD	6.5	Hexachlorobenzene	0.18	8 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-10	BDT-1-N-10-12BPC	N	9.5	Hexachlorobenzene	0.25	5 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-10	BDT-1-N-10-8BPC	N	6.5	Hexachlorobenzene	0.13	8 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-10	BDT-1-N-10-10BPC	N	8.5	Hexachlorobenzene	0.42	2 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-5	BDT-1-S-5-4BPC	N	4	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-5	BDI-1-S-5-6BPC	N	6	Hexachlorobenzene	0.03	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
BDT-1-S-5	BDT-1-S-5-2BPC	N	2	Hexachlorobenzene	0.03	3 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAI3-06	SSAI3-06-4BPC	N	5	Hexachlorobenzene	0.029	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAL4-01	SSAL4-01-1BPC	N	1	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
BDT-1-N-20	BDT-1-N-20-8BPC	N	8	Hexachlorobenzene	0.029	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-20	BDI-1-N-20-6BPC	N	6	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAI3-01	SSAI3-01-3BPC	N	0	Hexachlorobenzene	0.033	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK3-05	SSAK3-05-1BPC	N	3.5	Hexachlorobenzene	0.3	8 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-20	BDT-1-S-20-8BPC	N	8	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAI3-02	SSAI3-02-14BPC	N	0	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAK8-04	SSAK8-04-5BPC	N	3	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-05	SSAJ3-05-12BPC	N	4	Hexachlorobenzene	0.029	) mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAM5-03	SSAM5-03-10BPC	N	5.5	Hexachlorobenzene	0.49	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAM5-03	SSAM5-03-6BPC	N	1.5	Hexachlorobenzene	0.57	′ mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAM5-03	SSAM5-03-8BPC	N	3.5	Hexachlorobenzene	0.38	3 mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-S-10	BDI-1-S-10-8BPC	N	8	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method
SSAJ2-02	SSAJ2-02-4BPC	N	0	Hexachlorobenzene	0.76	s mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK5-02	SSAK5-02-1BPC	N	1	Hexachlorobenzene	0.032	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAK4	RSAK4-3BPC	N	3	Hexachlorobenzene	0.87	′ mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK3-02	SSAK3-02-1BPC	N	0	Hexachlorobenzene	0.03	8 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ2-02	SSAJ2-02-5BPC	N	0	Hexachlorobenzene	0.092	2 mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-05	SSAJ3-05-16BPC	N	8	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-05	SSAJ3-05-8BPC	N	0	Hexachlorobenzene	0.03	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSA04-02	SSAO4-02-3BPC	N	0	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSA03-03	SSAO3-03-9BPC	N	4	Hexachlorobenzene	0.029	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA106	SA106-12B	N	8.5	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK7-05	SSAK7-05-1BPC	N	1	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK4-02	SSAK4-02-1BPC	N	0.5	Hexachlorobenzene	0.11	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAO4-01	SSAU4-01-1BPC	N	0	Hexachlorobenzene	0.083	s mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAN6	RSAN6009-10B	FD	9.5	Hexachlorobenzene	0.0073	s mg/kg	N	0J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
RSAN6	RSAN6-10B	N	9.5	Hexachlorobenzene	0.037	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAP3-01	SSAP3-01-1BPC	N	0	Hexachlorobenzene	0.7	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSA07-01	SSAU7-01-1BPC	N	0	Hexachlorobenzene	0.032	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-20	BDT-1-N-20-2BPC	N	2.5	Hexachlorobenzene	0.13	s mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
BDT-1-N-20	BD1-1-N-20-4BPC	N	4	Hexachlorobenzene	0.03	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-07	SSAJ3-07-5BPC	N	0	Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-07	SSAJ3-07-12BPC	N	6	Hexachlorobenzene	0.029	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAH3-01	SSAH3-01-1BPC	N	1	Hexachlorobenzene	0.03	s mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAM7-04	SSAM7-04-1BPC	N	0	Hexachlorobenzene	0.09	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK5-01	SSAK5-01-1BPC	N	0.5	Hexachlorobenzene	0.31	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSA06-04	SSAU6-04-1BPC	N	1	Hexachlorobenzene	0.16	mg/kg	N	UJ	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAJ3-07	SSAJ3-07-8BPC	N	2	Hexachlorobenzene	0.029	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAK4-01	SSAK4-01-1BPC	N	1	Hexachlorobenzene	1.2	² mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAN6-04	55AN6-04-18PC	IN N	1	Hexachlorobenzene	0.031	mg/kg	IN N	ļ	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081.
RSAS8	RSAS8-0.5B	N	1.5	Hexachlorobenzene	0.0072	2 mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SA154	SA154-10B	IN N	2.5	Hexachlorobenzene	0.0058	s mg/kg	Y	J	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081.
SSAK8-05	55AK8-05-18PC	IN N	0.5	Hexachlorobenzene	0.031	mg/kg	N	l	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081.
SSAK3-04	55AK3-04-18PC	IN N	1.5	Hexachlorobenzene	0.072	с mg/кg	Y	J	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081.
SSAI3-05	55AI3-05-10BPC	IN N	1	Hexachlorobenzene	0.41	mg/kg	Y	ļ	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081. The data from EPA Method 82/0 and 8081.
SSAJ8-02	SSAJ8-02-18PC	IN N	1	Hexachlorobenzene	0.031	mg/kg	IN N	ļ	Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 and 8081. The data from EPA Method 8270 and 8081.
RSAU6		IN N	8.5	Hexachlorobenzene	0.0071	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth
SSAL2-03	55AL2-03-18PC	N	1 1	Hexachlorobenzene	0.2	rmg/kg	Y	J	Analyzed by both EPA Method 8270 and 8081. The data from EPA Meth

#### on for Exclusion nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excluded hod 8270 was excluded nod 8270 was excluded nod 8270 was excluded nod 8270 was excluded hod 8270 was excludec nod 8270 was excluded nod 8270 was excluded

Sample Location	Sample ID	Sample Type	Start Depth Chemical (ft)	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
SSAL2-02	SSAL2-02-4BPC	Ν	4.5 Hexachlorobenzene	0.65	mg/kg	Υ		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAK6-01	SSAK6-01-1BPC	Ν	1 Hexachlorobenzene	0.11	mg/kg	Y J		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAM5-04	SSAM5-04-5BPC	Ν	1 Hexachlorobenzene	0.03	mg/kg	Ν		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAM5-04	SSAM5-04-5BPC_FD	FD	1 Hexachlorobenzene	0.05	mg/kg	Y J		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAM5-04	SSAM5-04-10BPC	Ν	6 Hexachlorobenzene	0.032	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAK2-01	SSAK2-01-1BPC	N	3.5 Hexachlorobenzene	0.031	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
SSAK3-03	SSAK3-03-1BPC	N	2 Hexachlorobenzene	0.03	mg/kg	N		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
RSAN7	RSAN7-0.5B	N		0.29	mg/kg	Y		Analyzed by both EPA Method 8270 and 8081. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	IN N	4 Fluoranthene	0.096	mg/kg		)	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
CS-DC-1	CS-DC-1 CS-DC-2	IN N		0.097	mg/kg		) I	Analyzed by both EPA Mietinod 8270 and 8270 Sim. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2 CS-DC-1	N		0.007	mg/kg	N I	, I	Analyzed by both EPA Method 8270 and 8270 Smit. The data from EPA Method 8270 was excluder.
CS-DC-2	CS-DC-2	N	4 Acenaphthylene	0.000	ma/ka	N I	, I	Analyzed by both EFA Method 8270 and 8270 Sim. The data from EFA Method 8270 was excluder
CS-DC-1	CS-DC-1	N	0 Acenaphthylene	0.087	ma/ka	N L	, J	Analyzed by both Er A Method 8270 and 8270 SM. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	N	4 Pyrene	0.11	mg/kg	Y J		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	N	4 Benzo(g,h,i)perylene	0.11	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-1	CS-DC-1	N	0 Pyrene	0.096	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-1	CS-DC-1	Ν	0 Benzo(g,h,i)perylene	0.11	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	Ν	4 2-Methylnaphthalene	0.086	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-1	CS-DC-1	Ν	0 2-Methylnaphthalene	0.087	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	N	4 Acenaphthene	0.084	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-1	CS-DC-1	N	0 Acenaphthene	0.085	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-2	CS-DC-2	N	4 Phenanthrene	0.095	mg/kg	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
CS-DC-1	CS-DC-1	N	0 Phenanthrene	0.096	mg/kg	N L	)	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
CS-DC-2	CS-DC-2		4 Fluorene	0.093	mg/kg	N L	)	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
SA42	SA42009-10B		5.5 Naphthalene	0.0057	mg/kg	N N		Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded
SA42 SA212	SA42-10B SA212-0 5B	IN N		0.0053	mg/kg	N		Analyzed by both EPA Mitched \$260 and \$270. The data from EPA Mitched \$260 was excluded
BSA04	BSA04-0 5B	N		0.0057	mg/kg	N		Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
RSAP7	RSAP7-0.5B	N	0.5 Naphthalene	0.0057	ma/ka	N I	1.1	Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluder
SA156	SA156-10B	N	9.5 Naphthalene	0.0054	ma/ka	N C		Analyzed by both Er A Method 8260 and 8270. The data from Er A Method 8260 was excluded
CS-DC-1	CS-DC-1	N	0 Fluorene	0.094	ma/ka	N L	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA211	SA211-0.5B	N	1 Naphthalene	0.0049	mg/kg	N L	IJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA51	SA51009-10B	FD	7 Naphthalene	0.0053	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA136	SA136-0.5B	N	1.5 Naphthalene	0.0064	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA51	SA51-10B	Ν	7 Naphthalene	0.005	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAK4	RSAK4009-0.5B	FD	0.5 Naphthalene	0.0047	mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA74	SA74009-0.5B	FD	0 Naphthalene	0.0036	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA74	SA74-10B	N	9.5 Naphthalene	0.0072	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA185	SA185-0.5B	N	0.5 Naphthalene	0.0048	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA131	SA131-10B	N	9.5 Naphthalene	0.0047	mg/kg	N		Analyzed by both EPA Method 8260 and 82/0. The data from EPA Method 8260 was excluded
RSAL6	RSAL6-10B	N	9.5 Naphthalene	0.0045	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAL0	RSAL0-0.5B	IN N		0.0033	mg/kg	IN N		Analyzed by both EPA Method 2500 and 2700. The data from EPA Method 2500 was excluded
BSAN3	BSAN3-10B	N	0.5 Naphthalana	0.0043	mg/kg	N		Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
RSAN3	RSAN3-05B	N		0.004	mg/kg	N		Analyzed by both Er A method 8260 and 8270. The data from ErA method 8260 was excluder
SA74	SA74-0.5B	N	0 Naphthalene	0.004	ma/ka	N		Analyzed by both Er A Method 8260 and 8270. The data from Er A Method 8260 was excluded
RSA03	RSAO3-0.5B	N	0 Naphthalene	0.005	ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAM4	RSAM4-0.5B	N	0.5 Naphthalene	0.0037	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA171	SA171-5B	N	0 Naphthalene	0.0053	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA103	SA103-0.5B	N	0.5 Naphthalene	0.0071	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA114	SA114-10B	Ν	8 Naphthalene	0.0052	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA175	SA175-10B	Ν	2.5 Naphthalene	0.0041	mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAL4	RSAL4-0.5B	Ν	0.5 Naphthalene	0.0033	mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA109	SA109-0.5B	N	2.5 Naphthalene	0.0051	mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA67	SA67-0.5B	N	0.5 Naphthalene	0.0047	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA180	SA180-0.5B	N	0 Naphthalene	0.0053	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA180	SA180-10B	N	8.5 Naphthalene	0.0048	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
KSAM/	RSAM7009-10B		9.5 Naphthalene	0.0041	mg/kg	IN N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAM7	ROAMZ OF	IN N		0.0039	mg/kg	IN N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
DSA 12		IN N	5 Naphthalene	0.0055	mg/kg	N		Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
RSAK8	RSAK8-10B	N	7 Naphthalene	0.0040	ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAI 4	RSAL4009-0.5B	FD	0.5 Naphthalene	0.0033	ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAK6	RSAK6-10B	N	9.5 Naphthalene	0.0039	ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
		1.		0.0000	J J	1 -		

Sample Location	Sample ID	Sample Type	Start Depth (ft)	epth Chemical R		Unit	Detection Flag	Qualifier	Reaso
RSAK6	RSAK6-0.5B	Ν	0	Naphthalene	0.0045	5 mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAJ2	RSAJ2-10B	Ν	2.5	Naphthalene	0.0073	8 mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA64	SA64-0.5B	N	0	Naphthalene	0.0047	′ mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA121	SA121-0.5B	N	0.5	Naphthalene	0.0044	l mg/kg	N	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA121	SA121009-0.5B	FD	0.5	Naphthalene	0.0056	8 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAQ6	RSAQ6-0.5B	N	0	Naphthalene	0.005/	′ mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAQ6	RSAQ6-10B	N	/	Naphthalene	0.0054	l mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA157	SA157-0.5B	N	0.5	Naphthalene	0.01	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA157	SA157009-0.5B	FD	0.5	Naphthalana	0.0052	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAS5	RSAS5-0.5B	N	1	Naphthalana	0.0053	s mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA214	SA214-0.5B	N	1	Naphthalana	0.005	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA43	SA43-10B	IN N	5.5	Naphthalana	0.004	/mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA151	SA 151-0.5B	IN N	0.5	Naphthalana	0.005	s mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
DONIS	DSA120-10D	N	05	Naphthalana	0.0032	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSALS DSAME		N	0.3	Naphthalana	0.0032	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAIVIO SA107	SA107000 10P		0.3	Naphthalana	0.0030	a mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA197	SA197009-10B		9.0	Naphthalana	0.0042	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA197	SA 197-10B		9.5	Naphthalana	0.0037	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
5A104	SA 104009-10B		9	Naphthalana	0.0046	o mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
ROALI SA107	ROAL7-0.5D		0.5	Naphthalana	0.0040	nig/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA 197		IN NI	0	Naphthalana	0.0050	sing/kg	N N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
R3AU2	RSA02-0.5B		0.5	Naphthalana	0.000	l mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA 150	SA 150-0.5B		0.5	Naphthalana	0.0044	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
DSANA	SA 150-10D BSAN4000 10P		9.5	Naphthalana	0.005	sing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
R3AN4 8425	R3AN4009-10B		9.0	Naphthalana	0.0040	l mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA35	SA35-0.5B	N	0	Naphthalana	0.0032	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
DEANA		N	7	Naphthalana	0.0030	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAN4		N	0.5	Naphthalana	0.004	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
ROAN4	SAGO O EP	N	9.0	Naphthalana	0.003	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA09 SA85	SA09-0.5B	N	0.5	Naphthalana	0.002	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
DSAM3	BSAM3-0.5B	N	0.0	Naphthalana	0.0053	lmg/kg	N	111	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAN2	RSAN2-0.5B	N	15	Naphthalana	0.0054	Filig/kg	N	00	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA207	SA207-20B	N	1.3	Naphthalana	0.0052	ring/kg Img/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA207 SA138	SA138-0 5B	N	05	Naphthalene	0.0057	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
BSA07	RSA07-0 5B	N	0.5	Naphthalene	0.0050	alma/ka	N	111	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAD5	RSAP5-0.5B	N	0.0	Naphthalene	0.0053	ama/ka	N	05	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SAAA	SA44-10B	N	5	Naphthalene	0.003	sing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAM5	RSAM5-10B	N	1	Naphthalene	0.000	Sma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA76	SA76-10B	N	4	Naphthalene	0.0040	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAL8	RSAL 8-0.5B	N	0.5	Naphthalene	0.0002	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA182	SA182-10B	N	5.5	Naphthalene	0.000	Sma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA77	SA77-0 5B	N	0.0	Naphthalene	0.0000	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA126	SA126-0 5B	N	1	Naphthalene	0.000	R ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA70	SA70-0 5B	N	1	Naphthalene	0.0040	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA92	SA92-10B	N	75	Naphthalene	0.0000	/ mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA152	SA152-10B	N	9.5	Naphthalene	0.0007	ing/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA88	SA88-10B	N	5.5	Naphthalene	0.000	/ mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA201	SA201-10B	N	55	Naphthalene	0.0047	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA48	SA48-10B	N	7.5	Naphthalene	0.0002	a mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA92	SA92-0 5B	N	7.0	Naphthalene	0.0000	Sma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA100	SA100-0 5B	N	1	Naphthalene	0.0040	mg/kg	N	111	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA152	SA152-0 5B	N	0	Naphthalene	0.0040	) mg/kg	N	00	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA152	SA152009-0 5B	FD	0	Naphthalene	0.0040	Sma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA200	SA200-0 5B	N	1	Naphthalene	0.0040	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA55	SA55-0 5B	N	0.5	Naphthalene	0.000	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA144	SA144-0 5B	N	0.0	Naphthalene	0.0040	l ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270. The data from EPA Method 8260 and 8270.
SA104	SA104-10B	N	1	Naphthalene	0.003-	5 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAM8	RSAM8-10B	N		Naphthalene	0.0040	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA86	SA86-10B	N	55	Naphthalene	0.004	l ma/ka	Y	J	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA86	SA86009-10B	FD	5.5	Naphthalene	0.003-	) mg/kg	Ň	-	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAM8	RSAM8-0.5B	N	0.5	Naphthalene	0.003	3 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAN5	RSAN5-0.5B	N	05	Naphthalene	0.0000	l ma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
RSAM2	RSAM2-0.5B	N	0.5	Naphthalene	0.0047	/mg/kg	N	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
SA166	SA166-0.5B	N	0.5	Naphthalene	0.0052	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Meth
			0.0	[·····	0.0002		<u> </u>	ļ	

#### on for Exclusion nod 8260 was excluded hod 8260 was excludec hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excludec hod 8260 was excluded nod 8260 was excluded hod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excludec nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded nod 8260 was excluded hod 8260 was excluded nod 8260 was excluded

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
SA75	SA75-0.5B	Ν	(	) Naphthalene	0.0049	) mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA176	SA176-10B	Ν	9.5	5 Naphthalene	0.0036	8 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA75	SA75-10B	N	9.5	5 Naphthalene	0.0044	l mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA50	SA50-0.5B	N	(	Naphthalene	0.0051	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA82	SA82-0.5B	N	1	Naphthalene	0.0078	3 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA176	SA176-0.5B	N	(	J Naphthalene	0.0048	s mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAK4	RSAK4-0.5B	N	0.5		0.0052	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSALZ	RSAL2-TUB		9.5	Naphthalene	0.0048	sing/kg	N N	UJ	Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded
SA100	SA160-0.3BD		(	Naphthalono	0.0036	l mg/kg	N V	1	Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
SSA00-10	SSA07-06-0BPC	N		Naphthalene	0.0004	Ring/kg	V	3	Analyzed by both Er A method 8260 and 8270. The data from EPA Method 8260 was excluded
SSAN7-07	SSAN7-07-0BPC	N		Naphthalene	0.00047	/mg/kg	N	5	Analyzed by both ETA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SSA08-05	SSAO8-05-0BPC	N	(	Naphthalene	0.00047	Sma/ka	N		Analyzed by both EFA Method 8260 and 8270. The data from EPA Method 8260 was excluder
SSA08-12	SSAO8-12-0BPC	N	(	Naphthalene	0.0075	img/kg	Y	J	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SSAO8-12	SSAO8-12-0BPC FD	FD	(	) Naphthalene	0.0039	) ma/ka	Y	J	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAO3	RSAO3-10B	N	8	3 Naphthalene	0.0036	8 mg/kg	N	-	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA189	SA189-0.5B	Ν	0.5	Naphthalene	0.0045	5 mg/kg	Ν	UJ	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SSAO8-11	SSAO8-11-0BPC	N	(	Naphthalene	0.00067	′ mg/kg	Y	J	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SSAO8-08	SSAO8-08-0BPC	Ν	(	Naphthalene	0.00051	mg/kg	Y	J	Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SSAN8-04	SSAN8-04-0BPC	N	(	Naphthalene	0.00048	8 mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
CS-DC-1	CS-DC-1	N	(	Naphthalene	0.084	l mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA60	SA60-10B	Ν	9.5	5 Naphthalene	0.0047	′ mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA106	SA106-12B	Ν	8.5	5 Naphthalene	0.0057	′ mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA198	SA198-10B	Ν	2	2 Naphthalene	0.0044	l mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAN6	RSAN6009-10B	FD	9.5	5 Naphthalene	0.004	l mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAN6	RSAN6-10B	N	9.5	5 Naphthalene	0.0048	8 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA165	SA165-10B	N	8.5	5 Naphthalene	0.0053	8 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA150	SA150-10B	N	9.5	5 Naphthalene	0.0059	9 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
CS-DC-2	CS-DC-2	N	2	1 Naphthalene	0.083	8 mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA186	SA186-0.5B	N	0.5	Naphthalene	0.0052	2 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSA05	RSAU5-0.5B	N	0.5	Naphthalene	0.0047	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SATT	SA71-0.5B	N N	0.5	Naphthalene	0.0061	mg/kg	N N		Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded
SA54 SA57	SA54-0.5B	N	1.0		0.0040	sing/kg	IN N		Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded
SA37 SA47	SA7-10B	N		1 Naphthalene	0.0050	mg/kg	N	-	Analyzed by both Er A method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAO6	RSA06-0.5B	N	(	Naphthalene	0.0031	/ mg/kg	N		Analyzed by both Er A method 8260 and 8270. The data from EPA Method 8260 was excluder
SA102	SA102-10B	N	8.5	5 Naphthalene	0.0056	Sma/ka	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluder
SA57	SA57-10BD	FD	0.0	5 Naphthalene	0.0059	) mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAS8	RSAS8-0.5B	N	1.5	5 Naphthalene	0.0053	3 mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA154	SA154-10B	N	2.5	Naphthalene	0.0055	i mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA64	SA64-10B	N	9.5	Naphthalene	0.0044	l mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAN7	RSAN7-0.5B	Ν	1	1 Naphthalene	0.0057	′ mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA134	SA134-10B	Ν	8.5	5 Naphthalene	0.0041	mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA65	SA65-20B	Ν	8.5	5 Naphthalene	0.0055	5 mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA65	SA65-10B	N	(	) Naphthalene	0.0091	mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
SA31	SA31-0.5B	Ν	0.5	5 Naphthalene	0.0052	2 mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAO6	RSAO6-10B	N	8.5	5 Naphthalene	0.0034	l mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAH3	RSAH3009-0.5B	FD	1	1 Naphthalene	0.0024	l mg/kg	Ν		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAH3	RSAH3-0.5B	N	1	Naphthalene	0.0029	) mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
RSAH3	RSAH3-0.5B	N	1	Hexachlorobenzene	0.22	2 mg/kg	Y	J	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 was excluded
RSAH3	RSAH3009-0.5B	FD	1	Hexachlorobenzene	0.031	mg/kg	Ŷ	J	Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 was excluded
RSAU7	RSAU7-9B	N	(		0.00088	s mg/kg	N		Analyzed by both EPA Method 8260 and 8270. The data from EPA Method 8260 was excluded
R5AU7	RSAU7-9B	N	0.6		0.00088	s mg/kg	N		Analyzed by both EPA Method 82/0 and 8081. The data from EPA Method 82/0 was excluded
SA141	SA141-14B		0.8	Naphthalene	0.00095	mg/kg	IN N		Analyzed by both EPA Method 9260 and 9270. The data from EPA Method 9260 was excluded
SSAN7-06	SSAN7-06-08PC	N	0.8	Nanhthalene	0.0011	ling/kg	N	<u> </u>	Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
SSAN7-00	SSA07-00-0BPC	N	(	Naphthalono	0.00030	l mg/kg	N		Analyzed by both EFA Method 8260 and 8270. The data from EFA Method 8260 was excluded
RISB-09	RISB-09-5 0-20141211	N	E E	5 Benzo(a)anthracene	0.00034	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N	0.6	5 Benzo(a)anthracene	0.072	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N	0.5	5 Chrysene	0.076	ing/ka	N	- U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	N		5 Chrysene	0.078	3 mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N	0.5	Benzo(k)fluoranthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	N	5	Benzo(k)fluoranthene	0.072	2 mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-0.5-20141211	N	0.5	5 Benzo(b)fluoranthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-09	RISB-09-5.0-20141211	Ν	5	Benzo(b)fluoranthene	0.072	2 mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reaso
RISB-09	RISB-09-5.0-20141211	Ν	5	Dibenz(a,h)anthracene	0.1	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-09	RISB-09-0.5-20141211	N	0.5	Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-09	RISB-09-5.0-20141211	Ν	5	Benzo(a)pyrene	0.069	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-09	RISB-09-0.5-20141211	N	0.5	Benzo(a)pyrene	0.068	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-09	RISB-09-0.5-20141211	N	0.5	Indeno(1,2,3-cd)pyrene	0.13	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-09	RISB-09-5.0-20141211	N	5	Indeno(1,2,3-cd)pyrene	0.13	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Benzo(k)fluoranthene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Benzo(k)fluoranthene	0.076	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Chrysene	0.081	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Chrysene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Benzo(a)anthracene	0.076	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Benzo(a)anthracene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Dibenz(a,h)anthracene	0.11	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Dibenz(a,h)anthracene	0.1	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Benzo(a)pyrene	0.068	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Benzo(a)pyrene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Benzo(b)fluoranthene	0.076	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Benzo(b)fluoranthene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-0.5-20141203	N	0.5	Indeno(1,2,3-cd)pyrene	0.13	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
M-161D	M-161D-5.0-20141203	N	5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Benzo(a)anthracene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Benzo(a)anthracene	3.9	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Chrysene	4.2	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Chrysene	0.077	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Benzo(k)fluoranthene	3.9	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Benzo(k)fluoranthene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Benzo(b)fluoranthene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Benzo(b)fluoranthene	3.9	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Benzo(a)pyrene	3.7	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Benzo(a)pyrene	0.069	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Dibenz(a,h)anthracene	5.6	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-0.5-20141215	N	0.5	Indeno(1,2,3-cd)pyrene	0.13	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-12	RISB-12-2.5-20141216	N	2.5	Indeno(1,2,3-cd)pyrene	7.2	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	N	0	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.000099	mg/kg	Y	JK	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	2,3,7,8-Tetrachlorodibenzo-p-dioxin	0.000086	mg/kg	Ν	U	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.0021	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,6,7,8-Hexachlorodibenzofuran	0.0016	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	0.000065	mg/kg	Ν	U	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,5,6,7,8-Octachlorodibenzofuran	0.015	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.0001	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.00024	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,7,8-Hexachlorodibenzofuran	0.0018	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,7,8-Pentachlorodibenzofuran	0.00064	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	2,3,4,6,7,8-Hexachlorodibenzofuran	0.00037	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	2,3,4,7,8-Pentachlorodibenzofuran	0.00045	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	Ν	0	1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.0042	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxin	0.00035	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	2,3,7,8-Tetrachlorodibenzofuran	0.0012	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	0.0000081	mg/kg	Ν	U	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	Benzo(k)fluoranthene	0.12	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	N	0	Chrysene	0.086	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	N	0	1,2,3,7,8,9-Hexachlorodibenzofuran	0.00031	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-1	CS-DC-1	N	0	Benzo(a)pyrene	0.096	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	N	0	Benzo(a)anthracene	0.094	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	N	0	Dibenz(a,h)anthracene	0.1	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	Ν	0	Benzo(b)fluoranthene	0.097	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-1	CS-DC-1	Ν	0	Indeno(1,2,3-cd)pyrene	0.098	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Benzo(k)fluoranthene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Benzo(k)fluoranthene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	N	5	Benzo(a)anthracene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	N	0.5	Benzo(a)anthracene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Chrysene	0.081	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Chrysene	0.079	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Benzo(a)pyrene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA

#### on for Exclusion Method 8270 was excluded ethod 8280A was excluded Method 8270 was excluded Method 8270 was excluded ethod 8280A was excluded Method 8270 was excluded

Sample Location	Sample ID	Sample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reaso
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Benzo(a)pyrene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Benzo(b)fluoranthene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Benzo(b)fluoranthene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Dibenz(a,h)anthracene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-5.0-20141217	Ν	5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-11	RISB-11-0.5-20141217	Ν	0.5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Benzo(k)fluoranthene	0.074	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	Ν	5	Benzo(k)fluoranthene	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Benzo(a)anthracene	0.074	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	N	5	Benzo(a)anthracene	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Chrysene	0.08	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	N	5	Chrysene	0.082	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Dibenz(a,h)anthracene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	Ν	5	Dibenz(a,h)anthracene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Benzo(b)fluoranthene	0.074	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	N	5	Benzo(b)fluoranthene	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	N	0.5	Benzo(a)anthracene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	N	5	Benzo(a)anthracene	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	N	5	Chrysene	0.08	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	N	0.5	Chrysene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Benzo(k)fluoranthene	0.04	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	Ν	5	Benzo(k)fluoranthene	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	N	0.5	Benzo(k)fluoranthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	N	5	Benzo(a)pyrene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	N	0.5	Benzo(a)pyrene	0.068	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Chrysene	0.043	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Benzo(a)anthracene	0.02	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Benzo(a)pyrene	0.02	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Benzo(b)fluoranthene	0.026	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-C10B-1	CS-C10B-1	N	0	Indeno(1,2,3-cd)pyrene	0.022	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	Ν	0.5	Benzo(b)fluoranthene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	Ν	5	Benzo(b)fluoranthene	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	Ν	5	Dibenz(a,h)anthracene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	Ν	0.5	Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-0.5-20141120	N	0.5	Indeno(1,2,3-cd)pyrene	0.13	mg/kg	N	UJ	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-47	RISB-47-5.0-20141120	N	5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	N	0.5	Benzo(a)pyrene	0.071	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	N	5	Benzo(a)pyrene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-5.0-20141218	N	5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
RISB-13	RISB-13-0.5-20141217	Ν	0.5	Indeno(1,2,3-cd)pyrene	0.14	mg/kg	N	UJ	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.00017	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	N	4	2,3,7,8-Tetrachlorodibenzo-p-dioxin	0.000049	mg/kg	N	U	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.0005	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,5,6,7,8-Octachlorodibenzofuran	0.016	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.00018	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	0.000082	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	Benzo(k)fluoranthene	0.14	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	Chrysene	0.12	mg/kg	Y	J	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	Benzo(a)pyrene	0.095	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	Benzo(a)anthracene	0.093	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	N	4	Benzo(b)fluoranthene	0.096	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	Indeno(1,2,3-cd)pyrene	0.097	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.0023	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,6,7,8-Hexachlorodibenzofuran	0.002	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxin	0.0016	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	N	4	2,3,4,7,8-Pentachlorodibenzofuran	0.00061	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.0046	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	N	4	2,3,4,6,7,8-Hexachlorodibenzofuran	0.00036	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,4,7,8-Hexachlorodibenzofuran	0.0024	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	1,2,3,7,8-Pentachlorodibenzofuran	0.001	mg/kg	Y	J	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	N	4	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	0.00011	mg/kg	Y	JK	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	N	4	1,2,3,7,8,9-Hexachlorodibenzofuran	0.00035	mg/kg	Y	JK	Analyzed by both EPA Method 8280A and 8290. The data from EPA Me
CS-DC-2	CS-DC-2	Ν	4	2,3,7,8-Tetrachlorodibenzofuran	0.0018	mg/kg	Y		Analyzed by both EPA Method 8280A and 8290. The data from EPA Me

#### on for Exclusion Method 8270 was excluded ethod 8280A was excluded Method 8270 was excluded ethod 8280A was excluded

Sample Location	Sample ID	Sample Type	Start Depth (ft) Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
RISB-14	RISB-14-0.5-20141216	N	0.5 Chrysene	0.078	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	Ν	5 Chrysene	0.082	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	Ν	0.5 Benzo(a)anthracene	0.073	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	5 Benzo(a)anthracene	0.077	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	5 Benzo(k)fluoranthene	0.077	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N	0.5 Benzo(k)fluoranthene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-0.5-20141216	N	0.5 Benzo(a)pyrene	0.07	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	5 Benzo(a)pyrene	0.073	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-14	RISB-14-5.0-20141216	N	5 Benzo(b)fluoranthene	0.077	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-14	RISB-14-0.5-20141216	N		0.073	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISB-14	RISB-14-0.5-20141216	IN N	0.5 Dibenz(a,h)anthracene	0.1	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISD-14	RISB-14-5.0-20141210	IN NI	5 Underse (1, 2, 2, ed)systems	0.11	mg/kg	N		Analyzed by both EPA Method 9270 and 9270 Sint. The data from EPA Method 9270 was excluded
RISD-14 RISB-14	RISB-14-5.0-20141216 RISB-14-0 5-20141216	IN N		0.14	mg/kg	N		Alialyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0 5-20141210	N		0.13	mg/kg	N		Analyzed by both Er A method 8270 and 8270 Sint. The data from EPA Method 8270 was excluder.
M-162D	M-162D-5.0-20141209	N	5 Benzo(k)fluoranthene	0.072	mg/kg	N	<u> </u>	Analyzed by both ETA Method 8270 and 8270 GMs. The data from ETA Method 8270 was excluder
M-162D	M-162D-5.0-20141209	N	5 Benzo(a)anthracene	0.076	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
M-162D	M-162D-0.5-20141209	N	0.5 Benzo(a)anthracene	0.072	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	5 Chrysene	0.081	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N	0.5 Chrysene	0.077	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N	0.5 Dibenz(a,h)anthracene	0.1	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	5 Dibenz(a,h)anthracene	0.11	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	5 Benzo(b)fluoranthene	0.076	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N	0.5 Benzo(b)fluoranthene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	N	0.5 Benzo(a)pyrene	0.069	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	N	5 Benzo(a)pyrene	0.072	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-0.5-20141209	Ν	0.5 Indeno(1,2,3-cd)pyrene	0.13	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
M-162D	M-162D-5.0-20141209	Ν	5 Indeno(1,2,3-cd)pyrene	0.14	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5	Ν	1.5 Benzo(k)fluoranthene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5D	FD	1.5 Benzo(k)fluoranthene	0.35	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5D	FD	1.5 Chrysene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5	N	1.5 Chrysene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5 Benzo(a)anthracene	0.07	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	5 Benzo(a)anthracene	0.075	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5 Chrysene	0.075	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was excluded
RISB-10	RISB-10-5.0-20141215	N	5 Chrysene	0.08	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
RISB-10	RISB-10-0.5-20141215	N		0.07	mg/kg	N	0	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
RISD-10 SA03	RISB-10-5.0-20141215			0.075	mg/kg	N		Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
SA03	SA3-0.5D	FD	1.5 Benzo(a)pyrene	0.35	mg/kg	N		Analyzed by both Er A method 8270 and 8270 Sint. The data from EPA Method 8270 was excluder.
SA03	SA3-0.5	N	1.5 Benzo(a)anthracene	0.35	mg/kg	N		Analyzed by both Er A method 8270 and 8270 Sim: The data from EPA Method 8270 was excluder
SA03	SA3-0.5	N	1.5 Indeno(1.2.3-cd)pyrene	0.00	mg/kg	N	<u> </u>	Analyzed by both EFA Method 8270 and 8270 SM. The data from EFA Method 8270 was excluder
SA03	SA3-0.5D	FD	1.5 Dibenz(a,h)anthracene	0.35	ma/ka	N	<u>U</u>	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
SA03	SA3-0.5	N	1.5 Dibenz(a,h)anthracene	0.35	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5D	FD	1.5 Benzo(b)fluoranthene	0.35	ma/ka	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5	N	1.5 Benzo(b)fluoranthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5	N	1.5 Benzo(a)pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA03	SA3-0.5D	FD	1.5 Indeno(1,2,3-cd)pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5 Dibenz(a,h)anthracene	0.1	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	5 Dibenz(a,h)anthracene	0.11	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	5 Benzo(b)fluoranthene	0.075	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	Ν	0.5 Benzo(b)fluoranthene	0.07	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	Ν	5 Benzo(a)pyrene	0.071	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	Ν	0.5 Benzo(a)pyrene	0.067	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-5.0-20141215	N	5 Indeno(1,2,3-cd)pyrene	0.14	mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
RISB-10	RISB-10-0.5-20141215	N	0.5 Indeno(1,2,3-cd)pyrene	0.13	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N	0.5 Benzo(a)anthracene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N	0.5 Benzo(k)fluoranthene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N	0.5 2,3,7,8-Tetrachlorodibenzo-p-dioxin	8.965E-06	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA7-0.5	N	0.5 1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	2.17E-05	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA7-0.5	N	0.5 1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	8.545E-05	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SAU7	SA/-0.5	N	0.5 1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	1.945E-05	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SAU7	SA7-0.5	N N	U.5 1,2,3,6,7,8-Hexachlorodibenzoturan	0.0002496	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SUREEN. The data from EPA Method 8290 SUREEN was excluded
SAU/	SA1-U.S	IN N	0.5 Chrysons	0.0003921	ing/kg	Y NI		Analyzed by both EPA Method 8250 and 8250 SUKEEN. The data from EPA Method 8250 SUKEEN Was excluded
SAUT	SAT-0.5	IN	0.5 Univsene	0.35	шу/ку	IN	U	Anaryzeu by bourt EFA welliou oz/o anu oz/o anu oz/o anin. The uata itom EFA welliou oz/o was exclude(

Sample Location		Sample ID Sa	ample Type	Start Depth (ft)	Chemical	Result	Unit	Detection Flag	Qualifier	Reason for Exclusion
SA07	SA7-0.5	N		0.5	Benzo(a)pyrene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N		0.5	Benzo(b)fluoranthene	0.35	i mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N		0.5	Dibenz(a,h)anthracene	0.35	i mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N		0.5	Indeno(1,2,3-cd)pyrene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Benzo(k)fluoranthene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N		0.5	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	8.841E-06	6 mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA7-0.5	N		0.5	1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.0009271	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA7-0.5	N		0.5	2,3,4,6,7,8-Hexachlorodibenzofuran	0.0001125	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SAU7	SA7-0.5	N		0.5	2,3,4,7,8-Pentachlorodibenzofuran	9.293E-05	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was exclude:
SA07	SA7-0.5	IN N		0.5	1,2,3,7,0-Pentachiorodibenzolurar	0.0001997	mg/kg	T V		Alialyzed by both EPA Method 9200 and 9200 SCREEN. The data from EPA Method 9290 SCREEN was excluded
SA07	SA7-0.5	N		0.5	1,2,3,4,7,0-1 exaction of the nzofuran	3 135E-05	mg/kg	V		Analyzed by both Er A method 8200 and 8200 SCREEN. The data from EPA Method 8200 SCREEN was excluder
SA07	SA7-0.5	N		0.5	1,2,3,7,8,9-nexacillorodibenzo-n-dioxin	1.618E-05	img/kg	Y		Analyzed by both ETA Method 8250 and 8250 SCREEN. The data from ETA Method 8290 SCREEN was excluder
SA07	SA7-0.5	N		0.5	2.3.7.8-Tetrachlorodibenzofuran	0.0003692	2 mg/kg	Ŷ		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA7-0.5	N		0.5	1.2.3.4.5.6.7.8-Octachlorodibenzo-p-dioxir	0.0001919	) ma/ka	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA05	SA5-0.5	N		1	Benzo(a)anthracene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Chrysene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Dibenz(a,h)anthracene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Benzo(a)pyrene	0.43	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Benzo(b)fluoranthene	0.43	s mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA05	SA5-0.5	N		1	Indeno(1,2,3-cd)pyrene	0.43	s mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA07	SA7-0.5	N		0.5	1,2,3,4,5,6,7,8-Octachlorodibenzofuran	0.0025021	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA21	SA21-0.5	N		0.5	Benzo(k)fluoranthene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA21	SA21-0.5	N		0.5	Benzo(a)anthracene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA21	SA21-0.5	N		0.5	Chrysene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA21	SA21-0.5	N		0.5	Benzo(a)pyrene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA21	SA21-0.5	N		0.5	Indeno(1,2,3-cd)pyrene	0.34	mg/kg	N	UJ	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
SA21	SA21-0.5	N		0.5	Benzo(b)fluoranthene	0.34	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
SAZI	SA21-0.5			0.5	Dibenz(a,n)anthracene	0.34	mg/kg	N	0	Analyzed by both EPA Method 82/0 and 82/0 Silk. The data from EPA Method 82/0 was exclude:
SA67	SA07-0.30			0.5		0.45E-00	mg/kg	T V		Alialyzed by both EPA Method 8200 and 8200 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B			0.5	1 2 3 4 7 8-Hexachlorodibenzo-p-dioxin	3 34E-06	mg/kg	Y		Analyzed by both Er A method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluder
SA67	SA67-0.5B	B N		0.5	1,2,3,4,7,6-1 exaciliorodibenzofuran	0.000921	ma/ka	Y		Analyzed by both ETA Method 8250 and 8250 SCREEN. The data from ETA Method 8290 SCREEN was excluder
SA67	SA67-0.5B	3 N		0.5	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	6.49E-06	ing/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluder
SA67	SA67-0.5B	3 N		0.5	1.2.3.4.6.7.8-Heptachlorodibenzo-p-dioxin	0.0000285	img/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	з N		0.5	1,2,3,6,7,8-Hexachlorodibenzofuran	0.0000947	′mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	в N		0.5	1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.000144	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA06	SA6-0.5	N		1	Benzo(k)fluoranthene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5D	FD	)	1	Benzo(k)fluoranthene	0.38	s mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5	N		1	Chrysene	0.35	i mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5D	FD	)	1	Chrysene	0.38	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5D	FD	)	1	Benzo(a)anthracene	0.38	s mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5	N		1	Benzo(a)anthracene	0.35	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was excluded
SA06	SA6-0.5	N		1	Benzo(a)pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 82/0 and 82/0 SIM. The data from EPA Method 82/0 was exclude:
SAUG	SA6-0.5D	FD	,	1	Benzo(a)pyrene	0.38	mg/kg	IN N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
SAUG	SA6-0.5D	FD N	)	1	Benzo(b)fluorantnene	0.38	mg/kg	IN N	0	Analyzed by both EPA Miethod 8270 and 8270 SIM. The data from EPA Method 8270 was exclude:
SA06	SA6-0.5			1	Dihenz(a h)anthracene	0.30	mg/kg	N	<u> </u>	Analyzed by both End Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA06	SA6-0.5		,	1	Dibenz(a,n)antinacene	0.30	ling/kg	N		Analyzed by both Er A method 8270 and 8270 Sim: The data from EPA Method 8270 was excluder
SA06	SA6-0.5	N	,	1	Indeno(1,2,3-cd)pyrene	0.00	img/kg	N	U.I	Analyzed by both EFA Method 8270 and 8270 SM. The data from EFA Method 8270 was excluder
SA06	SA6-0.5D	FD	)	1	Indeno(1,2,3-cd)pyrene	0.38	l mg/kg	N	UJ	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluder
SA08	SA8-0.5	N		1	Benzo(a)pyrene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA08	SA8-0.5	N		1	Chrysene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA08	SA8-0.5	N		1	Benzo(a)anthracene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA08	SA8-0.5	N		1	Benzo(k)fluoranthene	0.35	i mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA08	SA8-0.5	N		1	Indeno(1,2,3-cd)pyrene	0.35	i mg/kg	Ν	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA08	SA8-0.5	N		1	Dibenz(a,h)anthracene	0.35	mg/kg	N	U	Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA67	SA67-0.5B	З N		0.5	1,2,3,7,8-Pentachlorodibenzofuran	0.0000947	′ mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	B N		0.5	1,2,3,4,7,8-Hexachlorodibenzofuran	0.000184	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	8 N		0.5	1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.00031	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	3 <u>N</u>		0.5	1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxin	0.0000549	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	H N		0.5	2,3,4,6,7,8-Hexachlorodibenzofuran	0.0000507	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA67	SA67-0.5B	B N		0.5	2,3,4,7,8-Pentachlorodibenzoturan	0.00004	mg/kg	Y		Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA07	SA07-0.58			0.5	1,2,3,7,8-Pentachiorodibenzo-p-dioxin	5.35E-06	ling/kg			Analyzed by both EPA Method 8290 and 8290 SUREEN. The data from EPA Method 8290 SUREEN Was excluded
5701	0701-0.5B	אן א		0.5		0.0000143	, mg/kg	1		Intraryzed by both LFA IVIEUTOU 0230 and 0230 SONELIN. THE data HUTH EFA IVIEUTOU 0230 SONEEIN WAS EXCIDENT

Sample Location	Sample ID	Sample Type	Start Depth (ft) Chemical	Result Unit	Detection Flag	Qualifier Reason for Exclusion
SA67	SA67-0.5B	Ν	0.5 2,3,7,8-Tetrachlorodibenzofuran	0.000138 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA08	SA8-0.5	Ν	1 Benzo(b)fluoranthene	0.35 mg/kg	N	U Analyzed by both EPA Method 8270 and 8270 SIM. The data from EPA Method 8270 was excluded
SA87	SA87-0.5B	Ν	0.5 2,3,7,8-Tetrachlorodibenzo-p-dioxin	0.0000166 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.00005 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.000603 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,6,7,8-Hexachlorodibenzofuran	0.000383 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.0000539 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.000236 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,5,6,7,8-Octachlorodibenzofuran	0.00438 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	0.0000359 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 2,3,4,6,7,8-Hexachlorodibenzofuran	0.000195 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxin	0.000519 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.00111 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 2,3,4,7,8-Pentachlorodibenzofuran	0.000177 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,4,7,8-Hexachlorodibenzofuran	0.000909 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,7,8-Pentachlorodibenzofuran	0.000445 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 2,3,7,8-Tetrachlorodibenzofuran	0.000603 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,7,8-Pentachlorodibenzo-p-dioxin	0.0000405 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA87	SA87-0.5B	Ν	0.5 1,2,3,7,8,9-Hexachlorodibenzofuran	0.0000514 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.0001407 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 2,3,7,8-Tetrachlorodibenzo-p-dioxin	2.371E-05 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.0004437 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.0001227 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	4.939E-05 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.0029508 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,6,7,8-Hexachlorodibenzofuran	0.001626 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,6,7,8-Heptachlorodibenzofuran	0.0060334 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 2,3,4,7,8-Pentachlorodibenzofuran	0.0005924 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 2,3,4,6,7,8-Hexachlorodibenzofuran	0.0008592 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,7,8-Pentachlorodibenzofuran	0.0012874 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,7,8-Hexachlorodibenzofuran	0.0026503 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,7,8-Pentachlorodibenzo-p-dioxin	8.789E-05 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,7,8,9-Hexachlorodibenzofuran	0.0002463 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 2,3,7,8-Tetrachlorodibenzofuran	0.0015187 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxir	0.0004427 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA16	SA16-0.5	Ν	0 1,2,3,4,5,6,7,8-Octachlorodibenzofuran	0.016112 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 2,3,7,8-Tetrachlorodibenzo-p-dioxin	2.9E-08 mg/kg	Ν	U Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	1.17E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	5.16E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,6,7,8-Hexachlorodibenzofuran	1.45E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,4,7,8,9-Heptachlorodibenzofuran	2.093E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	6.3E-08 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	1.3E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	Ν	0 1,2,3,7,8-Pentachlorodibenzofuran	9.85E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,4,7,8-Hexachlorodibenzofuran	2.577E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 2,3,4,6,7,8-Hexachlorodibenzofuran	4.88E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,4,6,7,8-Heptachlorodibenzofuran	6.645E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 2,3,4,7,8-Pentachlorodibenzofuran	2.65E-07 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 2,3,7,8-Tetrachlorodibenzofuran	1.36E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,7,8,9-Hexachlorodibenzofuran	1.85E-07 mg/kg	Ν	U Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,7,8-Pentachlorodibenzo-p-dioxin	7E-08 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,4,5,6,7,8-Octachlorodibenzo-p-dioxin	1.964E-06 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded
SA18	SA18-0.5	N	0 1,2,3,4,5,6,7,8-Octachlorodibenzofuran	1.441E-05 mg/kg	Y	Analyzed by both EPA Method 8290 and 8290 SCREEN. The data from EPA Method 8290 SCREEN was excluded

<u>Notes:</u> ft = feet

mg/kg = milligram per kilogram pCi/g = picocurie per gram EPA = Environmental Protetion Agency

FD = Field duplicate

N = Normal (Sample Type) N = Not detected (Detection Flag) SIM = Selective ion monitoring Y = Detected (Detection Flag)

J = The result is an estimated quantity. the associated numerical value is the approximate concentration of the analyte in the sample.

K = Qualified as estimated maximum possible concentration (dioxin congeners)

U = Not detected

UJ = The nondetected analyte was qualified as estimated at the sample quantitation limit. The reported sample quantitation limit is approximate and may be inaccurate or imprecise.

#### TABLE B-3. Summary of Rejected Parcel Soil Data Nevada Environmental Response Trust Site Henderson, Nevada

#### Phase A Investigation DVSR, September 2007

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
SA15-10	DU-3	EPA 350.1	Ammonia (as N)	< 0.64	mg/kg	R	m
SA18-0.5	DU-1	EPA 9012A	Cyanide	< 0.55	mg/kg	R	h
SA18-0.5D	DU-1	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h
SA18-10	DU-1	EPA 9012A	Cyanide	< 0.54	mg/kg	R	h
SA21-0.5	DU-1	EPA 9012A	Cyanide	< 0.52	mg/kg	R	h
SA3-0.5	DU-2	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h
SA3-0.5D	DU-2	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h
SA5-0.5	DU-2	EPA 9012A	Cyanide	< 0.65	mg/kg	R	h, m
SA6-0.5	DU-2	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h
SA6-0.5D	DU-2	EPA 9012A	Cyanide	< 0.58	mg/kg	R	h
SA6-0.5D	DU-2	SW 846 8081	Heptachlor	< 0.0020	mg/kg	R	m
SA7-0.5	DU-2	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h
SA7-0.5	DU-2	SW 846 9056	Nitrite	< 0.042	mg/kg	R	m
SA8-0.5	DU-2	EPA 9012A	Cyanide	< 0.53	mg/kg	R	h

#### Phase B Investigation Area I DVSR, January 2010

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
RSAO2-0.5B	DU-1	SW 846 9056	Chloride	222	mg/kg	R	0
RSAL2-10B	DU-1	SW 846 9056	Bromide	11	mg/kg	R	m
RSAL4-0.5B	DU-1	SW6010	Antimony	< 0.50	mg/kg	R	m
RSAL4009-0.5B	DU-1	SW6010	Antimony	< 0.50	mg/kg	R	m
SA69-0.5B	DU-1	SW6010	Antimony	< 0.50	mg/kg	R	m

#### Phase B Investigation Area II DVSR, February 2010

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
RSAL6-0.5B	DU-1	EPA 350.1	Ammonia (as N)	< 0.080	mg/kg	R	m
RSAL6-10B	DU-1	EPA 350.1	Ammonia (as N)	< 0.080	mg/kg	R	m
RSAO5-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.50	mg/kg	R	m
SA109-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.50	mg/kg	R	m
SA185-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.42	mg/kg	R	m
SA186-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.50	mg/kg	R	m
SA42009-10B	DU-1	SW 846 9012A	Cyanide	< 0.50	mg/kg	R	m
SA42-10B	DU-1	SW 846 9012A	Cyanide	< 0.42	mg/kg	R	m
SA44-10B	DU-1	SW 846 9012A	Cyanide	< 0.50	mg/kg	R	m
SA50-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.42	mg/kg	R	m
SA54-0.5B	DU-1	SW 846 9012A	Cyanide	< 0.42	mg/kg	R	m
SA73-0.5B	DU-1	EPA 350.1	Ammonia (as N)	< 0.10	mg/kg	R	m

#### Additional Pre-Confirmation Sampling DVSR, April 2011

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
SSAO8-05-0BPC	DU-1	SW 846 8270C	Benzo[a]pyrene	< 20	ug/kg	R	i
SSAO8-05-0BPC	DU-1	SW 846 8270C	Benzo[b]fluoranthene	< 27	ug/kg	R	i
SSAO8-05-0BPC	DU-1	SW 846 8270C	Benzo[g,h,i]perylene	< 16	ug/kg	R	i
SSAO8-05-0BPC	DU-1	SW 846 8270C	Benzo[k]fluoranthene	< 41	ug/kg	R	i
SSAO8-05-0BPC	DU-1	SW 846 8270C	Dibenzo(a,h)-anthracene	< 19	ug/kg	R	i
SSAO8-05-0BPC	DU-1	SW 846 8270C	Indeno[1,2,3-cd]pyrene	< 22	ug/kg	R	i

#### TABLE B-3. Summary of Rejected Parcel Soil Data Nevada Environmental Response Trust Site Henderson, Nevada

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
RISB-10-0.5-20141215	DU-1	EPA 8270C	Benzidine	< 0.66	mg/kg	R	m
RISB-11-0.5-20141217	DU-1	EPA 8270C	Benzidine	< 0.69	mg/kg	R	m
RISB-11-0.5-20141217	DU-1	EPA 6020	Niobium	< 1.9	mg/kg	R	m
RISB-11-5.0-20141217	DU-1	EPA 6020	Niobium	< 1.8	mg/kg	R	m
RISB-13-0.5-20141217	DU-1	EPA 6020	Niobium	< 1.9	mg/kg	R	m
RISB-13-5.0-20141218	DU-1	EPA 8270C	Benzoic acid	< 0.37	mg/kg	R	m
RISB-13-5.0-20141218	DU-1	EPA 6020	Niobium	< 2.0	mg/kg	R	m
RISB-47-0.5-20141120	DU-1	EPA 8270C	Benzidine	< 0.67	mg/kg	R	m
RISB-47-0.5-20141120	DU-1	EPA 6020	Niobium	< 1.9	mg/kg	R	m
RISB-47-5.0-20141120	DU-1	EPA 8270C	Benzidine	< 0.70	mg/kg	R	m
RISB-47-5.0-20141120	DU-1	EPA 6020	Niobium	< 1.9	mg/kg	R	m
RISB-51-0.5-20141030	DU-1	EPA 8081A	4,4'-DDD	< 0.0077	mg/kg	R	m
RISB-51-0.5-20141030	DU-1	EPA 8081A	gamma-Chlordane	< 0.0077	mg/kg	R	m
RISB-51-0.5-20141030	DU-1	EPA 8081A	Heptachlor	< 0.010	mg/kg	R	m
RISB-51-0.5-20141030	DU-1	EPA 8081A	Heptachlor epoxide	< 0.010	mg/kg	R	m

#### Phase I Remedial Investigation DVSR (ongoing, not submitted yet)

#### Tronox Parcels C, D, F, and G Investigation DVSR, February 2008

Sample ID	Decision Unit	Method	Analyte	Result	Unit	Validation Qualifier	Reason Code
TSB-GJ-03-0	DU-2	EPA 300	Nitrite (as N)	< 0.20	mg/kg	R	m
TSB-GJ-03-5	DU-2	EPA 300	Nitrite (as N)	< 0.21	mg/kg	R	m
TSB-GJ-04-0	DU-2	EPA 300	Nitrite (as N)	< 0.22	mg/kg	R	m
TSB-GJ-04-5	DU-2	EPA 300	Nitrite (as N)	< 0.22	mg/kg	R	m
TSB-GR-02-0	DU-2	EPA 300	Nitrite (as N)	< 0.21	mg/kg	R	m
TSB-GR-02-0 FD	DU-2	EPA 300	Nitrite (as N)	< 0.21	mg/kg	R	m
TSB-GR-02-5	DU-2	EPA 300	Nitrite (as N)	< 0.21	mg/kg	R	m

#### Notes:

DDD = Dichlorodiphenyldichloroethane

DU = Decision unit

DVSR = Data Validation Summary Report

EPA = Environmental Protection Agency

FD = Field duplicate

h = Qualified due to holding time exceedance

i = Qualified due to internal standard areas

m = Qualified due to matrix spike recoveries

mg/kg = milligram per kilogram

o = Other

R = Rejected value

µg/kg = microgram per kilogram

Phase A	Investigation	DVSR	Senter	her	2007
I Hase A	mvesugation	DV011,	oepten	INCI	2001

					RPD or		Quantitation	Final
Sample ID	Decision Unit	Analyte	Result	Unit	Difference	Limit	Limit	Qualifier
SA3-0.5	DU-2	Calcium	40,900	mg/kg	70	≤50		J
SA3-0.5D	DU-2	Calcium	19,800	mg/kg	70	≤50		J
SA6-0.5	DU-2	Titanium	361	mg/kg	52	≤50		J
SA6-0.5D	DU-2	Titanium	616	mg/kg	52	≤50		J
SA9-10	DU-1	Perchlorate	696	ug/kg	52	≤50		J
SA9-10D	DU-1	Perchlorate	408	ug/kg	52	≤50		L
SA11-0.5	DU-1	4,4-DDE	0.012	mg/kg	0.010	≤0.0019		J+
SA11-0.5D	DU-1	4,4-DDE	< 0.0019	mg/kg	0.010	≤0.0019	0.0019	UJ
SA11-0.5	DU-1	Beta-BHC	0.030	mg/kg	0.028	≤0.0019		J+
SA11-0.5D	DU-1	Beta-BHC	< 0.0019	mg/kg	0.028	≤0.0019	0.0019	UJ
SA11-0.5	DU-1	Sulfate	94	mg/kg	72	≤50		L
SA11-0.5D	DU-1	Sulfate	199	mg/kg	72	≤50		J
SA15-10	DU-3	Chloride	746	mg/kg	95	≤50		J
SA15-10D	DU-3	Chloride	266	mg/kg	95	≤50		J
SA15-10	DU-3	Nitrate	515	mg/kg	169	≤50		J
SA15-10D	DU-3	Nitrate	43	mg/kg	169	≤50		L
SA18-0.5	DU-1	Calcium	32,300	mg/kg	64	≤50		J
SA18-0.5D	DU-1	Calcium	16,600	mg/kg	64	≤50		J
SA18-0.5	DU-1	Chloride	543	mg/kg	62	≤50		J
SA18-0.5D	DU-1	Chloride	287	mg/kg	62	≤50		J
SA18-0.5	DU-1	Perchlorate	3,850	ug/kg	83	≤50		J
SA18-0.5D	DU-1	Perchlorate	1,590	ug/kg	83	≤50		J
SA18-0.5	DU-1	Sodium	800	mg/kg	60	≤50		J-
SA18-0.5D	DU-1	Sodium	433	mg/kg	60	≤50		J-

#### Phase B Investigation Area I DVSR, January 2010

					RPD or		Quantitation	Final
Sample ID	Decision Unit	Analyte	Result	Unit	Difference	Limit	Limit	Qualifier
SA152-0.5B	DU-1	1,2,3,4,5,6,7,8-Octachlorodibenzofuran	518	ng/kg	512	≤9.5		J
SA152009-0.5B	DU-1	1,2,3,4,5,6,7,8-Octachlorodibenzofuran	6.5	ng/kg	512	≤9.5	9.5	J
SA152-0.5B	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	187	ng/kg	184	≤4.8		J
SA152009-0.5B	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	2.5	ng/kg	184	≤4.8	4.8	J
SA152-0.5B	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	14	ng/kg	12	≤4.8	4.8	J
SA152009-0.5B	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	1.5	ng/kg	12	≤4.8	4.8	J
SA152-0.5B	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	92	ng/kg	91	≤4.8		J
SA152009-0.5B	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.94	ng/kg	91	≤4.8	4.8	JK
SA152-0.5B	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	107	ng/kg	106	≤4.8		J
SA152009-0.5B	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	0.95	ng/kg	106	≤4.8	4.8	J
SA152-0.5B	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	63	ng/kg	62	≤4.8		J
SA152009-0.5B	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	0.63	ng/kg	62	≤4.8	4.8	JK
SA152-0.5B	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	10	ng/kg	9.7	≤4.8	4.8	J
SA152009-0.5B	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	< 0.31	ng/kg	9.7	≤4.8	4.8	UJ
SA152-0.5B	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	5.9	ng/kg	5.3	≤4.8	4.8	J
SA152009-0.5B	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.61	ng/kg	5.3	≤4.8	4.8	J
SA152-0.5B	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	52	ng/kg	52	≤4.8		J
SA152009-0.5B	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	< 0.15	ng/kg	52	≤4.8	4.8	UJ
SA152-0.5B	DU-1	2,3,4,6,7,8-Hexachlorodibenzofuran	16	ng/kg	16	≤4.8	4.8	JK
SA152009-0.5B	DU-1	2,3,4,6,7,8-Hexachlorodibenzofuran	< 0.27	ng/kg	16	≤4.8	4.8	UJ
SA152-0.5B	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	28	ng/kg	28	≤4.8		J
SA152009-0.5B	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	< 0.15	ng/kg	28	≤4.8	4.8	UJ
SA180-0.5B	DU-1	4,4'-DDD	< 76	ug/kg	68	≤3.8	76	UJ
SA180-0.5BD	DU-1	4,4-DDD	7.9	ug/kg	68	≤3.8	3.8	J
RSAH3-0.5B	DU-1	4,4-DDE	73	ug/kg	39	≤34	34	J
RSAH3009-0.5B	DU-1	4,4-DDE	< 34	ug/kg	39	≤34	34	
RSAH3-0.5B	DU-1	4,4-DDT	69	ug/kg	35	≤34	34	J
RSAH3009-0.5B	DU-1	4,4-DDT	< 34	ug/kg	35	≤34	34	
SA57-10B	DU-1	Acetone	17	ug/kg	23	≤22	22	J
SA57-10BD	DU-1	Acetone	40	ug/kg	23	≤22	22	J
RSAH3-0.5B	DU-1	Ammonia (as N)	< 0.070	mg/kg	2.0	≤0.52	0.52	UJ
RSAH3009-0.5B	DU-1	Ammonia (as N)	2.1	mg/kg	2.0	≤0.52	0.52	J
SA180-0.5B	DU-1	Benz(a)anthracene	160	ug/kg	131	≤40	40	J
SA180-0.5BD	DU-1	Benz(a)anthracene	29	ug/kg	131	≤40	40	J
SA180-0.5B	DU-1	Benzo(a)pyrene	110	ug/kg	78	≤40	40	J
SA180-0.5BD	DU-1	Benzo(a)pyrene	32	ug/kg	78	≤40	40	J
SA180-0.5B	DU-1	Benzo(b)fluoranthene	110	ug/kg	78	≤40	40	J
SA180-0.5BD	DU-1	Benzo(b)fluoranthene	32	ug/kg	78	≤40	40	J
SA180-0.5B	DU-1	Benzo(g,h,i)perylene	74	ug/kg	47	≤40	40	J
SA180-0.5BD	DU-1	Benzo(g,h,i)perylene	27	ug/kg	47	≤40	40	J

SA180-0.5B	DU-1	Benzo(k)fluoranthene	110	ug/kg	77	≤40	40	J
SA180-0.5BD	DU-1	Benzo(k)fluoranthene	33	ug/kg	77	≤40	40	J
RSAH3-0.5B	DU-1	Beta-BHC	230	ug/kg	165	≤18		J
RSAH3009-0.5B	DU-1	Beta-BHC	65	ug/kg	165	≤18	18	J
RSAK4-0.5B	DU-1	Beta-BHC	330	ug/kg	53	≤50		J
RSAK4009-0.5B	DU-1	Beta-BHC	570	ug/kg	53	≤50		J
RSAH3-0.5B	DU-1	Calcium	19,700	mg/kg	57	≤50		J
RSAH3009-0.5B	DU-1	Calcium	35,300	mg/kg	57	≤50		J
SA180-0.5B	DU-1	Chrysene	190	ug/kg	144	≤40	40	J
SA180-0.5BD	DU-1	Chrysene	46	ug/kg	144	≤40	40	J
SA180-0.5B	DU-1	Fluoranthene	350	ug/kg	279	≤40		J
SA180-0.5BD	DU-1	Fluoranthene	71	ug/kg	279	≤40	40	J
RSAH3-0.5B	DU-1	Hexachlorobenzene	180	ug/kg	156	≤18		J
RSAH3009-0.5B	DU-1	Hexachlorobenzene	24	ug/kg	156	≤18	18	J
SA180-0.5B	DU-1	Indeno(1,2,3-cd)pyrene	69	ug/kg	46	≤40	40	J
SA180-0.5BD	DU-1	Indeno(1,2,3-cd)pyrene	23	ug/kg	46	≤40	40	J
RSAH3-0.5B	DU-1	Octachlorostyrene	79	ug/kg	67	≤6.8		J
RSAH3009-0.5B	DU-1	Octachlorostyrene	12	ug/kg	67	≤6.8	6.8	J
SA180-0.5B	DU-1	Phenanthrene	190	ug/kg	149	≤40	40	J
SA180-0.5BD	DU-1	Phenanthrene	41	ug/kg	149	≤40	40	J
SA180-0.5B	DU-1	Pyrene	270	ug/kg	219	≤40		J
SA180-0.5BD	DU-1	Pyrene	51	ug/kg	219	≤40	40	J
SA74-0.5B	DU-1	Radium-226	0.84	pCi/g	0.55	≤0.50	0.50	J
SA74009-0.5B	DU-1	Radium-226	0.29	pCi/g	0.55	≤0.50	0.50	J
RSAH3-0.5B	DU-1	Radium-228	1.5	pCi/g	1.25	≤0.50	0.50	J
RSAH3009-0.5B	DU-1	Radium-228	2.7	pCi/g	1.25	≤0.50		J
RSAK4-0.5B	DU-1	Radium-228	1.5	pCi/g	0.75	≤0.50	0.50	J
RSAK4009-0.5B	DU-1	Radium-228	0.79	pCi/g	0.75	≤0.50	0.50	J
SA74-0.5B	DU-1	Total Phosphorus-P	731	mg/kg	51	≤50		J
SA74009-0.5B	DU-1	Total Phosphorus-P	1,230	mg/kg	51	≤50		J
RSAH3-0.5B	DU-1	Tungsten	0.43	mg/kg	0.30	≤0.10	0.10	J
RSAH3009-0.5B	DU-1	Tungsten	0.13	mg/kg	0.30	≤0.10	0.10	J
RSAN4-10B	DU-1	Tungsten	0.70	mg/kg	0.47	≤0.10		J
RSAN4009-10B	DU-1	Tungsten	0.23	mg/kg	0.47	≤0.10	0.10	J

#### Phase B Investigation Area II DVSR, February 2010

					RPD or		Quantitation	Final
Sample ID	Decision Unit	Analyte	Result	Unit	Difference	Limit	Limit	Qualifier
SA86-10B	DU-3	4,4-DDE	18	ug/kg	13	≤3.8	3.8	J
SA86009-10B	DU-3	4,4-DDE	4.6	ug/kg	13	≤3.8	3.8	J
SA86-10B	DU-3	4,4-DDT	11	ug/kg	6.8	≤3.8	3.8	J
SA86009-10B	DU-3	4,4-DDT	4.2	ug/kg	6.8	≤3.8	3.8	J
SA104-10B	DU-3	Acetone	52	ug/kg	36	≤19	19	J
SA104009-10B	DU-3	Acetone	16	ug/kg	36	≤19	19	J
RSAN6-10B	DU-1	Acetone	33	ug/kg	26	≤19	19	J
RSAN6009-10B	DU-1	Acetone	< 7.4	ug/kg	26	≤19	19	UJ
SA104-10B	DU-3	Ammonia (as N)	1.8	mg/kg	0.86	≤0.55	0.55	J
SA104009-10B	DU-3	Ammonia (as N)	0.93	mg/kg	0.86	≤0.55	0.55	J
SA86-10B	DU-3	Arsenic	2.4	mg/kg	54	≤50		J
SA86009-10B	DU-3	Arsenic	4.1	mg/kg	54	≤50		J
SA86-10B	DU-3	Beta-BHC	< 0.94	ug/kg	25	≤1.9	1.9	UJ
SA86009-10B	DU-3	Beta-BHC	26	ug/kg	25	≤1.9		J
RSAN6-10B	DU-1	Beta-BHC	5.8	ug/kg	4.9	≤1.9	1.9	J
RSAN6009-10B	DU-1	Beta-BHC	< 0.93	ug/kg	4.9	≤1.9	1.9	UJ
SA104-10B	DU-3	Chromium-hexavalent	0.66	mg/kg	0.47	≤0.44	0.44	J
SA104009-10B	DU-3	Chromium-hexavalent	< 0.19	mg/kg	0.47	≤0.44	0.44	UJ
SA86-10B	DU-3	Endrin Ketone	20	ug/kg	17	≤3.8		J
SA86009-10B	DU-3	Endrin Ketone	3.0	ug/kg	17	≤3.8	3.8	J
SA86-10B	DU-3	Hexachlorobenzene	290	ug/kg	192	≤20		J
SA86009-10B	DU-3	Hexachlorobenzene	98	ug/kg	192	≤20	20	J
RSAN6-10B	DU-1	Hexachlorobenzene	24	ug/kg	23	≤1.9		J
RSAN6009-10B	DU-1	Hexachlorobenzene	< 0.93	ug/kg	23	≤1.9	1.9	UJ
SA104-10B	DU-3	Molybdenum	0.74	mg/kg	0.35	≤0.33	0.33	J
SA104009-10B	DU-3	Molybdenum	0.39	mg/kg	0.35	≤0.33	0.33	J
SA86-10B	DU-3	Octachlorostyrene	68	ug/kg	46	≤7.6		J
SA86009-10B	DU-3	Octachlorostyrene	22	ug/kg	46	≤7.6	7.6	J
SA104-10B	DU-3	Octachlorostyrene	34	ug/kg	30	≤7.3	7.3	L
SA104009-10B	DU-3	Octachlorostyrene	64	ug/kg	30	≤7.3		J
SA86-10B	DU-3	PCB-105	< 165	ng/kg	135	≤40	40	UJ
SA86009-10B	DU-3	PCB-105	<30	ng/kg	135	≤40	40	UJ

SA86-10B	DU-3	PCB-118	< 353	ng/kg	274	≤101	101	UJ
SA86009-10B	DU-3	PCB-118	< 79	ng/kg	274	≤101	101	UJ
SA42-10B	DU-1	Phenanthrene	< 29	ug/kg	971	≤110	110	UJ
SA42009-10B	DU-1	Phenanthrene	1,000	ug/kg	971	≤110		J
SA86-10B	DU-3	Radium-226	1.0	pCi/g	0.55	≤0.50	0.50	J
SA86009-10B	DU-3	Radium-226	1.6	pCi/g	0.55	≤0.50	0.50	J
SA105-10B	DU-3	Radium-228	1.6	pCi/g	0.53	≤0.50	0.50	J
SA105009-10B	DU-3	Radium-228	1.0	pCi/g	0.53	≤0.50	0.50	J
RSAM7-10B	DU-1	Radium-228	1.2	pCi/g	0.76	≤0.50	0.50	J
RSAM7009-10B	DU-1	Radium-228	0.42	pCi/g	0.76	≤0.50	0.50	J
RSAM7-10B	DU-1	Uranium-235	0.055	pCi/g	0.049	≤0.040	0.040	J
RSAM7009-10B	DU-1	Uranium-235	0.10	pCi/g	0.049	≤0.040	0.040	J
SA86-10B	DU-3	Uranium (total)	1.6	mg/kg	76	≤50		J
SA86009-10B	DU-3	Uranium (total)	3.6	mg/kg	76	≤50		J

#### Phase B Investigation Area III DVSR, February 2010

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
SA141-14B	DU-1	Calcium	51,300	mg/kg	110	≤50		J
SA141009-14B	DU-1	Calcium	15,000	mg/kg	110	≤50		J
SA157-0.5B	DU-1	Chromium-hexavalent	< 0.19	mg/kg	1.8	≤0.43	0.43	UJ
SA157009-0.5B	DU-1	Chromium-hexavalent	2.0	mg/kg	1.8	≤0.43	0.43	J
SA157-0.5B	DU-1	Manganese	1,550	mg/kg	101	≤50		J
SA157009-0.5B	DU-1	Manganese	508	mg/kg	101	≤50		J
SA141-14B	DU-1	Octachlorodibenzofuran	13	pg/g	7.6	≤5.4	5.4	J
SA141009-14B	DU-1	Octachlorodibenzofuran	5.4	pg/g	7.6	≤5.4	5.4	J
SA157-0.5B	DU-1	Radium-228	1.8	pCi/g	0.98	≤0.50	0.50	J
SA157009-0.5B	DU-1	Radium-228	0.78	pCi/g	0.98	≤0.50	0.50	J
SA157-0.5B	DU-1	Uranium-235	0.090	pCi/g	0.047	≤0.040	0.040	J
SA157009-0.5B	DU-1	Uranium-235	0.044	pCi/g	0.047	≤0.040	0.040	J

#### Phase B Investigation Area IV DVSR, March 2010

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
SA121-0.5B	DU-2	Chromium-hexavalent	1.6	mg/kg	1.4	≤0.42	0.42	J
SA121009-0.5B	DU-2	Chromium-hexavalent	< 0.19	mg/kg	1.4	≤0.42	0.42	UJ
SA121-0.5B	DU-2	Hexachlorobenzene	3.5	ug/kg	2.3	≤1.9	1.9	J
SA121009-0.5B	DU-2	Hexachlorobenzene	5.8	ug/kg	2.3	≤1.9	1.9	J
SA121-0.5B	DU-2	Radium-228	0.86	pCi/g	0.54	≤0.50	0.50	J
SA121009-0.5B	DU-2	Radium-228	< 0.32	pCi/g	0.54	≤0.50	0.50	UJ

#### Phase B Supplemental Sampling DVSR, July 2010

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
SA114-1BR	DU-1	Mercury	0.012	mg/kg	158	≤50		J
SA114009-1BR	DU-1	Mercury	0.10	mg/kg	158	≤50		J

#### Additional Pre-Confirmation Sampling DVSR, April 2011

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
BDT-2-S-5-12BPC	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	110	pg/g	114	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	400	pg/g	114	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	8.5	pg/g	4.7	≤2.7	2.7	J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	3.8	pg/g	4.7	≤2.7	2.7	J
SSAK7-05-1BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	160	pg/g	76	≤50		J
SSAK7-05-1BPC_FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	72	pg/g	76	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	250	pg/g	126	≤50		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	57	pg/g	126	≤50		J
SSAM5-04-5BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	4.9	pg/g	4.7	≤2.7	2.7	J
SSAM5-04-5BPC_FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	9.6	pg/g	4.7	≤2.7	2.7	J
BDT-2-N-5-12BPC	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzofuran	17	pg/g	11	≤2.7		J
BDT-2-N-5-12BPC_FD	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzofuran	6.2	pg/g	11	≤2.7	2.7	J

BDT-2-S-5-12BPC	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzofuran	1,300	pg/g	110	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,4,6,7,8-Heptachlorodibenzofuran	4,500	pg/g	110	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	110	pg/g	80	≤50		J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	47	pg/g	80	≤50		J
SSAJ8-03-3BPC	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	51	pg/g	41	≤2.8		J
SSAJ8-03-3BPC FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	10	pq/q	41	≤2.8	2.8	J
SSAK7-05-1BPC	DU-1	1.2.3.4.6.7.8-Heptachlorodibenzofuran	1.900	pa/a	77	≤50		J
SSAK7-05-1BPC FD	DU-1	1.2.3.4.6.7.8-Heptachlorodibenzofuran	840	pa/a	77	≤50		J
SSAK8-08-3BPC	DU-1	1 2 3 4 6 7 8-Heptachlorodibenzofuran	2 700	pg/g	125	_sc ≤50		J
SSAK8-08-3BPC ED	DU-1	1 2 3 4 6 7 8-Heptachlorodibenzofuran	620	pg/g	125	<50		• ا
SSAM5-04-5BPC	DUL1	1 2 3 4 6 7 8-Heptachlorodibenzofuran	60	pg/g	67	<50		Г
	DU-1	1,2,3,4,0,7,9-1 epitechiorodiberizoidian	120	pg/g	67	<u> </u>		J 1
SSAM5-04-50FC_FD	DU-1		6.7	pg/g	19	<u>-200</u>	2.6	J
SSAF3-04-1_01_BFC	DU-1		0.7	pg/g	10	<u>-</u> 2.0	2.0	J
SSAP3-04-1_01_BPC_PD	DU-1		25	pg/g	10	≥2.0 <2.7	2.7	J
BDT-2-N-5-12BPC	DU-3		9.6	pg/g	0.1	≤2.7	2.7	J
BDT-2-N-5-12BPC_FD	DU-3	1,2,3,4,7,8,9-Heptachlorodibenzofuran	3.5	pg/g	6.1	\$2.7	2.7	J
BD1-2-S-5-12BPC	DU-3	1,2,3,4,7,8,9-Heptachlorodibenzofuran	620	pg/g	105	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,4,7,8,9-Heptachlorodibenzofuran	2,000	pg/g	105	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	50	pg/g	90	≤50		J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	19	pg/g	90	≤50		J
SSAJ8-03-3BPC	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	21	pg/g	16	≤2.8		J
SSAJ8-03-3BPC_FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	4.6	pg/g	16	≤2.8	2.8	J
SSAK7-05-1BPC	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	880	pg/g	75	≤50		J
SSAK7-05-1BPC_FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	400	pg/g	75	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	1,100	pg/g	124	≤50		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	260	pg/g	124	≤50		J
SSAM5-04-5BPC	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	27	pg/g	71	≤50		J
SSAM5-04-5BPC_FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	57	pg/g	71	≤50		JK
SSAP3-04-1 01 BPC	DU-1	1.2.3.4.7.8.9-Heptachlorodibenzofuran	3.1	pa/a	8.9	≤2.6	2.6	J
SSAP3-04-1 01 BPC FD	DU-1	1.2.3.4.7.8.9-Heptachlorodibenzofuran	12	pa/a	8.9	≤2.6	2.6	J
BDT-2-S-5-12BPC	DU-3	1 2 3 4 7 8-Hexachlorodibenzo-p-dioxin	19	pg/g	106	<50	2.0	
BDT-2-S-5-12BPC ED	DU-3	1 2 3 4 7 8-Hexachlorodibenzo-p-dioxin	62	pg/g	106	<50		
BDT-3-S-10-4BPC	DU-1	1 2 3 4 7 8-Heyachlorodibenzo-p-dioxin	15	pg/g	4.0	<2.6		Г
BDT-3-S-10-4BPC ED		1,2,3,4,7,8 Hexachlorodibenzo-p-dioxin	11	pg/g	4.0	<2.0	2.6	J
SSAK7 05 1990		1,2,3,4,7,0-1 lexachlorodibenzo p dioxin	17	pg/g	4.0	-22.0 <2.6	2.0	J 1
SSAKT-05-TBPC	DU-1		0.7	pg/g	7.3	<u>-22.0</u>	26	J
SSAK7-05-TBPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	9.7	pg/g	7.3	≤2.0	2.0	J
SSAK8-08-3BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	40	pg/g	32	≤2.7	0.7	J
SSAK8-08-3BPC_FD	DU-1	1,2,3,4,7,8-Hexachiorodibenzo-p-dioxin	7.9	pg/g	32	\$2.7	2.7	J
BDT-2-N-5-12BPC	DU-3	1,2,3,4,7,8-Hexachlorodibenzoturan	5.8	pg/g	3.4	≤2.7	2.7	J
BDT-2-N-5-12BPC_FD	DU-3	1,2,3,4,7,8-Hexachlorodibenzofuran	2.4	pg/g	3.4	≤2.7	2.7	J
BDT-2-S-5-12BPC	DU-3	1,2,3,4,7,8-Hexachlorodibenzofuran	410	pg/g	91	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,4,7,8-Hexachlorodibenzofuran	1,100	pg/g	91	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	33	pg/g	64	≤50		J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	17	pg/g	64	≤50		J
SSAJ8-03-3BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	16	pg/g	13	≤2.8		J
SSAJ8-03-3BPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	3.0	pg/g	13	≤2.8	2.8	J
SSAK7-05-1BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	800	pg/g	58	≤50		J
SSAK7-05-1BPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	440	pg/g	58	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	820	pg/g	128	≤50		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	180	pg/g	128	≤50		J
SSAM5-04-5BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	16	pg/g	69	≤50		J
SSAM5-04-5BPC_FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	33	pg/g	69	≤50		J
SSAP3-04-1_01_BPC	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	2.3	pg/a	5.0	≤2.6	2.6	J
SSAP3-04-1 01 BPC FD	DU-1	1.2.3.4.7.8-Hexachlorodibenzofuran	7.3	pa/a	5.0	≤2.6	2.6	L
BDT-2-S-5-12BPC	DU-3	1 2 3 6 7 8-Hexachlorodibenzo-n-dioxin	33	6.6.4	108	≤50		,i
BDT-2-S-5-12BPC ED	DU-3	1 2 3 6 7 8-Hexachlorodibenzo-n-diovin	110	P9/9	108	<50		.1
SSAK7-05-18PC		1 2 3 6 7 8-Hexachlorodibenzo-n-diovin	20	P9/9	52	<50		1
		1 2 3 6 7 8-Heyzohlorodihanza a diavia	23	P9/9	52	-50 <50		Г
		1,2,0,0,7,0-FIEXACHIOTOUIDEHZO-p-UIOXIII	23	pg/g	125	-300 <50	l	J
SSANO-UO-3070			18	pg/g	120	 ∠50	l	J
DDT ON 5 10000	DU-1		18	pg/g	125	>00≤	~ 7	J
BD1-2-N-5-12BPC	DU-3	1,2,3,6,7,8-Hexachlorodibenzofuran	5.2	pg/g	3.1	≤2.7	2.7	J

BDT-2-N-5-12BPC_FD	DU-3	1,2,3,6,7,8-Hexachlorodibenzofuran	2.1	pg/g	3.1	≤2.7	2.7	JK
BDT-2-S-5-12BPC	DU-3	1,2,3,6,7,8-Hexachlorodibenzofuran	320	pg/g	103	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,6,7,8-Hexachlorodibenzofuran	1,000	pg/g	103	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	29	pg/g	16	≤2.7		J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	13	pg/g	16	≤2.7	2.7	J
SSAJ8-03-3BPC	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	13	pg/g	10	≤2.8	2.8	J
SSAJ8-03-3BPC FD	DU-1	1.2.3.6.7.8-Hexachlorodibenzofuran	2.7	pa/a	10	≤2.8	2.8	J
SSAK7-05-1BPC	DU-1	1.2.3.6.7.8-Hexachlorodibenzofuran	580	pa/a	61	≤50	-	J
SSAK7-05-1BPC FD	DU-1	1.2.3.6.7.8-Hexachlorodibenzofuran	310	pa/a	61	≤50		J
SSAK8-08-3BPC	DU-1	1 2 3 6 7 8-Hexachlorodibenzofuran	720	pa/a	120	≤50		J
SSAK8-08-3BPC ED	DU-1	1 2 3 6 7 8-Hexachlorodibenzofuran	180	pg/g	120	<50		.1
SSAM5-04-5BPC	DU-1	1 2 3 6 7 8-Hexachlorodibenzofuran	14	pg/g	67	<50		Т
SSAM5-04-5BPC ED	DU-1	1 2 3 6 7 8-Hexachlorodibenzofuran	28	pg/g	67	_00 <50		J
SSAR3-04-1 01 BPC	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	1.8	pg/g	33	0 <2.6	2.6	J
SSAP3-04-1_01_BPC_ED	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	5.1	pg/g	3.3	<u>-</u> 2.0	2.0	J
	DU-1		10	pg/g	3.3	<u>52.0</u>	2.0	J
	DU-3		19	pg/g	114	≥50 <50		JK
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	69	pg/g	114	<u>≤</u> 50		J
SSAK7-05-1BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	37	pg/g	64	≤50		J
SSAK7-05-1BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	19	pg/g	64	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	56	pg/g	43	≤2.7		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	13	pg/g	43	≤2.7	2.7	J
BDT-2-S-5-12BPC	DU-3	1,2,3,7,8,9-Hexachlorodibenzofuran	82	pg/g	91	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,7,8,9-Hexachlorodibenzofuran	220	pg/g	91	≤50		J
BDT-3-N-15-6BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	11	pg/g	4.4	≤2.7	2.7	J
BDT-3-N-15-6BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	6.6	pg/g	4.4	≤2.7	2.7	J
SSAK7-05-1BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	84	pg/g	60	≤50		J
SSAK7-05-1BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	45	pg/g	60	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	110	pg/g	128	≤50		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	24	pg/g	128	≤50		J
SSAM5-04-5BPC	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	2.6	pg/g	3.1	≤2.7	2.7	JK
SSAM5-04-5BPC_FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	5.7	pg/g	3.1	≤2.7	2.7	J
BDT-2-S-5-12BPC	DU-3	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	27	pg/g	100	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	81	pg/g	100	≤50		J
SSAK8-08-3BPC	DU-1	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	52	pg/g	39	≤2.7		J
SSAK8-08-3BPC_FD	DU-1	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	13	pg/g	39	≤2.7	2.7	J
BDT-2-S-5-12BPC	DU-3	1,2,3,7,8-Pentachlorodibenzofuran	290	pg/g	100	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	1,2,3,7,8-Pentachlorodibenzofuran	870	pg/g	100	≤50		J
BDT-3-N-15-6BPC	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	56	pg/g	67	≤50		J
BDT-3-N-15-6BPC_FD	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	28	pg/g	67	≤50		J
BDT-3-N-20-8BPC	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	20	pg/g	11	≤2.7		J
BDT-3-N-20-8BPC_FD	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	8.6	pg/g	11	≤2.7	2.7	J
SSAJ8-03-3BPC	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	11	pq/q	8.8	≤2.8	2.8	J
SSAJ8-03-3BPC FD	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	2.2	pq/q	8.8	≤2.8	2.8	J
SSAK8-08-3BPC	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	540	pq/q	122	≤50		J
SSAK8-08-3BPC FD	DU-1	1.2.3.7.8-Pentachlorodibenzofuran	130	pa/a	122	≤50		J
SSAM5-04-5BPC	DU-1	1.2.3.7.8-Pentachlorodibenzofuran	11	pa/a	11	≤2.7	2.7	J
SSAM5-04-5BPC FD	DU-1	1.2.3.7.8-Pentachlorodibenzofuran	22	pa/a	11	≤2.7		J
BDT-2-S-5-12BPC	DU-3	2 3 4 6 7 8-Hexachlorodibenzofuran	66	pg/g	104	≤50		J
BDT-2-S-5-12BPC FD	DU-3	2 3 4 6 7 8-Hexachlorodibenzofuran	210	pg/g	104	<50		.1
BDT-3-N-15-6BPC	DU-1	2 3 4 6 7 8-Hexachlorodibenzofuran	10	pg/g	3.1	<27	27	Т
BDT-3-N-15-6BPC ED	DU-1	2 3 4 6 7 8-Hexachlorodibenzofuran	69	pg/g	3.1	<2.7	2.7	о 1
BDT-3-N-20-8BPC	DU-1	2,3,4,6,7,8 Hexachlorodibenzofuran	5.7	pg/g	2.0	<2.7	2.1	J
	DU 1		2.0	pg/g	2.9	-22.7	2.7	J
	DU-1		2.0	pg/g	2.9	<u> </u>	2.1	J
SSANT-UD-10PU			150	pg/g	00	 		J
554K1-05-18PC_FD	DU-1		84	pg/g	00	≥5U <50		J
SSAND-UB-JBPU	DU-1		150	pg/g	105	<u>≤</u> 50		J
SSAKO-US-JEPC_FD	DU-1		4/	pg/g	105	<u>≤</u> 50	0.7	J
SSAM5-04-5BPC	DU-1	2,3,4,6,7,8-Hexachlorodibenzoturan	3.4	pg/g	3.7	≤2.7	2.7	J
SSAM5-04-5BPC_FD	DU-1	2,3,4,6,7,8-Hexachlorodibenzofuran	7.1	pg/g	3.7	≤2.7	2.7	J
ВUT-2-S-5-12BPC	DU-3	2,3,4,7,8-Pentachlorodibenzofuran	130	pg/g	102	≤50		J
BDT-2-S-5-12BPC FD	DU-3	2,3,4,7,8-Pentachlorodibenzofuran	400	pg/g	102	≤50		J

BDT-3-N-15-6BPC	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	30	pg/g	67	≤50		J
BDT-3-N-15-6BPC_FD	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	15	pg/g	67	≤50		J
BDT-3-N-20-8BPC	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	11	pg/g	6.5	≤2.7	2.7	J
BDT-3-N-20-8BPC_FD	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	4.5	pg/g	6.5	≤2.7	2.7	J
SSAJ8-03-3BPC	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	6.0	pg/g	4.9	≤2.8	2.8	J
SSAJ8-03-3BPC FD	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	1.1	pq/q	4.9	≤2.8	2.8	J
 SSAK8-08-3BPC	DU-1	2.3.4.7.8-Pentachlorodibenzofuran	300	pa/a	126	≤50		J
SSAK8-08-3BPC FD	DU-1	2.3.4.7.8-Pentachlorodibenzofuran	68	pa/a	126	≤50		J
SSAM5-04-5BPC	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	5.1	pg/g	6.9	_00 ≤2.7	27	JK
SSAM5-04-5BPC ED	DU-1	2 3 4 7 8-Pentachlorodibenzofuran	12	pg/g	6.9	<2.7	27	
BDT-2-S-5-12BPC	DU-3	2 3 7 8-Tetrachlorodibenzo-p-dioxin	8.2	P9/9	0.0	<50	2.1	
	DU-3	2,3,7,0-Tetrachlorodibenzo-p-dioxin	24	pg/g	90	 <50		J
BDT 2 N 45 CPDC	DU-3	2,3,7,8-Tetrachlorodiberizo-p-dioxin	1 5	pg/g	90	<u>-</u> 500	0.55	J
	DU-1	2,3,7,8-Tetrachlorodiberizo-p-dioxin	0.02	pg/g	0.08	≤0.00 <0.55	0.55	J
	DU-1	2,3,7,6-Tetrachlorodibenzo-p-dioxin	0.02	pg/g	0.80	≤0.55 <0.52	0.55	J
	DU-1	2,3,7,8-1 etrachlorodibenzo-p-dioxin	1.1	pg/g	0.80	S0.52	0.52	J
BD1-3-N-5-8BPC_FD	DU-1	2,3,7,8-1 etrachiorodibenzo-p-dioxin	1.9	pg/g	0.80	≤0.52	0.52	J
SSAK8-08-3BPC	DU-1	2,3,7,8-1 etrachlorodibenzo-p-dioxin	13	pg/g	95	≤50		J
SSAK8-08-3BPC_FD	DU-1	2,3,7,8-Tetrachlorodibenzo-p-dioxin	4.6	pg/g	95	≤50		J
BDT-2-S-5-12BPC	DU-3	2,3,7,8-Tetrachlorodibenzofuran	150	pg/g	93	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	2,3,7,8-Tetrachlorodibenzofuran	410	pg/g	93	≤50		J
BDT-3-N-15-6BPC	DU-1	2,3,7,8-Tetrachlorodibenzofuran	41	pg/g	88	≤50		J
BDT-3-N-15-6BPC_FD	DU-1	2,3,7,8-Tetrachlorodibenzofuran	16	pg/g	88	≤50		J
BDT-3-N-20-8BPC	DU-1	2,3,7,8-Tetrachlorodibenzofuran	12	pg/g	93	≤50		J
BDT-3-N-20-8BPC_FD	DU-1	2,3,7,8-Tetrachlorodibenzofuran	4.4	pg/g	93	≤50		J
BDT-3-N-5-8BPC	DU-1	2,3,7,8-Tetrachlorodibenzofuran	24	pg/g	95	≤50		J
BDT-3-N-5-8BPC_FD	DU-1	2,3,7,8-Tetrachlorodibenzofuran	67	pg/g	95	≤50		J
SSAJ8-03-3BPC	DU-1	2,3,7,8-Tetrachlorodibenzofuran	6.0	pg/g	4.8	≤0.55		J
SSAJ8-03-3BPC_FD	DU-1	2,3,7,8-Tetrachlorodibenzofuran	1.2	pg/g	4.8	≤0.55	0.55	J
SSAK8-08-3BPC	DU-1	2.3.7.8-Tetrachlorodibenzofuran	290	pa/a	116	≤50		J
SSAK8-08-3BPC FD	DU-1	2.3.7.8-Tetrachlorodibenzofuran	77	pa/a	116	≤50		J
SSAM5-04-5BPC	DU-1	2 3 7 8-Tetrachlorodibenzofuran	6.2	pg/g	64	<50		
SSAM5-04-5BPC ED	DU-1	2 3 7 8-Tetrachlorodibenzofuran	12	pg/g	64	<50		
BDT-4-S-20-8BPC	DU-1		4.4	P9/9	29	<1.8	1.8	1
BDT-4-S-20-8BPC ED	DU-1	4,4 DDE	1.5	ug/kg	2.0	<1.8	1.0	1
	DU-1	4,4-DDL	1.3	ug/kg	2.3	21.0	1.0	J 1
	DU-1		4.3	ug/kg	23	21.0	1.0	J
BDT-1-5-15-2BPC_FD	DU-1	beta-BHC	27	ug/kg	23	51.8		J
BDT-4-N-15-2BPC	DU-1	Hexachlorobenzene	21	ug/kg	78	≥50		J
BDT-4-N-15-2BPC_FD	DU-1	Hexachiorobenzene	9.2	ug/kg	78	<u>≤</u> 50		J
SSAM5-04-5BPC	DU-1	Hexachlorobenzene	51	ug/kg	99	≤50		J
SSAM5-04-5BPC_FD	DU-1	Hexachlorobenzene	150	ug/kg	99	≤50		J
SSAO8-12-0BPC	DU-1	Naphthalene	7.5	ug/kg	3.6	≤3.1	3.1	J
SSAO8-12-0BPC_FD	DU-1	Naphthalene	3.9	ug/kg	3.6	≤3.1	3.1	J
BDT-2-S-5-12BPC	DU-3	Octachlorodibenzodioxin	440	pg/g	93	≤50		J
BDT-2-S-5-12BPC_FD	DU-3	Octachlorodibenzodioxin	1,200	pg/g	93	≤50		J
BDT-3-N-20-8BPC	DU-1	Octachlorodibenzodioxin	11	pg/g	6.0	≤5.4	5.4	J
BDT-3-N-20-8BPC_FD	DU-1	Octachlorodibenzodioxin	< 5.0	pg/g	6.0	≤5.4	5.4	UJ
SSAK7-05-1BPC	DU-1	Octachlorodibenzodioxin	150	pg/g	88	≤50		J
SSAK7-05-1BPC_FD	DU-1	Octachlorodibenzodioxin	58	pg/g	88	≤50		J
SSAK8-08-3BPC	DU-1	Octachlorodibenzodioxin	270	pg/g	125	≤50		J
SSAK8-08-3BPC_FD	DU-1	Octachlorodibenzodioxin	62	pg/g	125	≤50		J
SSAM5-04-5BPC	DU-1	Octachlorodibenzodioxin	6.9	pg/g	8.1	≤5.4	5.4	J
SSAM5-04-5BPC_FD	DU-1	Octachlorodibenzodioxin	15	pg/g	8.1	≤5.4	5.4	J
BDT-2-N-5-12BPC	DU-3	Octachlorodibenzofuran	41	pg/g	25	≤5.4		J
BDT-2-N-5-12BPC FD	DU-3	Octachlorodibenzofuran	16	pa/a	25	≤5.4	5.4	J
BDT-2-S-5-12BPC	DU-3	Octachlorodibenzofuran	3 900	0/0 9	113	≤50		
BDT-2-S-5-12BPC FD	DU-3	Octachlorodibenzofuran	14 000	P9/9	113	<50		.1
BDT-3-N-20-8BPC		Octachlorodibenzofuran	260	P9/9	۵ <u>۵</u>	<50		1
		Octachlorodiberzofuran	200	P9/9	90		<sup> </sup>	J 1
			33	P9/9	90	 ∠E E		J
55AJ0-03-30PC FD			00	pg/g	90	≥0.0 <5.5		J
SSAJ8-U3-3BPC_FD	DU-1		20	pg/g	90	<u>≤</u> 5.5	5.5	J
SSAK7-05-1BPC	DU-1	Octachlorodibenzofuran	7,000	pg/g	111	≤50		J

SSAK7-05-1BPC_FD	DU-1	Octachlorodibenzofuran	2,000	pg/g	111	≤50		J
SSAK8-04-5BPC	DU-1	Octachlorodibenzofuran	340	pg/g	51	≤50		J
SSAK8-04-5BPC_FD	DU-1	Octachlorodibenzofuran	570	pg/g	51	≤50		J
SSAK8-08-3BPC	DU-1	Octachlorodibenzofuran	6,000	pg/g	120	≤50		J
SSAK8-08-3BPC_FD	DU-1	Octachlorodibenzofuran	1,500	pg/g	120	≤50		J
SSAM5-04-5BPC	DU-1	Octachlorodibenzofuran	140	pg/g	70	≤50		J
SSAM5-04-5BPC_FD	DU-1	Octachlorodibenzofuran	290	pg/g	70	≤50		J
SSAP3-04-1_01_BPC	DU-1	Octachlorodibenzofuran	24	pg/g	106	≤5.2	5.2	J
SSAP3-04-1_01_BPC_FD	DU-1	Octachlorodibenzofuran	130	pg/g	106	≤5.2		J

#### Phase I Remedial Investigation DVSR (ongoing, not submitted yet)

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
RISB-33-5.0-20141119	DU-1	Chlorate	1.1	mg/kg	58	≤50		J
RISB-33-5.0-20141119-FD	DU-1	Chlorate	2.0	mg/kg	58	≤50		J
RISB-51-5.0-20141030	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	240	pg/g	240	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzo-p-dioxin	0.26	pg/g	240	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	2,700	pg/g	2,699	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,4,6,7,8-Heptachlorodibenzofuran	1.5	pg/g	2,699	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	1,100	pg/g	1,099	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,4,7,8,9-Heptachlorodibenzofuran	0.72	pg/g	1,099	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	28	pg/g	28	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzo-p-dioxin	< 0.041	pg/g	28	≤5.4	5.4	UJ
RISB-51-5.0-20141030	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	1,200	pg/g	1,199	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,4,7,8-Hexachlorodibenzofuran	0.75	pg/g	1,199	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	60	pg/g	60	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,6,7,8-Hexachlorodibenzo-p-dioxin	0.19	pg/g	60	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	660	pg/g	660	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,6,7,8-Hexachlorodibenzofuran	0.43	pg/g	660	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	59	pg/g	59	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzo-p-dioxin	0.27	pg/g	59	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	110	pg/g	110	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,7,8,9-Hexachlorodibenzofuran	0.11	pg/g	110	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	35	pg/g	35	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,7,8-Pentachlorodibenzo-p-dioxin	< 0.068	pg/g	35	≤5.4	5.4	UJ
RISB-51-5.0-20141030	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	500	pg/g	500	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	1,2,3,7,8-Pentachlorodibenzofuran	0.38	pg/g	500	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	2,3,4,6,7,8-Hexachlorodibenzofuran	140	pg/g	140	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	2,3,4,6,7,8-Hexachlorodibenzofuran	0.13	pg/g	140	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	260	pg/g	260	≤5.4		J
RISB-51-5.0-20141030-FD	DU-1	2,3,4,7,8-Pentachlorodibenzofuran	0.14	pg/g	260	≤5.4	5.4	J
RISB-51-5.0-20141030	DU-1	2,3,7,8-Tetrachlorodibenzo-p-dioxin	11	pg/g	11	≤1.1		J
RISB-51-5.0-20141030-FD	DU-1	2,3,7,8-Tetrachlorodibenzo-p-dioxin	< 0.051	pg/g	11	≤1.1	1.1	UJ
RISB-51-5.0-20141030	DU-1	2,3,7,8-Tetrachlorodibenzofuran	300	pg/g	300	≤1.1		J
RISB-51-5.0-20141030-FD	DU-1	2,3,7,8-Tetrachlorodibenzofuran	0.44	pg/g	300	≤1.1	1.1	J
RISB-51-5.0-20141030	DU-1	Octachlorodibenzodioxin	250	pg/g	249	≤11		J
RISB-51-5.0-20141030-FD	DU-1	Octachlorodibenzodioxin	1.0	pg/g	249	≤11	11	J
RISB-51-5.0-20141030	DU-1	Octachlorodibenzofuran	8,200	pg/g	8,195	≤11		J
RISB-51-5.0-20141030-FD	DU-1	Octachlorodibenzofuran	5.0	pa/a	8.195	≤11	11	J

#### Tronox Parcels C, D, F, and G Investigation DVSR, February 2008

Sample ID	Decision Unit	Analyte	Result	Unit	RPD or Difference	Limit	Quantitation Limit	Final Qualifier
TSB-GR-02-0	DU-2	1,2,3,4,6,7,8-Heptachlorodibenzofuran	< 2.2	pg/g	2.5	≤2.2	2.2	UJ
TSB-GR-02-0 FD	DU-2	1,2,3,4,6,7,8-Heptachlorodibenzofuran	4.7	pg/g	2.5	≤2.2	2.2	J
TSB-GR-02-0	DU-2	1,2,3,4,7,8-Hexachlorodibenzofuran	< 1.9	pg/g	2.3	≤1.9	1.9	UJ
TSB-GR-02-0 FD	DU-2	1,2,3,4,7,8-Hexachlorodibenzofuran	4.2	pg/g	2.3	≤1.9	1.9	J
TSB-GR-02-0	DU-2	2,3,7,8-Tetrachlorodibenzofuran	1.5	pg/g	70	≤50		J
TSB-GR-02-0 FD	DU-2	2,3,7,8-Tetrachlorodibenzofuran	3.1	pg/g	70	≤50		J
TSB-GR-02-0	DU-2	4,4-DDE	2.3	ug/kg	23	≤1.8	1.8	J

TSB-GR-02-0 FD	DU-2	4,4-DDE	25	ug/kg	23	≤1.8		J
TSB-GR-02-0	DU-2	4,4-DDT	3.4	ug/kg	61	≤1.8	1.8	J
TSB-GR-02-0 FD	DU-2	4,4-DDT	69	ug/kg	66	≤1.7		J
TSB-GR-02-0	DU-2	beta-BHC	5.1	ug/kg	13	≤1.8	1.8	J
TSB-GR-02-0 FD	DU-2	beta-BHC	18	ug/kg	13	≤1.8		J
TSB-GR-02-0	DU-2	Octachlorodibenzodioxin	< 1.7	pg/g	8.3	≤1.7	1.7	UJ
TSB-GR-02-0 FD	DU-2	Octachlorodibenzodioxin	10	pg/g	8.3	≤1.7		J
TSB-GR-02-0	DU-2	Octachlorodibenzofuran	5.5	pg/g	74	≤50		J
TSB-GR-02-0 FD	DU-2	Octachlorodibenzofuran	12	pg/g	74	≤50		J
TSB-GR-02-0	DU-2	Perchlorate	3,760	ug/kg	110	≤50		J
TSB-GR-02-0 FD	DU-2	Perchlorate	13,000	ug/kg	110	≤50		J

#### Notes:

mg/kg = milligram per kilogram ng/kg = nanogram per kilogram pCi/g = picocurie per gram

pg/g = picogram per gram ug/kg = microgram per kilogram

BHC = Hexachlorocyclohexane

DDE = Dichlorodiphenyldichloroethylene

DDT = Dichlorodiphenyltrichloroethane

DU = Decision unit

FD = Field duplicate

J = Estimated value

K = Qualified as estimated maximum possible concentration (dioxin congeners)

PCB = Polychlorinated biphenyl

RPD = Relative percent differenct

UJ = Non-detect estimated quantitation limit

# TABLE B-5. Revisions of Censored Data for Blank Contamination Nevada Environmental Response Trust Site Henderson, Nevada

	Decision			Original	Corrected	Original	Corrected	Original	Corrected
Sample ID	Unit	Analyte	Unit	Result	Result	Detection Flag	Detection Flag	Qualifier	Qualifier
RSAJ2-10B	DU-1	Tin	ma/ka	10.7	4.3	N	Y		
RSA.I3-10B	DU-1	Boron	mg/kg	10.7	4.0	N	V		
RSA 13-10B	DU-1	Selenium	mg/kg	10.0	1.5	N	I V		J
ROADS-10D		Tin	mg/kg	4.2	1.5	N N	I V		J
ROAJO-TUD	DU-1	TIII Doron	mg/kg	10.0	4.2	N N	ř ř		J
ROAK4-0.5D	DU-1	DUIUII	mg/kg	10.2	5.5	N	Y		J
RSAK4-0.5B	DU-1	Tin	mg/kg	10.2	4.2	N	Ŷ		J
RSAK4009-0.5B	DU-1	Boron	mg/kg	10.1	5.3	N	Y		J
RSAK4009-0.5B	DU-1	Tin	mg/kg	10.1	4	N	Y		J
RSAK6-0.5B	DU-1	Boron	mg/kg	10.4	8.3	N	Y		J
RSAK6-0.5B	DU-1	Tin	mg/kg	10.4	4.1	N	Y		J
RSAK6-10B	DU-1	Boron	mg/kg	11	7.5	N	Y	UJ	J
RSAK6-10B	DU-1	Tin	ma/ka	11	4.5	N	Y		J
RSAK8-10B	DU-1	Boron	mg/kg	11	9.9	N	Ý	111	ı I
RSAL4-0.5B	DU-1	Boron	mg/kg	10.3	5.0	N	V		J
	DU-1	Tin	mg/kg	10.3	2.0	N	V V	00	J
RSAL4-0.3D	DU-1	Paran	mg/kg	10.3	3.9	N	ř V		J
R3AL4009-0.5B	DU-1	BUIUII	mg/kg	10.2	4.6	N	Y Y	UJ	J
RSAL4009-0.5B	DU-1	Tin	mg/kg	10.2	4	N	Y		J
RSAL5-0.5B	DU-1	Boron	mg/kg	10.8	7	N	Y		J
RSAL5-0.5B	DU-1	Tin	mg/kg	10.8	5	N	Y		J
RSAL6-0.5B	DU-1	Boron	mg/kg	10.5	4.3	N	Y		J
RSAL6-0.5B	DU-1	Chloride	mg/kg	2.1	1.9	N	Y		J
RSAL6-0.5B	DU-1	Tin	mg/kg	10.5	4.4	N	Y		J
RSAL6-10B	DU-1	Boron	ma/ka	10.9	10	N	Y		J
RSAL6-10B	DU-1	Tin	ma/ka	10.9	47	N	Ý		
RSAL7-0.5B	DU-1	Boron	mg/kg	10.2	4.5	N	v		, j
RSAL 7-0.5B	DU-1	Tin	mg/kg	10.2	4.5	N	V		J
	DU 1	Tin	mg/kg	10.2	3.7	N	I V		J
ROALO-U.JD	DU-1	TIII Denen	під/кд	10.3	4.3	IN N	ř		J
RSAM2-0.5B	DU-1	Boron	mg/kg	10.3	3.8	N	Y		J
RSAM2-0.5B	DU-1	Tin	mg/kg	10.2	3.4	N	Ŷ		J
RSAM3-0.5B	DU-1	Boron	mg/kg	10.4	4.2	N	Y		J
RSAM3-0.5B	DU-1	Tin	mg/kg	10.3	3.7	N	Y		J
RSAM4-0.5B	DU-1	Boron	mg/kg	10.4	8	N	Y		J
RSAM4-0.5B	DU-1	Tin	mg/kg	10.4	4.3	N	Y		J
RSAM5-10B	DU-3	Boron	mg/kg	10.9	5.6	N	Y		J
RSAM5-10B	DU-3	Tin	ma/ka	10.9	4.4	N	Y		J
RSAM6-0.5B	DU-1	Boron	ma/ka	10.2	6.7	N	Y		J
RSAM6-0.5B	DU-1	Tin	mg/kg	10.2	4.5	N	Ŷ		.l
RSAM7-0.5B	DU-1	Boron	mg/kg	10.2	5	N	V		<u> </u>
RSAM7-0.5B		Tin	mg/kg	10.2	4.2	N	V		J 1
RSAM7-0.3D		Poron	mg/kg	10.2	4.3	IN N	1 V		J
RSAM7009-10B	DU-1	BUIUII	mg/kg	10.7	9.4	N	ř.		J
RSAM7009-10B	DU-1	Tin	mg/kg	10.7	4.5	N	Y		J
RSAM7-10B	DU-1	Boron	mg/kg	10.6	8	N	Y		J
RSAM7-10B	DU-1	lin	mg/kg	10.6	4.1	N	Y		J
RSAM8-0.5B	DU-1	Tin	mg/kg	10.2	4.5	N	Y		J
RSAM8-10B	DU-1	Boron	mg/kg	10.7	7.2	N	Y		J
RSAM8-10B	DU-1	Tin	mg/kg	10.7	4.4	N	Y		J
RSAN3-0.5B	DU-1	Antimony	mg/kg	2.1	0.6	N	Y	UJ	J
RSAN3-0.5B	DU-1	Boron	ma/ka	10.7	8.7	N	Y		J
RSAN3-0.5B	DU-1	Tin	ma/ka	10.7	4.8	N	Y		, L
RSAN3-10B	DU-1	Tin	ma/ka	10.4	4.4	N	Ŷ	1	, i
RSAN4-0 5B	 DU-1	Boron	ma/ka	10.4	50	N	V		1
RSAN4-0 5B		Tin	mg/kg	10.0	4.2	N	I V	ł	
DEANI4000 40D	DU-1	T III Doron	mg/kg	10.0	4.2	IN N	ř V		J
RSAN4009-10B	DU-1	DUIUII	mg/kg	10.8	8.7	N	Y		J
RSAN4009-10B	DU-1	Tin	mg/kg	10.8	4.3	N	Y		J
RSAN4-10B	DU-1	Boron	mg/kg	10.6	7.3	N	Y		J
RSAN4-10B	DU-1	Selenium	mg/kg	4.2	1.1	N	Y		J
RSAN4-10B	DU-1	Tin	mg/kg	10.6	4.4	N	Y		J
RSAN5-0.5B	DU-1	Boron	mg/kg	10.3	7.3	N	Y		J
RSAN5-0.5B	DU-1	Tin	mg/kg	10.3	4.2	N	Y		J
RSAN6009-10B	DU-1	Boron	ma/ka	10.8	6.3	N	Y		J
RSAN6009-10B	DU-1	Platinum	ma/ka	0.11	0.012	N	Ŷ	1	,i
RSAN6009-10B	DU-1	Tin	ma/ka	10.8	4 1	N	Ý	1	
RSAN6-10B		Boron	ma/ka	11	70	N	V		1
RSANG-10B		Platinum	mg/kg	0.11	0.011	N	I V		J 1
		Solonium	mg/Kg	0.11	0.011	IN N	T V		J
ROAND-TUB			mg/kg	4.4	0.8	N	Y		J
ROANT CER	DU-1		mg/kg	11	4.1	N	Y		J
KSAN/-0.5B	DU-1	Antimony	mg/kg	2	1.1	N	Y	UJ	J
KSAN7-0.5B	DU-1	Platinum	mg/kg	0.1	0.023	N	Y		J

# TABLE B-5. Revisions of Censored Data for Blank Contamination Nevada Environmental Response Trust Site Henderson, Nevada

	Decision			Original	Corrected	Original	Corrected	Original	Corrected
Sample ID	Unit	Analyte	Unit	Result	Result	Detection Flag	Detection Flag	Qualifier	Qualifier
PSANZ 0.5B	DLL-1	Tin	ma/ka	10.1	2.0	N	v	quanto	- uuuiiioi
RSANT-0.3D	DU-1	Boron	mg/kg	10.1	5.0	IN NI	1 V		J
RSAU3-0.5D	DU-1	DUIUII	mg/kg	10.1	5.6	N	ř V		J
RSAU3-0.56	DU-1	1 In Dana a	mg/kg	10.1	4	N	Y		J
RSAU3-10B	DU-1	Boron	mg/kg	11	7.9	N	Y		J
RSA03-10B	DU-1	Mercury	mg/kg	0.019	0.01	N	Y		J
RSAO3-10B	DU-1	Tin	mg/kg	11	4.4	N	Y		J
RSAO5-0.5B	DU-1	Boron	mg/kg	10.3	10.2	N	Y		J
RSAO5-0.5B	DU-1	Tin	mg/kg	10.3	4	N	Y		J
RSAO6-0.5B	DU-1	Selenium	mg/kg	4.1	1.1	Ν	Y		J
RSAO6-0.5B	DU-1	Tin	mg/kg	10.2	4.2	Ν	Y		J
RSAO6-10B	DU-1	Boron	ma/ka	10.7	8	N	Y		J
RSAO6-10B	DU-1	Mercury	mg/kg	0.018	0.008	N	Ŷ		
RSAO6-10B	DU-1	Platinum	mg/kg	0.010	0.000	N	Y		ů I
RSA06-10B	DU-1	Tin	mg/kg	10.7	3.4	N	V		J
PSA08-21 5B	DU-1	Ammonio	mg/kg	0.54	0.12	N	V V		J
RSA00-21.5D	DU-1	Tin	mg/kg	0.54	0.13	IN N	ř V		J
R3A06-21.3B	D0-1	TIII Dana a	mg/kg	10.7	5.2	N	ř V		J
RSAP5-0.5B	DU-2	Boron	mg/kg	10.8	7.9	N	Y		J
RSAP5-0.5B	DU-2	lin	mg/kg	10.8	4	N	Y		J
RSAP7-0.5B	DU-2	Antimony	mg/kg	2.1	1.6	N	Y	UJ	J
RSAP7-0.5B	DU-2	Boron	mg/kg	10.5	6.7	N	Y		J
RSAP7-0.5B	DU-2	Platinum	mg/kg	0.1	0.032	N	Y		J
RSAP7-0.5B	DU-2	Tin	mg/kg	10.5	3.7	N	Y		J
RSAQ4-0.5B	DU-2	Antimony	mg/kg	2.1	1.3	Ν	Y	UJ	J
RSAQ4-0.5B	DU-2	Tin	ma/ka	10.3	4.4	Ν	Y		J
RSAQ6-0.5B	DU-2	Boron	ma/ka	10.7	4.8	N	Y		J
RSAQ6-0.5B	DU-2	Tin	mg/kg	10.7	2.8	N	Ŷ		J
RSAO6-10B	DU-2	Boron	mg/kg	10.0	4.7	N	V		ů I
RSAO6-10B	DU-2	Tin	mg/kg	10.0	2.0	N	V		J 1
ROACT 0 5P		Antimony	mg/kg	10.0	2.0	N	I V		J
RSAQT-0.5D	DU-2	Anumony	mg/kg	2.1	1.2	IN N	Ť	UJ	J
RSAQ7-0.5D	DU-2	DUIUII	mg/kg	10.6	6.6	N	Y		J
RSAQ7-0.5B	DU-2	Tin Di fi	mg/kg	10.6	3.6	N	Y		J
RSAS5-0.5B	DU-2	Platinum	mg/kg	0.1	0.011	N	Y		J
RSAS5-0.5B	DU-2	Tin	mg/kg	10.5	4.6	N	Y		J
RSAS8-0.5B	DU-2	Boron	mg/kg	10.3	6.4	N	Y		J
RSAS8-0.5B	DU-2	Tin	mg/kg	10.3	4.1	N	Y		J
SA100-0.5B	DU-1	Boron	mg/kg	10.1	3.7	N	Y		J
SA100-0.5B	DU-1	Tin	mg/kg	10.1	3.5	Ν	Y		J
SA102-10B	DU-1	Ammonia	mg/kg	0.54	0.18	Ν	Y		J
SA102-10B	DU-1	Platinum	mg/kg	0.09	0.02	Ν	Y		J
SA102-10B	DU-1	Tin	ma/ka	87	3.7	N	Y		J
SA103-0.5B	DU-2	Platinum	mg/kg	0.11	0.015	N	Ŷ		J
SA103-0 5B	DU-2	Tin	mg/kg	10.6	5.4	N	V		J
SA10/000-10B	DU-3	Boron	mg/kg	10.0	7.4	N	V I		J 1
SA104000-10B	DU-3	Tin	mg/kg	10.7	1.1	N	I V		J
SA104003-10D		Poron	mg/kg	10.7	4.7	IN N	1 V		J
SA104-10B	DU-3	DUIUII	під/кд	10.6	0.9	IN N	Ť.		J
SA104-10B	DU-3	Tin	mg/kg	10.8	4.8	N	Y		J
SA105009-10B	DU-3	Boron	mg/kg	10.8	5.9	N	Y		J
SA105009-10B	DU-3	lin	mg/kg	10.8	5	N	Y		J
SA105-10B	DU-3	Boron	mg/kg	11	5.5	N	Y		J
SA105-10B	DU-3	Tin	mg/kg	11	5.1	N	Y		J
SA106-12B	DU-1	Mercury	mg/kg	0.018	0.008	Ν	Y		J
SA106-12B	DU-1	Tin	mg/kg	10.4	4.9	Ν	Y		J
SA108-20B	DU-1	Antimony	mg/kg	2.1	1.5	Ν	Y	UJ	J
SA108-20B	DU-1	Tin	ma/ka	10.6	4.5	N	Y		J
SA109-0.5B	DU-1	Boron	ma/ka	10.5	8.7	N	Y		J
SA109-0.5B	DU-1	Tin	ma/ka	10.5	3.8	N	Ŷ		
SA114-10B	DU-1	Ammonia	mg/kg	0.54	0.33	N	Y		ů I
SA11/-10B	DU-1	Boron	mg/kg	10.6	4.0	N	V		J
SA114-10D		Molyhdonum	mg/kg	10.0	4.0	IN NI	I V		J
SA114-10B	DU-1		mg/kg	0.32	0.15	N	ř V		J
SA114-10B	DU-1		mg/kg	10.6	4.4	N	Y		J
SA121-0.5B	DU-2	Antimony	mg/kg	2.1	1.9	N	Y	UJ	J
SA121-0.5B	DU-2	Boron	mg/kg	10.7	8	N	Y		J
SA121-0.5B	DU-2	Tin	mg/kg	10.7	4.1	N	Y		J
SA121009-0.5B	DU-2	Boron	mg/kg	10.8	8.4	N	Y		J
SA121009-0.5B	DU-2	Tin	mg/kg	10.8	4.7	N	Y		J
SA122-0.5B	DU-2	Molybdenum	mg/kg	0.31	0.29	Ν	Y		J
SA122-0.5B	DU-2	Platinum	mg/kg	0.1	0.006	Ν	Y	1	J
SA122-0.5B	DU-2	Tin	mg/kg	10.4	5.6	N	Y		J

# TABLE B-5. Revisions of Censored Data for Blank Contamination Nevada Environmental Response Trust Site Henderson, Nevada

	Decision			Original	Corrected	Original	Corrected	Original	Corrected
Sample ID	Unit	Analyte	Unit	Result	Result	Detection Flag	Detection Flag	Qualifier	Qualifier
SA123-0 5B	DU-1	Boron	ma/ka	10.4	4.4	N	V		
SA123-0.5B	DU-1	Selenium	mg/kg	10.4	4.4	N	I V		J
SA123-0.3D	DU-1	Tin	mg/kg	4.2	0.9	IN N	ř V		J
0A120-0.5D	DU-1	TIII Danan	mg/kg	10.4	4.5	IN N	ř		J
SA126-0.5B	DU-2	Boron	mg/kg	10	7.5	N	Y		J
SA126-0.5B	DU-2	Platinum	mg/kg	0.1	0.011	N	Y		J
SA126-0.5B	DU-2	lin	mg/kg	10	4	N	Y		J
SA128-10B	DU-3	Boron	mg/kg	10.8	6.1	N	Y		J
SA128-10B	DU-3	Selenium	mg/kg	4.3	1	N	Y		J
SA128-10B	DU-3	Tin	mg/kg	10.8	5	Ν	Y		J
SA131-10B	DU-1	Boron	mg/kg	10.9	7.2	Ν	Y		J
SA131-10B	DU-1	Tin	ma/ka	10.9	4.8	N	Y		J
SA134-10B	DU-1	Tin	ma/ka	10.9	44	N	Y		J
SA136-0.5B	DU-2	Tin	ma/ka	11	4.5	N	Ŷ		- J
SA138-0 5B	DU-2	Boron	mg/kg	10.6	0.8	N	V		ů I
SA138-0 5B	DU-2	Platinum	mg/kg	0.1	0.029	N	I V		J
SA138-0.5B		Tin	mg/kg	10.6	0.020	IN N	1 V		J
3A130-0.3B	D0-2	1111	mg/kg	10.6	4	N	Y		J
SA143-24B	DU-1	Boron	mg/kg	10.9	5.9	N	Y		J
SA143-24B	DU-1	Platinum	mg/kg	0.11	0.008	N	Y		J
SA143-24B	DU-1	lin	mg/kg	10.9	3.9	N	Y		J
SA144-0.5B	DU-1	Tin	mg/kg	10.4	4.3	N	Y		J
SA145-0.5B	DU-1	Tin	mg/kg	10.2	4.1	N	Y		J
SA145-10B	DU-1	Boron	mg/kg	10.8	10.7	Ν	Y		J
SA145-10B	DU-1	Tin	mg/kg	10.8	4.1	Ν	Y		J
SA150-10B	DU-1	Antimony	ma/ka	2.4	2.1	N	Y	UJ	J
SA150-10B	DU-1	Boron	ma/ka	11.9	7.2	N	Ŷ		J
SA150-10B	DU-1	Platinum	mg/kg	0.12	0.012	N	Ŷ		
SA150-10B	DU-1	Tin	mg/kg	11.0	4.3	N	V		ů I
SA151-0.5B	DU-1	Tin	mg/kg	10.4	4.5	N	I V		J
SA152.0.5D	DU-1	Deren	mg/kg	10.4	4.5	IN NI	1 V		J
SA 152-0.5B	DU-1	DUIUII	mg/kg	10.5	4.4	N	ř V		J
SA152-0.5B	DU-1	Tin	mg/kg	10.3	3.6	N	Y		J
SA152009-0.5B	DU-1	Boron	mg/kg	10.3	4.1	N	Y		J
SA152009-0.5B	DU-1	Tin	mg/kg	10.2	3.8	N	Y		J
SA152-10B	DU-1	Boron	mg/kg	10.7	6.4	N	Y		J
SA152-10B	DU-1	Tin	mg/kg	10.7	3.9	Ν	Y		J
SA154-10B	DU-1	Antimony	mg/kg	2.1	1.1	N	Y	UJ	J
SA154-10B	DU-1	Boron	mg/kg	10.7	7.7	Ν	Y		J
SA154-10B	DU-1	Platinum	ma/ka	0.11	0.01	Ν	Y		J
SA154-10B	DU-1	Tin	ma/ka	10.7	4.1	N	Y		J
SA156-10B	DU-2	Boron	mg/kg	10.5	9.4	N	Ŷ		
SA156-10B	DU-2	Selenium	mg/kg	4.2	1.5	N	Ý		
SA156-10B	DU-2	Tin	mg/kg	4.2	1.5	N	I V		J
SA157 0 5P		Boron	mg/kg	10.5	4.4	IN N	1 V		J
SA157-0.5D	DU-1	Doron	mg/kg	10.0	4.4	IN N	1 V		J
SA 157-0.3D	DU-1	1ID Dener	mg/kg	10.8	4.5	N	Y		J
SA157009-0.5B	D0-1	Boron	mg/kg	10.7	3.7	N	Y		J
SA157009-0.5B	DU-1	Selenium	mg/kg	4.3	1.3	N	Y	UJ	J
SA157009-0.5B	DU-1	Tin	mg/kg	10.7	4.1	N	Y		J
SA158-0.5B	DU-1	Tin	mg/kg	10.3	4.6	N	Y		J
SA158-10B	DU-1	Boron	mg/kg	10.7	8.2	N	Y		J
SA158-10B	DU-1	Tin	mg/kg	10.7	4	N	Y		J
SA165-10B	DU-3	Selenium	mg/kg	4.2	0.7	Ν	Y		J
SA165-10B	DU-3	Tin	mg/kg	10.6	4.8	Ν	Y		J
SA166-0.5B	DU-1	Boron	ma/ka	10.4	6.7	Ν	Y		J
SA166-0.5B	DU-1	Tin	ma/ka	10.4	4.5	N	Y		
SA170-0.5B	DU-2	Tin	mg/kg	10.4	4.6	N	V		<u> </u>
SA171-5B	DU-1	Antimony	mg/kg	10.5	4.0	N	I V		J
SA171-5D	DU-1	Roron	mg/kg	2.1	1	IN N	ř V	UJ	J
3A171-3B	DU-1	D01011	mg/kg	10.6	8.4	N	ř.		J
SA171-5B	DU-1	TIN	mg/kg	10.6	4	N	Y		J
SA175-10B	DU-3	Tin	mg/kg	11	4.8	N	Y		J
SA176-0.5B	DU-1	lin	mg/kg	10.4	4.3	N	Y		J
SA176-10B	DU-1	Boron	mg/kg	10	7.2	N	Y		J
SA176-10B	DU-1	Tin	mg/kg	10	4.2	Ν	Y		J
SA182-10B	DU-1	Boron	mg/kg	10.9	6.8	N	Y		J
SA182-10B	DU-1	Tin	mg/kg	10.9	4.1	Ν	Y		J
SA185-0.5B	DU-1	Boron	ma/ka	10.5	5.6	Ν	Y		J
SA185-0.5B	DU-1	Tin	ma/ka	10.5	39	N	Ŷ	1	J
SA186-0.5B	DU-1	Boron	ma/ka	10.3	9.7	N	v V	l	.1
SA186-0 5B		Tin	ma/ka	10.3	<i>J</i> 1	N	i V		
SA180-0.5B		Boron	mg/kg	10.3		N	I V		J 1
0.1103-0.30	00-1	DOIOII	mg/kg	10.4	0	IN	ſ	1	J

# TABLE B-5. Revisions of Censored Data for Blank Contamination Nevada Environmental Response Trust Site Henderson, Nevada

	Decision			Original	Corrected	Original	Corrected	Original	Corrected
Sample ID	Unit	Analyte	Unit	Result	Result	Detection Flag	Detection Flag	Qualifier	Qualifier
SA189-0.5B	DU-1	Tin	ma/ka	10.5	39	N	Y		
SA197-0.5B	DU-3	Boron	ma/ka	10.0	4.5	N	Ŷ		J
SA197-0.5B	DU-3	Molybdenum	ma/ka	0.32	0.3	N	Ŷ		J
SA197-0.5B	DU-3	Tin	ma/ka	10.7	4.6	N	Ŷ		J
SA197009-10B	DU-3	Boron	ma/ka	10.5	6.7	N	Y		J
SA197009-10B	DU-3	Tin	mg/kg	10.5	4.5	Ν	Y		J
SA197-10B	DU-3	Boron	mg/kg	10.4	6.4	Ν	Y		J
SA197-10B	DU-3	Chloride	ma/ka	2.1	1.7	Ν	Y	UJ	J
SA197-10B	DU-3	Tin	mg/kg	10.4	4.3	Ν	Y		J
SA198-10B	DU-3	Boron	mg/kg	10.4	6.7	Ν	Y		J
SA198-10B	DU-3	Tin	mg/kg	10.4	4.4	Ν	Y		J
SA200-0.5B	DU-1	Tin	mg/kg	10.6	4.4	Ν	Y		J
SA201-10B	DU-1	Tin	mg/kg	10.7	4.7	Ν	Y		J
SA211-0.5B	DU-2	Antimony	mg/kg	2.1	1.5	Ν	Y	UJ	J
SA211-0.5B	DU-2	Boron	mg/kg	10.5	8	Ν	Y		J
SA211-0.5B	DU-2	Tin	mg/kg	10.5	4.2	Ν	Y		J
SA212-0.5B	DU-2	Antimony	mg/kg	2	1.1	Ν	Y	UJ	J
SA212-0.5B	DU-2	Boron	mg/kg	9.8	5.9	Ν	Y		J
SA212-0.5B	DU-2	Tin	mg/kg	9.8	3.7	Ν	Y		J
SA214-0.5B	DU-2	Antimony	mg/kg	2.1	1	Ν	Y	UJ	J
SA214-0.5B	DU-2	Boron	mg/kg	10.7	7.8	Ν	Y		J
SA214-0.5B	DU-2	Tin	mg/kg	10.7	4.1	Ν	Y		J
SA31-0.5B	DU-2	Boron	mg/kg	10.4	7.4	Ν	Y		J
SA31-0.5B	DU-2	Tin	ma/ka	10.4	5.8	N	Y		J
SA35-0.5B	DU-1	Boron	mg/kg	10.4	3.9	Ν	Y		J
SA35-0.5B	DU-1	Tin	mg/kg	10.4	3.9	Ν	Y		J
SA35-10B	DU-1	Boron	mg/kg	10.6	5.5	Ν	Y		J
SA35-10B	DU-1	Selenium	mg/kg	4.2	0.8	Ν	Y		J
SA35-10B	DU-1	Tin	mg/kg	10.6	4.5	Ν	Y		J
SA39-10B	DU-1	Boron	mg/kg	10.4	6.2	Ν	Y		J
SA39-10B	DU-1	Tin	mg/kg	10.4	3.8	Ν	Y		J
SA42009-10B	DU-1	Boron	mg/kg	10.9	7.7	Ν	Y		J
SA42009-10B	DU-1	Molybdenum	mg/kg	0.33	0.28	Ν	Y		J
SA42009-10B	DU-1	Platinum	mg/kg	0.11	0.011	Ν	Y		J
SA42009-10B	DU-1	Tin	mg/kg	10.9	4.9	Ν	Y		J
SA42-10B	DU-1	Boron	mg/kg	10.9	7.3	Ν	Y		J
SA42-10B	DU-1	Molybdenum	mg/kg	0.33	0.22	Ν	Y		J
SA42-10B	DU-1	Platinum	mg/kg	0.1	0.011	N	Y		J
SA42-10B	DU-1	Tin	mg/kg	10.9	4.4	N	Y		J
SA42-10B	DU-1	Tungsten	mg/kg	0.1	0.091	Ν	Y	UJ	J
SA43-10B	DU-1	Boron	mg/kg	10.7	8.6	Ν	Y		J
SA43-10B	DU-1	Molybdenum	mg/kg	0.32	0.17	N	Y		J
SA43-10B	DU-1	Tin	mg/kg	10.7	4.9	Ν	Y		J
SA44-10B	DU-1	Boron	mg/kg	10.7	6.4	N	Y		J
SA44-10B	DU-1	Molybdenum	mg/kg	0.32	0.24	Ν	Y		J
SA44-10B	DU-1	Tin	mg/kg	10.7	4.6	N	Y		J
SA49-10B	DU-3	Boron	mg/kg	10.7	8.9	N	Y		J
SA49-10B	DU-3	Tin	mg/kg	10.7	4.9	N	Y		J
SA50-0.5B	DU-1	Boron	mg/kg	10.2	9.7	N	Y		J
SA50-0.5B	DU-1	Tin	mg/kg	10.2	5.3	N	Y		J
SA51009-10B	DU-1	Tin	mg/kg	10.6	5.2	N	Y		J
SA51-10B	DU-1	Tin	mg/kg	10.8	4.8	N	Y		J
SA54-0.5B	DU-1	Boron	mg/kg	9.9	5.7	N	Y		J
SA54-0.5B	DU-1	Tin	mg/kg	9.9	3.9	N	Y		J
SA55-0.5B	DU-1	Boron	mg/kg	10.3	5.8	N	Y		J
SA55-0.5B	DU-1	Tin	mg/kg	10.3	4	N	Y		J
SA60-10B	DU-3	Antimony	mg/kg	2.2	2.1	N	Y	UJ	J
SA60-10B	DU-3	Boron	mg/kg	11	10.6	N	Y		J
SA60-10B	DU-3	Platinum	mg/kg	0.11	0.014	N	Y		J
SA60-10B	DU-3	Tin	mg/kg	11	4.2	N	Y		J
SA62-0.5B	DU-1	Tin	mg/kg	10.3	4.7	N	Y		J
SA64-0.5B	DU-3	Boron	mg/kg	10.7	5.9	N	Y		J
SA64-0.5B	DU-3	Tin	mg/kg	10.7	4.7	N	Y		J
SA64-10B	DU-3	Boron	mg/kg	10.8	6.9	N	Y		J
SA64-10B	DU-3	Platinum	mg/kg	0.11	0.085	N	Y		J
SA64-10B	DU-3	Tin	mg/kg	10.8	4.2	N	Y		J
SA65-10B	DU-3	Boron	mg/kg	10.7	6.1	N	Y		J
SA65-10B	DU-3	Platinum	mg/kg	0.11	0.014	N	Y		J

#### TABLE B-5. Revisions of Censored Data for Blank Contamination Nevada Environmental Response Trust Site

Henderson, Nevada

	Decision	)		Original	Corrected	Original	Corrected	Original	Corrected
Sample ID	Unit	Analyte	Unit	Result	Result	Detection Flag	<b>Detection Flag</b>	Qualifier	Qualifier
SA65-10B	DU-3	Tin	mg/kg	10.7	4	Ν	Y		J
SA65-20B	DU-3	Boron	mg/kg	10.6	7	Ν	Y		J
SA65-20B	DU-3	Platinum	mg/kg	0.11	0.007	Ν	Y		J
SA65-20B	DU-3	Tin	mg/kg	10.6	4.1	Ν	Y		J
SA69-0.5B	DU-1	Boron	mg/kg	10.5	4.5	Ν	Y	UJ	J
SA69-0.5B	DU-1	Tin	mg/kg	10.5	4.2	Ν	Y		J
SA70-0.5B	DU-1	Tin	mg/kg	10.5	4.5	Ν	Y		J
SA71-0.5B	DU-1	Tin	mg/kg	10.2	4.6	Ν	Y		J
SA73-0.5B	DU-1	Boron	mg/kg	10.8	5	Ν	Y		J
SA73-0.5B	DU-1	Tin	mg/kg	10.8	4.5	Ν	Y		J
SA73-10B	DU-1	Ammonia	mg/kg	0.56	0.11	Ν	Y	UJ	J
SA73-10B	DU-1	Boron	mg/kg	10.9	6.8	Ν	Y		J
SA73-10B	DU-1	Tin	mg/kg	10.9	4.8	Ν	Y		J
SA74-0.5B	DU-1	Boron	mg/kg	10.8	4.8	Ν	Y		J
SA74-0.5B	DU-1	Tin	mg/kg	10.8	4.5	Ν	Y		J
SA74009-0.5B	DU-1	Boron	mg/kg	11	5.5	Ν	Y		J
SA74009-0.5B	DU-1	Tin	mg/kg	11	5.2	Ν	Y		J
SA74-10B	DU-1	Boron	mg/kg	11	8.6	Ν	Y		J
SA74-10B	DU-1	Tin	mg/kg	11	4.5	Ν	Y		J
SA75-0.5B	DU-1	Boron	mg/kg	10.6	2.5	Ν	Y	UJ	J
SA75-0.5B	DU-1	Tin	mg/kg	10.6	4.3	Ν	Y		J
SA75-10B	DU-1	Boron	mg/kg	10.9	6.3	Ν	Y	UJ	J
SA75-10B	DU-1	Tin	mg/kg	10.9	4	Ν	Y		J
SA76-10B	DU-1	Boron	mg/kg	10.7	9	Ν	Y	UJ	J
SA76-10B	DU-1	Tin	mg/kg	10.7	4.2	Ν	Y		J
SA77-0.5B	DU-2	Boron	mg/kg	10	4.6	Ν	Y		J
SA77-0.5B	DU-2	Tin	mg/kg	10	4.9	Ν	Y		J
SA82-0.5B	DU-1	Ammonia	mg/kg	0.51	0.22	Ν	Y		J
SA82-0.5B	DU-1	Boron	mg/kg	9.8	9.1	Ν	Y		J
SA82-0.5B	DU-1	Tin	mg/kg	9.8	4	Ν	Y		J
SA85-0.5B	DU-1	Boron	mg/kg	10.3	8	Ν	Y		J
SA85-0.5B	DU-1	Tin	mg/kg	10.3	4.9	Ν	Y		J
SA86009-10B	DU-3	Boron	mg/kg	11.4	5	Ν	Y		J
SA86009-10B	DU-3	Molybdenum	mg/kg	0.34	0.3	Ν	Y		J
SA86009-10B	DU-3	Tin	mg/kg	11.4	6.2	Ν	Y		J
SA86-10B	DU-3	Boron	mg/kg	11.1	4.9	Ν	Y		J
SA86-10B	DU-3	Molybdenum	mg/kg	0.33	0.27	Ν	Y		J
SA86-10B	DU-3	Tin	mg/kg	11.1	6	Ν	Y		J
SA86-10B	DU-3	Tungsten	mg/kg	0.11	0.08	Ν	Y	UJ	J
SA88-10B	DU-1	Ammonia	mg/kg	0.55	0.33	Ν	Y		J
SA88-10B	DU-1	Boron	mg/kg	10.5	10.3	Ν	Y		J
SA88-10B	DU-1	Tin	mg/kg	10.5	5.6	Ν	Y		J
SA92-0.5B	DU-1	Tin	mg/kg	10.3	4.3	Ν	Y		J
SA92-10B	DU-1	Boron	mg/kg	11	6.8	Ν	Y		J
SA92-10B	DU-1	Tin	mg/kg	11	4.4	Ν	Y		J
RSAH3-0.5B	DU-1	Boron	mg/kg	10.2	6.4	Ν	Y		J
RSAH3009-0.5B	DU-1	Boron	mg/kg	10.2	5.6	Ν	Y		J
RSAH3-0.5B	DU-1	Tin	mg/kg	10.2	4.5	Ν	Y		J
RSAH3009-0.5B	DU-1	Tin	mg/kg	10.2	4.3	Ν	Y		J

#### Notes:

mg/kg = milligram per kilogram DU = Decision unit

N = Not detected

Y = Detected

J = The result is an estimated quantity. the associated numerical value is the approximate concentration of the analyte in the sample.

UJ = The nondetected analyte was qualified as estimated at the sample quantitation limit. The reported sample quantitation limit is approximate and may be inaccurate or imprecise.

Decision		Maximum		Maximum Detected	Sereening	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit	-	Qualified Data		BHRA Data Set	Levei	
DU-1	1,1-Dichloroethene	0.0012	J	0.0012	1,100	mg/kg
DU-1	1,2,3-Trichlorobenzene	0.0013	J	0.0013	151	mg/kg
DU-1	1,2,4-Trichlorobenzene	0.0037	J	0.0037	125	mg/kg
DU-1	1,2,4-Trimethylbenzene	0.0014	J	0.0014	218	mg/kg
DU-1	1,2-Dichlorobenzene	0.00039	J	0.00039	376	mg/kg
DU-1	1,3,5-Trimethylbenzene	0.00050	J	0.00050	182	mg/kg
DU-1	1,4-Dichlorobenzene	0.0017	J	0.0017	475	mg/kg
DU-1	1-Methylnaphthalene	0.0033	J	5.6	81	mg/kg
DU-1	2,3,7,8-TCDD TEQ	0.019	J	0.019	0.0027	mg/kg
DU-1	2,4'-DDE	0.088	J	0.088	9.5	mg/kg
DU-1	2-Butanone	0.0043	J	0.027	28,400	mg/kg
DU-1	2-Methylnaphthalene	0.10	J	7.9	368	mg/kg
DU-1	4,4'-DDD	0.032	J	0.032	15	mg/kg
DU-1	4,4'-DDE	0.18	J	6.0	9.5	mg/kg
DU-1	4,4'-DDT	0.069	J	2.3	7.5	mg/kg
DU-1	Acenaphthene	0.057	J	0.70	118	mg/kg
DU-1	Acenaphthylene	0.22	J	0.22	118	mg/kg
DU-1	Acetone	0.040	J	0.15	1,040,000	mg/kg
DU-1	alpha-BHC	0.00080	J	0.0013	0.49	mg/kg
DU-1	Aluminum	9,640	J	12,200	1,240,000	mg/kg
DU-1	Ammonia	3.5	J	3.5	6,140	mg/kg
DU-1	Anthracene	0.30	J	0.30	4.3	mg/kg
DU-1	Antimony	2.1	J	2.4	519	mg/kg
DU-1	Arsenic	5.4	J	34	7.2	mg/kg
DU-1	Barium	440	J	1,780	238,000	mg/kg
DU-1	Benzo(g,h,i)perylene	0.16	J	0.38	25,300	mg/kg
DU-1	Beryllium	0.56	J	0.70	2,540	mg/kg
DU-1	beta-BHC	0.57	J	0.87	1.7	mg/kg
DU-1	bis(2-Ethylhexyl)phthalate	0.38	J	0.82	183	mg/kg
DU-1	Boron	11	J	1,510	259,000	mg/kg
DU-1	Bromide	4.6	J	4.6	441,000	mg/kg
DU-1	Bromodichloromethane	0.00069	J	0.00069	1.4	mg/kg
DU-1	Bromoform	0.0017	J	0.0017	104	mg/kg
DU-1	Butylbenzylphthalate	0.053	J	0.053	1,350	mg/kg
DU-1	Cadmium	0.90	J	1.8	1,260	mg/kg
DU-1	Calcium	56,600	J	62,500	NA	mg/kg
DU-1	Carbon tetrachloride	0.00063	J	0.00063	3.2	mg/kg
DU-1	Chlorate	48	J	20,900	38,900	mg/kg
DU-1	Chlordane (total)	0.0030	J	0.0030	7.3	mg/kg
DU-1	Chloride	735	J	6,670	113,000	mg/kg
DU-1	Chlorobenzene	0.0012	J	0.0012	18,300	mg/kg
DU-1	Chloroform	0.0044	J	0.15	1.5	mg/kg
DU-1	Chromium (total)	35	J	102	1,950,000	mg/kg
DU-1	Chromium VI	2.0	J	106	7.0	mg/kg
DU-1	Cobalt	71	J	284	385	mg/kg
DU-1	Copper	25	J	160	36,700	mg/kg
DU-1	Cyanide (total)	0.48	J	1.3	179	mg/kg
DU-1	delta-BHC	0.0011	J	0.0011	334	mg/kg
DU-1	Dieldrin	0.059	J	0.059	0.16	mg/kg
DU-1	Diethylphthalate	0.065	J	0.065	733,000	mg/kg
DU-1	Dimethylphthalate	0.22	J	0.79	9,160.000	mg/kg
DU-1	Di-n-butylphthalate	1.0	J	3.0	91,600	mg/kg
DU-1	Di-n-octylphthalate	0.088	J	0.088	9,160	mg/kg
DU-1	Endosulfan I	0.0015	J	0.0015	5,500	mg/kg
DU-1	Endrin	0.00070	J	0.0054	30	mg/kg
DU-1	Endrin ketone	0.0018	J	0.0045	30	ma/ka
I						

Decision		Maximum Maximum Dete		Maximum Detected	Screening	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit	2	Qualified Data		BHRA Data Set	Level	
DU-1	Ethyl tert-butyl ether	0.00038	J	0.00038	70.900	mg/kg
DU-1	Fluoranthene	0.35	J	0.35	33,700	mg/kg
DU-1	Fluorene	0.0010	J	1.1	93	ma/ka
DU-1	gamma-BHC	0.0013	J	0.0013	2.8	ma/ka
DU-1	gamma-Chlordane	0.0014	J	0.0014	7.3	ma/ka
DU-1	Hexachlorobenzene	0.33	J	4.7	0.23	ma/ka
DU-1	Hexachlorobutadiene	0.0045	J	0.0045	6.1	ma/ka
DU-1	Iron	20.300	J	24.000	908.000	mg/kg
DU-1	Lead	62	J	267	800	ma/ka
DU-1	m.p-Xylene	0.0015	J	0.0023	387	ma/ka
DU-1	Magnesium	18,000	J	71,000	5,200,000	mg/kg
DU-1	Manganese	9,100	J	29,200	28,100	mg/kg
DU-1	Mercury	0.10	J	1.9	389	mg/kg
DU-1	Methoxychlor	0.31	J	0.38	4,580	mg/kg
DU-1	Methylene Chloride	0.0048	J	0.0082	1,550	mg/kg
DU-1	Molybdenum	2.3	J	55	6,490	mg/kg
DU-1	Naphthalene	0.25	J	3.1	18	mg/kg
DU-1	Nickel	26	J	164	24,700	mg/kg
DU-1	Nitrate	57	J	173	2,080,000	mg/kg
DU-1	Nitrite	2.0	J	4.2	130,000	mg/kg
DU-1	Octachlorostyrene	0.27	J	2.1	NA	mg/kg
DU-1	ortho-Phosphate	2,900	J	2,900	30,400,000	mg/kg
DU-1	o-Xylene	0.00074	J	0.00074	434	mg/kg
DU-1	p-Cymene	0.00055	J	0.00055	647	mg/kg
DU-1	Perchlorate	156	J	1.500	908	mg/kg
DU-1	Phenanthrene	1.0	J	1.5	25	mg/kg
DU-1	Phosphorus (total)	1.230	J	1.600	9.630.000	mg/kg
DU-1	Platinum	0.054	J	0.054	649	ma/ka
DU-1	Potassium	3.280	J	6.120	NA	ma/ka
DU-1	Pyrene	0.46	J	1.3	44	ma/ka
DU-1	Radium-226	2.0	J	2.5	0.023	pCi/a
DU-1	Radium-228	3.3	J	3.3	0.041	pCi/q
DU-1	Selenium	1.5	J	1.5	6.490	mg/kg
DU-1	Silicon	250	J	250	NA	mg/kg
DU-1	Silver	0.40	J	7.6	6.490	ma/ka
DU-1	Sodium	7,000	J	11,700	NA	mg/kg
DU-1	Strontium	440	J	750	779.000	mg/kg
DU-1	Styrene	0.00028	J	0.00028	867	mg/kg
DU-1	Sulfate	645	J	14,600	NA	mg/kg
DU-1	Sulfur	1,900	J	14,000	NA	mg/kg
DU-1	tert Butyl alcohol	0.0076	J	0.0076	21,300	mg/kg
DU-1	Thallium	0.98	J	8.4	13	mg/kg
DU-1	Thorium-228	2.8	J	2.8	0.025	pCi/g
DU-1	Thorium-230	1.5	J	4.3	8.4	pCi/g
DU-1	Thorium-232	2.5	J	2.5	7.4	pCi/g
DU-1	Tin	5.9	J	12	779,000	mg/kg
DU-1	Titanium	926	J	1,270	5,190,000	mg/kg
DU-1	Toluene	0.0022	J	0.0022	817	mg/kg
DU-1	Toxaphene	0.62	J	0.62	2.3	mg/kg
DU-1	Trichlorofluoromethane	0.0017	J	0.0017	1,210	mg/kg
DU-1	Tungsten	0.97	J	8.5	1,040	mg/kg
DU-1	Uranium (total)	1.0	J	3.1	3,830	ma/ka
DU-1	Úranium-234	2.0	J	2.8	11	pCi/a
DU-1	Uranium-235	0.12	J	0.25	0.35	pCi/a
DU-1	Uranium-238	2.2	J	2.5	1.4	pCi/a
DU-1	Vanadium	49	J	78	6.420	ma/ka
<u> </u>			-		5,125	

Decision		Maximum		Maximum Detected	Saraaning	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit	_	Qualified Data		BHRA Data Set	Levei	
DU-1	Vinyl chloride	0.00028	J	0.00028	2.2	mg/kg
DU-1	Zinc	73	J	300	389,000	mg/kg
DU-1	1,1-Dichloroethene	0.00055	J-	0.0012	1,100	mg/kg
DU-1	1,2,4-Trichlorobenzene	0.0024	J-	0.0037	125	mg/kg
DU-1	1,2,4-Trimethylbenzene	0.00064	J-	0.0014	218	mg/kg
DU-1	2-Butanone	0.0032	J-	0.027	28,400	mg/kg
DU-1	4,4'-DDE	0.74	J-	6.0	9.5	mg/kg
DU-1	4,4'-DDT	0.16	J-	2.3	7.5	mg/kg
DU-1	Acetone	0.023	J-	0.15	1,040,000	mg/kg
DU-1	Antimony	1.0	J-	2.4	519	mg/kg
DU-1	Arsenic	4.1	J-	34	7.2	mg/kg
DU-1	Benzo(g,h,i)perylene	0.38	J-	0.38	25,300	mg/kg
DU-1	beta-BHC	0.27	J-	0.87	1.7	mg/kg
DU-1	Boron	15	J-	1,510	259,000	mg/kg
DU-1	Chlorate	11	J-	20,900	38,900	mg/kg
DU-1	Chloroform	0.0011	J-	0.15	1.5	mg/kg
DU-1	Chromium (total)	23	J-	102	1,950,000	mg/kg
DU-1	Cobalt	131	J-	284	385	mg/kg
DU-1	Copper	10	J-	160	36,700	mg/kg
DU-1	Hexachlorobenzene	0.22	J-	4.7	0.23	mg/kg
DU-1	Lead	9.9	J-	267	800	mg/kg
DU-1	Magnesium	10,200	J-	71,000	5,200,000	mg/kg
DU-1	Manganese	471	J-	29,200	28,100	mg/kg
DU-1	Naphthalene	2.0	J-	3.1	18	mg/kg
DU-1	Nickel	18	J-	164	24,700	mg/kg
DU-1	Perchlorate	3.7	J-	1.500	908	mg/kg
DU-1	Phosphorus (total)	959	J-	1.600	9.630.000	mg/kg
DU-1	Radium-226	1.8	J-	2.5	0.023	pCi/a
DU-1	Radium-228	1.1	J-	3.3	0.041	pCi/q
DU-1	Sodium	3,730	J-	11,700	NA	mg/kg
DU-1	Sulfate	927	J-	14.600	NA	mg/kg
DU-1	Thorium-230	0.67	J-	4.3	8.4	pCi/q
DU-1	Titanium	1.160	J-	1.270	5.190.000	mg/kg
DU-1	Trichloroethene	0.00042	J-	0.00042	6.9	mg/kg
DU-1	Tungsten	8.5	J-	8.5	1,040	mg/kg
DU-1	Uranium-238	1.7	J-	2.5	1.4	pCi/q
DU-1	Vanadium	50	J-	78	6.420	mg/kg
DU-1	Zinc	34	J-	300	389,000	mg/kg
DU-1	4,4'-DDD	0.0027	J+	0.032	15	mg/kg
DU-1	4,4'-DDE	0.085	J+	6.0	9.5	mg/kg
DU-1	4,4'-DDT	0.033	J+	2.3	7.5	mg/kg
DU-1	Acetone	0.15	J+	0.15	1,040,000	mg/kg
DU-1	Aldrin	0.00052	J+	0.00052	0.21	mg/kg
DU-1	alpha-BHC	0.0013	J+	0.0013	0.49	mg/kg
DU-1	Ammonia	1.8	J+	3.5	6,140	mg/kg
DU-1	Beryllium	0.60	J+	0.70	2,540	mg/kg
DU-1	beta-BHC	0.059	J+	0.87	1.7	mg/kg
DU-1	bis(2-Ethylhexyl)phthalate	0.46	J+	0.82	183	ma/ka
DU-1	Bromide	0.70	J+	4.6	441.000	mg/kg
DU-1	Chlorate	31	J+	20.900	38.900	ma/ka
DU-1	Chloride	2.070	J+	6.670	113.000	ma/ka
DU-1	Copper	21	J+	160	36.700	ma/ka
DU-1	delta-BHC	0.00080	J+	0.0011	334	ma/ka
DU-1	Di-n-butvlphthalate	1.7	J+	3.0	91,600	ma/ka
DU-1	Endosulfan I	0.00024	J+	0.0015	5.500	ma/ka
DU-1	Endosulfan sulfate	0.016	J+	0.016	5,500	ma/ka
		0.010	- ·	0.010	3,000	

Decision		Maximum		Maximum Detected	Sereening	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit	_	Qualified Data		BHRA Data Set	Levei	
DU-1	Endrin ketone	0.0045	J+	0.0045	30	mg/kg
DU-1	gamma-BHC	0.00083	J+	0.0013	2.8	mg/kg
DU-1	Hexachlorobenzene	0.098	J+	4.7	0.23	mg/kg
DU-1	Lead	75	J+	267	800	mg/kg
DU-1	Manganese	943	J+	29,200	28,100	mg/kg
DU-1	Methoxychlor	0.011	J+	0.38	4,580	mg/kg
DU-1	Nickel	17	J+	164	24,700	mg/kg
DU-1	Nitrate	173	J+	173	2,080,000	mg/kg
DU-1	Perchlorate	170	J+	1,500	908	mg/kg
DU-1	Pyrene	0.14	J+	1.3	44	mg/kg
DU-1	Radium-226	1.7	J+	2.5	0.023	pCi/g
DU-1	Radium-228	2.0	J+	3.3	0.041	pCi/g
DU-1	Sulfate	990	J+	14,600	NA	mg/kg
DU-1	Titanium	441	J+	1,270	5,190,000	mg/kg
DU-1	Uranium (total)	0.86	J+	3.1	3,830	mg/kg
DU-1	Uranium-235	0.021	J+	0.25	0.35	pCi/g
DU-1	Vanadium	44	J+	78	6,420	mg/kg
DU-1	Zinc	269	J+	300	389,000	mg/kg
DU-2	1,1,1-Trichloroethane	0.00095	J	0.00095	638	mg/kg
DU-2	1,1-Dichloroethane	0.0030	J	0.0030	17	mg/kg
DU-2	1,2,4-Trimethylbenzene	0.00082	J	0.00082	218	mg/kg
DU-2	1,4-Dichlorobenzene	0.016	J	0.016	475	mg/kg
DU-2	2,3,7,8-TCDD TEQ	0.0016	J	0.0016	0.0027	mg/kg
DU-2	2-Butanone	0.014	J	0.014	28,400	mg/kg
DU-2	4,4'-DDE	0.025	J	0.91	9.5	mg/kg
DU-2	4,4'-DDT	0.069	J	0.61	7.5	mg/kg
DU-2	Acenaphthylene	0.0015	J	0.0015	118	mg/kg
DU-2	Acetone	0.090	J	0.090	1,040,000	mg/kg
DU-2	alpha-BHC	0.0066	J	0.012	0.49	mg/kg
DU-2	Anthracene	0.021	J	0.021	4.3	mg/kg
DU-2	Antimony	1.9	J	2.2	519	mg/kg
DU-2	Aroclor-1248	0.091	J	0.091	1.1	mg/kg
DU-2	Aroclor-1260	0.034	J	0.034	1.1	mg/kg
DU-2	Arsenic	6.3	J	6.3	7.2	mg/kg
DU-2	Barium	191	J	217	238,000	mg/kg
DU-2	Benzo(g,h,i)perylene	0.075	J	0.075	25,300	mg/kg
DU-2	Beryllium	0.56	J	0.60	2,540	mg/kg
DU-2	beta-BHC	0.057	J	0.29	1.7	mg/kg
DU-2	bis(2-Ethylhexyl)phthalate	0.097	J	61	183	mg/kg
DU-2	Boron	9.8	J	49	259,000	mg/kg
DU-2	Bromide	1.1	J	1.1	441,000	mg/kg
DU-2	Cadmium	0.080	J	8.9	1,260	mg/kg
DU-2	Calcium	40,900	J	43,900	NA	mg/kg
DU-2	Chlorate	0.54	J	1,510	38,900	mg/kg
DU-2	Chloride	5.0	J	2,470	113,000	mg/kg
DU-2	Chromium (total)	9.4	J	19	1,950,000	mg/kg
DU-2	Chromium VI	1.6	J	2.4	7.0	mg/kg
DU-2	Copper	19	J	32	36,700	mg/kg
DU-2	Dimethoate	0.013	J	0.013	183	mg/kg
DU-2	Dimethylphthalate	0.033	J	0.033	9,160,000	mg/kg
DU-2	Di-n-butylphthalate	0.077	J	7.5	91,600	mg/kg
DU-2	Fluoranthene	0.28	J	0.28	33,700	mg/kg
DU-2	Fluoride	0.52	J	0.52	51,900	mg/kg
DU-2	Hexachlorobenzene	0.075	J	0.52	0.23	mg/kg
DU-2	Iron	15,700	J	18,400	908,000	mg/kg
DU-2	Lead	14	J	51	800	mg/kg

Decision		Maximum		Maximum Detected	Sorooning	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit		Qualified Data		BHRA Data Set	Level	
DU-2	Lithium	10	J	14	2,600	mg/kg
DU-2	m,p-Xylene	0.0015	J	0.0015	387	mg/kg
DU-2	Manganese	483	J	1,290	28,100	mg/kg
DU-2	Mercury	0.017	J	0.056	389	mg/kg
DU-2	Methylene Chloride	0.00034	J	0.00034	1,550	mg/kg
DU-2	Molybdenum	0.49	J	0.92	6,490	mg/kg
DU-2	Nickel	18	J	18	24,700	mg/kg
DU-2	Nitrite	0.21	J	4.7	130,000	mg/kg
DU-2	Octachlorostyrene	0.21	J	0.21	NA	mg/kg
DU-2	ortho-Phosphate	2.4	J	7.2	30,400,000	mg/kg
DU-2	Perchlorate	34	J	34	908	mg/kg
DU-2	Phenanthrene	0.15	J	0.15	25	mg/kg
DU-2	Phosphorus (total)	727	J	1,110	9,630,000	mg/kg
DU-2	Platinum	0.077	J	0.077	649	mg/kg
DU-2	Pyrene	0.26	J	0.26	44	mg/kg
DU-2	Radium-226	1.3	J	1.3	0.023	pCi/g
DU-2	Radium-228	1.3	J	2.4	0.041	pCi/g
DU-2	Selenium	1.5	J	1.5	6,490	mg/kg
DU-2	Silver	0.22	J	0.22	6,490	mg/kg
DU-2	Sodium	1,730	J	1,920	NA	mg/kg
DU-2	Strontium	202	J	309	779,000	mg/kg
DU-2	Sulfate	1,060	J	2,260	NA	mg/kg
DU-2	Sulfur	785	J	1,290	NA	mg/kg
DU-2	Thallium	0.15	J	0.15	13	mg/kg
DU-2	Thorium-228	1.7	J	2.2	0.025	pCi/g
DU-2	Thorium-230	0.99	J	2.0	8.4	pCi/g
DU-2	Thorium-232	1.5	J	2.3	7.4	pCi/g
DU-2	Tin	5.8	J	5.8	779,000	mg/kg
DU-2	Titanium	744	J	1,140	5,190,000	mg/kg
DU-2	Toluene	0.0011	J	0.0011	817	mg/kg
DU-2	Trichlorofluoromethane	0.0017	J	0.0017	1,210	mg/kg
DU-2	Tungsten	0.58	J	1.4	1,040	mg/kg
DU-2	Uranium (total)	1.5	J	1.7	3,830	mg/kg
DU-2	Uranium-234	1.4	J	1.9	11	pCi/g
DU-2	Uranium-235	0.096	J	0.21	0.35	pCi/g
DU-2	Uranium-238	0.98	J	1.7	1.4	pCi/g
DU-2	Zirconium	22	J	25	104	mg/kg
DU-2	4,4'-DDE	0.0056	J-	0.91	9.5	mg/kg
DU-2	4,4'-DDT	0.0047	J-	0.61	7.5	mg/kg
DU-2	Acetone	0.020	J-	0.090	1,040,000	mg/kg
DU-2	Antimony	2.2	J-	2.2	519	mg/kg
DU-2	Beryllium	0.40	J-	0.60	2,540	mg/kg
DU-2	Boron	49	J-	49	259,000	mg/kg
DU-2		29,700	J-	43,900	NA	mg/kg
DU-2	Chlorate	18	J-	1,510	38,900	mg/kg
DU-2		5.4	J-	2,470	113,000	mg/kg
DU-2	Chromium (total)	19	J-	19	1,950,000	mg/kg
DU-2	Copper	8.8	J-	8.8	385	mg/kg
	Copper	1/	J-	32	36,700	mg/kg
DU-2	Magnasium	12,000	J-	18,400	908,000	mg/kg
	Magnesium	8,360	J-	28,800	5,200,000	mg/kg
	Marauny	532	J-	1,290	28,100	mg/kg
DU-2	INIERCUTY Niekol	0.019	J-	0.056	389	mg/kg
DU-2		14	J-	18	24,700	mg/kg
DU-2		4.7	J-	4.7	130,000	mg/kg
DU-2	Perchiorate	2.6	J-	34	908	mg/kg
DU-2	Potassium	2,200	J-	2,480	NA	mg/kg
# TABLE B-6. Summary of J Qualified Soil DataNevada Environmental Response Trust SiteHenderson, Nevada

Dunit     Analyte     Concentration of Qualified Data     Concentration in Soil     Level     Unit       DU-2     Radum-228     1.1     J-     1.3     0.023     C/20       DU-2     Radum-228     1.8     J-     2.4     0.041     D/20       DU-2     Sodium     1.790     J-     1.820     NA     mgkpg       DU-2     Yungstein     1.4     J-     1.4     1.44     1.44     1.44     1.44     1.44     1.44     1.44     0.400     mgkpg       DU-2     Vanadum     2.01     J+     0.62     6.420     mgkpg       DU-2     Acetone     0.026     J+     0.030     1.40.4000     mgkpg       DU-2     Calcium     9.930     J+     2.17     238.000     mgkpg       DU-2     Calcium     9.930     J+     1.280     NA     mgkpg       DU-2     Nohum     9.22     J+     8.2     1.300     mgkpg       DU-2     Nohum     9.2     J+ <t< th=""><th>Decision</th><th></th><th>Maximum</th><th></th><th>Maximum Detected</th><th>Scrooning</th><th></th></t<>	Decision		Maximum		Maximum Detected	Scrooning	
Unit     Dul-2     Radum-28     1.1     J-     1.3     0.023     Radum-28       DU-2     Radum-28     1.8     J-     2.4     0.041     DC/2       DU-2     Radum-28     1.8     J-     1.42     0.041     DC/2       DU-2     Tungsten     1.4     J-     1.42     1.44     1.490     mgkg       DU-2     Tungsten     1.4     J-     1.62     6.420     mgkg       DU-2     Zanc     39     J-     6.6     380,000     mgkg       DU-2     Zanc     39     J-     6.6     380,000     mgkg       DU-2     Barum     2017     J+     2.470     113,000     mgkg       DU-2     Chorate     1.08     J+     1.510     38,800     mgkg       DU-2     Nicotae     2.0     J+     2.470     113,000     mgkg       DU-2     Nicotae     2.0     J+     3.2     J+     2.2     2.000     mgkg       DU-2	Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
DU-2     Radium-228     1.1     J-     1.3     0.023     pC/g0       DU-2     Radium-228     1.8     J-     2.4     0.041     DC/g0       DU-2     Sodium     1.790     J-     1.920     NA     mgkg0       DU-2     Tungsten     1.4     J-     1.4     1.44 <td< th=""><th>Unit</th><th>-</th><th>Qualified Data</th><th></th><th>BHRA Data Set</th><th>Level</th><th></th></td<>	Unit	-	Qualified Data		BHRA Data Set	Level	
DU-2     Redum-228     1.8     J-     2.4     0.041     pC/g       DU-2     Tungaten     1.790     J-     1920     NA     mg/kg       DU-2     Tungaten     1.4     J-     1.4     1.4     mg/kg       DU-2     Zinc     39     J-     68     989.000     mg/kg       DU-2     Zinc     39     J-     68     989.000     mg/kg       DU-2     Zactone     0.026     J+     0.090     1.404.000     mg/kg       DU-2     Cactor     9.930     J+     43.900     NA     mg/kg       DU-2     Chiorité     108     J+     1.510     38.900     mg/kg       DU-2     Chiorité     108     J+     1.200     28.100     mg/kg       DU-2     Nindamagenese     378     J+     1.22.91     113.000     mg/kg       DU-2     Nindamagenese     378     J+     1.200     28.100     mg/kg       DU-2     Nintain     2.2     2.00.000	DU-2	Radium-226	1.1	J-	1.3	0.023	pCi/a
DU-2     Sodium     1,780     J-     1,920     NA     mg/kg       DU-2     Vanadium     43     J-     62     6,420     mg/kg       DU-2     Zunc     33     J-     68     980.000     mg/kg       DU-2     Acatone     0.026     J+     0.090     1.040.000     mg/kg       DU-2     Acatone     0.026     J+     0.300     mg/kg       DU-2     Calcium     9.930     J+     43.900     NA     mg/kg       DU-2     Chioride     49.53     J+     1.200     28.100     mg/kg       DU-2     Chioride     49.7     J+     9.2     13.0     mg/kg       DU-2     Ntrate     20     J+     52     2.080.000     mg/kg       DU-2     Phosphorus (total)     1.010     J+     1.110     9.830.000     mg/kg       DU-2     Strontium     226     J+     309     779.000     mg/kg       DU-2     Strontium     226     J+     3	DU-2	Radium-228	1.8	J-	2.4	0.041	pCi/q
DU-2     Tungsten     1.4     J-     1.4     1.4     1.4     mg/kg       DU-2     Zinc     33     J-     68     980.000     mg/kg       DU-2     Zinc     33     J-     68     980.000     mg/kg       DU-2     Bartum     201     J+     0.026     J+     0.020     mg/kg       DU-2     Chiorate     108     J+     43,900     NA     mg/kg       DU-2     Chiorate     108     J+     1,510     38,900     mg/kg       DU-2     Chiorate     108     J+     1,200     28,100     mg/kg       DU-2     Nicobum     9,2     J+     9,2     28,000     mg/kg       DU-2     Nicobum     9,2     J+     9,2     28,000     mg/kg       DU-2     Nicobum     9,2     J+     9,2     28,000     mg/kg       DU-2     Nicobum     9,2     J+     9,2     20,000     mg/kg       DU-2     Sitiisto     6,7	DU-2	Sodium	1.790	J-	1.920	NA	ma/ka
DU-2     Vanadum     43     J-     62     6.420     mg/kg       DU-2     Acetone     0.026     J+     0.080     mg/kg       DU-2     Acetone     0.026     J+     0.090     mg/kg       DU-2     Calcium     201     J+     43.000     mg/kg       DU-2     Calcium     9.930     J+     43.000     mg/kg       DU-2     Chioriate     108     J+     2.160     mg/kg       DU-2     Chioriate     108     J+     2.270     113.000     mg/kg       DU-2     Minganese     378     J+     9.2     13.000     mg/kg       DU-2     Nitrate     20     J+     9.2     2.800.00     mg/kg       DU-2     Pinchiorate (otal)     1.010     J+     1.110     9.830.00     mg/kg       DU-2     Strontium     226     J+     309     779.000     mg/kg       DU-2     Strontium     2268     J+     1.40     5190.000     mg/kg	DU-2	Tungsten	1.4	J-	1.4	1.040	ma/ka
DU-2     Zinc     39     J-     68     398,000     mg/kg       DU-2     Barium     201     J+     0.190     1.940,000     mg/kg       DU-2     Barium     201     J+     217     238,000     mg/kg       DU-2     Chiorate     108     J+     1.510     38,900     mg/kg       DU-2     Chiorate     108     J+     1.2470     130.00     mg/kg       DU-2     Nicolum     9.2     J+     9.2     2.810.00     mg/kg       DU-2     Nicolum     9.2     J+     9.2     2.000     mg/kg       DU-2     Nicolum     9.2     J+     9.2     2.000     mg/kg       DU-2     Picrhiorate     6.7     J+     3.4     9.000     mg/kg       DU-2     Picrhiorate     6.7     J+     3.4     9.000     mg/kg       DU-2     Silfate     2.26     J+     1.000     mg/kg       DU-2     Silfate     2.26     J+     2.00012	DU-2	Vanadium	43	J-	62	6,420	mg/kg
DU-2     Acetone     0.026     J+     0.030     1.040,000     mg/kg       DU-2     Calcium     9.930     J+     43.900     NA     mg/kg       DU-2     Calcium     9.930     J+     43.900     mg/kg       DU-2     Chiorate     108     J+     1.510     38.900     mg/kg       DU-2     Nicolum     9.2     J+     9.2     1.300     mg/kg       DU-2     Nicolum     9.2     J+     9.2     1.30     mg/kg       DU-2     Nicolum     9.2     J+     9.2     1.30     mg/kg       DU-2     Nicolum     9.1     J+     3.4     9.08     mg/kg       DU-2     Phosphorus (total)     1.010     J+     1.110     9.830,000     mg/kg       DU-2     Strontium     2.26     J+     3.09     mg/kg       DU-2     Strontium     2.26     J+     1.140     9.830,000     mg/kg       DU-3     1.2.4 Tichiorbonzene     0.0012     J     0.00	DU-2	Zinc	39	J-	68	389,000	mg/kg
DU-2     Barium     201     J+     217     238,000     mg/kg       DU-2     Chiorate     108     J+     1,510     38,900     mg/kg       DU-2     Chiorate     108     J+     1,510     38,900     mg/kg       DU-2     Chiorate     108     J+     1,220     28,100     mg/kg       DU-2     Nicolum     9,2     J+     9,2     2,800     mg/kg       DU-2     Nicolum     9,2     J+     9,2     2,800,00     mg/kg       DU-2     Nicolum     9,2     J+     5,2     2,060,00     mg/kg       DU-2     Procisions     6,7     J+     3,4     908     mg/kg       DU-2     Silicon     1,80     J+     1,110     9,530,00     mg/kg       DU-2     Silicon     180     J+     1,200     NA     mg/kg       DU-2     Silicon     160,012     125     mg/kg     0.012     125     mg/kg       DU-3     Silifate     226	DU-2	Acetone	0.026	J+	0.090	1,040,000	mg/kg
DU-2     Calcium     9,330     J+     43,900     NA     mg/kg       DU-2     Chiorate     108     J+     1,510     38,900     mg/kg       DU-2     Chiorate     495     J+     2,470     113,000     mg/kg       DU-2     Nitobium     5.2     J+     9.2     130     mg/kg       DU-2     Nitrate     20     J+     5.2     2,080,000     mg/kg       DU-2     Prosphorus (total)     1,010     J+     4,110     9,630,000     mg/kg       DU-2     Strontium     226     J+     309     T79,000     mg/kg       DU-2     Strontium     226     J+     309     T79,000     mg/kg       DU-2     Strontium     266     J+     1,400     NA     mg/kg       DU-2     Strontium     606     J+     1,140     5,190,000     mg/kg       DU-3     1,2,4-trichorbonzene     0,0012     125     mg/kg     DU-3     1,2,4-trichorbonzene     0,0021     0,0027 <t< td=""><td>DU-2</td><td>Barium</td><td>201</td><td>J+</td><td>217</td><td>238,000</td><td>mg/kg</td></t<>	DU-2	Barium	201	J+	217	238,000	mg/kg
DU-2     Chiorate     108     J+     1.510     38,900     mg/kg       DU-2     Manganese     378     J+     1.290     28,100     mg/kg       DU-2     Nitrate     20     J+     5.2     2,080,000     mg/kg       DU-2     Nitrate     20     J+     5.2     2,080,000     mg/kg       DU-2     Picrohiorate     6.7     J+     34     908     mg/kg       DU-2     Picrohiorate     6.7     J+     34     908     mg/kg       DU-2     Silicon     180     J+     1,110     9,650,000     mg/kg       DU-2     Stilicon     180     J+     1,40     5,190,000     mg/kg       DU-2     Stilicon     606     J+     1,140     5,190,000     mg/kg       DU-3     1,2.4.Trichiorobenzene     0.0012     J     0.0021     0.0027     mg/kg       DU-3     2.4.Methyinapithalene     0.026     J     0.026     mg/kg       DU-3     2.4.Methyinapithalene	DU-2	Calcium	9,930	J+	43,900	NA	mg/kg
DU-2     Chloride     495     J+     2.470     113.000     mg/kg       DU-2     Nindium     9.2     J+     9.2     130     mg/kg       DU-2     Nintale     20     J+     9.2     130     mg/kg       DU-2     Prechlorate     6.7     J+     34     908     mg/kg       DU-2     Prechlorate     6.7     J+     34     908     mg/kg       DU-2     Strontium     226     J+     309     779.000     mg/kg       DU-2     Strontium     226     J+     309     779.000     mg/kg       DU-2     Strontium     226     J+     140     5.190.000     mg/kg       DU-3     1.2-Dichlorobenzene     0.00037     J     0.00037     376     mg/kg       DU-3     2.4-Trianium     606     J     0.0021     0.0027     mg/kg       DU-3     2.4-Indirobenzene     0.0052     J     0.0021     mg/kg       DU-3     2.4-ArtoDT     0.0052     J<	DU-2	Chlorate	108	J+	1,510	38,900	mg/kg
DU-2     Manganese     378     J+     1,280     28,100     mg/kg       DU-2     Nitrate     20     J+     52     2,080,000     mg/kg       DU-2     Nitrate     20     J+     52     2,080,000     mg/kg       DU-2     Prechlorate     6.7     J+     34     908     mg/kg       DU-2     Prechlorate     6.7     J+     34     908     mg/kg       DU-2     Sticon     180     J+     180     NA     mg/kg       DU-2     Stican     226     J+     309     779,000     mg/kg       DU-3     1,2,4-Trichlorobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     1,2,4-Trichlorobenzene     0.0021     J     0.0021     0.027     mg/kg       DU-3     1,2,4-Trichlorobenzene     0.0045     J     0.0021     0.027     mg/kg       DU-3     2-Methylnaphthalene     0.026     J     0.026     mg/kg       DU-3     2-Methylnaphthalene	DU-2	Chloride	495	J+	2,470	113,000	mg/kg
DU-2     Nitrate     9.2     J+     9.2     2.0     mg/kg       DU-2     Perchlorate     6.7     J+     52     2.080.000     mg/kg       DU-2     Perchlorate     6.7     J+     34     908     mg/kg       DU-2     Sticon     180     J+     180     NA     mg/kg       DU-2     Stirontum     226     J+     309     779.000     mg/kg       DU-2     Stirontum     226     J+     309     779.000     mg/kg       DU-2     Stirate     228     J+     2.0012     114     5190.000     mg/kg       DU-3     1.2.4-Trichlorobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     2.7.8-TCDD TEQ     0.0021     J     0.0027     mg/kg       DU-3     2.4.4-tDDT     0.052     J     0.020     9.5     mg/kg       DU-3     2.4-deltympithalene     0.026     J     0.026     1.8     mg/kg       DU-3     Acetone     0	DU-2	Manganese	378	J+	1,290	28,100	mg/kg
DU-2     Nitrate     20     J+     52     2,080,000     mg/kg       DU-2     Perchlorate     6,7     J+     34     908     mg/kg       DU-2     Pinsphorus (total)     1,010     J+     1,110     9,630,000     mg/kg       DU-2     Silicon     180     J+     180     NA     mg/kg       DU-2     Suffate     226     J+     309     779,000     mg/kg       DU-2     Stafate     228     J+     2,260     NA     mg/kg       DU-3     1,2.4-Trichlorobenzene     0.0012     J     0.00037     376     mg/kg       DU-3     2.8,7,8-TCDD TEQ     0.0021     J     0.0021     0.0021     0.0021     0.0021       DU-3     2.4 Methylnaphtalene     0.030     J     0.030     388     mg/kg       DU-3     2.4 Methylnaphtalene     0.026     J     0.026     1.040,000     mg/kg       DU-3     A.4 DDE     0.0652     J     0.026     1.040,000     mg/kg	DU-2	Niobium	9.2	J+	9.2	130	mg/kg
DU-2     Perchlorate     6.7     J+     34     908     mg/kg       DU-2     Phosphorus (total)     1.010     J+     1.110     9.630.000     mg/kg       DU-2     Strontium     226     J+     309     779.000     mg/kg       DU-2     Strontium     226     J+     3.09     779.000     mg/kg       DU-2     Titanium     606     J+     1.140     5.190.000     mg/kg       DU-3     1.2.4-Trichforobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     1.2.2-Chtlorobenzene     0.00021     J     0.00027     mg/kg       DU-3     2.4.7-trichforobenzene     0.0021     J     0.0027     mg/kg       DU-3     2.4.2-trichtorobenzene     0.0025     J     0.020     mg/kg       DU-3     2.4.4-DDT     0.065     J     0.20     9.5     mg/kg       DU-3     A.4+DDT     0.062     J     0.052     1.04.000     mg/kg       DU-3     Atetone     0.026	DU-2	Nitrate	20	J+	52	2,080,000	mg/kg
DU-2     Phosphorus (total)     1,010     J+     1,110     9,630,000     mg/kg       DU-2     Siticon     180     J+     180     NA     mg/kg       DU-2     Siticon     226     J+     309     779,000     mg/kg       DU-2     Stufate     226     J+     309     779,000     mg/kg       DU-3     1,2,4-Trichlorobenzene     0.0012     J     0.00037     J     0.00037     376     mg/kg       DU-3     1,2,4-Trichlorobenzene     0.00012     J     0.00021     0.0027     mg/kg       DU-3     2,3,7.8-TCDD TEO     0.0045     J     0.0045     28,400     mg/kg       DU-3     4,4'-DDE     0.065     J     0.033     37.5     mg/kg       DU-3     4,4'-DDE     0.052     J     0.052     1,040,000     mg/kg       DU-3     Acetaoe     0.052     J     0.052     1,040,000     mg/kg       DU-3     Acetaoe     0.56     J     11.7     7.40,000     mg/kg <td>DU-2</td> <td>Perchlorate</td> <td>6.7</td> <td>J+</td> <td>34</td> <td>908</td> <td>mg/kg</td>	DU-2	Perchlorate	6.7	J+	34	908	mg/kg
DU-2     Silicon     180     J+     180     NA     mg/kg       DU-2     Strontum     226     J+     309     779.000     mg/kg       DU-2     Strontum     258     J+     2,260     NA     mg/kg       DU-3     1,2-1richiorobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     1,2-2richiorobenzene     0.00037     J     0.0021     0.0027     mg/kg       DU-3     2,3-78-TCDD TEQ     0.0021     J     0.0045     28,400     mg/kg       DU-3     2,3-78-TCDD TEQ     0.0065     J     0.020     9.5     mg/kg       DU-3     2,4-trongenthalene     0.030     J     0.030     368     mg/kg       DU-3     4,4'-DDT     0.052     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.026     14.4'O.000     mg/kg       DU-3     Aluminum     7,600     J     11.700     1,240,000     mg/kg       DU-3	DU-2	Phosphorus (total)	1,010	J+	1,110	9,630,000	mg/kg
DU-2     Strontium     226     J+     309     779,000     mg/kg       DU-2     Sulfate     258     J+     2.260     NA     mg/kg       DU-3     1,2.4-Trichorobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     1,2.4-Trichorobenzene     0.00037     J     0.00037     376     mg/kg       DU-3     2,3.7.8-TCDD TEQ     0.0021     J     0.0045     28,400     mg/kg       DU-3     2.Methylnaphtalene     0.030     J     0.030     388     mg/kg       DU-3     4.4'-DDE     0.065     J     0.20     9.5     mg/kg       DU-3     4.4'-DDT     0.052     J     0.033     7.5     mg/kg       DU-3     Actone     0.026     J     0.026     1.040,000     mg/kg       DU-3     Actone     0.052     J     0.025     0.49     mg/kg       DU-3     Alpha-BHC     0.025     J     0.025     0.49     mg/kg       DU-3     Acteton	DU-2	Silicon	180	J+	180	NA	mg/kg
DU-2     Sulfate     258     J+     2,260     NA     mg/kg       DU-2     Titanium     606     J+     1,140     5,190,000     mg/kg       DU-3     1,2,4-Tichlorobenzene     0.00037     J     0.00037     376     mg/kg       DU-3     2,3,7,8-TCDD TEQ     0.00021     J     0.0021     0.0027     mg/kg       DU-3     2.5-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     2.4-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     4.4'-DDT     0.052     J     0.026     118     mg/kg       DU-3     Acenaphthene     0.026     J     0.0025     0.49     mg/kg       DU-3	DU-2	Strontium	226	J+	309	779,000	mg/kg
DU-3     Titanium     606     J+     1,140     5,190,000     mg/kg       DU-3     1,2-Dichlorobenzene     0.00037     J     0.00037     376     mg/kg       DU-3     2,3.7,8-TCDD TEQ     0.0021     J     0.0021     0.0027     mg/kg       DU-3     2.8-Butanone     0.0045     J     0.0021     0.0027     mg/kg       DU-3     2.4-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     2.4-Methylnaphthalene     0.0652     J     0.20     9.5     mg/kg       DU-3     Acenaphthene     0.052     J     0.33     7.5     mg/kg       DU-3     Acetone     0.052     J     0.052     1.040.000     mg/kg       DU-3     Aluminum     7.600     J     11.700     1.240.000     mg/kg       DU-3     Animonia     563     J     563     6,140     mg/kg       DU-3     Animony     2.1     J     2.1     238.000     mg/kg       DU-3 <td>DU-2</td> <td>Sulfate</td> <td>258</td> <td>J+</td> <td>2,260</td> <td>NA</td> <td>mg/kg</td>	DU-2	Sulfate	258	J+	2,260	NA	mg/kg
DU-3     1,2,4-Trichlorobenzene     0.0012     J     0.0012     125     mg/kg       DU-3     1,2,51chlorobenzene     0.00037     J     0.00021     0.0027     mg/kg       DU-3     2,3,7,8-TCDD TEQ     0.0021     J     0.0045     28,400     mg/kg       DU-3     2.4Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     4.4'-DDE     0.065     J     0.20     9.5     mg/kg       DU-3     4.4'-DDT     0.052     J     0.026     118     mg/kg       DU-3     Acenaphthene     0.026     J     0.052     1.040.000     mg/kg       DU-3     Acenaphthene     0.052     J     0.052     1.040.000     mg/kg       DU-3     Akerone     0.052     J     0.0052     0.49     mg/kg       DU-3     Akerone     0.025     J     0.0052     0.49     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3	DU-2	Titanium	606	J+	1,140	5,190,000	mg/kg
DU-3     1,2-Dichlorobenzene     0.00037     J     0.00037     376     mg/kg       DU-3     2.3,7.8-TCDD TEQ     0.0021     J     0.0021     0.0027     mg/kg       DU-3     2-Butanone     0.0045     J     0.0020     388     mg/kg       DU-3     2-Methylnaphthalene     0.030     J     0.030     388     mg/kg       DU-3     4.4'-DDT     0.052     J     0.33     7.5     mg/kg       DU-3     Acetaphthene     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.025     1.040,000     mg/kg       DU-3     Aluminum     7.600     J     11,700     1,240,000     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Antimony     2.1     J     2.1     2.1     2.1     80.00     mg/kg       DU-3     Barium     191     J     2.1     2.38.000     mg/kg	DU-3	1,2,4-Trichlorobenzene	0.0012	J	0.0012	125	mg/kg
DU-3     2,3,7,8,TCDD TEQ     0.0021     J     0.0021     0.0027     mg/kg       DU-3     2-Butanone     0.0045     J     0.0045     28,400     mg/kg       DU-3     2-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     4,4*DDF     0.065     J     0.20     9.5     mg/kg       DU-3     A,4*DDT     0.052     J     0.033     7.5     mg/kg       DU-3     A,4*DDT     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.025     J     0.0025     0.49     mg/kg       DU-3     Alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Antiminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Antiminum     7,600     J     111     7.2     mg/kg       DU-3     Arsenic     5.6     J     111     7.2     mg/kg       DU-3     Barium     0.71 <td>DU-3</td> <td>1,2-Dichlorobenzene</td> <td>0.00037</td> <td>J</td> <td>0.00037</td> <td>376</td> <td>mg/kg</td>	DU-3	1,2-Dichlorobenzene	0.00037	J	0.00037	376	mg/kg
DU-3     2-Butanone     0.0045     J     0.0045     28,400     mg/kg       DU-3     2-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     4,4'-DDE     0.065     J     0.20     9,5     mg/kg       DU-3     4,4'-DDT     0.052     J     0.033     7.5     mg/kg       DU-3     Acetone     0.026     J     0.052     1,040,000     mg/kg       DU-3     Alpha-BHC     0.025     J     0.0025     0.49     mg/kg       DU-3     Alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Alminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Antimony     2.1     J     2.21     238,00     mg/kg       DU-3     Barium     191     J     221     23,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     Ber/lium	DU-3	2,3,7,8-TCDD TEQ	0.0021	J	0.0021	0.0027	mg/kg
DU-3     2-Methylnaphthalene     0.030     J     0.030     368     mg/kg       DU-3     4,4'-DDT     0.065     J     0.20     9.5     mg/kg       DU-3     4,4'-DDT     0.052     J     0.33     7.5     mg/kg       DU-3     Acenaphthene     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.026     1,040,000     mg/kg       DU-3     Acenaphthene     0.025     J     0.026     1,040,000     mg/kg       DU-3     Atuminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Barzo(g,h,i)perylene     0.19     J     0.19     25,300     mg/kg       DU-3     Berzo(g,h,i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     Berzo(g,h,i	DU-3	2-Butanone	0.0045	J	0.0045	28,400	mg/kg
DU-3     4,4'-DDE     0.065     J     0.20     9.5     mg/kg       DU-3     4,4'-DDT     0.052     J     0.33     7.5     mg/kg       DU-3     Acenaphthene     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.052     1,040,000     mg/kg       DU-3     alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Animony     2.1     J     2.63     6,140     mg/kg       DU-3     Animony     2.1     J     2.1     519     mg/kg       DU-3     Barlum     191     J     2.1     238,000     mg/kg       DU-3     Beryllium     0.71     J     0.71     2,540     mg/kg       DU-3     Berylliphyliphthalate     0.17     J     0.652     1.7     mg/kg       DU-3     Boron     11	DU-3	2-Methylnaphthalene	0.030	J	0.030	368	mg/kg
DU-3     4,4'-DDT     0.052     J     0.33     7.5     mg/kg       DU-3     Acenaphthene     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.0652     1,040,000     mg/kg       DU-3     alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Ammonia     563     J     563     6,140     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Barium     191     J     2.21     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     Berl/limm     0.71     J     0.85     183     mg/kg       DU-3     beta-BHC     0.026     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11	DU-3	4,4'-DDE	0.065	J	0.20	9.5	mg/kg
DU-3     Acetone     0.026     J     0.026     118     mg/kg       DU-3     Acetone     0.052     J     0.052     1,040,000     mg/kg       DU-3     alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.19     25,300     mg/kg       DU-3     Berz/lium     0.71     J     0.85     183     mg/kg       DU-3     bis(2-Ethylexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     0.35     1,260     mg/kg       DU-3     Botrylbenzylphthalate <td< td=""><td>DU-3</td><td>4,4'-DD1</td><td>0.052</td><td>J</td><td>0.33</td><td>7.5</td><td>mg/kg</td></td<>	DU-3	4,4'-DD1	0.052	J	0.33	7.5	mg/kg
DU-3     Acetone     0.052     J     0.052     1,040,000     mg/kg       DU-3     alpha-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Ammonia     563     J     563     6,140     mg/kg       DU-3     Arsenic     5.6     J     11     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Berzo(g,h,i)perylene     0.19     J     0.11     7.2     mg/kg       DU-3     Berzyllium     0.711     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     0.35     1,260     mg/kg       DU-3     Cadmium     0.11	DU-3	Acenaphthene	0.026	J	0.026	118	mg/kg
DU-3     alpna-BHC     0.0025     J     0.0025     0.49     mg/kg       DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Anmonia     563     J     563     6,140     mg/kg       DU-3     Antmony     2.1     J     2.1     519     mg/kg       DU-3     Arsenic     5.6     J     111     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Cadmium     0.11     J	DU-3	Acetone	0.052	J	0.052	1,040,000	mg/kg
DU-3     Aluminum     7,600     J     11,700     1,240,000     mg/kg       DU-3     Antmonia     563     J     563     6,140     mg/kg       DU-3     Antmony     2.1     J     2.1     519     mg/kg       DU-3     Arsenic     5.6     J     111     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.83     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Cadmium     0.15     J<	DU-3	alpha-BHC	0.0025	J	0.0025	0.49	mg/kg
DU-3     Antimonia     563     J     563     6,140     mg/kg       DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Arsenic     5.6     J     11     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Berzo(g,h.i)perylene     0.19     J     0.71     2,540     mg/kg       DU-3     Beryllium     0.71     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloroform     0.0039	DU-3	Aluminum	7,600	J	11,700	1,240,000	mg/kg
DU-3     Antimony     2.1     J     2.1     519     mg/kg       DU-3     Arsenic     5.6     J     11     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.19     25,300     mg/kg       DU-3     Beryllium     0.71     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Botylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloroform     0.0039 <td>DU-3</td> <td>Ammonia</td> <td>563</td> <td>J</td> <td>563</td> <td>6,140</td> <td>mg/kg</td>	DU-3	Ammonia	563	J	563	6,140	mg/kg
DU-3     Arsenic     5.6     J     11     7.2     mg/kg       DU-3     Barium     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.19     225,300     mg/kg       DU-3     Beryllium     0.71     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorotenzene     0.0	DU-3	Antimony	2.1	J	2.1	519	mg/kg
DU-3     Banzo(g,h,i)perylene     191     J     221     238,000     mg/kg       DU-3     Benzo(g,h,i)perylene     0.19     J     0.19     25,300     mg/kg       DU-3     Beryllium     0.71     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.855     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.355     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloride     746     J     3,140     113,000     mg/kg       DU-3     Chloroform<	DU-3	Arsenic	5.6	J	11	7.2	mg/kg
DU-3     Benzolg,n,jperylene     0.19     J     0.19     25,300     mg/kg       DU-3     Beryllium     0.71     J     0.71     2,540     mg/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     0.11     J     0.35     1,260     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorate     0.0019     J     0.0039     1.5     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chloroform	DU-3	Barium Banzo(z. b. i)zozulozo	191	J	221	238,000	mg/kg
DU-3     Definition     D.71     J     D.71     Z,540     Ing/kg       DU-3     beta-BHC     0.026     J     0.052     1.7     mg/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chloroform     0.0039     J     0.0019     18,300     mg/kg       DU-3     Chloroform	DU-3	Benzo(g,n,i)perviene	0.19	J	0.19	25,300	mg/kg
DU-3     Deta-Bric     0.026     J     0.052     1.7     Ing/kg       DU-3     bis(2-Ethylhexyl)phthalate     0.17     J     0.85     183     mg/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Choroinum (total)     27     J     27     1,950,000     mg/kg       DU-3     Choroinum VI	DU-3	beta BHC	0.71	J	0.71	2,540	mg/kg
DU-3     Display Lethylphinatate     0.17     J     J     185     Ing/kg       DU-3     Boron     11     J     13     259,000     mg/kg       DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Choroium (total)     27     J     27     1,950,000     mg/kg       DU-3     Choroium VI     0.666     J     1.3     7.0     mg/kg       DU-3     Copper	DU-3	bis(2 Ethylboxyl)phthalato	0.026	J	0.052	1.7	mg/kg
DU-3     Butylbenzylphthalate     0.024     J     0.024     1,350     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloride     746     J     3,140     113,000     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0039     1.5     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     Diethylphthalate     0.3		Boron	0.17	3	0.85	250,000	mg/kg
DU-3     Cadmium     0.024     J     0.024     1,350     Ingrkg       DU-3     Cadmium     0.11     J     0.35     1,260     mg/kg       DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloride     746     J     3,140     113,000     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Di-n-butylphthalate     0.35<	DU-3	Butylbenzylphthalate	0.024	J	0.024	259,000	mg/kg
DU-3     Calcium     40,400     J     56,500     NA     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Choroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Di-n-butylphthalate     0.087	DU-3	Cadmium	0.024	J	0.35	1,350	mg/kg
DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chlorate     0.15     J     107     38,900     mg/kg       DU-3     Chloride     746     J     3,140     113,000     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate	DU-3	Calcium	40.400		56 500	1,200 ΝΔ	mg/kg
DU-3     Chloride     746     J     3,140     113,000     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.020     J     0.020     30     mg/kg       DU-3	DU-3	Chlorate	40,400		107	38 000	mg/kg
DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chlorobenzene     0.0019     J     0.0019     18,300     mg/kg       DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.020     J     0.020     30     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg	DU-3	Chloride	746	J	3 140	113 000	mg/kg
DU-3     Chloroform     0.0039     J     0.0039     1.5     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Chlorobenzene	0.0019	J	0.0019	18 300	ma/ka
DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium (total)     27     J     27     1,950,000     mg/kg       DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Chloroform	0.0010	J.	0.0039	15	ma/ka
DU-3     Chromium VI     0.66     J     1.3     7.0     mg/kg       DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Chromium (total)	27	J	27	1.950.000	ma/ka
DU-3     Copper     13     J     25     36,700     mg/kg       DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Chromium VI	0.66	J	13	7.0	ma/ka
DU-3     delta-BHC     0.0015     J     0.0015     334     mg/kg       DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Copper	13	J	25	36,700	ma/ka
DU-3     Diethylphthalate     0.35     J     0.35     733,000     mg/kg       DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-ctylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	delta-BHC	0.0015	J	0.0015	334	ma/ka
DU-3     Di-n-butylphthalate     0.087     J     0.65     91,600     mg/kg       DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Diethylphthalate	0.35	J	0.35	733.000	ma/ka
DU-3     Di-n-octylphthalate     0.084     J     0.084     9,160     mg/kg       DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Di-n-butylphthalate	0.087	J	0.65	91,600	ma/ka
DU-3     Endrin ketone     0.020     J     0.020     30     mg/kg       DU-3     Fluoranthene     0.071     J     0.071     33,700     mg/kg	DU-3	Di-n-octylphthalate	0.084	J	0.084	9,160	ma/ka
DU-3 Fluoranthene 0.071 J 0.071 33,700 mg/kg	DU-3	Endrin ketone	0.020	J	0.020	30	mg/kg
	DU-3	Fluoranthene	0.071	J	0.071	33,700	mg/kg

## TABLE B-6. Summary of J Qualified Soil DataNevada Environmental Response Trust SiteHenderson, Nevada

Desision		Maximum		Maximum Detected	Saraaning	
Decision	Analyte	Concentration of	Qualifier	Concentration in Soil	Screening	Unit
Unit	-	Qualified Data		BHRA Data Set	Level	
DU-3	Hexachlorobenzene	0.29	J	1.1	0.23	mg/kg
DU-3	Hexachlorobutadiene	0.0020	J	0.0020	6.1	mg/kg
DU-3	Iron	14,600	J	17,700	908,000	mg/kg
DU-3	Magnesium	22,000	J	22,000	5,200,000	mg/kg
DU-3	Manganese	390	J	1,500	28,100	mg/kg
DU-3	Mercury	0.017	J	0.31	389	mg/kg
DU-3	Methylene Chloride	0.0018	J	0.0018	1.550	mg/kg
DU-3	Molybdenum	0.74	J	0.97	6.490	mg/kg
DU-3	Naphthalene	0.0011	J	0.0011	18	ma/ka
DU-3	Nickel	16	J	17	24,700	mg/kg
DU-3	Nitrate	515	J	515	2,080,000	mg/kg
DU-3	Nitrite	2.3	J	77	130,000	mg/kg
DU-3	Octachlorostyrene	0.16	J	0.68	NA	mg/kg
DU-3	ortho-Phosphate	21	J	21	30,400,000	mg/kg
DU-3	Phenanthrene	0.089	J	0.089	25	mg/kg
DU-3	Platinum	0.085	J	0.16	649	mg/kg
DU-3	Potassium	2,050	J	2,290	NA	mg/kg
DU-3	Pyrene	0.24	J	0.24	44	mg/kg
DU-3	Radium-226	1.6	J	2.0	0.023	pCi/g
DU-3	Radium-228	1.7	J	1.9	0.041	pCi/g
DU-3	Selenium	1.0	J	1.0	6,490	mg/kg
DU-3	Silver	0.30	J	0.30	6,490	mg/kg
DU-3	Sodium	924	J	4.020	NA	ma/ka
DU-3	Strontium	805	J	805	779.000	ma/ka
DU-3	Sulfate	748	J	15.300	NA	ma/ka
DU-3	Tetrachloroethene	0.00068	J	0.00068	117	ma/ka
DU-3	Thallium	0.18	J	0.18	13	ma/ka
DU-3	Thorium-228	0.87	J	3.0	0.025	pCi/a
DU-3	Thorium-230	1.2	J	3.3	8.4	pCi/a
DU-3	Thorium-232	0.80	J	2.4	7.4	pCi/a
DU-3	Tin	6.2	J	6.2	779.000	ma/ka
DU-3	Titanium	747	J	996	5.190.000	ma/ka
DU-3	Tungsten	0.080	J	0.37	1.040	ma/ka
DU-3	Uranium (total)	3.6	J	3.6	3.830	ma/ka
DU-3	Uranium-234	1.0	J	3.4	11	pCi/a
DU-3	Uranium-235	0.17	J	0.17	0.35	pCi/a
DU-3	Uranium-238	3.3	J	3.3	1.4	pCi/a
DU-3	Vanadium	39	J	54	6.420	ma/ka
DU-3	Zinc	44	J	45	389.000	ma/ka
DU-3	Antimony	0.16	J-	2.1	519	ma/ka
DU-3	Benzo(g,h,i)perylene	0.038	J-	0.19	25.300	ma/ka
DU-3	Boron	5.9	J-	13	259.000	ma/ka
DU-3	Chlorate	16	J-	107	38,900	ma/ka
DU-3	Chromium (total)	11	J-	27	1.950.000	ma/ka
DU-3	Cobalt	7.7	J-	8.4	385	ma/ka
DU-3	Magnesium	10,600	J-	22.000	5,200,000	ma/ka
DU-3	Manganese	154	J-	1.500	28,100	ma/ka
DU-3	Nickel	17	- J-	17	24,700	ma/ka
DU-3	Nitrite	5.3	J-	77	130 000	ma/ka
DU-3	Perchlorate	1,100	J-	2,620	908	ma/ka
DU-3	Sodium	1,550	J-	4 020	NA	ma/ka
DU-3	Sulfate	15 300	J-	15,300	NA	ma/ka
DU-3	Tungsten	0.37	ر ا-	0.37	1 040	ma/ka
DU-3	Vanadium	39	.J-	54	6 420	ma/ka
DU-3	Zinc	27	,I-	45	389 000	ma/ka
DU-3	4 4'-DDF	0.037	.]+	0.20	9 5	ma/ka
	.,	0.007		0.20	9.0	iiig/ikg

## TABLE B-6. Summary of J Qualified Soil DataNevada Environmental Response Trust SiteHenderson, Nevada

Decision Unit	Analyte	Maximum Concentration of Qualified Data	Qualifier	Maximum Detected Concentration in Soil BHRA Data Set	Screening Level	Unit
DU-3	4,4'-DDT	0.057	J+	0.33	7.5	mg/kg
DU-3	Acetone	0.036	J+	0.052	1,040,000	mg/kg
DU-3	Ammonia	3.7	J+	563	6,140	mg/kg
DU-3	Arsenic	4.5	J+	11	7.2	mg/kg
DU-3	beta-BHC	0.014	J+	0.052	1.7	mg/kg
DU-3	Chlorate	0.086	J+	107	38,900	mg/kg
DU-3	Chloride	1,660	J+	3,140	113,000	mg/kg
DU-3	Manganese	239	J+	1,500	28,100	mg/kg
DU-3	Nitrate	65	J+	515	2,080,000	mg/kg
DU-3	Perchlorate	4.7	J+	2,620	908	mg/kg
DU-3	Radium-226	1.4	J+	2.0	0.023	pCi/g
DU-3	Radium-228	1.9	J+	1.9	0.041	pCi/g
DU-3	Sulfate	254	J+	15,300	NA	mg/kg
DU-3	Thorium-228	1.9	J+	3.0	0.025	pCi/g

#### Notes:

mg/kg = milligram per kilogram

pCi/g = picocurie per gram

BHC = Hexachlorocyclohexane

BHRA = Baseline Health risk assessment

DDD = Dichlorodiphenyldichloroethane

DDE = Dichlorodiphenyldichloroethylene

DDT = Dichlorodiphenyltrichloroethane

DU = Decision unit

J = Estimated value

J- = Estimate value, biased low

J+ = Estimate value, biased high

NA = Not available

TCDD = Tetrachlorodibenzo-p-dioxin

TEQ = Toxicity equivalent

#### APPENDIX C SOIL BHRA DATA SET

> APPENDIX C-1 SOIL BHRA DATA SET - CHEMICALS AND RADIONUCLIDES (CD)

> APPENDIX C-2 SOIL BHRA DATA SET - ASBESTOS

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Total Long Asbestos Protocol Structures Count (s/sample)	Short Amphibole Protocol Structures Count (s/sample)	Short Chrysotile Ptotocol Structures Count (s/sample)	Total Short Asbestos Protocol Structures Count (s/sample)	Total Amphibole Protocol Structures Count (s/sample)	Total Chrysotile Protocol Structures Count (s/sample)	Total Asbestos Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
RISB-09-0.5-20141211	DU-1	Ν	12/11/2014	0.5	0	0	0	0	0	0	0	0	0	8960000	8960000
RISB-09-5.0-20141211	DU-1	Ν	12/11/2014	5	0	0	0	0	0	0	0	0	0	8900000	8900000
RISB-10-0.5-20141215	DU-1	Ν	12/15/2014	0.5	1	0	1	0	0	0	1	0	1	8970000	8970000
RISB-10-5.0-20141215	DU-1	Ν	12/15/2014	5	0	0	0	0	0	0	0	0	0	8860000	8860000
RISB-11-0.5-20141217	DU-1	Ν	12/17/2014	0.5	0	0	0	0	0	0	0	0	0	8900000	8900000
RISB-11-5.0-20141217	DU-1	Ν	12/17/2014	5	0	0	0	0	0	0	0	0	0	8970000	8970000
RISB-12-0.5-20141215	DU-1	Ν	12/15/2014	0.5	1	2	3	0	3	3	1	5	6	8900000	8900000
RISB-12-5.0-20141216	DU-1	Ν	12/16/2014	5	0	0	0	0	0	0	0	0	0	8900000	8900000
RISB-13-0.5-20141217	DU-1	Ν	12/17/2014	0.5	0	0	0	0	0	0	0	0	0	8970000	8970000
RISB-13-5.0-20141218	DU-1	Ν	12/18/2014	5	0	0	0	0	0	0	0	0	0	8940000	8940000
RISB-14-0.5-20141216	DU-1	Ν	12/16/2014	0.5	0	0	0	0	0	0	0	0	0	8970000	8970000
RISB-14-5.0-20141216	DU-1	Ν	12/16/2014	5	1	0	1	0	0	0	1	0	1	8920000	8920000
RSAH3-0.0	DU-1	Ν	6/11/2008	0.5	0	1	1	NA	NA	NA	0	1	1	2998000	2998000
RSAK4-0.0	DU-1	Ν	6/12/2008	0	0	1	1	NA	NA	NA	0	1	1	2991000	2991000
RSAK6-0.0	DU-1	Ν	6/17/2008	0	0	0	0	NA	NA	NA	0	0	0	2976000	2976000
RSAL4-0.0	DU-1	Ν	6/12/2008	0	0	0	0	NA	NA	NA	0	0	0	2999026	2999026
RSAL5-0.0	DU-1	Ν	6/12/2008	0	0	1	1	NA	NA	NA	0	1	1	2966000	2966000
RSAL6-0.0B	DU-1	Ν	9/16/2009	0	0	0	0	0	1	1	0	1	1	8860000	8860000
RSAL7-0.0	DU-1	Ν	6/17/2008	0	0	0	0	NA	NA	NA	0	0	0	2981000	2981000
RSAL8-0.0	DU-1	Ν	6/17/2008	0	0	0	0	NA	NA	NA	0	0	0	2991000	2991000
RSAM2-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2959000	2959000
RSAM3-0.0	DU-1	Ν	6/18/2008	0.5	0	0	0	NA	NA	NA	0	0	0	2966000	2966000
RSAM4-0.0	DU-1	Ν	6/18/2008	0	0	0	0	0	1	1	0	1	1	2969000	2969000
RSAM6-0.0B	DU-1	Ν	7/17/2009	0	0	0	0	0	0	0	0	0	0	8860000	8860000
RSAM7-1.00BPC	DU-1	N	6/16/2010	0.5	0	0	0	< 1	0	< 1	< 1	0	< 1	8224442	8224442
RSAN2-0.0	DU-1	Ν	6/18/2008	1	0	0	0	NA	NA	NA	0	0	0	2985000	2985000
RSAN3-0.0	DU-1	N	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2983000	2983000
RSAN4-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2983000	2983000
RSAN5-0.0B	DU-1	Ν	7/28/2009	0	0	0	0	0	0	0	0	0	0	8850000	8850000
RSAN7-0.0B	DU-1	Ν	8/4/2009	0.5	0	0	0	0	0	0	0	0	0	8850000	8850000
RSAO2-0.0	DU-1	Ν	6/16/2008	0	0	0	0	NA	NA	NA	0	0	0	2974627	2974627
RSAO6-0.33BPC	DU-1	Ν	4/15/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000
RSAO6-0.33BPC-FD	DU-1	FD	4/15/2010	0	0	0	0	0	0	0	0	0	0	8910000	8910000
RSAP5-0.0B	DU-2	Ν	8/7/2009	0.5	0	2	2	0	3	3	0	5	5	8910000	8910000
RSAP7-0.0B	DU-2	Ν	10/27/2009	0	0	0	0	0	0	0	0	0	0	8970000	8970000
RSAQ4-0.0B	DU-2	N	8/7/2009	0.5	0	0	0	0	0	0	0	0	0	8850000	8850000
RSAQ7-0.0B	DU-2	N	10/2/2009	0	0	0	0	0	1	1	0	1	1	8850000	8850000
RSAS5-0.0B	DU-2	N	9/24/2009	0.5	0	< 1	< 1	0	0	0	0	< 1	< 1	8870000	8870000
RSAS8-0.33BPC	DU-2	N	4/8/2010	1.5	0	0	0	0	1	1	0	1	1	8880000	8880000
S2-PG-1-1-0.0	DU-2	N	4/8/2010	0	0	0	0	0	0	0	0	0	0	8940000	8940000
S2-PG-1-1-0.0-FD	DU-2	FD	4/8/2010	0	0	0	0	0	0	0	0	0	0	8910000	8910000

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Total Long Asbestos Protocol Structures Count (s/sample)	Short Amphibole Protocol Structures Count (s/sample)	Short Chrysotile Ptotocol Structures Count (s/sample)	Total Short Asbestos Protocol Structures Count (s/sample)	Total Amphibole Protocol Structures Count (s/sample)	Total Chrysotile Protocol Structures Count (s/sample)	Total Asbestos Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SA05-033BPC	DU-2	Ν	5/13/2010	1	0	0	0	0	0	0	0	0	0	8310000	8310000
SA100-0.0	DU-1	N	6/18/2008	0.5	0	0	0	NA	NA	NA	0	0	0	2969000	2969000
SA103-0.0B	DU-2	Ν	8/7/2009	0	0	2	2	0	5	5	0	7	7	8870000	8870000
SA11-0.33BPC	DU-1	Ν	4/12/2010	0	0	0	0	0	1	1	0	1	1	8860000	8860000
SA121-0.33BPC	DU-2	Ν	4/7/2010	0.5	0	3	3	0	0	0	0	3	3	2980000	2980000
SA122-0.0B	DU-2	N	9/11/2009	3.5	0	1	1	0	0	0	0	1	1	8930000	8930000
SA123-0.0B	DU-1	N	7/13/2009	1	0	1	1	0	< 1	< 1	0	2	2	8850000	8880000
SA126-0.0B	DU-2	Ν	8/5/2009	0.5	0	0	0	0	0	0	0	0	0	8920000	8920000
SA136-0.33BPC	DU-2	Ν	4/8/2010	1	0	0	0	0	0	0	0	0	0	8930000	8930000
SA138-0.0B	DU-2	Ν	8/7/2009	0	0	0	0	0	0	0	0	0	0	8970000	8970000
SA144-0.0B	DU-1	Ν	7/29/2009	0.5	0	0	0	0	0	0	0	0	0	8910000	8910000
SA144009-0.0B	DU-1	FD	7/29/2009	0.5	0	0	0	0	0	0	0	0	0	8910000	8910000
SA145-0.0B	DU-1	Ν	7/23/2009	0	0	0	0	0	1	1	0	1	1	8850000	8850000
SA151-0.0B	DU-1	Ν	7/28/2009	0	0	0	0	0	1	1	0	1	1	8910000	8910000
SA151-0.33BPC	DU-1	Ν	4/12/2010	0.5	0	0	0	0	0	0	0	0	0	8870000	8870000
SA151009-0.0B	DU-1	FD	7/28/2009	0	0	0	0	0	0	0	0	0	0	8910000	8910000
SA152-0.0	DU-1	Ν	6/16/2008	0	0	0	0	0	2	2	0	2	2	2826974	2826974
SA157-0.0B	DU-1	Ν	10/2/2009	0	0	0	0	0	0	0	0	0	0	8860000	8860000
SA158-0.0B	DU-1	Ν	8/7/2009	0	0	0	0	0	0	0	0	0	0	8910000	8910000
SA166-0.0	DU-1	Ν	6/18/2008	0	0	0	0	0	2	2	0	2	2	2969000	2969000
SA170-0.0B	DU-2	Ν	8/11/2009	1.5	0	0	0	0	0	0	0	0	0	8860000	8860000
SA170009-0.0B	DU-2	FD	8/11/2009	1.5	0	0	0	0	0	0	0	0	0	8920000	8920000
SA176-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2991000	2991000
SA18	DU-1	Ν	12/3/2006	0	0	2	2	NA	NA	NA	0	2	2	2995000	5990000
SA185-0.0B	DU-1	Ν	7/31/2009	0.5	0	1	1	0	1	1	0	2	2	8930000	8930000
SA186-0.0B	DU-1	Ν	7/31/2009	0	0	0	0	0	0	0	0	0	0	8920000	8920000
SA189-0.33BPC	DU-1	Ν	5/3/2010	0.5	0	0	0	0	0	0	0	0	0	8960000	8960000
SA197-0.0B	DU-3	Ν	7/13/2009	0	0	0	0	0	0	0	0	0	0	8920000	8920000
SA200-0.0B	DU-1	Ν	7/29/2009	0.5	0	0	0	0	0	0	0	0	0	8850000	8850000
SA21	DU-1	Ν	12/2/2006	0	0	0	0	NA	NA	NA	0	0	0	2935000	2935000
SA211-0.0B	DU-2	Ν	8/7/2009	0.5	0	0	0	0	0	0	0	0	0	8940000	8940000
SA211009-0.0B	DU-2	FD	8/7/2009	0.5	0	0	0	0	0	0	0	0	0	8940000	8940000
SA212-0.0B	DU-2	Ν	8/7/2009	0	0	1	1	0	0	0	0	1	1	8850000	8850000
SA31-0.0B	DU-2	Ν	9/15/2009	0	0	0	0	0	0	0	0	0	0	8910000	8910000
SA47-0.0	DU-1	N	6/19/2008	0.5	0	0	0	NA	NA	NA	0	0	0	2969000	2969000
SA54-0.0B	DU-1	N	7/29/2009	1	0	0	0	0	0	0	0	0	0	8870000	8870000
SA55-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2978000	2978000
SA6	DU-2	N	12/7/2006	0.5	0	0	0	NA	NA	NA	0	0	0	2846000	2846000
SA62-0.0B	DU-1	N	7/17/2009	0	0	0	0	0	0	0	0	0	0	8850000	8850000
SA64-0.0B	DU-3	Ν	7/13/2009	0	0	0	0	0	3	3	0	3	3	8960000	8960000
SA67-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2978000	2978000

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Total Long Asbestos Protocol Structures Count (s/sample)	Short Amphibole Protocol Structures Count (s/sample)	Short Chrysotile Ptotocol Structures Count (s/sample)	Total Short Asbestos Protocol Structures Count (s/sample)	Total Amphibole Protocol Structures Count (s/sample)	Total Chrysotile Protocol Structures Count (s/sample)	Total Asbestos Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SA69-0.0	DU-1	N	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2991000	2991000
SA7	DU-2	N	12/7/2006	0	0	1	1	NA	NA	NA	0	1	1	2988000	2990000
SA70-0.0B	DU-1	N	7/13/2009	0.5	0	0	0	0	0	0	0	0	0	8910000	8910000
SA71-0.0B	DU-1	N	7/17/2009	0	0	0	0	0	1	1	0	1	1	8860000	8860000
SA73-0.0B	DU-1	N	9/16/2009	0	0	0	0	0	0	0	0	0	0	8920000	8920000
SA74-0.0	DU-1	Ν	6/12/2008	0	0	0	0	NA	NA	NA	0	0	0	2969000	2969000
SA75-0.0	DU-1	Ν	6/12/2008	0	0	0	0	NA	NA	NA	0	0	0	2978000	2978000
SA77-0.33BPC	DU-2	Ν	4/20/2010	1	0	0	0	1	0	1	1	0	1	8930000	8930000
SA77-0.33BPC_FD	DU-2	FD	4/20/2010	1	0	0	0	0	0	0	0	0	0	8930000	8930000
SA8	DU-2	N	12/7/2006	0.5	0	2	2	NA	NA	NA	0	2	2	2997000	5990000
SA85-0.0	DU-1	N	6/18/2008	0	0	1	1	0	1	1	0	2	2	2991000	2991000
SA87-0.0	DU-1	Ν	6/18/2008	0	0	0	0	NA	NA	NA	0	0	0	2982000	2982000
SSAK2-01-0.00BPC	DU-1	Ν	4/15/2010	2.5	0	0	0	0	0	0	0	0	0	8900000	8900000
SSAK2-01-0.00BPC_FD	DU-1	FD	4/15/2010	2.5	0	0	0	0	0	0	0	0	0	8900000	8900000
SSAK5-02-0.00BPC	DU-1	Ν	4/21/2010	0	0	0	0	0	0	0	0	0	0	8940000	8940000
SSAK5-03-0.00BPC	DU-1	Ν	5/12/2010	0	0	0	0	0	< 1	< 1	0	< 1	< 1	8870000	8870000
SSAL2-03-0.00BPC	DU-1	Ν	4/28/2010	0	0	3	3	0	1	1	0	4	4	8940000	8940000
SSAL3-03-0.00BPC	DU-1	Ν	5/11/2010	0	0	3	3	0	11	11	0	14	14	7500000	7500000
SSAL4-02-0.00BPC	DU-1	Ν	4/15/2010	0	0	0	0	0	0	0	0	0	0	8960000	8960000
SSAL4-02-0.00BPC_FD	DU-1	FD	4/15/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000
SSAL4-03-0.00BPC	DU-1	Ν	4/15/2010	0	0	0	0	0	0	0	0	0	0	8940000	8940000
SSAL5-01-0.00BPC	DU-1	Ν	5/13/2010	0	0	0	0	0	0	0	0	0	0	8000000	8000000
SSAL5-02-0.00BPC	DU-1	Ν	5/13/2010	0	0	0	0	0	< 1	< 1	0	< 1	< 1	8130000	8130000
SSAL7-03-0.00BPC	DU-1	Ν	4/23/2010	0.5	0	0	0	0	0	0	0	0	0	8870000	8870000
SSAM4-01-0.67BPC	DU-1	Ν	8/3/2010	1	0	0	0	0	0	0	0	0	0	8940000	8940000
SSAM4-03-0.00BPC	DU-1	Ν	8/3/2010	0	0	0	0	0	0	0	0	0	0	8900000	8900000
SSAM4-04-0.00BPC	DU-1	Ν	6/28/2010	0.5	0	0	0	0	0	0	0	0	0	8910000	8910000
SSAM7-01-0.00BPC	DU-1	Ν	4/22/2010	0	0	0	0	0	0	0	0	0	0	8870000	8870000
SSAM7-02-0.33BPC	DU-1	Ν	4/19/2010	0.5	0	0	0	0	0	0	0	0	0	8840000	8840000
SSAM7-08-0.00BPC	DU-1	Ν	8/23/2010	0.5	0	< 1	< 1	0	< 2	< 2	0	3	3	8108643	8108643
SSAN3-01-0.00 BPC	DU-1	Ν	4/9/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000
SSAN3-02-0.00BPC	DU-1	Ν	4/8/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000
SSAN4-01-0.00_1_BPC	DU-1	Ν	10/28/2010	2	0	< 2	< 2	0	0	0	0	< 2	< 2	7516394	7516394
SSAN5-01-0.00BPC	DU-1	Ν	4/12/2010	0	0	0	0	0	0	0	0	0	0	8850000	8850000
SSAN6-05-0.00BPC	DU-1	Ν	4/12/2010	0	0	1	1	0	0	0	0	1	1	8920000	8920000
SSAN7-03-0.00BPC	DU-1	N	5/12/2010	0.5	0	0	0	0	3	3	0	3	3	7650000	7650000
SSAN7-03-0.00BPC_FD	DU-1	FD	5/12/2010	0.5	0	0	0	0	0	0	0	0	0	7380000	7380000
SSAO5-01-0.00BPC	DU-1	Ν	4/20/2010	0.5	0	0	0	0	0	0	0	0	0	8960000	8960000
SSAO5-01-0.00BPC_FD	DU-1	FD	4/20/2010	0.5	0	0	0	0	0	0	0	0	0	8930000	8930000
SSAO6-04-0.00BPC	DU-1	N	4/12/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000
SSA07-01-0.00BPC	DU-1	N	4/12/2010	0	0	0	0	0	0	0	0	0	0	8930000	8930000

Sample ID	DU	Sample Type	Sample Date	Start Depth (ft bgs)	Long Amphibole Protocol Structures Count (s/sample)	Long Chrysotile Protocol Structures Count (s/sample)	Total Long Asbestos Protocol Structures Count (s/sample)	Short Amphibole Protocol Structures Count (s/sample)	Short Chrysotile Ptotocol Structures Count (s/sample)	Total Short Asbestos Protocol Structures Count (s/sample)	Total Amphibole Protocol Structures Count (s/sample)	Total Chrysotile Protocol Structures Count (s/sample)	Total Asbestos Protocol Structures Count (s/sample)	Long Amphibole Analytical Sensitivity (s/g PM <sub>10</sub> )	Long Chrysotile Analytical Sensitivity (s/g PM <sub>10</sub> )
SSAQ3-04-0.00_1_BPC	DU-2	Ν	9/24/2010	0	0	< 2	< 2	0	0	0	0	< 2	< 2	7993975	7993975
SSAQ4-01-0.00BPC	DU-2	Ν	4/7/2010	0	0	3	3	0	3	3	0	6	6	2990000	2990000
SSAQ4-02-0.00 BPC	DU-2	Ν	4/9/2010	0	0	0	0	0	2	2	0	2	2	8910000	8910000
SSAS8-01-0.00BPC	DU-2	Ν	5/19/2010	1	0	0	0	0	0	0	0	0	0	7910000	7910000
SSAS8-02-0.00BPC	DU-2	Ν	8/18/2010	0	0	1	1	1	2	3	1	3	4	8870000	8870000
SSAS8-03-0.33BPC	DU-2	Ν	8/18/2010	0	0	0	0	0	0	0	0	0	0	8870000	8870000
SSAS8-04-0.00BPC	DU-2	Ν	8/16/2010	0	3	0	3	10	0	10	13	0	13	8880000	8850000
SSAS8-04-0.33BPC	DU-2	N	8/16/2010	0.33	0	0	0	0	0	0	0	0	0	8860000	8860000
TSB-GJ-03-0	DU-2	Ν	11/19/2007	0	0	0	0	NA	NA	NA	0	0	0	2987831	2987831
TSB-GR-02-0	DU-2	Ν	11/19/2007	0	0	1	1	0	2	2	0	3	3	2968961	2968961

#### Notes:

bgs = below ground surface ft = feet s/g PM<sub>10</sub> = fiber per gram of particulate matter (< 10 micrometer) s/sample = fiber per sample BHRA = Baseline Health Risk Assessment DU = Decision unit DVSR = Data Validation Summary Report FD = Field Duplicate N = Normal Sample NA = Not Available indicate values reported as "< X" in indicate that analytical sensitivity is

indicate values reported as "< X" in the DVSRs or analytical reports. These data will not be included in the BHRA.

indicate that analytical sensitivity is different between long amphibole and long chrysotile as reported in the DVSRs or analytical reports. The higher analytical sensitivity will be conservatively used in the risk calculation of BHRA.



	TSB-GJ-04 TSB-GJ-03 TSB-GJ-03	
The states		
1		
Legend		
Soil Sample Location for BHRA	Parcel H	Server and
NERT Site Boundary		in the second
Parcel Boundary		3 3
Excavation Control Area (ECA)		and benchmark
Remediation Zone A		- CHINER
operations area	Source: Esti, DigitalGlobe, GeoEye, Earthstar Geographics, CNES/Airbus DS, USDA, US	SGS, AEX,
RAMBOLL ENVIRON	Soil Sampling Locations Included in the BHRA Nevada Environmental Response Trust Site, Henderson, Nevada	Figure C-1
2200 Powell St., Suite 700, Emeryville, CA 94608	Drafter: RS Date: 10/20/2017 Contract Number: 21-41400C-M06A Approved by: Re	vised:

> APPENDIX D SOIL DATA SUMMARY STATISTICS

## TABLE D-1. Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics, and Radionuclides in Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

						Nonde	etects				Detects			
Group	Analyte	Unit	No. of Samples	No. of Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Chlorine	Chlorate	mg/kg	254	198	78	0.044	5.8	0.045	20,900	3.0	204	1,610	7.9	SA106
Oxyanions	Perchlorate	mg/kg	337	321	95	0.035	0.43	0.012	2,620	8.4	100	288	2.9	RSAM5
Metals	Aluminum	mg/kg	260	260	100			3,900	12,200	9,020	8,930	1,420	0.16	SA43
	Antimony	mg/kg	257	77	30	0.50	2.3	0.11	2.4	0.32	0.70	0.70	1.0	SA114
	Arsenic	mg/kg	571	571	100			0.58	34	3.3	3.7	2.2	0.58	EE-C25-1
	Barium	mg/kg	260	260	100			65	1,780	177	190	110	0.58	SA123
	Beryllium	mg/kg	195	195	100			0.22	0.71	0.46	0.46	0.074	0.16	SA86
	Boron	mg/kg	260	236	91	1.4	13	2.5	1,510	7.3	24	135	5.6	SA62
	Cadmium	mg/kg	260	139	53	0.0050	0.51	0.040	8.9	0.13	0.26	0.77	2.9	SA103
	Calcium	mg/kg	195	195	100			9,930	62,500	26,700	27,500	9,280	0.34	RSAM2
	Chromium (total)	mg/kg	262	262	100			3.5	102	9.4	12	11	0.93	SA106
	Chromium VI	mg/kg	215	48	22	0.11	0.49	0.11	106	0.79	5.8	18	3.1	SA106
	Cobalt	mg/kg	310	310	100			3.2	284	7.6	13	29	2.3	RSAO8
	Copper	mg/kg	260	260	100			8.0	160	18	20	12	0.61	RISB-12
	Iron	mg/kg	260	260	100			7,050	24,000	15,000	14,900	2,690	0.18	RISB-12
L	Lead	mg/kg	321	321	100			3.6	267	8.9	13	19	1.5	SA92
	Lithium	mg/kg	7	7	100			10	14	11	12	1.5	0.13	TSB-GR-02
	Magnesium	mg/kg	309	309	100			5,300	71,000	9,700	10,700	5,070	0.48	DS-C39B-1
	Manganese	mg/kg	410	410	100			133	29,200	390	962	2,220	2.3	CS-C44-1
	Mercury	mg/kg	262	230	88	0.0067	0.040	0.0030	1.9	0.017	0.035	0.13	3.7	M-162D
	Molybdenum	mg/kg	260	194	75	0.052	2.0	0.15	55	0.48	1.1	4.9	4.5	RISB-12
	Nickel	mg/kg	260	260	100			6.6	164	15	16	11	0.66	RSAO8
	Niobium	mg/kg	19	1	5.3	0.76	2.1	9.2	9.2	9.2	9.2			TSB-GR-02
	Palladium	mg/kg	25	7	28	0.048	0.060	0.33	0.52	0.44	0.43	0.072	0.17	TSB-GJ-03
	Phosphorus (total)	mg/kg	187	187	100			456	1,600	853	861	188	0.22	RISB-09
	Platinum	mg/kg	195	143	73	0.010	0.24	0.0050	0.16	0.011	0.015	0.017	1.1	SA64
	Potassium	mg/kg	195	195	100			1,230	6,120	2,160	2,320	649	0.28	SA141
	Selenium	mg/kg	260	16	6.2	0.16	4.7	0.70	1.5	1.0	1.0	0.24	0.24	RSAJ3
	Silicon	mg/kg	25	25	100			41	250	140	132	53	0.40	RISB-10
	Silver	mg/kg	260	49	19	0.20	1.5	0.020	7.6	0.12	0.30	1.1	3.6	SA201
	Sodium	mg/kg	195	195	100			198	11,700	756	1,160	1,290	1.1	SA106
	Strontium	mg/kg	213	213	100			73	805	186	208	98	0.47	SA15
	Sulfur	mg/kg	25	16	64	211	430	498	14,000	798	1,680	3,300	2.0	RISB-14
	Thallium	mg/kg	260	176	68	0.10	0.28	0.054	8.4	0.10	0.19	0.64	3.3	SA180
l f	Tin	mg/kg	195	184	94	9.4	12	0.40	12	4.2	3.8	1.6	0.43	RSAK8
	Titanium	mg/kg	195	195	100			361	1,270	751	743	179	0.24	SA166
	Tungsten	mg/kg	213	173	81	0.10	5.6	0.080	8.5	0.23	0.42	0.80	1.9	RSAK8
	Uranium (total)	mg/kg	213	213	100			0.55	3.6	1.0	1.2	0.52	0.45	SA86
	Vanadium	mg/kg	195	195	100			22	78	42	42	9.0	0.21	RSAK8

#### TABLE D-1. Summary Statistics for Chlorine Oxyanions, Metals, Other Inorganics, and Radionuclides in Soil (0-10 ft bgs)

#### Nevada Environmental Response Trust Site

Henderson, Nevada

Chomical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Metals	Zinc	mg/kg	260	260	100			18	300	33	37	25	0.69	RISB-37
	Zirconium	mg/kg	25	25	100			15	31	22	22	3.9	0.18	RISB-14
Other Inorganics	Ammonia	mg/kg	199	36	18	0.080	6.5	0.11	563	1.6	39	120	3.0	RSAM5
	Bromide	mg/kg	209	24	11	0.063	28	0.20	83	1.1	4.7	17	3.6	SA15
	Chloride	mg/kg	206	196	95	2.1	85	0.90	6,670	75	402	840	2.1	RSAJ2
	Chlorite	mg/kg	7	0	0	0.040	0.040							
	Cyanide (total)	mg/kg	132	2	1.5	0.13	1.2	0.48	1.3	0.89	0.89	0.58	0.65	RSAJ2
	Fluoride	mg/kg	7	1	14	0.25	0.25	0.52	0.52	0.52	0.52			TSB-GR-02
	Nitrate	mg/kg	210	187	89	0.048	6.1	0.27	515	5.6	20	46	2.4	SA15
	Nitrate/Nitrite	mg/kg	18	17	94	1.2	1.2	2.7	37	11	11	8.1	0.73	RISB-10
	Nitrite	mg/kg	202	40	20	0.080	22	0.090	77	0.47	3.3	12	3.7	SA64
	ortho-Phosphate	mg/kg	48	6	13	1.1	57	2.4	2,900	6.5	490	1,180	2.4	SA11
	Sulfate	mg/kg	210	208	99	2.1	22	6.7	15,300	187	839	2,350	2.8	SA65
Radionuclides	Radium-226	pCi/g	208	208	100			0.20	2.5	0.92	0.95	0.36	0.38	SA92
	Radium-228	pCi/g	208	208	100			0	3.3	1.2	1.3	0.49	0.39	SA70
	Thorium-228	pCi/g	205	205	100			0.48	3.0	1.8	1.8	0.36	0.20	SA65
	Thorium-230	pCi/g	205	205	100			0.43	4.3	1.1	1.2	0.44	0.37	SA74
	Thorium-232	pCi/g	205	205	100			0.54	2.5	1.6	1.6	0.33	0.21	SA189
ī	Uranium-234	pCi/g	187	187	100			0.27	3.4	1.0	1.1	0.44	0.39	SA128
	Uranium-235	pCi/g	187	187	100			-0.029	0.25	0.052	0.062	0.043	0.70	RSAK6
	Uranium-238	pCi/g	205	205	100			0.24	3.3	0.96	1.0	0.38	0.37	SA128

#### Notes:

-- = No value

bgs = below ground surface

ft = feet

mg/kg = milligram per kilogram

pCi/g = picocurie per gram

#### Nevada Environmental Response Trust Site

Chemical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Dioxin/Furans	2,3,7,8-TCDD TEQ*	mg/kg	466	463	99	0.000044	0.0037	0.000000055	0.019	0.000018	0.00030	0.0012	3.8	RISB-50
Other Organics	Benzenesulfonic acid	mg/kg	27	0	0	0.50	0.50							
	4-Chlorobenzenesulfonic acid	mg/kg	27	0	0	0.50	0.50			-				
	o,o-Dimethyl Phosphorodithioate	mg/kg	27	0	0	2.5	2.5			-				
	Diethylphosphorodithioate	mg/kg	27	0	0	0.50	0.50							
	Phthalic acid	mg/kg	52	0	0	0.25	70							
PAHs	Acenaphthene	mg/kg	474	7	1.5	0.00016	1.1	0.00062	0.70	0.026	0.12	0.26	2.2	EE-C25-1
	Acenaphthylene	mg/kg	474	6	1.3	0.00017	1.8	0.00066	0.22	0.0014	0.038	0.089	2.4	EE-C25-1
	Anthracene	mg/kg	474	14	3.0	0.00072	1.8	0.00055	0.30	0.0035	0.028	0.079	2.8	EE-C25-1
	BaPEq*	mg/kg	474	83	18	0.00090	2.4	0.0069	0.40	0.016	0.047	0.075	1.6	SSAM5-03
	Benzo(g,h,i)perylene	mg/kg	473	53	11	0.0011	1.7	0.0014	0.38	0.014	0.036	0.062	1.7	SSAO6-04
	Fluoranthene	mg/kg	474	68	14	0.0010	3.9	0.0017	0.35	0.010	0.038	0.063	1.7	SA180
	Fluorene	mg/kg	474	3	0.60	0.00047	1.9	0.00052	1.1	0.0010	0.37	0.63	1.7	EE-C25-1
	1-Methylnaphthalene	mg/kg	26	4	15	0.00026	8.3	0.00062	5.6	0.0022	1.4	2.8	2.0	EE-C25-1
	2-Methylnaphthalene	mg/kg	474	10	2.1	0.00031	3.9	0.00077	7.9	0.0092	0.81	2.5	3.1	EE-C25-1
	Naphthalene	mg/kg	536	23	4.3	0.00032	3.3	0.0010	3.1	0.0014	0.23	0.75	3.2	EE-C25-1
	Phenanthrene	mg/kg	474	64	14	0.0011	1.8	0.0017	1.5	0.0055	0.057	0.22	3.9	EE-C25-1
	Pyrene	mg/kg	474	90	19	0.0011	1.3	0.0011	1.3	0.011	0.054	0.15	2.8	EE-C25-1
PCBs	Aroclor-1016	mg/kg	48	0	0	0.034	0.37							
	Aroclor-1221	mg/kg	48	0	0	0.034	0.74							
	Aroclor-1232	mg/kg	48	0	0	0.034	0.37							
	Aroclor-1242	mg/kg	48	0	0	0.034	0.37							
	Aroclor-1248	mg/kg	48	1	2.1	0.034	0.37	0.091	0.091	0.091	0.091			RSAS5
	Aroclor-1254	mg/kg	48	0	0	0.034	0.37							
	Aroclor-1260	mg/kg	66	2	3.0	0.017	0.37	0.034	0.061	0.048	0.048	0.019	0.40	RISB-12
Pesticides - OCPs	Aldrin	mg/kg	281	2	0.70	0.000088	0.092	0.00049	0.00052	0.00051	0.00051	0.000021	0.042	SSAL2-05
	alpha-BHC	mg/kg	281	15	5.3	0.000096	0.092	0.00024	0.012	0.00059	0.0019	0.0032	1.7	RSAQ4
	beta-BHC	mg/kg	281	161	57	0.00035	0.15	0.00072	0.87	0.011	0.044	0.10	2.3	SA67
	delta-BHC	mg/kg	281	7	2.5	0.000083	0.092	0.00048	0.0015	0.00059	0.00079	0.00038	0.48	SA86
	gamma-BHC	mg/kg	281	3	1.1	0.000083	0.11	0.00083	0.0019	0.0013	0.0013	0.00054	0.40	RSAQ4
	alpha-Chlordane	mg/kg	281	0	0	0.00010	0.092							
	gamma-Chlordane	mg/kg	279	1	0.40	0.000086	0.092	0.0014	0.0014	0.0014	0.0014			RSAO7
	trans/gamma-Chlordane	mg/kg	1	0	0	0.00055	0.00055							
	Chlordane (total)	mg/kg	247	1	0.40	0.00021	0.45	0.0030	0.0030	0.0030	0.0030			SA66
	2,4'-DDD	mg/kg	7	0	0	0.00011	0.00011							
	4,4'-DDD	mg/kg	280	10	3.6	0.00016	0.18	0.0014	0.032	0.0046	0.0078	0.0093	1.2	SSAL3-04
	2,4'-DDE	mg/kg	32	9	28	0.000089	0.015	0.0018	0.088	0.0060	0.027	0.031	1.1	RISB-52
	4,4'-DDE	mg/kg	281	153	54	0.00024	0.18	0.00040	6.0	0.015	0.27	0.83	3.1	SSAM3-01

#### Nevada Environmental Response Trust Site

Chamical			No of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Pesticides - OCPs	4,4'-DDT	mg/kg	281	122	43	0.00043	0.18	0.00066	2.3	0.013	0.10	0.27	2.6	SSAM2-01
	Dieldrin	mg/kg	281	4	1.4	0.000073	0.18	0.00027	0.059	0.016	0.023	0.028	1.2	SSAM2-01
	Endosulfan I	mg/kg	281	2	0.70	0.000083	0.092	0.00024	0.0015	0.00087	0.00087	0.00089	1.0	SSAL3-01
	Endosulfan II	mg/kg	281	0	0	0.00015	0.18							
	Endosulfan sulfate	mg/kg	281	2	0.70	0.00012	0.18	0.0042	0.016	0.010	0.010	0.0083	0.83	BDT-4-S-15
	Endrin	mg/kg	281	2	0.70	0.000083	0.18	0.00070	0.0054	0.0031	0.0031	0.0033	1.1	SA180
	Endrin aldehyde	mg/kg	281	0	0	0.00011	0.18							
	Endrin ketone	mg/kg	281	10	3.6	0.00038	0.18	0.00061	0.020	0.0012	0.0035	0.0059	1.7	SA86
	Heptachlor	mg/kg	279	0	0	0.00021	0.092							
	Heptachlor epoxide	mg/kg	280	0	0	0.00012	0.098							
	Hexachlorobenzene	mg/kg	674	385	57	0.00028	10	0.00032	4.7	0.053	0.19	0.37	2.0	SSAK3-05
	Methoxychlor	mg/kg	281	16	5.7	0.00043	0.92	0.00050	0.38	0.0021	0.064	0.12	1.9	SSAM2-01
	Toxaphene	mg/kg	281	1	0.40	0.0071	3.6	0.62	0.62	0.62	0.62			SSAL3-04
	2,4,5-TP	mg/kg	1	0	0	0.021	0.021							
Pesticides - OPPs	Atrazine	mg/kg	16	0	0	0.012	0.15							
	Chlorpyrifos	mg/kg	57	0	0	0.0062	0.082							
	Coumaphos	mg/kg	57	0	0	0.0027	0.036							
	Dasanit	mg/kg	57	0	0	0.0078	0.10							
	Demeton (O + S)	mg/kg	16	0	0	0.0072	0.096							
	Demeton-O	mg/kg	57	0	0	0.0051	0.067							
	Demeton-S	mg/kg	57	0	0	0.0047	0.062							
	Diazinon	mg/kg	57	0	0	0.0070	0.093							
	Dibrom	mg/kg	57	0	0	0.022	0.29							
	Dichlorovos	mg/kg	57	0	0	0.0071	0.094							
	Dimethoate	mg/kg	57	3	5.3	0.0068	0.090	0.011	0.013	0.012	0.012	0.0010	0.083	SA05
	Disulfoton	mg/kg	57	0	0	0.0074	0.098							
	Ethoprop	mg/kg	57	0	0	0.0047	0.063							
	Ethyl p-nitrophenyl benzenethiophosphate	mg/kg	57	0	0	0.0035	0.047							
	Famphur	mg/kg	57	0	0	0.0031	0.041							
	Fenthion	mg/kg	57	0	0	0.0084	0.11							
	Guthion	mg/kg	57	0	0	0.0034	0.045							
	Malathion	mg/kg	57	0	0	0.0045	0.059							
	Merphos	mg/kg	57	0	0	0.0049	0.065							
	Methyl parathion	mg/kg	57	0	0	0.0061	0.081							
	Mevinphos	mg/kg	57	0	0	0.0044	0.059							
	Parathion	mg/kg	57	0	0	0.0051	0.067							
	Phorate	mg/kg	57	0	0	0.0055	0.073							
	Prothiophos	mg/kg	57	0	0	0.0038	0.050							

#### Nevada Environmental Response Trust Site

Chemical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
Pesticides - OPPs	Ronnel	mg/kg	57	0	0	0.015	0.19							
	Simazine	mg/kg	16	0	0	0.021	0.28							
	Stirophos	mg/kg	57	1	1.8	0.0042	0.055	0.041	0.041	0.041	0.041			SA166
	Sulfotepp	mg/kg	57	0	0	0.0060	0.080							
	Sulprofos	mg/kg	57	0	0	0.0041	0.054							
	Thionazin	mg/kg	57	0	0	0.0054	0.071							
	o-Ethyl o-2,4,5-trichlorophenyl ethyl-phosphonothioate	mg/kg	57	0	0	0.006	0.080							
SVOCs	Acetophenone	mg/kg	7	0	0	0.033	0.033				-			
	Aniline	mg/kg	25	0	0	0.033	4.7							
	Azobenzene	mg/kg	7	0	0	0.033	0.033							
	Benzenethiol	mg/kg	7	0	0	0.12	0.12							
	Benzidine	mg/kg	14	0	0	0.67	37							
	Benzoic acid	mg/kg	24	0	0	0.033	19							
	Benzyl alcohol	mg/kg	25	0	0	0.033	8.3							
k k	bis(2-Chloro-1-methylethyl) ether	mg/kg	7	0	0	0.033	0.033							
	bis(2-Chloroethoxy)methane	mg/kg	25	0	0	0.033	7.4							
	bis(2-Chloroethyl) ether	mg/kg	25	0	0	0.033	3.9							
	bis(2-Ethylhexyl)phthalate	mg/kg	469	97	21	0.033	5.0	0.058	61	0.097	0.79	6.2	7.8	SSAP4-01
	bis(4-Chlorophenyl) disulfide	mg/kg	7	0	0	0.20	0.20							
	bis(4-Chlorophenyl) sulfone	mg/kg	7	0	0	0.33	0.33							
	4-Bromophenyl-phenyl ether	mg/kg	25	0	0	0.033	4.2							
	Butylbenzylphthalate	mg/kg	469	7	1.5	0.0025	4.6	0.0033	0.053	0.0056	0.017	0.018	1.1	RSAL2
	Carbazole	mg/kg	7	0	0	0.033	0.033							
	4-Chloro-3-methylphenol	mg/kg	25	0	0	0.033	3.9							
	4-Chloroaniline	mg/kg	25	0	0	0.033	7.4							
	2-Chloronaphthalene	mg/kg	25	0	0	0.033	3.7							
	2-Chlorophenol	mg/kg	25	0	0	0.033	3.9							
	4-Chlorophenyl-phenyl ether	mg/kg	25	0	0	0.033	4.7							
	4-Chlorothioanisole	mg/kg	7	0	0	0.0076	0.0076							
	4-Chlorothiophenol	mg/kg	7	0	0	0.19	0.19							
	Di-n-butylphthalate	mg/kg	469	33	7.0	0.027	5.0	0.035	7.5	0.076	0.59	1.4	2.4	SSAP4-01
	Di-n-octylphthalate	mg/kg	469	2	0.40	0.0012	5.0	0.084	0.088	0.086	0.086	0.0028	0.033	SSAO4-01
	Dibenzofuran	mg/kg	25	0	0	0.033	3.7							
	3,3'-Dichlorobenzidine	mg/kg	25	0	0	0.033	8.3							
	2,2'-/4,4'-Dichlorobenzil	mg/kg	7	0	0	0.070	0.70							
	2,4-Dichlorophenol	mg/kg	25	0	0	0.033	3.7							
	Diethylphthalate	mg/kg	469	5	1.1	0.024	5.3	0.042	0.35	0.062	0.11	0.13	1.2	SA86

#### Nevada Environmental Response Trust Site

Chemical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
SVOCs	2,4-Dimethylphenol	mg/kg	25	0	0	0.033	7.2							
	Dimethylphthalate	mg/kg	469	54	12	0.00093	3.7	0.0015	0.79	0.042	0.12	0.17	1.4	BDT-1-S-10
	2,4-Dinitrophenol	mg/kg	25	0	0	0.33	18							
	2,4-Dinitrotoluene	mg/kg	25	0	0	0.033	4.5							
	2,6-Dinitrotoluene	mg/kg	25	0	0	0.033	5.3							
	1,4-Dioxane	mg/kg	449	0	0	0.0052	7.1							
	Diphenyl disulfide	mg/kg	7	0	0	0.029	0.029							
	Diphenyl sulfide	mg/kg	7	0	0	0.0035	0.0035							
	Diphenyl sulfone	mg/kg	7	0	0	0.0067	0.0067							
	1,2-Diphenylhydrazine	mg/kg	7	0	0	0.033	0.033							
	Hexachlorobutadiene	mg/kg	256	5	2.0	0.00028	0.033	0.00041	0.0045	0.00095	0.0017	0.0017	1.0	SA11
	Hexachlorocyclopentadiene	mg/kg	25	0	0	0.13	7.4							
	Hexachloroethane	mg/kg	25	0	0	0.033	7.4							
	Hydroxymethyl phthalimide	mg/kg	7	0	0	0.043	0.043							
	Isophorone	mg/kg	25	0	0	0.033	3.7							
	2-Methylphenol	mg/kg	25	0	0	0.08	4.5							
	3&4-Methylphenol	mg/kg	25	0	0	0.067	7.4							
	2-Nitroaniline	mg/kg	25	0	0	0.033	3.7							
	3-Nitroaniline	mg/kg	25	0	0	0.033	7.4							
	4-Nitroaniline	mg/kg	25	0	0	0.13	7.4							
	Nitrobenzene	mg/kg	469	0	0	0.0019	3.9							
	2-Nitrophenol	mg/kg	25	0	0	0.033	7.4							
	4-Nitrophenol	mg/kg	25	0	0	0.14	7.8							
	n-Nitroso-di-n-propylamine	mg/kg	25	0	0	0.033	3.9							
	n-Nitrosodiphenylamine	mg/kg	25	0	0	0.033	4.5							
	Octachlorostyrene	mg/kg	467	78	17	0.0037	130	0.0021	2.1	0.067	0.11	0.25	2.2	SSAK3-05
	Pentachlorobenzene	mg/kg	7	0	0	0.033	0.033							
	Pentachlorophenol	mg/kg	25	0	0	0.33	19							
	Phenol	mg/kg	25	0	0	0.033	5.0							
	Pyridine	mg/kg	469	0	0	0.028	14							
	1,2,4,5-Tetrachlorobenzene	mg/kg	7	0	0	0.033	0.033							
	2,4,5-Trichlorophenol	mg/kg	25	0	0	0.033	7.2							
	2,4,6-Trichlorophenol	mg/kg	25	0	0	0.033	4.2							
VOCs	Acetone	mg/kg	256	102	40	0.0017	0.041	0.0027	0.15	0.012	0.022	0.027	1.2	SA106
	Acetonitrile	mg/kg	7	0	0	0.0020	0.0020							
	t-Amyl methyl ether	mg/kg	249	0	0	0.00011	0.010							
	Benzene	mg/kg	256	0	0	0.00017	0.010							
	Bromobenzene	mg/kg	256	0	0	0.00023	0.010							

#### Nevada Environmental Response Trust Site

Chemical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
VOCs	Bromochloromethane	mg/kg	256	0	0	0.00015	0.010							
	Bromodichloromethane	mg/kg	256	2	0.80	0.00011	0.010	0.00040	0.00069	0.00055	0.00055	0.00021	0.38	SSAO8-10
	Bromoform	mg/kg	256	1	0.40	0.00012	0.010	0.0017	0.0017	0.0017	0.0017			SA102
	Bromomethane	mg/kg	256	0	0	0.00025	0.013							
	2-Butanone	mg/kg	256	75	29	0.00073	0.020	0.00065	0.027	0.0016	0.0034	0.0056	1.7	SSAO7-06
	tert-Butyl alcohol	mg/kg	249	1	0.40	0.0052	0.20	0.0076	0.0076	0.0076	0.0076			RSAM4
	n-Butylbenzene	mg/kg	256	0	0	0.00028	0.010							
	sec-Butylbenzene	mg/kg	256	0	0	0.00025	0.010							
	tert-Butylbenzene	mg/kg	256	0	0	0.00025	0.010							
	Carbon disulfide	mg/kg	7	0	0	0.00055	0.00055							
	Carbon tetrachloride	mg/kg	256	1	0.40	0.00032	0.010	0.00063	0.00063	0.00063	0.00063			RSAN3
	Chlorobenzene	mg/kg	256	6	2.3	0.00012	0.010	0.00064	0.0019	0.0011	0.0011	0.00044	0.39	SA15
	Chloroethane	mg/kg	256	0	0	0.00035	0.010							
	Chloroform	mg/kg	256	85	33	0.00014	0.0091	0.00031	0.15	0.0014	0.0064	0.022	3.4	SA11
	Chloromethane	mg/kg	256	0	0	0.00039	0.010							
	2-Chlorotoluene	mg/kg	256	0	0	0.00026	0.010							
	4-Chlorotoluene	mg/kg	256	0	0	0.00039	0.010							
4	Cumene	mg/kg	256	0	0	0.00018	0.010							-
	p-Cymene	mg/kg	256	1	0.40	0.00024	0.010	0.00055	0.00055	0.00055	0.00055			SSAN8-04
	1,2-Dibromo-3-chloropropane	mg/kg	256	0	0	0.00030	0.010							
	Dibromochloromethane	mg/kg	256	0	0	0.00029	0.010							
	1,2-Dibromoethane	mg/kg	249	0	0	0.00026	0.010							
	Dibromomethane	mg/kg	256	0	0	0.00032	0.010							
	1,2-Dichlorobenzene	mg/kg	256	5	2.0	0.00015	0.010	0.00026	0.00039	0.00038	0.00036	0.000055	0.15	RSAM4
	1,3-Dichlorobenzene	mg/kg	256	0	0	0.00013	0.010							
	1,4-Dichlorobenzene	mg/kg	256	4	1.6	0.00011	0.010	0.00056	0.016	0.0012	0.0047	0.0075	1.6	SA08
	Dichlorodifluoromethane	mg/kg	256	0	0	0.00026	0.010							
	1,1-Dichloroethane	mg/kg	256	1	0.40	0.00011	0.010	0.0030	0.0030	0.0030	0.0030			SA08
	1,2-Dichloroethane	mg/kg	256	0	0	0.00034	0.010							
	1,1-Dichloroethene	mg/kg	256	4	1.6	0.00030	0.010	0.00055	0.0012	0.00081	0.00084	0.00027	0.32	SSAN8-04
	1,2-Dichloroethene	mg/kg	7	0	0	0.00054	0.00054							
	cis-1,2-Dichloroethene	mg/kg	256	1	0.40	0.00028	0.010	0.0041	0.0041	0.0041	0.0041			RISB-57
	trans-1,2-Dichloroethene	mg/kg	256	0	0	0.00020	0.010							
	1,2-Dichloropropane	mg/kg	256	0	0	0.00028	0.010							
	1,3-Dichloropropane	mg/kg	256	0	0	0.00018	0.010							
	2,2-Dichloropropane	mg/kg	256	0	0	0.00017	0.010							
	1,1-Dichloropropene	mg/kg	256	0	0	0.00027	0.010							
	cis-1,3-Dichloropropene	mg/kg	256	0	0	0.00033	0.010							

#### Nevada Environmental Response Trust Site

Chemical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	No. of Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
VOCs	trans-1,3-Dichloropropene	mg/kg	256	0	0	0.00020	0.010							
	Diisopropyl ether	mg/kg	249	0	0	0.00023	0.010							
	Dimethyl disulfide	mg/kg	7	0	0	0.00021	0.00021							
	2,2-Dimethylpentane	mg/kg	7	0	0	0.00028	0.00028			-				
	2,3-Dimethylpentane	mg/kg	7	0	0	0.00022	0.00022			-				
	2,4-Dimethylpentane	mg/kg	7	0	0	0.00019	0.00019							
	3,3-Dimethylpentane	mg/kg	7	0	0	0.00020	0.00020							
	Ethanol	mg/kg	13	0	0	0.19	60							
	Ethyl benzene	mg/kg	228	0	0	0.00019	0.010			-				
	Ethyl tert-butyl ether	mg/kg	249	1	0.40	0.00023	0.010	0.00038	0.00038	0.00038	0.00038			RSAO2
	Ethylene glycol	mg/kg	6	0	0	53	60							
	3-Ethylpentane	mg/kg	7	0	0	0.00021	0.00021			-				
	Formaldehyde	mg/kg	3	0	0	0.21	0.22			-				
	n-Heptane	mg/kg	7	0	0	0.00016	0.00016							
	2-Hexanone	mg/kg	256	0	0	0.00028	0.020							
	lodomethane	mg/kg	7	0	0	0.00026	0.00026							
	Methanol	mg/kg	6	0	0	53	60							
	Methyl tert-butyl ether	mg/kg	256	0	0	0.00017	0.010							
	4-Methyl-2-pentanone	mg/kg	256	0	0	0.00063	0.020							
	Methylene Chloride	mg/kg	256	68	27	0.00037	0.010	0.00034	0.0082	0.0013	0.0016	0.0012	0.75	RSAM8
	2-Methylhexane	mg/kg	7	0	0	0.00020	0.00020							
	3-Methylhexane	mg/kg	7	0	0	0.00014	0.00014							
	2-Nitropropane	mg/kg	7	0	0	0.0017	0.0017							
	n-Nonyl aldehyde	mg/kg	7	0	0	0.00088	0.00088							
	n-Propylbenzene	mg/kg	256	0	0	0.00029	0.010							
	Styrene	mg/kg	256	1	0.40	0.00032	0.010	0.00028	0.00028	0.00028	0.00028			SA55
	1,1,1,2-Tetrachloroethane	mg/kg	256	0	0	0.00022	0.010							
	1,1,2,2-Tetrachloroethane	mg/kg	256	0	0	0.00014	0.010							
	Tetrachloroethene	mg/kg	256	1	0.40	0.00027	0.010	0.00068	0.00068	0.00068	0.00068			SA128
	Toluene	mg/kg	256	53	21	0.00013	0.010	0.00023	0.0022	0.00075	0.00084	0.00040	0.48	RSAM3
	1,1,2-Trichloro-1,2,2- trifluoroethane	mg/kg	11	0	0	0.00054	0.0055							
	1,2,3-Trichlorobenzene	mg/kg	256	2	0.80	0.00038	0.010	0.00081	0.0013	0.0011	0.0011	0.00035	0.33	SA11
	1,2,4-Trichlorobenzene	mg/kg	256	6	2.3	0.00037	0.010	0.00065	0.0037	0.0012	0.0017	0.0012	0.71	SA11
	1,3,5-Trichlorobenzene	mg/kg	7	0	0	0.00068	0.00068							
	1,1,1-Trichloroethane	mg/kg	256	1	0.40	0.00015	0.010	0.00095	0.00095	0.00095	0.00095			SA08
	1,1,2-Trichloroethane	mg/kg	256	0	0	0.00028	0.010							
	Trichloroethene	mg/kg	256	2	0.80	0.00012	0.010	0.00042	0.0021	0.0013	0.0013	0.0012	0.94	RISB-57
	Trichlorofluoromethane	mg/kg	256	5	2.0	0.00032	0.010	0.00035	0.0017	0.0016	0.0013	0.00057	0.44	SA136

Nevada Environmental Response Trust Site

#### Henderson, Nevada

Chamical			No. of	No. of		Nonde	etects				Detects			
Group	Analyte	Unit	Samples	Detects	% Detects	Minimum	Maximum	Minimum	Maximum	Median	Mean	Standard Deviation	Coefficient of Variation	Location of Maximum
VOCs	1,2,3-Trichloropropane	mg/kg	256	0	0	0.00041	0.010							
	1,2,4-Trimethylbenzene	mg/kg	256	6	2.3	0.00022	0.010	0.00042	0.0014	0.00084	0.00085	0.00033	0.39	SSAO8-11
	1,3,5-Trimethylbenzene	mg/kg	256	1	0.40	0.00021	0.010	0.00050	0.00050	0.00050	0.00050			SSAO8-11
	2,2,3-Trimethylbutane	mg/kg	7	0	0	0.00021	0.00021							
	Vinyl acetate	mg/kg	7	0	0	0.00018	0.00018							
	Vinyl chloride	mg/kg	256	1	0.40	0.00024	0.010	0.00028	0.00028	0.00028	0.00028			RSAM7
	m,p-Xylene	mg/kg	233	7	3.0	0.00053	0.010	0.00064	0.0023	0.0012	0.0013	0.00053	0.41	RISB-54
	o-Xylene	mg/kg	233	3	1.3	0.00031	0.010	0.00046	0.00074	0.00057	0.00059	0.00014	0.24	RISB-54
	Xylenes (total)	mg/kg	30	0	0	0.00086	0.013							

#### Notes:

-- = No value

bgs = below ground surface

ft = feet

mg/kg = milligram per kilogram

BaPEq = Benzo(a)pyrene equivalent

BHC = Hexachlorocyclohexane

DDD = Dichlorodiphenyldichloroethane

DDE = Dichlorodiphenyldichloroethylene

DDT = Dichlorodiphenyltrichloroethane

OCP = Organochlorine pesticide

OPP = Organophosphorus pesticide

PAH = Polycyclic aromatic hydrocarbon

PCB = Polychlorinated biphenyl

SVOC = Semivolatile organic compound

TCDD = Tetrachlorodibenzo-p-dioxin

TEQ = Toxicity equivalent

TP = Trichlorophenoxy

VOC = Volatile organic compound

\* Methodology for equivalent calculations explained in text

> APPENDIX E RZ-A AND BRC/TIMET SOIL BACKGROUND DATA SETS (CD)



> APPENDIX F BACKGROUND EVALUATION FOR METALS AND RADIONUCLIDES

## TABLE F-1. Summary Statistics for Metals in RZ-A Background Soils and Study Area Soils (0-10 feet bgs)Nevada Environmental Response Trust SiteHenderson, Nevada

		No. of	No. of		Non-Dete	cts (mg/kg)			Detects	(mg/kg)		Shapiro	-Wilk Test
Chemical Name	Location	Samples	NO. 01 Detects	% Detects	Minimun	Maximum	Minimum	Modian	Moan	Maximum	Standard	Normal	Lognormal
		oumpies	Deletio		winning	Waximum	Willing	Weulan	Weall	Waximum	Deviation	(p-value)	(p-value)
Aluminum	Background	31	31	100%	NA	NA	7,340	8,970	9,020	11,400	890	0.6	0.9
	Study Area	260	260	100%	NA	NA	3,900	9,020	8,930	12,200	1,420	0.03	<0.001
Antimony	Background	31	3	9.7%	2.0	2.2	0.60	0.90	1.6	3.4	1.5	<0.001	<0.001
	Study Area	257	77	30%	0.50	2.3	0.11	0.32	0.70	2.4	0.70	<0.001	<0.001
Arsenic	Background	31	31	100%	NA	NA	1.6	2.4	2.4	4.3	0.54	0.02	0.5
	Study Area	571	571	100%	NA	NA	0.58	3.3	3.7	34	2.2	<0.001	<0.001
Barium	Background	31	31	100%	NA	NA	111	162	166	213	23	0.6	0.4
	Study Area	260	260	100%	NA	NA	65	177	190	1,780	110	<0.001	<0.001
Beryllium	Background	31	31	100%	NA	NA	0.36	0.46	0.46	0.59	0.048	0.6	0.7
	Study Area	195	195	100%	NA	NA	0.22	0.46	0.46	0.71	0.074	0.2	<0.001
Boron	Background	31	7	23%	10	11	3.6	6.2	6.7	12	2.7	<0.001	<0.001
	Study Area	260	236	91%	1.4	13	2.5	7.4	24	1,510	135	<0.001	<0.001
Cadmium	Background	31	25	81%	0.10	0.11	0.11	0.19	0.20	0.48	0.085	0.003	0.02
	Study Area	260	139	53%	0.0050	0.51	0.040	0.13	0.26	8.9	0.77	<0.001	<0.001
Calcium	Background	31	31	100%	NA	NA	19,200	28,200	29,000	43,300	6,580	0.2	0.6
	Study Area	195	195	100%	NA	NA	9,930	26,700	27,500	62,500	9,280	<0.001	0.7
Chromium (total)	Background	31	31	100%	NA	NA	5.6	7.5	7.7	11	1.2	0.4	0.7
	Study Area	262	262	100%	NA	NA	3.5	9.4	12	102	11	<0.001	<0.001
Chromium VI	Background	31	1	3.2%	0.41	0.43	0.29	0.29	0.29	0.29	NA	<0.001	<0.001
	Study Area	215	48	22%	0.11	0.49	0.11	0.79	5.8	106	18	<0.001	<0.001
Cobalt	Background	31	31	100%	NA	NA	5.4	7.3	7.3	9.1	0.76	0.5	0.4
	Study Area	310	310	100%	NA	NA	3.2	7.6	13	284	29	<0.001	<0.001
Copper	Background	31	31	100%	NA	NA	16	19	23	140	22	<0.001	<0.001
	Study Area	260	260	100%	NA	NA	8.0	18	20	160	12	<0.001	<0.001
Iron	Background	31	31	100%	NA	NA	11,300	15,700	15,500	20,600	2,140	0.5	0.3
	Study Area	260	260	100%	NA	NA	7,050	15,000	14,900	24,000	2,690	0.2	0.006
Lead	Background	31	31	100%	NA	NA	7.1	8.9	11	73	12	<0.001	<0.001
	Study Area	321	321	100%	NA	NA	3.6	8.9	13	267	19	<0.001	<0.001
Magnesium	Background	31	31	100%	NA	NA	7,700	9,810	9,990	13,000	1,320	0.8	1
	Study Area	309	309	100%	NA	NA	5,300	9,700	10,700	71,000	5,070	<0.001	<0.001
Manganese	Background	31	31	100%	NA	NA	262	360	366	537	61	0.03	0.4
	Study Area	410	410	100%	NA	NA	133	390	962	29,200	2,220	<0.001	<0.001
Mercury	Background	31	27	87%	0.017	0.019	0.0060	0.016	0.036	0.36	0.069	<0.001	<0.001
	Study Area	262	230	88%	0.0067	0.040	0.0030	0.017	0.035	1.9	0.13	<0.001	<0.001
Molybdenum	Background	31	30	97%	0.31	0.31	0.31	0.49	1.7	33	5.9	<0.001	<0.001
	Study Area	260	194	75%	0.052	2.0	0.15	0.48	1.1	55	4.9	<0.001	<0.001
Nickel	Background	31	31	100%	NA	NA	13	16	16	21	1.8	0.08	0.5
	Study Area	260	260	100%	NA	NA	6.6	15	16	164	11	<0.001	<0.001

#### TABLE F-1. Summary Statistics for Metals in RZ-A Background Soils and Study Area Soils (0-10 feet bgs) Nevada Environmental Response Trust Site Henderson, Nevada

					Non-Dete	cts (mg/kg)			Detects	(mg/kg)		Shapiro	-Wilk Test
Chemical Name	Location	No. of Samples	No. of Detects	% Detects	Minimun	Maximum	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal (p-value)
Platinum	Background	31	19	61%	0.10	0.11	0.0060	0.010	0.012	0.046	0.0085	<0.001	<0.001
	Study Area	195	143	73%	0.010	0.24	0.0050	0.011	0.015	0.16	0.017	<0.001	<0.001
Potassium	Background	31	31	100%	NA	NA	1,450	2,080	2,180	4,210	658	<0.001	0.02
	Study Area	195	195	100%	NA	NA	1,230	2,160	2,320	6,120	649	<0.001	0.02
Selenium	Background	31	3	9.7%	4.1	4.4	0.80	0.80	0.83	0.90	0.058	<0.001	<0.001
	Study Area	260	16	6.2%	0.16	4.7	0.70	1.0	1.0	1.5	0.24	<0.001	<0.001
Silver	Background	31	0	0%	0.50	0.50	NA	NA	NA	NA	NA	NA	NA
	Study Area	260	49	19%	0.20	1.5	0.020	0.12	0.30	7.6	1.1	<0.001	<0.001
Sodium	Background	31	31	100%	NA	NA	307	630	621	1,050	194	0.3	0.3
	Study Area	195	195	100%	NA	NA	198	756	1,160	11,700	1,290	<0.001	<0.001
Strontium	Background	31	31	100%	NA	NA	129	214	222	339	57	0.4	0.3
	Study Area	213	213	100%	NA	NA	73	186	208	805	98	<0.001	0.03
Thallium	Background	31	31	100%	NA	NA	0.071	0.092	0.11	0.19	0.033	<0.001	0.003
	Study Area	260	176	68%	0.10	0.28	0.054	0.10	0.19	8.4	0.64	<0.001	<0.001
Tin	Background	31	0	0%	10	11	NA	NA	NA	NA	NA	0.4	0.4
	Study Area	195	184	94%	9.4	12	0.40	4.2	3.8	12	1.6	<0.001	<0.001
Titanium	Background	31	31	100%	NA	NA	480	829	793	1,080	162	0.2	0.04
	Study Area	195	195	100%	NA	NA	361	751	743	1,270	179	0.2	<0.001
Tungsten	Background	31	30	97%	0.11	0.11	0.12	0.17	0.21	0.62	0.11	<0.001	0.04
	Study Area	213	173	81%	0.10	5.6	0.080	0.23	0.42	8.5	0.80	<0.001	<0.001
Uranium (total)	Background	31	31	100%	NA	NA	0.66	0.98	1.1	1.9	0.36	0.002	0.05
	Study Area	213	213	100%	NA	NA	0.55	1.0	1.2	3.6	0.52	<0.001	<0.001
Vanadium	Background	31	31	100%	NA	NA	28	46	44	55	7.6	0.08	0.02
	Study Area	195	195	100%	NA	NA	22	42	42	78	9.0	0.09	0.05
Zinc	Background	31	31	100%	NA	NA	26	33	40	254	40	<0.001	<0.001
	Study Area	260	260	100%	NA	NA	18	33	37	300	25	<0.001	<0.001

#### Notes:

bgs = below ground surface

mg/kg = milligram per kilogram

NA = value not available

p-values < 0.01 are shown in italic.

Background dataset is from RZ-A, excluding the six borings in LOU 62.

Shapiro Wilk tests use 1/2 the detection limit (DL) for non-detects.

## TABLE F-2. Background Comparisons for Metals in Study Area Soils (0-10 feet bgs)

Nevada Environmental Response Trust Site

		t-test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Aluminum	N	0.7	0.8	0.5	0.2	0.3	No
Antimony	NP	1	1	1	0.3	1	LDF
Arsenic	NP	<0.001	<0.001	<0.001	<0.001	<0.001	Yes
Barium	NP	0.001	0.002	0.004	0.05	0.02	Yes
Beryllium	N	0.5	0.6	0.5	0.4	0.3	No
Boron	NP	0.02	<0.001	1	0.007	0.005	LDF
Cadmium	NP	0.4	1	0.9	1	0.3	No
Calcium	LN	0.9	1	0.9	0.7	0.2	No
Chromium (total)	NP	<0.001	<0.001	<0.001	<0.001	<0.001	Yes
Chromium VI	NP	0.02	<0.001	0.1	0.007	0.4	LDF
Cobalt	NP	<0.001	<0.001	0.09	0.008	0.002	Yes
Copper	NP	0.8	0.9	1	0.8	0.9	No
Iron	N	0.9	0.9	0.9	0.8	0.4	No
Lead	NP	0.2	0.3	0.5	0.2	0.7	No
Magnesium	NP	0.03	0.2	0.6	0.1	0.03	No
Manganese	NP	<0.001	<0.001	0.02	<0.001	<0.001	Yes
Mercury	NP	0.5	0.7	0.7	0.4	0.9	No
Molybdenum	NP	0.7	0.8	0.1	0.9	0.8	No
Nickel	NP	0.3	0.8	0.9	0.4	0.4	No
Platinum	NP	0.7	0.8	0.8	0.04	0.5	No
Potassium	LN	0.1	0.1	0.05	0.2	0.7	No
Selenium	NP	1	1	1	0.9	0.8	LDF
Silver	NP	0.06	0.8	0.2	0.001	NA	LDF
Sodium	NP	<0.001	<0.001	0.003	<0.001	<0.001	Yes

## TABLE F-2. Background Comparisons for Metals in Study Area Soils (0-10 feet bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

	Distribution	<i>t</i> -test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
Chemical Name	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Strontium	LN	0.9	1	1	0.6	0.08	No
Thallium	NP	0.03	0.004	<0.001	0.6	0.04	Yes
Tin	NP	1	1	1	0.001	NA	LDF
Titanium	N	0.9	1	1	1	0.5	No
Tungsten	NP	<0.001	<0.001	0.001	0.2	0.03	Yes
Uranium (total)	NP	0.1	0.2	0.3	0.8	0.09	No
Vanadium	N, LN	0.9	0.9	0.9	0.9	0.2	No
Zinc	NP	0.7	0.7	0.6	0.4	0.8	No

#### Notes:

bgs = below ground surface

NA = value not available

LDF = Low detection frequency (<25%) in either site or background datasets. Background comparison results may not be applicable.

p-values in italics indicate p < 0.025

Background comparison tests use 1/2 the detection limit (DL) for non-detects in the parametric test (t-test) and the DL for non-parametric tests (Gehan test, quantile test, and slippage test).

Background dataset is from RZ-A, excluding the six borings in LOU 62.

Final background determination is based on non-parametric tests only due to large sample size.

Distibution:

N = Study area data and background data consistent with normal distribution

LN = Study area data and background data consistent with log-normal distribution

NP = Study area data or background data is not consistent with both normal distribution and log-normal distribution.

#### TABLE F-3. Summary Statistics for Radionuclides in RZ-A Background Soils and Study Area Soils (0-10 feet bgs) Nevada Environmental Response Trust Site Henderson, Nevada

	Chamical		No. of	No. of				Detects	(pCi/g)		Shapiro	-Wilk Test
Chain	Name	Location	NO. Of Samples	No. of Detects	% Detects	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal ( <i>p</i> -value)
Uranium-238	Uranium-238	Background	31	31	100%	0.36	1.0	1.0	1.6	0.21	0.004	<0.001
		Study Area	205	205	100%	0.24	0.96	1.0	3.3	0.38	<0.001	<0.001
	Uranium-234	Background	31	31	100%	0.39	1.0	1.1	1.7	0.30	0.09	0.04
		Study Area	187	187	100%	0.27	1.0	1.1	3.4	0.44	<0.001	<0.001
	Thorium-230	Background	31	31	100%	0.51	1.1	1.1	1.7	0.28	0.7	0.7
		Study Area	205	205	100%	0.43	1.1	1.2	4.3	0.44	<0.001	<0.001
	Radium-226	Background	31	31	100%	0	0.89	0.95	1.7	0.35	0.4	<0.001
		Study Area	208	208	100%	0.20	0.92	0.95	2.5	0.36	<0.001	0.2
Thorium-232	Thorium-232	Background	31	31	100%	1.0	1.5	1.5	2.1	0.24	1	1
		Study Area	205	205	100%	0.54	1.6	1.6	2.5	0.33	0.4	<0.001
	Radium-228	Background	31	31	100%	0.46	1.2	1.3	2.5	0.54	0.3	0.8
		Study Area	208	208	100%	0	1.2	1.3	3.3	0.49	0.02	NA
	Thorium-228	Background	31	31	100%	1.2	1.7	1.7	2.9	0.36	0.03	0.5
		Study Area	205	205	100%	0.48	1.8	1.8	3.0	0.36	<0.001	<0.001
Uranium-235	Uranium-235	Background	31	31	100%	-0.077	0.049	0.051	0.20	0.044	0.003	NA
		Study Area	187	187	100%	-0.029	0.052	0.062	0.25	0.043	<0.001	NA

#### Notes:

bgs = below ground surface

pCi/g = picocurie per gram

NA = value not available

p-values < 0.01 are shown in italic.

Background dataset is from RZ-A, excluding the six borings in LOU 62.

Shapiro Wilk tests use 1/2 the detection limit (DL) for non-detects.

## TABLE F-4. Background Comparisons for Radionuclides in Study Area Soils (0-10 feet bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

Chain	Chemical	Distribution	<i>t</i> -test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
Chain	Name	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Uranium-238	Uranium-238	NP	0.5	0.7	0.9	0.4	0.2	No
	Uranium-234	NP	0.2	0.4	0.4	0.8	0.08	No
	Thorium-230	NP	0.03	0.05	0.1	0.4	0.1	No
	Radium-226	LN	0.5	0.4	0.7	0.9	0.4	No
Thorium-232	Thorium-232	N	0.06	0.1	0.07	0.1	0.1	No
	Radium-228	N, LN	0.6	0.6	0.5	0.6	0.8	No
	Thorium-228	NP	0.2	0.2	0.1	0.2	0.9	No
Uranium-235	Uranium-235	NP	0.1	0.2	0.2	0.2	0.6	No

#### Notes:

bgs = below ground surface

p-values in italics indicate p < 0.025

Background comparison tests use 1/2 the detection limit (DL) for non-detects in the parametric test (t-test) and the DL for non-parametric tests (Gehan test, quantile test, and slippage test).

Background dataset is from RZ-A, excluding the six borings in LOU 62.

Final background determination is based on non-parametric tests only due to large sample size.

Distibution:

N = Study area data and background data consistent with normal distribution

LN = Study area data and background data consistent with log-normal distribution

NP = Study area data or background data is not consistent with both normal distribution and log-normal distribution.

#### TABLE F-5A. Equivalence Test for Secular Equilibrium of Uranium Decay Series<sup>1</sup> Nevada Environmental Response Trust Remediation Project Site Henderson, Nevada

Location	p-value	Conclusion <sup>2</sup>	Delta	Sample Size <sup>3</sup>	Number Missing <sup>4</sup>	Analyte	Mean Proportions of Radioactivity	95% Confi	d. Intervals	Shifts⁵
				0120	incomg		••••••••••••••••••••••••••••••••••••••	Lower	Upper	
All	<0.0001	in Secular	0.1	218	21	Ra-226	0.2215	0.2061	0.2370	0
	Equilic	Equilibrium				Th-230	0.2779	0.2670	0.2889	0
						U-234	0.2580	0.2483	0.2677	0
						U-238	0.2426	0.2331	0.2520	0
Background	<0.0001	in Secular	0.1	31	0	Ra-226	0.2267	0.1909	0.2625	0
		Equilibrium				Th-230	0.2626	0.2314	0.2939	0
						U-234	0.2572	0.2283	0.2861	0
						U-238	0.2534	0.2276	0.2793	0
Study Area	<0.0001	in Secular	0.1	187	21	Ra-226	0.2207	0.2034	0.2379	0
		Equilibrium				Th-230	0.2805	0.2686	0.2924	0
						U-234	0.2581	0.2476	0.2686	0
						U-238	0.2408	0.2305	0.2510	0

#### Notes:

1. Analyzed in top 10 feet bgs using the EnviroGISdT software tool from Neptune & Company, Inc.

2. Tool states "in Secular Equilibrium" if the computed *p*-value is less than a standard significance level of 0.05.

3. Sample dataset includes field duplicates

4. Count of sampes for which one or more results are unavailable. These samples are not counted in the sample size and are not included in the secular equilibrium calculation.

5. Data Shift - Lists the values of the data shift utilized by the tool in case of negative radioactivity measurements. All measurements values for that radioisotope are shifted upwards by the shift value so that all values are non-negative. A zero shift value indicates lack of negative measurements.

## TABLE F-5B. Equivalence Test for Secular Equilibrium of Thorium Decay Series1Nevada Environmental Response Trust Remediation Project SiteHenderson, Nevada

Location	p-value	Conclusion <sup>2</sup>	Delta	Sample Size <sup>3</sup>	Number Missing⁴	Analyte	Mean Proportions of Radioactivity	95% Confid. Intervals		Shifts <sup>5</sup>
								Lower	Upper	
All	0.0115	in Secular	0.1	236	3	Ra-228	0.2722	0.2554	0.2891	0
	Equilibrium				Th-228	0.3831	0.3725	0.3938	0	
						Th-232	0.3447	0.3356	0.3537	0
Background	0.164	Not in Secular	0.1	31	0	Ra-228	0.2779	0.2341	0.3218	0
		Equilibrium				Th-228	0.3808	0.3557	0.4060	0
						Th-232	0.3413	0.3141	0.3684	0
Study Area	0.0288	in Secular	0.1	205	3	Ra-228	0.2714	0.2529	0.2898	0
		Equilibrium				Th-228	0.3835	0.3717	0.3952	0
						Th-232	0.3452	0.3354	0.3549	0

#### Notes:

1. Analyzed in top 10 feet bgs using the EnviroGISdT software tool from Neptune & Company, Inc.

2. Tool states "in Secular Equilibrium" if the computed *p*-value is less than a standard significance level of 0.05.

3. Sample dataset includes field duplicates

4. Count of samples for which one or more results are unavailable. These samples are not counted in the sample size and are not included in the secular equilibrium calculation.

5. Data Shift - Lists the values of the data shift utilized by the tool in case of negative radioactivity measurements. All measurements values for that radioisotope are shifted upwards by the shift value so that all values are non-negative. A zero shift value indicates lack of negative measurements.

TABLE F-6. Correlation Matrices for the Uranium Decay Series and the Thorium DecaySeries Nevada Environmental Response Trust SiteHenderson, Nevada

- **Uranium Decay Chain** Correl. Ra-226 Th-230 U-234 U-238 Ra-226 1 0.453 0.350 0.309 Th-230 0.453 1 0.687 0.667 U-234 0.350 0.687 1 0.894 U-238 0.309 0.667 0.894 1
- i) Study Area Soils (0-10 ft bgs)

Thorium Decay Chain					
Correl.	Ra-228	Th-228	Th-232		
Ra-228	1	-0.101	0.014		
Th-228	-0.101	1	0.690		
Th-232	0.014	0.690	1		

#### ii) Background (RZ-A) Soils

Uranium Decay Chain						
Correl.	Ra-226 Th-230		U-234	U-238		
Ra-226	1	0.680	0.313	0.321		
Th-230	0.680	1	0.351	0.271		
U-234	0.313	0.351	1	0.886		
U-238	0.321	0.271	0.886	1		

Thorium Decay Chain					
Correl.	Ra-228	Th-228	Th-232		
Ra-228	1	0.297	0.119		
Th-228	0.297	1	0.627		
Th-232	0.119	0.627	1		



## Figure F1–1. RZ–A Background vs. Study Area Boxplots Aluminum





## BCL = 519 mg/kg BCL = 519 mg/kgBCL = 519 mg/kg

## Figure F1–2. RZ–A Background vs. Study Area Boxplots Antimony



## Figure F1–3. RZ–A Background vs. Study Area Boxplots Arsenic




10<sup>1</sup>

RZ-A Background

#### Figure F1–4. RZ–A Background vs. Study Area Boxplots Barium

**Ramboll Environ** 

Study Area



## Figure F1–5. RZ–A Background vs. Study Area Boxplots Beryllium





#### Figure F1–6. RZ–A Background vs. Study Area Boxplots Boron



## Figure F1–7. RZ–A Background vs. Study Area Boxplots Cadmium

















#### Figure F1–11. RZ–A Background vs. Study Area Boxplots Cobalt





# Figure F1–12. RZ–A Background vs. Study Area Boxplots Copper







10<sup>0</sup>

RZ-A Background

#### Figure F1–14. RZ–A Background vs. Study Area Boxplots Lead

**Ramboll Environ** 

Study Area







## Figure F1–17. RZ–A Background vs. Study Area Boxplots Mercury





## Figure F1–18. RZ–A Background vs. Study Area Boxplots Molybdenum



## Figure F1–19. RZ–A Background vs. Study Area Boxplots Nickel





## Figure F1–20. RZ–A Background vs. Study Area Boxplots Platinum





## Figure F1–21. RZ–A Background vs. Study Area Boxplots Potassium











# Figure F1–23. RZ–A Background vs. Study Area Boxplots Silver



# Figure F1–24. RZ–A Background vs. Study Area Boxplots Sodium





# Figure F1–25. RZ–A Background vs. Study Area Boxplots Strontium





## Figure F1–26. RZ–A Background vs. Study Area Boxplots Thallium



 $10^{-1}$ 

RZ-A Background

Figure F1–27. RZ–A Background vs. Study Area Boxplots Tin

**Ramboll Environ** 

Study Area



## Figure F1–28. RZ–A Background vs. Study Area Boxplots Titanium





## Figure F1–29. RZ–A Background vs. Study Area Boxplots Tungsten

**Ramboll Environ** 

Study Area









## Figure F1–31. RZ–A Background vs. Study Area Boxplots Vanadium







Figure F1–32. RZ–A Background vs. Study Area Boxplots Zinc



# Figure F1–33. RZ–A Background vs. Study Area Boxplots Uranium–238





# Figure F1–34. RZ–A Background vs. Study Area Boxplots Uranium–234





# Figure F1–35. RZ–A Background vs. Study Area Boxplots Thorium–230





# Figure F1–36. RZ–A Background vs. Study Area Boxplots Radium–226





# Figure F1–37. RZ–A Background vs. Study Area Boxplots Thorium–232





# Figure F1–38. RZ–A Background vs. Study Area Boxplots Radium–228










### Figure F1–40. RZ–A Background vs. Study Area Boxplots Uranium–235



### Figure F2–1. Normal and Lognormal Q–Q Plots Aluminum





### Figure F2–2. Normal and Lognormal Q–Q Plots Antimony





### Figure F2–3. Normal and Lognormal Q–Q Plots Arsenic





# Figure F2–4. Normal and Lognormal Q–Q Plots Barium





# Figure F2–5. Normal and Lognormal Q–Q Plots Beryllium





### Figure F2–6. Normal and Lognormal Q–Q Plots Boron





# Figure F2–7. Normal and Lognormal Q–Q Plots Cadmium









# Figure F2–8. Normal and Lognormal Q–Q Plots Calcium

Figure F2–9. Normal and Lognormal Q–Q Plots Chromium (total)





#### Figure F2–10. Normal and Lognormal Q–Q Plots Chromium VI





### Figure F2–11. Normal and Lognormal Q–Q Plots Cobalt





# Figure F2–12. Normal and Lognormal Q–Q Plots Copper







#### Figure F2–13. Normal and Lognormal Q–Q Plots Iron



## Figure F2–14. Normal and Lognormal Q–Q Plots Lead



Normal Theoretical Quantiles

#### Figure F2–15. Normal and Lognormal Q–Q Plots Magnesium





#### Figure F2–16. Normal and Lognormal Q–Q Plots Manganese





### Figure F2–17. Normal and Lognormal Q–Q Plots Mercury





### Figure F2–18. Normal and Lognormal Q–Q Plots Molybdenum





## Figure F2–19. Normal and Lognormal Q–Q Plots Nickel





## Figure F2–20. Normal and Lognormal Q–Q Plots Platinum





## Figure F2–21. Normal and Lognormal Q–Q Plots Potassium











# Figure F2–23. Normal and Lognormal Q–Q Plots Silver





## Figure F2–24. Normal and Lognormal Q–Q Plots Sodium





# Figure F2–25. Normal and Lognormal Q–Q Plots Strontium





# Figure F2–26. Normal and Lognormal Q–Q Plots Thallium











## Figure F2–28. Normal and Lognormal Q–Q Plots Titanium





# Figure F2–29. Normal and Lognormal Q–Q Plots Tungsten





# Figure F2–30. Normal and Lognormal Q–Q Plots Uranium (total)





## Figure F2–31. Normal and Lognormal Q–Q Plots Vanadium





## Figure F2–32. Normal and Lognormal Q–Q Plots Zinc





#### Figure F2–33. Normal and Lognormal Q–Q Plots Uranium–238





#### Figure F2–34. Normal and Lognormal Q–Q Plots Uranium–234





#### Figure F2–35. Normal and Lognormal Q–Q Plots Thorium–230




# Figure F2–36. Normal and Lognormal Q–Q Plots Radium–226



# Figure F2–37. Normal and Lognormal Q–Q Plots Thorium–232





# Figure F2–38. Normal and Lognormal Q–Q Plots Radium–228



# Figure F2–39. Normal and Lognormal Q–Q Plots Thorium–228





# Figure F2–40. Normal and Lognormal Q–Q Plots Uranium–235





Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

#### APPENDIX G SPATIAL QUARTILE PLOTS
























































































Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

> APPENDIX H NDEP FLOWCHART FOR RADIONUCLIDE DATA USABILITY



COPCs indicates "chemicals of potential concern".  $U_{metal}$  denotes metallic uranium.

Figure 1. Flowchart describing the decision framework for radionuclide historical dataset usability for Sites within the BMI Complex and Common Areas, Henderson, NV.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

# APPENDIX I PROUCL OUTPUT FILES

## UCL Statistics for Uncensored Full Data Sets

User Selected Options Date/Time of Computation From File Full Precision Confidence Coefficient Number of Bootstrap Operations

## final\_numeric (brc/timet\_radium-226)

	General S	tatistics	
Total Number of Observations	104	Number of Distinct Observations	82
		Number of Missing Observations	0
Minimum	0.494	Mean	1.112
Maximum	2.36	Median	1.065
SD	0.347	Std. Error of Mean	0.0341
Coefficient of Variation	0.312	Skewness	0.987
	Normal G	OF Test	
Shapiro Wilk Test Statistic	0.938	Shapiro Wilk GOF Test	
5% Shapiro Wilk P Value	1.05/4E-4	Data Not Normal at 5% Significance Level	
	0.111	Lillietors GOF Test	
5% Lilliefors Critical Value	0.08/2	Data Not Normal at 5% Significance Level	
Data Not	Normal at 5%		
As	suming Norm	al Distribution	
95% Normal UCL	1 160	95% UCLS (Adjusted for Skewness)	1 170
95% Students-t OCL	1.109	95% Aujusted-CLT UCL (Chen-1995) 05% Modified t UCL (Johnson 1078)	1.172
		95% Modified-LOCE (Johnson-1978)	1.109
	Gamma G	OF Test	
A-D Test Statistic	0.643	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.752	Detected data appear Gamma Distributed at 5% Significance	e Level
K-S Test Statistic	0.071	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.0884	Detected data appear Gamma Distributed at 5% Significance	e Level
Detected data appear	Gamma Dist	ibuted at 5% Significance Level	
	Gamma S	tatistics	
k hat (MLE)	11.02	k star (bias corrected MLE)	10.71
Theta hat (MLE)	0.101	Theta star (bias corrected MLE)	0.104
nu hat (MLE)	2292	nu star (bias corrected)	2227
MLE Mean (bias corrected)	1.112	MLE Sd (bias corrected)	0.34
		Approximate Chi Square Value (0.05)	2118
Adjusted Level of Significance	0.0477	Adjusted Chi Square Value	2117
Ass	suming Gamn	na Distribution	
95% Approximate Gamma UCL (use when n>=50)	1.169	95% Adjusted Gamma UCL (use when n<50)	1.17
	Lognormal (	GOF Test	
Shapiro Wilk Test Statistic	0.979	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk P Value	0.468	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0658	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.0872	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal at	5% Significance Level	
	Lognormal	Statistics	
Minimum of Logged Data	-0.705	Mean of logged Data	0.0603
Maximum of Logged Data	0.859	SD of logged Data	0.305
Assi	iming Loanor	mal Distribution	
95% H-UCL	1.173	90% Chebyshev (MVUE) UCL	1.214
95% Chebyshev (MVUE) UCL	1.261	97.5% Chebyshev (MVUE) UCL	1.325
99% Chebyshev (MVUE) UCL	1.451		
Nonnarame	tric Distributio	n Free LICL Statistics	
Data appear to follow a I	Discernible Di	stribution at 5% Significance Level	

Nonparametric [	Distribution Free UCLs
Nonparamouro E	

95% CLT UCL	1.168	95% Jackknife UCL	1.169
95% Standard Bootstrap UCL	1.167	95% Bootstrap-t UCL	1.174

95% Hall's Bootstrap UCL	1.173	95% Percentile Bootstrap UCL	1.168
95% BCA Bootstrap UCL	1.172		
90% Chebyshev(Mean, Sd) UCL	1.214	95% Chebyshev(Mean, Sd) UCL	1.261
97.5% Chebyshev(Mean, Sd) UCL	1.325	99% Chebyshev(Mean, Sd) UCL	1.451

## Suggested UCL to Use

95% Approximate Gamma UCL 1.169

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (brc/timet\_radium-228)

	General Sta	itistics	
Total Number of Observations	84	Number of Distinct Observations	66
		Number of Missing Observations	0
Minimum	0.946	Mean	1 916
Maximum	2 94	Median	1 96
	0.405	Std. Error of Mean	0.0441
	0.405		0.0441
Coefficient of variation	0.211	Skewness	0.0338
	Normal GO	F Test	
Shapiro Wilk Test Statistic	0.982	Shapiro Wilk GOF Test	
5% Shapiro Wilk P Value	0.677	Data appear Normal at 5% Significance Level	
Lilliefors Test Statistic	0.0658	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.0008	Data appear Normal at 5% Significance Level	
Data appea	ar Normal at 59	6 Significance Level	
As: 95% Normal LICI	suming Norma	Distribution 95% LICLs (Adjusted for Skewness)	
	1 0 9 0	95% OCLS (Adjusted for Skewness)	1 000
95% Student's-t UCL	1.989	95% Adjusted-CLT UCL (Chen-1995)	1.988
		95% Modified-t UCL (Johnson-1978)	1.989
	Gamma GO	F Test	
A-D Test Statistic	0.652	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.75	Detected data appear Gamma Distributed at 5% Significance	e l evel
K S Test Statistic	0.0047	Kolmogorov Smirnov Gamma GOE Test	C LOVOI
EV K C Critical Value	0.0347	Detected data annear Commo Distributed at 5% Cignificance	
5% K-S United value	0.0972	Detected data appear Gamma Distributed at 5% Significand	e Levei
Detected data appear	Gamma Distri	buted at 5% Significance Level	
	Gamma Sta	atistics	
k hat (MLE)	21.49	k star (bias corrected MLE)	20.73
Theta hat (MLE)	0.0892	Theta star (bias corrected MLE)	0.0924
nu hat (MLE)	3610	nu star (bias corrected)	3482
MLE Mean (bias corrected)	1 916	MLE Sd (bias corrected)	0 421
		Approximate Chi Square Value (0.05)	3346
Adjusted Loval of Significance	0.0471	Adjusted Chi Square Value	2244
Adjusted Level of Significance	0.0471	Aujusted Chi Square Value	3344
Ass	suming Gamma	a Distribution	
95% Approximate Gamma UCL (use when n>=50))	1.994	95% Adjusted Gamma UCL (use when n<50)	1.995
	Lognormal G	OF Test	
Shapiro Wilk Test Statistic	0,966	Shaniro Wilk Lognormal GOF Test	
5% Shaniro Wilk P Value	0.11	Data appear Lognormal at 5% Significance Level	
J // Shapiro Wilk F Value	0.11	Lillioforn Lognormal COE Toot	
	0.108		
5% Lilliefors Critical Value	0.0968	Data Not Lognormal at 5% Significance Level	
Data appear Approx	ximate Lognorr	nal at 5% Significance Level	
	Lognormal S	tatistics	
Minimum of Logged Data	-0.0555	Mean of logged Data	0.627
Maximum of Logged Data	1.078	SD of logged Data	0.222
<b>.</b>		al Distribution	
Assu			2 050
95% H-UCL	2		2.059
95% Chebyshev (MVUE) UCL	2.123	97.5% Chebyshev (MVUE) UCL	2.212
99% Chebyshev (MVUE) UCL	2.386		

Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

95% CLT UCL	1.988	95% Jackknife UCL	1.989
95% Standard Bootstrap UCL	1.988	95% Bootstrap-t UCL	1.988
95% Hall's Bootstrap UCL	1.989	95% Percentile Bootstrap UCL	1.988
95% BCA Bootstrap UCL	1.989		
90% Chebyshev(Mean, Sd) UCL	2.048	95% Chebyshev(Mean, Sd) UCL	2.108
97.5% Chebyshev(Mean, Sd) UCL	2.191	99% Chebyshev(Mean, Sd) UCL	2.355

# Suggested UCL to Use

95% Student's-t UCL 1.989

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (brc/timet\_thorium-228)

	General Statistics		
Total Number of Observations	120	Number of Distinct Observations	76
		Number of Missing Observations	0
Minimum	1.07	Mean	1.687
Maximum	2.28	Median	1.705
SD	0.278	Std. Error of Mean	0.0253
Coefficient of Variation	0.165	Skewness	-0.136
	Normal GOF Test		
Shapiro Wilk Test Statistic	0.96	Shapiro Wilk GOF Test	
5% Shapiro Wilk P Value	0.0107	Data Not Normal at 5% Significance Level	
Lilliefors Test Statistic	0.0762	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.0812	Data appear Normal at 5% Significance Level	
Data appear App	roximate Normal at 5%	Significance Level	
As	suming Normal Distrib	ution	
95% Normal UCL		95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.729	95% Adjusted-CLT UCL (Chen-1995)	1.728
		95% Modified-t UCL (Johnson-1978)	1.729
	Gamma GOF Test		
A-D Test Statistic	1.132	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.75	Data Not Gamma Distributed at 5% Significance Leve	el
K-S Test Statistic	0.0899	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.0841	Data Not Gamma Distributed at 5% Significance Leve	el
Data Not Gam	ma Distributed at 5% S	ignificance Level	
	Gamma Statistics		
k hat (MLE)	35.8	k star (bias corrected MLE)	34.91
Theta hat (MLE)	0.0471	Theta star (bias corrected MLE)	0.0483
nu hat (MLE)	8591	nu star (bias corrected)	8378
MLE Mean (bias corrected)	1.687	MLE Sd (bias corrected)	0.285
		Approximate Chi Square Value (0.05)	8166
Adjusted Level of Significance	0.048	Adjusted Chi Square Value	8164
As	suming Gamma Distrit	pution	
95% Approximate Gamma UCL (use when n>=50))	1.73	95% Adjusted Gamma UCL (use when n<50)	1.731
	Lognormal GOF Tes	t	
Shapiro Wilk Test Statistic	0.951	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk P Value	9.1661E-4	Data Not Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0966	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.0812	Data Not Lognormal at 5% Significance Level	
Data Not L	ognormal at 5% Signif		
	Lognormal Statistics	3	0 505
Minimum of Logged Data	0.0677	Mean of logged Data	0.509
Maximum of Logged Data	0.824	SD of logged Data	0.17
Ass	uming Lognormal Distr	ibution	
95% H-UCL	1.733	90% Chebyshev (MVUE) UCL	1.767
95% Chebyshev (MVUE) UCL	1.802	97.5% Chebyshev (MVUE) UCL	1.852

99% Chebyshev (MVUE) UCL 1.95

### Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

## Nonparametric Distribution Free UCLs

95% CLT UCL	1.728
95% Standard Bootstrap UCL	1.728
95% Hall's Bootstrap UCL	1.728
95% BCA Bootstrap UCL	1.728
90% Chebyshev(Mean, Sd) UCL	1.763
97.5% Chebyshev(Mean, Sd) UCL	1.845

95% Jackknife UCL 1.729 95% Bootstrap-t UCL 1.728 95% Percentile Bootstrap UCL 1.728 95% Chebyshev(Mean, Sd) UCL 1.797 99% Chebyshev(Mean, Sd) UCL 1.939

## Suggested UCL to Use

95% Student's-t UCL 1.729

When a data set follows an approximate (e.g., normal) distribution passing one of the GOF test When applicable, it is suggested to use a UCL based upon a distribution (e.g., gamma) passing both GOF tests in ProUCL

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

Note: For highly negatively-skewed data, confidence limits (e.g., Chen, Johnson, Lognormal, and Gamma) may not be reliable. Chen's and Johnson's methods provide adjustments for positvely skewed data sets.

### final\_numeric (brc/timet\_thorium-230)

	General Statistics		
Total Number of Observations	120	Number of Distinct Observations	82
		Number of Missing Observations	0
Minimum	0.66	Mean	1.246
Maximum	3.01	Median	1 19
SD	0.383	Std Error of Mean	0.0349
Coefficient of Variation	0.307	Skewness	1 606
	Normal GOF Test		
Shapiro Wilk Test Statistic	0.89	Shapiro Wilk GOF Test	
5% Shapiro Wilk P Value	1 103F-12	Data Not Normal at 5% Significance Level	
Lilliefors Test Statistic	0 131	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.0812	Data Not Normal at 5% Significance Level	
Data Not	Normal at 5% Signific	ance Level	
As	suming Normal Distrit	pution	
95% Normal UCL	Ū	95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.304	95% Adjusted-CLT UCL (Chen-1995)	1.309
		95% Modified-t UCL (Johnson-1978)	1.305
	Gamma GOF Test		
A-D Test Statistic	1.051	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.751	Data Not Gamma Distributed at 5% Significance Leve	el
K-S Test Statistic	0.0944	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.0842	Data Not Gamma Distributed at 5% Significance Leve	el
Data Not Gam	na Distributed at 5% S	Significance Level	
	Gamma Statistics		
k hat (MLE)	12.57	k star (bias corrected MLE)	12.27
Theta hat (MLE)	0.0991	Theta star (bias corrected MLE)	0.102
nu hat (MLE)	3018	nu star (bias corrected)	2944
MLE Mean (bias corrected)	1.246	MLE Sd (bias corrected)	0.356
(**************************************		Approximate Chi Square Value (0.05)	2819
Adjusted Level of Significance	0.048	Adjusted Chi Square Value	2817
As	suming Gamma Distril	oution	
95% Approximate Gamma UCL (use when n>=50))	1.301	95% Adjusted Gamma UCL (use when n<50)	1.302
· · · · · · · · · · · · · · · · · · ·			
	Lognormal GOF Tes	st	
Shapiro Wilk Test Statistic	0.974	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk P Value	0.207	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0743	Lilliefors Lognormal GOF Test	

5% Lilliefors Critical Value	0.0812	Data appear Lognormal at 5% Significance Level
Data appear L	.ognormal a	t 5% Significance Level

### Lognormal Statistics

Minimum of Logged Data Maximum of Logged Data	-0.416 1.102	Mean of logged Data SD of logged Data	0.18 0.278
Assu	ming Lognormal Distribution		
95% H-UCL	1.3	90% Chebyshev (MVUE) UCL	1.34
95% Chebyshev (MVUE) UCL	1.383	97.5% Chebyshev (MVUE) UCL	1.444

95% Chebyshev (MVUE) UCL	1.383	97.5% Chebysh
99% Chebyshev (MVUE) UCL	1.563	

## Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

Nonpar	ametric Distribution Free UCLs		
95% CLT UCL	1.304	95% Jackknife UCL	1.304
95% Standard Bootstrap UCL	1.303	95% Bootstrap-t UCL	1.311
95% Hall's Bootstrap UCL	1.311	95% Percentile Bootstrap UCL	1.306
95% BCA Bootstrap UCL	1.307		
90% Chebyshev(Mean, Sd) UCL	1.351	95% Chebyshev(Mean, Sd) UCL	1.398
97.5% Chebyshev(Mean, Sd) UCL	1.464	99% Chebyshev(Mean, Sd) UCL	1.594
	Suggested UCL to Use		
95% Student's-t UCL	1.304	or 95% Modified-t UCL	1.305

95% Student's-t UCL or 95% H-UCL

MLE Mean (bias corrected)

1.614

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006).

1.3

However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

### ProUCL computes and outputs H-statistic based UCLs for historical reasons only.

H-statistic often results in unstable (both high and low) values of UCL95 as shown in examples in the Technical Guide.

It is therefore recommended to avoid the use of H-statistic based 95% UCLs.

Use of nonparametric methods are preferred to compute UCL95 for skewed data sets which do not follow a gamma distribution.

### final\_numeric (brc/timet\_thorium-232)

	General Sta	tistics	
Total Number of Observations	120	Number of Distinct Observations	73
		Number of Missing Observations	0
Minimum	1.05	Mean	1.614
Maximum	2.23	Median	1.57
SD	0.266	Std. Error of Mean	0.0243
Coefficient of Variation	0.165	Skewness	0.178
	Normal GO	= Test	
Shapiro Wilk Test Statistic	0.961	Shapiro Wilk GOF Test	
5% Shapiro Wilk P Value	0.0129	Data Not Normal at 5% Significance Level	
Lilliefors Test Statistic	0.0827	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.0812	Data Not Normal at 5% Significance Level	
Data Not	Normal at 5%	Significance Level	
Ass	suming Normal	Distribution	
95% Normal UCL		95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.654	95% Adjusted-CLT UCL (Chen-1995)	1.655
		95% Modified-t UCL (Johnson-1978)	1.655
	Gamma GO	F Test	
A-D Test Statistic	0.773	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.75	Data Not Gamma Distributed at 5% Significance Leve	el
K-S Test Statistic	0.0752	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.0841	Detected data appear Gamma Distributed at 5% Significanc	e Level
Detected data follow App	r. Gamma Dist	ribution at 5% Significance Level	
	Gamma Sta	tistics	
k hat (MLE)	37.02	k star (bias corrected MLE)	36.1
Theta hat (MLE)	0.0436	Theta star (bias corrected MLE)	0.0447
nu hat (MLE)	8885	nu star (bias corrected)	8664

MLE Sd (bias corrected)

0.269

		Approximate Chi Square Value (0.05)	8449
Adjusted Level of Significance	0.048	Adjusted Chi Square Value	8446
Ass	uming Gam	na Distribution	
95% Approximate Gamma UCL (use when n>=50)	1.655	95% Adjusted Gamma UCL (use when n<50)	1.656
	Lognormal	GOF Test	
Shapiro Wilk Test Statistic	0.965	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk P Value	0.0349	Data Not Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0794	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.0812	Data appear Lognormal at 5% Significance Level	
Data appear Approx	kimate Logno	rmal at 5% Significance Level	
	Lognormal	Statistics	
Minimum of Logged Data	0.0488	Mean of logged Data	0.465
Maximum of Logged Data	0.802	SD of logged Data	0.166
Assu	ming Lognoi	mal Distribution	
95% H-UCL	1.657	90% Chebyshev (MVUE) UCL	1.688
95% Chebyshev (MVUE) UCL	1.722	97.5% Chebyshev (MVUE) UCL	1.768
99% Chebyshev (MVUE) UCL	1.859		
Nonparame	tric Distributi	on Free UCL Statistics	
Data appear to follow a D	Discernible D	istribution at 5% Significance Level	
Nonpar	ametric Distr	ibution Free UCLs	
	4 05 4		4 05 4

95% CLT UCL	1.654	95% Jackknife UCL	1.654
95% Standard Bootstrap UCL	1.654	95% Bootstrap-t UCL	1.655
95% Hall's Bootstrap UCL	1.655	95% Percentile Bootstrap UCL	1.654
95% BCA Bootstrap UCL	1.653		
90% Chebyshev(Mean, Sd) UCL	1.687	95% Chebyshev(Mean, Sd) UCL	1.72
97.5% Chebyshev(Mean, Sd) UCL	1.766	99% Chebyshev(Mean, Sd) UCL	1.856

## Suggested UCL to Use

95% Approximate Gamma UCL 1.655

When a data set follows an approximate (e.g., normal) distribution passing one of the GOF test When applicable, it is suggested to use a UCL based upon a distribution (e.g., gamma) passing both GOF tests in ProUCL

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (brc/timet\_uranium-234)

	General Statistics		
Total Number of Observations	120	Number of Distinct Observations	74
		Number of Missing Observations	0
Minimum	0.47	Mean	1.109
Maximum	2.84	Median	0.99
SD	0.457	Std. Error of Mean	0.0417
Coefficient of Variation	0.412	Skewness	1.792
	Normal GOF Test		

#### Normal GOF Test

Shapiro Wilk Test Statistic	0.819
5% Shapiro Wilk P Value	0
Lilliefors Test Statistic	0.195
5% Lilliefors Critical Value	0.0812

Shapiro Wilk GOF Test Data Not Normal at 5% Significance Level Lilliefors GOF Test Data Not Normal at 5% Significance Level

## Data Not Normal at 5% Significance Level

Ass	uming Normal Distribution		
95% Normal UCL	-	95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.178	95% Adjusted-CLT UCL (Chen-1995)	1.185
		95% Modified-t UCL (Johnson-1978)	1.179

## Gamma GOF Test

A-D Test Statistic	3.149	Anderson-Darling Gamma GOF Test
5% A-D Critical Value	0.753	Data Not Gamma Distributed at 5% Significance Level
K-S Test Statistic	0.14	Kolmogorov-Smirnov Gamma GOF Test
5% K-S Critical Value	0.0843	Data Not Gamma Distributed at 5% Significance Level

## Data Not Gamma Distributed at 5% Significance Level

	Gamma	Statistics	
k hat (MLE)	7.606	k star (bias corrected MLE)	7.422
Theta hat (MLE)	0.146	Theta star (bias corrected MLE)	0.149
nu hat (MLE)	1825	nu star (bias corrected)	1781
MI E Mean (bias corrected)	1 109	MLF Sd (bias corrected)	0 407
		Approximate Chi Square Value (0.05)	1684
Adjusted Level of Significance	0.048	Adjusted Chi Square Value	1683
Adjusted Level of Olymineance	0.040	Adjusted Oni Oquale Value	1005
Ass	suming Gar	nma Distribution	
95% Approximate Gamma UCL (use when n>=50))	1 173	95% Adjusted Gamma UCL (use when n<50)	1 174
	Loanorma	al GOF Test	
Shapiro Wilk Test Statistic	0.947	Shaniro Wilk Lognormal GOF Test	
5% Shaniro Wilk P Value	2 / 870E_/	Data Not Lognormal at 5% Significance Level	
Lilliefore Test Statistic	0 115	Lilliefors Lognormal GOE Test	
Elilietors Test Statistic	0.115	Deta Net Lagnermal et E% Significance Lovel	
5% Lilieiors Childar Value	0.0612	t E% Significance Level	
Data Not L	ognormal a	it 5% Significance Level	
	Lognorm	al Statistics	
Minimum of Logged Data	-0 755	Mean of logged Data	0.0362
Maximum of Logged Data	1 044	SD of logged Data	0.353
Maximum of Logged Data	1.044	OD of logged Data	0.000
Assi	ımina Loan	ormal Distribution	
95% H-UCI	1 168	90% Chebyshev (MVUE) UC	1 2 1 2
95% Chebyshey (MVUE) UCL	1 262	97.5% Chebyshev (M\/LIE) LICI	1 331
99% Chebyshev (MVUE) UCL	1 / 66	37.5% Chebyshev (MVOE) COE	1.001
	1.400		
Nonparame Data do not fe	etric Distribution	ution Free UCL Statistics cernible Distribution (0.05)	
Nonpar	ametric Dis	stribution Free UCLs	
95% CLT UCL	1.178	95% Jackknife UCL	1.178
95% Standard Bootstrap UCL	1.177	95% Bootstrap-t UCL	1.187
95% Hall's Bootstrap UCL	1.188	95% Percentile Bootstrap UCL	1.179
95% BCA Bootstrap UCL	1.186		
90% Chebyshev(Mean, Sd) UCL	1.234	95% Chebyshev(Mean, Sd) UCL	1.291
97.5% Chebyshev(Mean, Sd) UCL	1.37	99% Chebyshev(Mean, Sd) UCL	1.524
	Suggested	I UCL to Use	
95% Student's-t UCL	1.178	or 95% Modified-t UCL	1.179
Note: Suggestions regarding the selection of a 95%	UCL are p	rovided to help the user to select the most appropriate 95% UCL.	
Recommendations are bas	ed upon da	ita size, data distribution, and skewness.	
These recommendations are based upon the resul	ts of the sir	nulation studies summarized in Singh, Maichle, and Lee (2006).	
However, simulations results will not cover all Real W	orld data se	ets; for additional insight the user may want to consult a statisticial	า.
final numeric (brc/timet uranium-235)			
	Genera	I Statistics	
Total Number of Observations	120	Number of Distinct Observations	76
	-	Number of Missing Observations	0
Minimum	0	Mean	0.0659
Mavimum	0 21	Median	0.059
	0.283	Std Error of Moon	0.000
Coofficient of Veriation	0.0002		0.00049
Coefficient of Variation	0.579	SKewness	0.öZ

### Normal GOF Test 0.953

0.00163

0.109

0.0812

Shapiro Wilk Test Statistic 5% Shapiro Wilk P Value Lilliefors Test Statistic 5% Lilliefors Critical Value

Data Not Data Not

Shapiro Wilk GOF Test Data Not Normal at 5% Significance Level Lilliefors GOF Test Data Not Normal at 5% Significance Level

Data Not Normal at 5% Significance Level

## Assuming Normal Distribution

95% Normal UCL

95% Student's-t UCL 0.0717

95% UCLs (Adjusted for Skewness) 95% Adjusted-CLT UCL (Chen-1995)

 95% Adjusted-CLT UCL (Chen-1995)
 0.0719

 95% Modified-t UCL (Johnson-1978)
 0.0717

Gamma Statistics Not Available Lognormal Statistics Not Available

### Nonparametric Distribution Free UCL Statistics Data do not follow a Discernible Distribution (0.05)

## Nonparametric Distribution Free UCLs

95% CLT UCL	0.0716	95% Jackknife UCL	0.0717
95% Standard Bootstrap UCL	0.0717	95% Bootstrap-t UCL	0.072
95% Hall's Bootstrap UCL	0.0719	95% Percentile Bootstrap UCL	0.0716
95% BCA Bootstrap UCL	0.0719		
90% Chebyshev(Mean, Sd) UCL	0.0764	95% Chebyshev(Mean, Sd) UCL	0.0811
97.5% Chebyshev(Mean, Sd) UCL	0.0877	99% Chebyshev(Mean, Sd) UCL	0.101

### Suggested UCL to Use

95% Chebyshev (Mean, Sd) UCL 0.0811

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

### final\_numeric (brc/timet\_uranium-238)

### General Statistics

Total Number of Observations	120	Number of Distinct Observations	75
		Number of Missing Observations	0
Minimum	0.45	Mean	1.085
Maximum	2.37	Median	1.015
SD	0.373	Std. Error of Mean	0.0341
Coefficient of Variation	0.344	Skewness	1.298

Shapiro Wilk GOF Test

Data Not Normal at 5% Significance Level Lilliefors GOF Test

Data Not Normal at 5% Significance Level

0 0286

### Normal GOF Test

 Shapiro Wilk Test Statistic
 0.895

 5% Shapiro Wilk P Value
 5.103E-12

 Lilliefors Test Statistic
 0.151

 5% Lilliefors Critical Value
 0.0812

 Data Not Normal at 5% Significance Level

### Assuming Normal Distribution

95% Normal UCL	Ū	95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.141	95% Adjusted-CLT UCL (Chen-1995)	1.145
		95% Modified-t UCL (Johnson-1978)	1.142
	Gamma GOF Test		
A-D Test Statistic	1.449	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.752	Data Not Gamma Distributed at 5% Significance Leve	el
K-S Test Statistic	0.106	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.0842	Data Not Gamma Distributed at 5% Significance Leve	el
Data Not Gamm	na Distributed at 5%	Significance Level	
	Gamma Statistics		0 454
K hat (MLE)	9.691	k star (bias corrected MLE)	9.454
I heta hat (MLE)	0.112	Theta star (bias corrected MLE)	0.115
nu hat (MLE)	2326	nu star (bias corrected)	2269
MLE Mean (bias corrected)	1.085	MLE Sd (bias corrected)	0.353
		Approximate Chi Square Value (0.05)	2159
Adjusted Level of Significance	0.048	Adjusted Chi Square Value	2158
Ass	uming Gamma Distri	bution	
95% Approximate Gamma UCL (use when n>=50))	1.14	95% Adjusted Gamma UCL (use when n<50)	1.14
	Lognormal GOF Te	st	
Shapiro Wilk Test Statistic	0.973	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk P Value	0.158	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0861	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.0812	Data Not Lognormal at 5% Significance Level	
Data appear Approx	kimate Lognormal at	5% Significance Level	
	Lognormal Statistic	8	

# Minimum of Logged Data -0.799 Mean of logged Data

Maximum of Logged Data	0.863	SD of logged Data	0.321

Assuming Lognormal Distribution

95% H-UCL	1.14	90% Chebyshev (MVUE) UCL	1.18
95% Chebyshev (MVUE) UCL	1.224	97.5% Chebyshev (MVUE) UCL	1.286
99% Chebyshev (MVUE) UCL	1.406		

## Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

### Nonparametric Distribution Free UCLs

95% CLT UCL	1.141	95% Jackknife UCL	1.141
95% Standard Bootstrap UCL	1.141	95% Bootstrap-t UCL	1.146
95% Hall's Bootstrap UCL	1.146	95% Percentile Bootstrap UCL	1.141
95% BCA Bootstrap UCL	1.144		
90% Chebyshev(Mean, Sd) UCL	1.187	95% Chebyshev(Mean, Sd) UCL	1.233
97.5% Chebyshev(Mean, Sd) UCL	1.297	99% Chebyshev(Mean, Sd) UCL	1.424
:	Suggested UCL to Use		
95% Student's-t UCL	1.141	or 95% Modified-t UCL	1.142

5% Student's-t UCL	1.141	or 95% Modified-t UCL
or 95% H-UCL	1.14	

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

### ProUCL computes and outputs H-statistic based UCLs for historical reasons only. H-statistic often results in unstable (both high and low) values of UCL95 as shown in examples in the Technical Guide. It is therefore recommended to avoid the use of H-statistic based 95% UCLs. Use of nonparametric methods are preferred to compute UCL95 for skewed data sets which do not follow a gamma distribution.

final\_numeric (rz-a\_radium-226)

	General Statistics		
Total Number of Observations	31	Number of Distinct Observations	29
		Number of Missing Observations	0
Minimum	0.0472	Mean	0.952
Maximum	1.72	Median	0.891
SD	0.354	Std. Error of Mean	0.0636
Coefficient of Variation	0.372	Skewness	0.101
	Normal GOF Test		
Shapiro Wilk Test Statistic	0.966	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Normal at 5% Significance Level	
Lilliefors Test Statistic	0.114	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Normal at 5% Significance Level	
Data appea	r Normal at 5% Signi	ficance Level	
Ass	suming Normal Distrit	oution	
95% Normal UCL	-	95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.06	95% Adjusted-CLT UCL (Chen-1995)	1.058
		95% Modified-t UCL (Johnson-1978)	1.06
	Gamma GOF Test		
A-D Test Statistic	1.162	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.748	Data Not Gamma Distributed at 5% Significance Leve	el
K-S Test Statistic	0.179	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.158	Data Not Gamma Distributed at 5% Significance Leve	əl
Data Not Gamm	na Distributed at 5% S	Significance Level	
	Gamma Statistics		
k hat (MLE)	4.548	k star (bias corrected MLE)	4.129
Theta hat (MLE)	0.209	Theta star (bias corrected MLE)	0.231
nu hat (MLE)	282	nu star (bias corrected)	256
MLE Mean (bias corrected)	0.952	MLE Sd (bias corrected)	0.469
· · · · · ·		Approximate Chi Square Value (0.05)	220
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	218.1
Ass	uming Gamma Distri	bution	
mate Gamma UCL (use when n>=50))	1.108	95% Adjusted Gamma UCL (use when n<50)	1.118

95% Approximate Gamma UCL (use when n>=50))

Lognormal GOF Test

Shapiro Wilk Test Statistic	0.669	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data Not Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.233	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.156	Data Not Lognormal at 5% Significance Level	
Data Not Lognormal at 5% Significance Level			

### Lognormal Statistics

Minimum of Logged Data Maximum of Logged Data	-3.053 0.542	Mean of logged Data SD of logged Data	-0.163 0.62
Assur	ning Lognormal Distribution		
95% H-UCL	1.296	90% Chebyshev (MVUE) UCL	1.387
95% Chebyshev (MVUE) UCL	1.552	97.5% Chebyshev (MVUE) UCL	1.781

95% Chebyshev (MVUE) UCL 99% Chebyshev (MVUE) UCL 2.231

### Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

### Nonparametric Distribution Free UCLs

1.057	95% Jackknife UCL	1.06
1.056	95% Bootstrap-t UCL	1.063
1.061	95% Percentile Bootstrap UCL	1.057
1.06		
1.143	95% Chebyshev(Mean, Sd) UCL	1.23
1.35	99% Chebyshev(Mean, Sd) UCL	1.585
	1.057 1.056 1.061 1.06 1.143 1.35	1.057         95% Jackknife UCL           1.056         95% Bootstrap-t UCL           1.061         95% Percentile Bootstrap UCL           1.06         1.143           1.35         95% Chebyshev(Mean, Sd) UCL

## Suggested UCL to Use

95% Student's-t UCL 1.06

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006).

However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (rz-a\_radium-228)

Total Number of Observations	<b>General Sta</b> 31	tistics Number of Distinct Observations Number of Missing Observations	30 0
Minimum	0.46	Mean	1.279
Maximum	2.46	Median	1.18
SD	0.542	Std. Error of Mean	0.0974
Coefficient of Variation	0.424	Skewness	0.51
	Normal GO	F Test	
Shapiro Wilk Test Statistic	0.957	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Normal at 5% Significance Level	
Lilliefors Test Statistic	0.0888	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Normal at 5% Significance Level	
Data appear	<sup>•</sup> Normal at 5%	6 Significance Level	
Ass	uming Normal	Distribution	
95% Normal UCL		95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.445	95% Adjusted-CLT UCL (Chen-1995)	1.449
		95% Modified-t UCL (Johnson-1978)	1.446
	Gamma GO	F Test	
A-D Test Statistic	0.13	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.747	Detected data appear Gamma Distributed at 5% Significance	e Level
K-S Test Statistic	0.0688	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.158	Detected data appear Gamma Distributed at 5% Significance	e Level
Detected data appear 0	Gamma Distrib	outed at 5% Significance Level	
	Gamma Sta	tistics	
k hat (MLE)	5.597	k star (bias corrected MLE)	5.076
Theta hat (MLE)	0.229	Theta star (bias corrected MLE)	0.252
nu hat (MLE)	347	nu star (bias corrected)	314.7
MLE Mean (bias corrected)	1.279	MLE Sd (bias corrected)	0.568
		Approximate Chi Square Value (0.05)	274.6

Adjusted Chi Square Value 272.6

Adjusted Level of Significance 0.0413

Assuming Gamma Distribution

1.466

	Lognormal GOF Test		
Shapiro Wilk Test Statistic	0.974	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0712	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal at 5% Signif	icance Level	
	Lognormal Statistics		0 4 5 4

95% Adjusted Gamma UCL (use when n<50)

1.477

Minimum of Logged Data	-0.///	Mean of logged Data	0.154
Maximum of Logged Data	0.9	SD of logged Data	0.447
Assu	ming Lognormal Distribution		
95% H-UCL	1.504	90% Chebyshev (MVUE) UCL	1.606
95% Chebyshev (MVUE) UCL	1.751	97.5% Chebyshev (MVUE) UCL	1.952
99% Chebyshev (MVUE) UCL	2.348		

## Nonparametric Distribution Free UCL Statistics

Data appear to follow a Discernible Distribution at 5% Significance Level

### Nonparametric Distribution Free UCLs

95% CLT UCL	1.44	95% Jackknife UCL	1.445
95% Standard Bootstrap UCL	1.438	95% Bootstrap-t UCL	1.453
95% Hall's Bootstrap UCL	1.446	95% Percentile Bootstrap UCL	1.444
95% BCA Bootstrap UCL	1.444		
90% Chebyshev(Mean, Sd) UCL	1.572	95% Chebyshev(Mean, Sd) UCL	1.704
97.5% Chebyshev(Mean, Sd) UCL	1.888	99% Chebyshev(Mean, Sd) UCL	2.248

## Suggested UCL to Use

95% Student's-t UCL 1.445

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006).

However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (rz-a\_thorium-228)

	General S	Statistics	
Total Number of Observations	31	Number of Distinct Observations	30
		Number of Missing Observations	0
Minimum	1.16	Mean	1.696
Maximum	2.88	Median	1.69
SD	0.361	Std. Error of Mean	0.0648
Coefficient of Variation	0.213	Skewness	1.023
	Normal G	OF Test	
Shapiro Wilk Test Statistic	0.929	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data Not Normal at 5% Significance Level	
Lilliefors Test Statistic	0.0999	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Normal at 5% Significance Level	
Data appear Appro	ximate Norr	nal at 5% Significance Level	
Ass	uming Norm	al Distribution	
95% Normal UCL		95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.806	95% Adjusted-CLT UCL (Chen-1995)	1.816
		95% Modified-t UCL (Johnson-1978)	1.808
	Gamma G	OF Test	
A-D Test Statistic	0.324	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.745	Detected data appear Gamma Distributed at 5% Significance	e Level
K-S Test Statistic	0.0823	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.157	Detected data appear Gamma Distributed at 5% Significance	e Level
Detected data appear (	Gamma Dist	ributed at 5% Significance Level	

### Gamma Statistics

k hat (MLE)	24.47	k star (bias corrected MLE)	22.12
Theta hat (MLE)	0.0693	Theta star (bias corrected MLE)	0.0767
nu hat (MLE)	1517	nu star (bias corrected)	1372

MLE Mean (bias corrected)	1 696	MLE Sd (bias corrected)	0 361
WEE Wear (blas corrected)	1.000	Approximate Chi Square Value (0.05)	1287
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	1282
Ass	uming Gamma Distril	pution	1.015
95% Approximate Gamma UCL (use when n>=50))	1.809	95% Adjusted Gamma UCL (use when n<50)	1.815
	Lognormal GOF Tes	st	
Shapiro Wilk Test Statistic	0.968	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0781	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal at 5% Sig	nificance Level	
	Lognomial Statistics		
Minimum of Loggod Data		Moon of logged Date	0 500
Minimum of Logged Data	0.148	Mean of logged Data	0.508
Maximum of Logged Data	1.050	SD of logged Data	0.204
Assu	ming Lognormal Dist	ibution	
95% H-UCL	1.811	90% Chebyshev (MVUE) UCL	1.884
95% Chebyshev (MVUE) UCL	1.969	97.5% Chebyshev (MVUE) UCL	2.087
99% Chebyshev (MVUE) UCL	2.319		
Nonparamet	ric Distribution Free	JCL Statistics	
Data appear to follow a D	iscernible Distribution	n at 5% Significance Level	
		•	
Nonpara	ametric Distribution F	ree UCLs	
Nonpara 95% CLT UCL	ametric Distribution F 1.803	ree UCLs 95% Jackknife UCL	1.806
<b>Nonpara</b> 95% CLT UCL 95% Standard Bootstrap UCL	ametric Distribution F 1.803 1.801	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL	1.806 1.821
<b>Nonpara</b> 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL	ametric Distribution F 1.803 1.801 1.836	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL	1.806 1.821 1.802
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL	ametric Distribution F 1.803 1.801 1.836 1.813	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL	1.806 1.821 1.802
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL	ametric Distribution F 1.803 1.801 1.836 1.813 1.891	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL	1.806 1.821 1.802 1.979
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL	1.806 1.821 1.802 1.979 2.341
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Li	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL	1.806 1.821 1.802 1.979 2.341
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL	1.806 1.821 1.802 1.979 2.341
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL se	1.806 1.821 1.802 1.979 2.341
Nonpara 95% CLT UCL 95% Standard Bootstrap UCL 95% BCA Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 95% Student's-t UCL When a data set follows an approxim	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 nate (e.g., normal) dia	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL se	1.806 1.821 1.802 1.979 2.341
Nonpar 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 95% Student's-t UCL When a data set follows an approxir When applicable, it is suggested to use a UCL bas	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 nate (e.g., normal) dised upon a distribution	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL se stribution passing one of the GOF test in (e.g., gamma) passing both GOF tests in ProUCL	1.806 1.821 1.802 1.979 2.341
Nonpar 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 97.5% Student's-t UCL When a data set follows an approxir When applicable, it is suggested to use a UCL bas Note: Suggestions regarding the selection of a 95%	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 nate (e.g., normal) dii sed upon a distributio UCL are provided to I	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL 56 stribution passing one of the GOF test n (e.g., gamma) passing both GOF tests in ProUCL help the user to select the most appropriate 95% UCL.	1.806 1.821 1.802 1.979 2.341
Nonpar 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 95% Student's-t UCL When a data set follows an approxim When applicable, it is suggested to use a UCL bas Note: Suggestions regarding the selection of a 95% Recommendations are base	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 mate (e.g., normal) dii sed upon a distribution UCL are provided to 1 ed upon data size, da	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL 36 stribution passing one of the GOF test in (e.g., gamma) passing both GOF tests in ProUCL help the user to select the most appropriate 95% UCL. ta distribution, and skewness.	1.806 1.821 1.802 1.979 2.341
Nonpar 95% CLT UCL 95% Standard Bootstrap UCL 95% Hall's Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 95% Student's-t UCL When a data set follows an approxir When applicable, it is suggested to use a UCL bas Note: Suggestions regarding the selection of a 95% Recommendations are based These recommendations are based upon the result	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 nate (e.g., normal) dis sed upon a distribution UCL are provided to I de upon data size, da s of the simulation stu	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL 39% stribution passing one of the GOF test in (e.g., gamma) passing both GOF tests in ProUCL help the user to select the most appropriate 95% UCL. ta distribution, and skewness. udies summarized in Singh, Maichle, and Lee (2006).	1.806 1.821 1.802 1.979 2.341
Nonpar 95% CLT UCL 95% Standard Bootstrap UCL 95% Bootstrap UCL 95% BCA Bootstrap UCL 90% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 97.5% Chebyshev(Mean, Sd) UCL 97.5% Student's-t UCL When a data set follows an approxir When applicable, it is suggested to use a UCL bas Note: Suggestions regarding the selection of a 95% Recommendations are base These recommendations are based upon the result However, simulations results will not cover all Real Wo	ametric Distribution F 1.803 1.801 1.836 1.813 1.891 2.101 Suggested UCL to Us 1.806 mate (e.g., normal) dis- sed upon a distribution UCL are provided to I ded upon data size, da s of the simulation stu- rid data sets; for add	ree UCLs 95% Jackknife UCL 95% Bootstrap-t UCL 95% Percentile Bootstrap UCL 95% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL 99% Chebyshev(Mean, Sd) UCL se stribution passing one of the GOF test in (e.g., gamma) passing both GOF tests in ProUCL help the user to select the most appropriate 95% UCL. ta distribution, and skewness. udies summarized in Singh, Maichle, and Lee (2006). itional insight the user may want to consult a statisticiar	1.806 1.821 1.802 1.979 2.341

## final\_numeric (rz-a\_thorium-230)

	•.		
		Number of Missing Observations	0
Minimum	0.509	Mean	1.079
Maximum	1.71	Median	1.07
SD	0.283	Std. Error of Mean	0.0507
Coefficient of Variation	0.262	Skewness	0.392
	Normal G	OF Test	
Shapiro Wilk Test Statistic	0.975	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Normal at 5% Significance Level	
Lilliefors Test Statistic	0.111	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Normal at 5% Significance Level	
Data appea	r Normal at §	5% Significance Level	
Ass	uming Norm	al Distribution	
95% Normal UCL	-	95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.165	95% Adjusted-CLT UCL (Chen-1995)	1.166
		95% Modified-t UCL (Johnson-1978)	1.166
	Gamma G	OF Test	
A-D Test Statistic	0.186	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.745	Detected data appear Gamma Distributed at 5% Significance	e Level
K-S Test Statistic	0.0809	Kolmogorov-Smirnov Gamma GOF Test	
		-	

Number of Distinct Observations

29

**General Statistics** 

31

Total Number of Observations

5% K-S Critical Value Detected data appear	0.158 Gamma Di	Detected data appear Gamma Distributed at 5% Significance stributed at 5% Significance Level	e Level
	Gamma	Statistics	
k hat (MLE)	14.73	k star (bias corrected MLE)	13.32
Theta hat (MLE)	0.0733	Theta star (bias corrected MLE)	0.081
nu hat (MLE)	913.1	nu star (bias corrected)	826.1
MLE Mean (bias corrected)	1.079	MLE Sd (bias corrected)	0.296
х, , , , , , , , , , , , , , , , , , ,		Approximate Chi Square Value (0.05)	760.4
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	756.9
Ass	uming Gan	nma Distribution	
95% Approximate Gamma UCL (use when n>=50))	1.172	95% Adjusted Gamma UCL (use when n<50)	1.178
	Loanorma	al GOF Test	
Shapiro Wilk Test Statistic	0.977	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.0776	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal	at 5% Significance Level	
	Lognorma	al Statistics	
Minimum of Logged Data	-0.675	Mean of logged Data	0.0418
Maximum of Logged Data	0.536	SD of logged Data	0.271
Assu	ming Logn	ormal Distribution	
95% H-UCL	1.181	90% Chebyshev (MVUE) UCL	1.24
95% Chebyshev (MVUE) UCL	1.313	97.5% Chebyshev (MVUE) UCL	1.413
99% Chebyshev (MVUE) UCL	1.611		
Nonparame	tric Distribu	tion Free UCL Statistics	
Data appear to follow a I	Discernible	Distribution at 5% Significance Level	
Nonpar	ametric Dis	tribution Free UCLs	
	4 4 0 0		4 4 0 5

95% CLT UCL	1.163	95% Jackknife UCL	1.165
95% Standard Bootstrap UCL	1.161	95% Bootstrap-t UCL	1.171
95% Hall's Bootstrap UCL	1.171	95% Percentile Bootstrap UCL	1.161
95% BCA Bootstrap UCL	1.165		
90% Chebyshev(Mean, Sd) UCL	1.231	95% Chebyshev(Mean, Sd) UCL	1.3
97.5% Chebyshev(Mean, Sd) UCL	1.396	99% Chebyshev(Mean, Sd) UCL	1.584

### Suggested UCL to Use

95% Student's-t UCL 1.165

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness. These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006).

However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

**General Statistics** 

### final\_numeric (rz-a\_thorium-232)

Total Number of Observations	31	Number of Distinct Observations	28
		Number of Missing Observations	0
Minimum	1.02	Mean	1.503
Maximum	2.07	Median	1.51
SD	0.238	Std. Error of Mean	0.0428
Coefficient of Variation	0.159	Skewness	0.137

### Normal GOF Test

0.99

0.059

0.929

1.576

Shapiro Wilk Test Statistic 5% Shapiro Wilk Critical Value Lilliefors Test Statistic 5% Lilliefors Critical Value Shapiro Wilk GOF Test Data appear Normal at 5% Significance Level Lilliefors GOF Test Data appear Normal at 5% Significance Level

tical Value 0.156 Data ap Data appear Normal at 5% Significance Level Assuming Normal Distribution

## 95% UCLs (Adjusted for Skewness)

95% Normal UCL 95% Student's-t UCL

95% Adjusted-CLT UCL (Chen-1995) 1.574

95% Modified-t UCL (Johnson-1978) 1.576

	Gamma GC	DF Test	
A-D Test Statistic	0.161	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.744	Detected data appear Gamma Distributed at 5% Significanc	e Level
K-S Test Statistic	0.0797	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.157	Detected data appear Gamma Distributed at 5% Significanc	e Level
Detected data appear	Gamma Distri	buted at 5% Significance Level	
	0	-11-11	
			20 54
K nat (MLE)	40.43	K Star (bias corrected MLE)	30.54
Ineta hat (MLE)	0.0372	Theta star (bias corrected MLE)	0.0411
nu nat (MLE)	2507	nu star (blas corrected)	2205
MLE Mean (bias corrected)	1.503	MLE Sd (blas corrected)	0.249
	0.0440	Approximate Chi Square Value (0.05)	2156
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	2150
Ass	suming Gamm	a Distribution	
95% Approximate Gamma UCL (use when n>=50))	1.579	95% Adjusted Gamma UCL (use when n<50)	1.584
	Lognormal G	OF Test	
Shapiro Wilk Test Statistic	0.985	Shapiro Wilk Lognormal GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.09	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal at	5% Significance Level	
	Lognormal S	Statistics	
Minimum of Logged Data	0.0198	Mean of logged Data	0.395
Maximum of Logged Data	0.728	SD of logged Data	0.161
Assu	imina Loanorm	nal Distribution	
95% H-UCI	1 582	90% Chebyshev (MVUF) UC	1 635
95% Chebyshey (MVUE) UCI	1 694	97.5% Chebyshev (MV/LIE) LICI	1 777
99% Chebyshev (MVUE) UCL	1.939		1
Nonparame	tric Distributio	n Free UCL Statistics	
Data appear to follow a I	Discernible Dis	tribution at 5% Significance Level	
Manna	o no otnio. Diotnik		
			1 576
95% CLT UCL	1.575	95% Jackkille UCL	1.576
95% Standard Bootstrap UCL	1.372	95% BOOISIIAP-LUCL	1.570
95% Hall's Bootstrap UCL	1.576	95% Percentile Bootstrap UCL	1.572
95% BCA Bootstrap UCL	1.575		1 000
90% Chebyshev(Mean, Sd) UCL	1.631	95% Chebysnev(Mean, Sd) UCL	1.689
97.5% Chebyshev(Mean, Sd) UCL	1.77	99% Chebyshev(Mean, Sd) UCL	1.929
	Suggested LIC	CL to Lise	
95% Student's-t UCL	1.576		
Note: Suggestions regarding the selection of a 95%	UCL are provi	ided to help the user to select the most appropriate 95% UCL.	
Recommendations are bas	ed upon data s	size, data distribution, and skewness.	
These recommendations are based upon the resul	ts of the simula	ation studies summarized in Singh, Maichle, and Lee (2006).	
However, simulations results will not cover all Real We	orld data sets;	for additional insight the user may want to consult a statisticiar	1.
final numeric (rz-a uranium-234)			
	Conoral St	atistics	

	General Statistics		
Total Number of Observations	31	Number of Distinct Observations	27
		Number of Missing Observations	0
Minimum	0.391	Mean	1.062
Maximum	1.74	Median	1.02
SD	0.297	Std. Error of Mean	0.0534
Coefficient of Variation	0.28	Skewness	0.571
	Normal GOF Test		
Shapiro Wilk Test Statistic	0.942	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data appear Normal at 5% Significance Level	
Lilliefors Test Statistic	0.135	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data appear Normal at 5% Significance Level	
		· · · · · · · · · · · · · · · · · · ·	

Data appear Normal at 5% Significance Level

Assuming Normal Distribution

95% Normal UCL		95% UCLs (Adjusted for Skewness)	
95% Student's-t UCL	1.153	95% Adjusted-CLT UCL (Chen-1995)	1.156
		95% Modified-t UCL (Johnson-1978)	1.154
	Gamma G	GOF Test	
A-D Test Statistic	0.518	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0.746	Detected data appear Gamma Distributed at 5% Significanc	e Level
K-S Test Statistic	0.102	Kolmogorov-Smirnov Gamma GOF Test	
5% K-S Critical Value	0.158	Detected data appear Gamma Distributed at 5% Significanc	e Level
Detected data appear	Gamma Dis	tributed at 5% Significance Level	
	Gamma S	Statistics	
k hat (MLE)	12.85	k star (bias corrected MLE)	11.62
Theta hat (MLE)	0.0827	Theta star (bias corrected MLE)	0.0914
nu hat (MLE)	796.5	nu star (bias corrected)	720.7
MLE Mean (bias corrected)	1.062	MLE Sd (bias corrected)	0.312
		Approximate Chi Square Value (0.05)	659.4
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	656.2
		· · · · · · · · · · · · · · · · · · ·	
Ass	suming Gam	ma Distribution	
95% Approximate Gamma UCL (use when n>=50))	1.161	95% Adjusted Gamma UCL (use when n<50)	1.167
	Lognormal	GOE Test	
Shaniro Wilk Test Statistic	0 033	Shaniro Wilk Lognormal GOE Test	
5% Shapiro Wilk Critical Value	0.000	Data appear Lognormal at 5% Significance Level	
Lilliefors Test Statistic	0.525	Lilliefors Lognormal GOF Test	
5% Lilliefors Critical Value	0.114	Data appear Lognormal at 5% Significance Level	
Data appear	Lognormal a	t 5% Significance Level	
	Lognormal	Statistics	
Minimum of Logged Data	-0.939	Mean of logged Data	0.0212
Maximum of Logged Data	0.554	SD of logged Data	0.293
Assi	imina Loanoi	rmal Distribution	
95% H-UCL	1.174	90% Chebyshev (MVUE) UCL	1.236
95% Chebyshev (MVUE) UCL	1.313	97.5% Chebyshev (MVUE) UCL	1.421
99% Chebyshev (MVUE) UCL	1.632		
Nonparame Data appear to follow a l	tric Distributi	on Free UCL Statistics	
Data appear to follow a t			
Nonpar	ametric Dist	ibution Free UCLs	
95% CLT UCL	1.15	95% Jackknife UCL	1.153
95% Standard Bootstrap UCL	1.149	95% Bootstrap-t UCL	1.161
95% Hall's Bootstrap UCL	1.161	95% Percentile Bootstrap UCL	1.15
95% BCA Bootstrap UCL	1.154		
90% Chebyshev(Mean, Sd) UCL	1.223	95% Chebyshev(Mean, Sd) UCL	1.295
97.5% Chebyshev(Mean, Sd) UCL	1.396	99% Chebyshev(Mean, Sd) UCL	1.594
	Queente d l		
	Suggested l		

95% Student's-t UCL

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

1.153

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

## final\_numeric (rz-a\_uranium-235)

### General Statistics

31

Minimum -0.0766 Maximum 0.203 SD 0.0442 Coefficient of Variation 0.862

## Normal GOF Test

Shapiro Wilk Test Statistic	0.892
5% Shapiro Wilk Critical Value	0.929
Lilliefors Test Statistic	0.132

Shapiro Wilk GOF Test							
Data Not Normal at 5% Significance Level							
Lilliefors GOF Test							

Number of Distinct Observations

Number of Missing Observations

29

0

0.0513

0.0486

0.00794

0.549

Mean

Median

Skewness

Std. Error of Mean

5% Lilliefors Critical Value 0.156 Data appear Normal at 5% Significance Level Data appear Approximate Normal at 5% Significance Level

Assuming Normal Distribution

95% Normal UCL 95% Student's-t UCL 0.0648 95% UCLs (Adjusted for Skewness)

 95% Adjusted-CLT UCL (Chen-1995)
 0.0652

 95% Modified-t UCL (Johnson-1978)
 0.0649

Gamma Statistics Not Available Lognormal Statistics Not Available

Nonparametric Distribution Free UCL Statistics Data appear to follow a Discernible Distribution at 5% Significance Level

Nonparametric Distribution Free UCLs

95% CLT UCL	0.0644	95% Jackknife UCL	0.0648
95% Standard Bootstrap UCL	0.0642	95% Bootstrap-t UCL	0.0658
95% Hall's Bootstrap UCL	0.0689	95% Percentile Bootstrap UCL	0.0643
95% BCA Bootstrap UCL	0.0648		
90% Chebyshev(Mean, Sd) UCL	0.0751	95% Chebyshev(Mean, Sd) UCL	0.0859
97.5% Chebyshev(Mean, Sd) UCL	0.101	99% Chebyshev(Mean, Sd) UCL	0.13

Suggested UCL to Use

95% Student's-t UCL 0.0648

When a data set follows an approximate (e.g., normal) distribution passing one of the GOF test When applicable, it is suggested to use a UCL based upon a distribution (e.g., gamma) passing both GOF tests in ProUCL

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006).

However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

### final\_numeric (rz-a\_uranium-238)

	General Statis	tics	
Total Number of Observations	31	Number of Distinct Observations	26
		Number of Missing Observations	0
Minimum	0.361	Mean	1.034
Maximum	1.59	Median	1.01
SD	0.214	Std. Error of Mean	0.0384
Coefficient of Variation	0.207	Skewness	-0.157
	Normal GOF T	est	
Shapiro Wilk Test Statistic	0.898	Shapiro Wilk GOF Test	
5% Shapiro Wilk Critical Value	0.929	Data Not Normal at 5% Significance Level	
Lilliefors Test Statistic	0.186	Lilliefors GOF Test	
5% Lilliefors Critical Value	0.156	Data Not Normal at 5% Significance Level	
Data Not	Normal at 5% Sig	nificance Level	
۵۹	suming Normal Di	stribution	
95% Normal UCI	Saming Horman Bi	95% UCLs (Adjusted for Skewness)	
95% Student's-t UC	1 099	95% Adjusted-CLT UCL (Chen-1995)	1 096
		95% Modified-t UCL (Johnson-1978)	1.099
	Gamma GOE 1	Feet	
A-D Test Statistic	1 524	Anderson-Darling Gamma GOF Test	
5% A-D Critical Value	0 745	Data Not Gamma Distributed at 5% Significance Leve	4
K-S Test Statistic	0.218	Kolmogorov-Smirnov Gamma GOF Test	,
5% K-S Critical Value	0.157	Data Not Gamma Distributed at 5% Significance Leve	el
Data Not Gamr	na Distributed at 5	i% Significance Level	
	Gamma Statis	tics	
k hat (MLE)	20.22	k star (bias corrected MLE)	18 29
Theta hat (MLE)	0.0511	Theta star (bias corrected MLE)	0.0565
nu hat (MLE)	1254	nu star (bias corrected)	1134
MI F Mean (bias corrected)	1 034	MI E Sd (bias corrected)	0 242
		Approximate Chi Square Value (0.05)	1057
Adjusted Level of Significance	0.0413	Adjusted Chi Square Value	1053
Ass	suming Gamma D	istribution	

95% Adjusted Gamma UCL (use when n<50)

1.113

Assuming Gamma Distribution
0)) 1.109

95% Approximate Gamma UCL (use when n>=50))

	Lognormal GO	F Test				
Shapiro Wilk Test Statistic	0.776	Shapiro Wilk Lognormal GOF Test				
5% Shapiro Wilk Critical Value	0.929	Data Not Lognormal at 5% Significance Level				
Lilliefors Test Statistic	0.242	Lilliefors Lognormal GOF Test				
5% Lilliefors Critical Value	0.156	Data Not Lognormal at 5% Significance Level				
Data Not Lo	ognormal at 5%	Significance Level				
	l ognormal Sta	tietice				
Minimum of Logged Data	_1 010	Mean of logged Data	0 00813			
Maximum of Logged Data	0.464	SD of logged Data	0.00013			
Maximum of Logged Data	0.404	SD of logged Data	0.244			
Assu	ming Lognormal	Distribution				
95% H-UCL	1.123	90% Chebyshev (MVUE) UCL	1.175			
95% Chebyshev (MVUE) UCL	1.238	97.5% Chebyshev (MVUE) UCL	1.325			
99% Chebyshev (MVUE) UCL	1.495					
Nonparamet	tric Distribution I	Free UCL Statistics				
Data do not fo	ollow a Discernib	le Distribution (0.05)				
Nonpara	ametric Distribut	ion Free UCLs				
95% CLT UCL	1.097	95% Jackknife UCL	1.099			
95% Standard Bootstrap UCL	1.096	95% Bootstrap-t UCL	1.099			
95% Hall's Bootstrap UCL	1.104	95% Percentile Bootstrap UCL	1.095			

95% Hall's Bootstrap UCL	1.104	95% Percentile Bootstrap UCL	1.095
95% BCA Bootstrap UCL	1.093		
90% Chebyshev(Mean, Sd) UCL	1.149	95% Chebyshev(Mean, Sd) UCL	1.201
97.5% Chebyshev(Mean, Sd) UCL	1.273	99% Chebyshev(Mean, Sd) UCL	1.415

# Suggested UCL to Use

95% Student's-t UCL 1.099 or 95% Modified-t UCL 1.099

Note: Suggestions regarding the selection of a 95% UCL are provided to help the user to select the most appropriate 95% UCL. Recommendations are based upon data size, data distribution, and skewness.

These recommendations are based upon the results of the simulation studies summarized in Singh, Maichle, and Lee (2006). However, simulations results will not cover all Real World data sets; for additional insight the user may want to consult a statistician.

Note: For highly negatively-skewed data, confidence limits (e.g., Chen, Johnson, Lognormal, and Gamma) may not be reliable. Chen's and Johnson's methods provide adjustments for positvely skewed data sets.

Interim Report Identification of COPCs and Decision Units for OU-1 Soils, Revision 1 Nevada Environmental Response Trust Site Henderson, Nevada

> APPENDIX J DECISION UNIT-SPECIFIC BACKGROUND EVALUATION FOR METALS AND RADIONUCLIDES

## TABLE J-1. Summary Statistics for Metals in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs) Nevada Environmental Response Trust Site Henderson, Nevada

Oh and a share	Location	No. of Samples	No. of Detects	% Detects	Non-Detects (mg/kg)		Detects (mg/kg)				Shapiro-Wilk Test		
Name					Minimun	Maximum	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal (p-value)
Arsenic	BRC/TIMET Regional Background	120	120	100%	NA	NA	2.1	3.9	4.1	7.2	1.1	0.002	0.2
	RZ-A Background	31	31	100%	NA	NA	1.6	2.4	2.4	4.3	0.54	0.02	0.5
	DU-1	457	457	100%	NA	NA	0.58	3.2	3.6	34	2.3	<0.001	<0.001
	DU-2	48	48	100%	NA	NA	1.2	2.8	2.9	6.3	0.89	<0.001	0.3
	DU-3	66	66	100%	NA	NA	1.7	4.5	4.9	11	1.8	<0.001	0.9
Chromium VI	BRC/TIMET Regional Background	104	0	0%	0.25	0.25	NA	NA	NA	NA	NA	NA	NA
	RZ-A Background	31	1	3.2%	0.41	0.43	0.29	0.29	0.29	0.29	NA	<0.001	<0.001
	DU-1	154	34	22%	0.11	0.49	0.13	1.3	7.9	106	21	<0.001	<0.001
	DU-2	36	6	17%	0.11	0.43	0.11	0.39	0.83	2.4	0.95	<0.001	<0.001
	DU-3	25	8	32%	0.11	0.44	0.12	0.54	0.54	1.3	0.36	<0.001	0.003
Cobalt	BRC/TIMET Regional Background	120	120	100%	NA	NA	3.7	8.3	8.2	16	2.5	0.05	0.01
	RZ-A Background	31	31	100%	NA	NA	5.4	7.3	7.3	9.1	0.76	0.5	0.4
	DU-1	249	249	100%	NA	NA	4.6	7.8	14	284	32	<0.001	<0.001
	DU-2	36	36	100%	NA	NA	5.4	7.1	7.1	8.8	0.95	0.3	0.2
	DU-3	25	25	100%	NA	NA	3.2	6.7	6.4	8.4	1.4	0.09	0.004
Manganese	BRC/TIMET Regional Background	120	120	100%	NA	NA	151	419	425	1,090	135	<0.001	0.003
	RZ-A Background	31	31	100%	NA	NA	262	360	366	537	61	0.03	0.4
	DU-1	347	347	100%	NA	NA	133	407	1,070	29,200	2,400	<0.001	<0.001
	DU-2	36	36	100%	NA	NA	249	368	395	1,290	168	<0.001	<0.001
	DU-3	27	27	100%	NA	NA	151	302	347	1,500	251	<0.001	0.007
Palladium	BRC/TIMET Regional Background	104	104	100%	NA	NA	0.14	0.40	0.46	1.5	0.24	<0.001	0.5
	RZ-A Background	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	DU-1	18	0	0%	0.048	0.060	NA	NA	NA	NA	NA	0.5	0.5
	DU-2	7	7	100%	NA	NA	0.33	0.44	0.43	0.52	0.072	0.7	0.6
	DU-3	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Thallium	BRC/TIMET Regional Background	120	42	35%	0.20	1.1	0.10	1.1	0.92	1.8	0.62	<0.001	<0.001
	RZ-A Background	31	31	100%	NA	NA	0.071	0.092	0.11	0.19	0.033	<0.001	0.003
	DU-1	199	133	67%	0.21	0.28	0.060	0.10	0.22	8.4	0.73	<0.001	<0.001
	DU-2	36	22	61%	0.10	0.26	0.074	0.10	0.11	0.15	0.025	0.04	0.001
	DU-3	25	21	84%	0.22	0.24	0.054	0.095	0.097	0.18	0.028	0.1	0.6
#### TABLE J-1. Summary Statistics for Metals in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs) Nevada Environmental Response Trust Site Henderson, Nevada

Chemical	Location	No. of	No. of Detects	% Detects	Non-Dete	cts (mg/kg)		D	etects (m	g/kg)		Shapiro-Wilk Test	
Name		Samples			Minimun	Maximum	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal (p-value)
Zirconium	BRC/TIMET Regional Background	104	104	100%	NA	NA	60	125	126	179	27	0.3	<0.001
	RZ-A Background	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	DU-1	18	18	100%	NA	NA	15	22	22	31	4.6	0.2	0.5
	DU-2	7	7	100%	NA	NA	21	22	22	25	1.5	0.8	0.9
	DU-3	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

#### Notes:

bgs = below ground surface

mg/kg = milligram per kilogram

DU = Decision unit

NA = value not available

*p*-values < 0.01 are shown in italic.

RZ-A Background dataset is from RZ-A, excluding the six borings in LOU 62.

Shapiro Wilk tests use 1/2 the detection limit (DL) for non-detects.

#### TABLE J-2. Background Comparisons for Metals in Decision Unit Soils (0-10 feet bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

	Location		t-test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
Chemical Name	Location	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Arsenic	DU-1	NP	1	1	1	1	0.05	No
	DU-2	LN	0.001	0.001	0.001	0.002	0.4	Yes
	DU-3	LN	0.001	0.002	0.002	0.05	0.005	Yes
Chromium VI	DU-1	NP	0.02	<0.001	<0.001	<0.001	NA	LDF
	DU-2	NP	0.2	0.8	0.8	0.2	0.8	LDF
	DU-3	NP	0.001	<0.001	<0.001	<0.001	NA	LDF
Cobalt	DU-1	NP	0.002	0.002	0.6	0.8	<0.001	Yes
	DU-2	N, LN	0.9	0.9	0.9	0.4	1	No
	DU-3	N	1	1	1	1	1	No
Manganese	DU-1	NP	<0.001	<0.001	0.1	<0.001	<0.001	Yes
	DU-2	NP	0.2	0.2	0.4	0.5	0.5	No
	DU-3	NP	0.9	1	1	1	0.2	No
Palladium	DU-1	N, LN	1	1	1	1	1	LDF
	DU-2	NA	NA	NA	NA	NA	NA	NA
	DU-3	NA	NA	NA	NA	NA	NA	NA
Thallium	DU-1	NP	1	1	1	1	0.6	No
	DU-2	N	0.9	0.9	0.02	0.7	1	Yes
	DU-3	N, LN	1	1	1	1	1	No

#### TABLE J-2. Background Comparisons for Metals in Decision Unit Soils (0-10 feet bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

	Location	Distribution	<i>t</i> -test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for	
Chemical Name	Location	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?	
Zirconium	DU-1	N, LN	1	1	1	1	1	No	
	DU-2	NA	NA	NA	NA	NA	NA	NA	
	DU-3	NA	NA	NA	NA	NA	NA	NA	

#### Notes:

bgs = below ground surface

DU = Decision unit

LDF = Low detection frequency (<25%) in either site or background datasets. Background comparison results may not be applicable.

NA = value not available

*p*-values in italics indicate p < 0.025

Background comparison tests use 1/2 the detection limit (DL) for non-detects in the parametric test (t-test) and the DL for non-parametric tests (Gehan test, quantile test, and slippage test).

RZ-A background dataset (excluding the six borings in LOU 62) is used for the southern portion of the Site (DU-2). BRC/TIMET regional background dataset is used for the northern portion of the Site (DU-1 and DU-3).

For large sample size (>100), final background determination is based only on non-parametric testing results. And for small sample size (<=100), final background determination is based on both parametric and non-parametric testing results.

Distibution:

N = Study area data and background data consistent with normal distribution

LN = Study area data and background data consistent with log-normal distribution

NP = Study area data or background data is not consistent with both normal distribution and log-normal distribution.

## TABLE J-3. Summary Statistics for Radionuclides in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs)Nevada Environmental Response Trust SiteHenderson, Nevada

	Chamical		No. of	No. of	0/			Detects	(pCi/g)		Shapiro-Wilk Test	
Chain	Name	Location	Samples	Detects	70 Detects	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal ( <i>p</i> -value)
Uranium-238	Uranium-238	BRC/TIMET Regional Background	120	120	100%	0.45	1.0	1.1	2.4	0.37	<0.001	0.1
		RZ-A Background	31	31	100%	0.36	1.0	1.0	1.6	0.21	0.004	<0.001
		DU-1	147	147	100%	0.24	0.95	1.0	2.5	0.31	<0.001	<0.001
		DU-2	36	36	100%	0.34	0.96	1.0	1.7	0.25	0.05	<0.001
		DU-3	22	22	100%	0.38	1.1	1.3	3.3	0.75	0.03	0.3
	Uranium-234	BRC/TIMET Regional Background	120	120	100%	0.47	0.99	1.1	2.8	0.46	<0.001	<0.001
		RZ-A Background	31	31	100%	0.39	1.0	1.1	1.7	0.30	0.09	0.04
		DU-1	129	129	100%	0.27	0.98	1.1	2.8	0.37	<0.001	<0.001
		DU-2	36	36	100%	0.77	1.0	1.1	1.9	0.27	<0.001	0.01
		DU-3	22	22	100%	0.37	1.3	1.4	3.4	0.80	0.2	0.3
	Thorium-230	BRC/TIMET Regional Background	120	120	100%	0.66	1.2	1.2	3.0	0.38	<0.001	0.07
		RZ-A Background	31	31	100%	0.51	1.1	1.1	1.7	0.28	0.7	0.7
		DU-1	147	147	100%	0.43	1.1	1.2	4.3	0.38	<0.001	<0.001
		DU-2	36	36	100%	0.74	1.1	1.1	2.0	0.22	<0.001	0.01
		DU-3	22	22	100%	0.53	1.3	1.6	3.3	0.77	0.03	0.3
	Radium-226	BRC/TIMET Regional Background	104	104	100%	0.49	1.1	1.1	2.4	0.35	<0.001	0.4
		RZ-A Background	31	31	100%	0.047	0.89	0.95	1.7	0.35	0.4	<0.001
		DU-1	147	147	100%	0.20	0.91	0.93	2.5	0.36	<0.001	0.2
		DU-2	36	36	100%	0.46	0.89	0.86	1.3	0.23	0.3	0.06
		DU-3	25	25	100%	0.45	1.2	1.2	2.0	0.43	0.8	0.6
Thorium-232	Thorium-232	BRC/TIMET Regional Background	120	120	100%	1.1	1.6	1.6	2.2	0.27	0.03	0.07
		RZ-A Background	31	31	100%	1.0	1.5	1.5	2.1	0.24	1	1
		DU-1	147	147	100%	0.66	1.6	1.6	2.5	0.32	0.2	<0.001
		DU-2	36	36	100%	1.1	1.4	1.5	2.3	0.32	0.01	0.2
		DU-3	22	22	100%	0.54	1.6	1.5	2.4	0.42	0.1	0.004
	Thorium-228	BRC/TIMET Regional Background	120	120	100%	1.1	1.7	1.7	2.3	0.28	0.03	0.004
		RZ-A Background	31	31	100%	1.2	1.7	1.7	2.9	0.36	0.03	0.5
		DU-1	147	147	100%	0.48	1.8	1.8	2.8	0.36	<0.001	<0.001
		DU-2	36	36	100%	1.2	1.7	1.6	2.2	0.23	0.5	0.2
		DU-3	22	22	100%	0.78	1.6	1.7	3.0	0.47	0.2	0.2
	Radium-228	BRC/TIMET Regional Background	84	84	100%	0.95	2.0	1.9	2.9	0.40	0.7	0.06
		RZ-A Background	31	31	100%	0.46	1.2	1.3	2.5	0.54	0.3	0.8
		DU-1	147	147	100%	0	1.2	1.2	3.3	0.50	0.01	NA
		DU-2	36	36	100%	0.32	1.5	1.5	2.4	0.49	0.5	0.001
		DU-3	25	25	100%	0.45	1.2	1.2	1.9	0.44	0.4	0.2

#### TABLE J-3. Summary Statistics for Radionuclides in Background (RZ-A and BRC/TIMET Regional) Soils and Decision Unit Soils (0-10 feet bgs) Nevada Environmental Response Trust Site Henderson, Nevada

	Chomical		No. of	No. of	9/			Detects (	(pCi/g)		Shapiro	-Wilk Test
Chain	Name	Location	Samples	Detects	Detects	Minimum	Median	Mean	Maximum	Standard Deviation	Normal (p-value)	Lognormal (p-value)
Uranium-235	Uranium-235	BRC/TIMET Regional Background	120	120	100%	0	0.059	0.066	0.21	0.038	<0.001	NA
		RZ-A Background	31	31	100%	-0.077	0.049	0.051	0.20	0.044	0.003	NA
		DU-1	129	129	100%	-0.026	0.053	0.061	0.25	0.041	<0.001	NA
		DU-2	36	36	100%	-0.029	0.047	0.056	0.21	0.048	0.02	NA
		DU-3	22	22	100%	0.0047	0.075	0.076	0.17	0.047	0.6	0.02

#### Notes:

bgs = below ground surface

pCi/g = picocurie per gram

DU = Decision unit

NA = value not available

*p*-values < 0.01 are shown in italic.

RZ-A Background dataset is from RZ-A, excluding the six borings in LOU 62.

Shapiro Wilk tests use 1/2 the detection limit (DL) for non-detects.

#### TABLE J-4. Background Comparisons for Radionuclides in Decision Unit Soils (0-10 ft bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

Chain	Chemical	Leastion	Distribution	<i>t</i> -test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
Chain	Name	Location	Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Uranium-238	Uranium-238	DU-1	NP	1	1	0.9	1	0.6	No
		DU-2	N	0.6	0.6	0.8	0.5	0.5	No
		DU-3	N, LN	0.09	0.3	0.08	0.1	0.02	Yes
	Uranium-234	DU-1	NP	0.8	0.8	0.6	0.7	1	No
		DU-2	LN	0.2	0.1	0.2	0.8	0.3	No
		DU-3	N, LN	0.04	0.1	0.02	0.01	0.2	Yes
	Thorium-230	DU-1	NP	1	1	1	1	0.6	No
		DU-2	LN	0.5	0.4	0.5	1	0.5	No
		DU-3	N, LN	0.02	0.03	0.03	0.01	0.2	Yes
	Radium-226	DU-1	LN	1	1	1	1	0.6	No
		DU-2	N, LN	0.9	0.6	0.8	1	1	No
		DU-3	N, LN	0.2	0.3	0.2	0.09	1	No
Thorium-232	Thorium-232	DU-1	N	0.6	0.7	0.4	0.6	0.3	No
		DU-2	NA	0.5	0.6	0.8	0.3	0.1	No
		DU-3	NA	0.8	0.9	0.8	0.9	0.02	Yes
	Thorium-228	DU-1	NP	0.003	0.02	0.005	0.008	<0.001	Yes
		DU-2	N, LN	0.8	0.8	0.8	1	1	No
		DU-3	N, LN	0.6	0.7	0.7	0.9	0.02	Yes
	Radium-228	DU-1	NP	1	1	1	1	0.6	No
		DU-2	NA	0.07	0.07	0.04	0.3	1	No
		DU-3	NA	1	1	1	1	1	No

#### TABLE J-4. Background Comparisons for Radionuclides in Decision Unit Soils (0-10 ft bgs)

Nevada Environmental Response Trust Site

#### Henderson, Nevada

Chain	Chemical	Location	Distribution	t-test	<i>t-</i> test (logged data)	Gehan Test	Quantile Test (0.8)	Slippage Test	Fail Statistical Testing for
Chain	Name		Distribution	(p-value)	(p-value)	(p-value)	(p-value)	(p-value)	Background Consistency?
Uranium-235	Uranium-235	DU-1	NA	0.8	0.5	0.9	1	0.3	No
		DU-2	NA	0.3	0.6	0.4	0.5	0.5	No
		DU-3	NA	0.2	0.3	0.2	0.1	1	No

#### Notes:

bgs = below ground surface

DU = Decision unit

LDF = Low detection frequency (<25%) in either site or background datasets. Background comparison results may not be applicable.

*p*-values in italics indicate p < 0.025

Background comparison tests use 1/2 the detection limit (DL) for non-detects in the parametric test (t-test) and the DL for non-parametric tests (Gehan test, quantile test, and slippage test).

RZ-A background dataset (excluding the six borings in LOU 62) is used for the southern portion of the Site (DU-2). BRC/TIMET regional background dataset is used for the northern portion of the Site (DU-1 and DU-3).

For large sample size (>100), final background determination is based only on non-parametric testing results. And for small sample size (<=100), final background determination is based on both parametric and non-parametric testing results.

#### Distibution:

N = Study area data and background data consistent with normal distribution

LN = Study area data and background data consistent with log-normal distribution

NP = Study area data or background data is not consistent with both normal distribution and log-normal distribution.

# Table J-5A. Equivalence Test for Secular Equilibrium of Uranium Decay Series (U-238 Chain)1Nevada Environmental Response Trust Remediation Project SiteHenderson, Nevada

Location	n valuo	Conclusion <sup>2</sup>	Dolta	Sample	Number	Analyta	Mean Proportions	95% Confi	d. Intervals	Chitta <sup>5</sup>
Location	p-value	Conclusion	Della	Size <sup>3</sup>	Missing⁴	Analyte	of Radioactivity	Lower	Upper	Shins
BRC/TIMET	<0.0001	in Secular	0.1	104	16	Ra-226	0.2401	0.2272	0.2529	0
Regional		Equilibrium				Th-230	0.2720	0.2608	0.2831	0
Баскугоини						U-234	0.2448	0.2341	0.2555	0
						U-238	0.2431	0.2339	0.2524	0
RZ-A	<0.0001	in Secular	0.1	31	0	Ra-226	0.2267	0.1909	0.2625	0
Background		Equilibrium				Th-230	0.2626	0.2314	0.2939	0
						U-234	0.2572	0.2283	0.2861	0
						U-238	0.2534	0.2276	0.2793	0
DU-1	<0.0001	in Secular	0.1	129	18	Ra-226	0.2215	0.1996	0.2434	0
		Equilibrium				Th-230	0.2831	0.2677	0.2985	0
						U-234	0.2551	0.2427	0.2674	0
						U-238	0.2402	0.2279	0.2526	0
DU-2	<0.0001	in Secular	0.1	36	0	Ra-226	0.2108	0.1865	0.2351	0
		Equilibrium				Th-230	0.2639	0.2460	0.2819	0
						U-234	0.2750	0.2560	0.2941	0
						U-238	0.2502	0.2276	0.2728	0
DU-3	0.0352	in Secular	0.1	22	3	Ra-226	0.2317	0.1591	0.3043	0
		Equilibrium				Th-230	0.2920	0.2518	0.3322	0
						U-234	0.2481	0.1999	0.2963	0
						U-238	0.2283	0.1872	0.2693	0

#### Notes:

DU = Decision unit

1. Analyzed in top 10 feet bgs using the EnviroGISdT software tool from Neptune & Company, Inc.

2. Tool states "in Secular Equilibrium" if the computed *p*-value is less than a standard significance level of 0.05.

3. Sample dataset includes field duplicates

4. Count of samples for which one or more results are unavailable. These samples are not counted in the sample size and are not included in the secular equilibrium calculation.

5. Data Shift - Lists the values of the data shift utililzed by the tool in case of negative radioactivity measurements. All measurements values for that radioisotope are shifted upwards by the shift value so that all values are non-negative. A zero shift value indicates lack of negative measurements.

# Table J-5B. Equivalence Test for Secular Equilibrium of Thorium Decay Series (Th-232 Chain)Nevada Environmental Response Trust Remediation Project SiteHenderson, Nevada

Location	p-value	Conclusion <sup>2</sup>	Delta	Sample	Number	Analvte	Mean Proportions	95% Confi	d. Intervals	Shifts⁵
	P	Constant		Size <sup>°</sup>	Missing⁴	· · · · · <b>,</b> · ·	of Radioactivity	Lower	Upper	onno
BRC/TIMET	<0.0001	in Secular	0.1	84	36	Ra-228	0.3599	0.3446	0.3752	0
Regional		Equilibrium				Th-228	0.3270	0.3174	0.3366	0
Background						Th-232	0.3130	0.3039	0.3222	0
RZ-A	0.164	Not in Secular	0.1	31	0	Ra-228	0.2779	0.2341	0.3218	0
Background		Equilibrium				Th-228	0.3808	0.3557	0.4060	0
						Th-232	0.3413	0.3141	0.3684	0
DU-1	0.3894	Not in Secular	0.1	147	0	Ra-228	0.2610	0.2393	0.2828	0
		Equilibrium				Th-228	0.3901	0.3757	0.4046	0
						Th-232	0.3489	0.3377	0.3600	0
DU-2	<0.0001	in Secular	0.1	36	0	Ra-228	0.3131	0.2785	0.3478	0
		Equilibrium				Th-228	0.3592	0.3351	0.3832	0
						Th-232	0.3277	0.3085	0.3469	0
DU-3	0.3869	Not in Secular	0.1	22	3	Ra-228	0.2720	0.1943	0.3496	0
		Equilibrium				Th-228	0.3790	0.3425	0.4155	0
						Th-232	0.3491	0.3022	0.3959	0

#### Notes:

DU = Decision unit

1. Analyzed in top 10 feet bgs using the EnviroGISdT software tool from Neptune & Company, Inc.

2. Tool states "in Secular Equilibrium" if the computed p-value is less than a standard significance level of 0.05.

3. Sample dataset includes field duplicates

4. Count of samples for which one or more results are unavailable. These samples are not counted in the sample size and are not included in the secular equilibrium calculation.

5. Data Shift - Lists the values of the data shift utililzed by the tool in case of negative radioactivity measurements. All measurements values for that radioisotope are shifted upwards by the shift value so that all values are non-negative. A zero shift value indicates lack of negative measurements.

Table J-6. Correlation Matrices for the Uranium Decay Series and the Thorium Decay SeriesNevada Environmental Response Trust Remediation Project SiteHenderson, Nevada

	Uranium Decay Chain											
Correl.	Ra-226	Th-230	U-234	U-238								
Ra-226	1	0.663	0.691	0.707								
Th-230	0.663	1	0.762	0.756								
U-234	0.691	0.762	1	0.880								
U-238	0.707	0.756	0.880	1								

#### i) BRC/TIMET Regional Background Soils

Thorium Decay Chain										
Correl.	Ra-228	Th-228	Th-232							
Ra-228	1	0.297	0.305							
Th-228	0.297	1	0.765							
Th-232	0.305	0.765	1							

#### ii) RZ-A Background Soils

	Uranium Decay Chain											
Correl.	Ra-226	Th-230	U-234	U-238								
Ra-226	1	0.680	0.313	0.321								
Th-230	0.680	1	0.351	0.271								
U-234	0.313	0.351	1	0.886								
U-238	0.321	0.271	0.886	1								

Thorium Decay Chain				
Correl.	Ra-228	Th-228	Th-232	
Ra-228	1	0.297	0.119	
Th-228	0.297	1	0.627	
Th-232	0.119	0.627	1	

#### iii) DU-1 Soils

Uranium Decay Chain				
Correl.	Ra-226	Th-230	U-234	U-238
Ra-226	1	0.295	0.279	0.237
Th-230	0.295	1	0.594	0.547
U-234	0.279	0.594	1	0.854
U-238	0.237	0.547	0.854	1

Thorium Decay Chain				
Correl.	Ra-228	Th-228	Th-232	
Ra-228	1	-0.104	0.040	
Th-228	-0.104	1	0.653	
Th-232	0.040	0.653	1	

Table J-6. Correlation Matrices for the Uranium Decay Series and the Thorium Decay SeriesNevada Environmental Response Trust Remediation Project SiteHenderson, Nevada

#### iv) DU-2 Soils

Uranium Decay Chain				
Correl.	Ra-226	Th-230	U-234	U-238
Ra-226	1	0.384	0.353	0.162
Th-230	0.384	1	0.532	0.385
U-234	0.353	0.532	1	0.553
U-238	0.162	0.385	0.553	1

Thorium Decay Chain				
Correl.	Ra-228	Th-228	Th-232	
Ra-228	1	0.385	0.470	
Th-228	0.385	1	0.607	
Th-232	0.470	0.607	1	

#### v) DU-3 Soils

Uranium Decay Chain				
Correl.	Ra-226	Th-230	U-234	U-238
Ra-226	1	0.736	0.389	0.407
Th-230	0.736	1	0.796	0.830
U-234	0.389	0.796	1	0.989
U-238	0.407	0.830	0.989	1

Thorium Decay Chain				
Correl.	Ra-228	Th-228	Th-232	
Ra-228	1	-0.358	-0.646	
Th-228	-0.358	1	0.878	
Th-232	-0.646	0.878	1	



### Figure J1–1. Background vs. Decision Unit Boxplots Arsenic





## Figure J1–2. Background vs. Decision Unit Boxplots Chromium VI





## Figure J1–3. Background vs. Decision Unit Boxplots Cobalt













## Figure J1–6. Background vs. Decision Unit Boxplots Thallium











## Figure J1–8. Background vs. Decision Unit Boxplots Uranium–238





## Figure J1–9. Background vs. Decision Unit Boxplots Uranium–234





## Figure J1–10. Background vs. Decision Unit Boxplots Thorium–230





Figure J1–11. Background vs. Decision Unit Boxplots Radium–226





## Figure J1–12. Background vs. Decision Unit Boxplots Thorium–232











## Figure J1–14. Background vs. Decision Unit Boxplots Thorium–228





## Figure J1–15. Background vs. Decision Unit Boxplots Uranium–235





Normal Theoretical Quantiles

Figure J2–1A. Normal Q–Q Plots Arsenic

## Figure J2–1B. Lognormal Q–Q Plots Arsenic



## Figure J2–2A. Normal Q–Q Plots Chromium VI







## Figure J2–3A. Normal Q–Q Plots Cobalt



## Figure J2–3B. Lognormal Q–Q Plots Cobalt





## Figure J2–4A. Normal Q–Q Plots Manganese

## Figure J2–4B. Lognormal Q–Q Plots Manganese







## Figure J2–5B. Lognormal Q–Q Plots Palladium


### Figure J2–6A. Normal Q–Q Plots Thallium



## Figure J2–6B. Lognormal Q–Q Plots Thallium





Figure J2–7A. Normal Q–Q Plots Zirconium

### Figure J2–7B. Lognormal Q–Q Plots Zirconium









Figure J2–8B. Lognormal Q–Q Plots Uranium-238



## Figure J2–9A. Normal Q–Q Plots Uranium–234



## Figure J2–9B. Lognormal Q–Q Plots Uranium–234







# Figure J2–10B. Lognormal Q–Q Plots Thorium–230



#### Figure J2–11A. Normal Q–Q Plots Radium–226



### Figure J2–11B. Lognormal Q–Q Plots Radium–226





### Figure J2–12B. Lognormal Q–Q Plots Thorium–232



## Figure J2–13B. Lognormal Q–Q Plots Radium–228







# Figure J2–14B. Lognormal Q–Q Plots Thorium–228



Figure J2–15A. Normal Q–Q Plots Uranium–235



Figure J2–15B. Lognormal Q–Q Plots Uranium–235